GLOBAL BANCORP 700 E. El Camino Real, Suite 110 Mountain View, California 94040 (650) 810-9400

July 22, 2011

PROPOSED MERGER – YOUR VOTE IS VERY IMPORTANT

The boards of directors of Global Bancorp, Global Trust Bank, and Bay Commercial Bank have approved a merger agreement in which Global Bancorp will first be merged with and into its subsidiary, Global Trust Bank, and then Global Trust Bank will merge with and into Bay Commercial Bank. For Global Bancorp shareholders, if the principal terms of the merger agreement are approved and both mergers are subsequently completed, each outstanding share of Global Bancorp common stock will be converted into the right to receive either (i) .75678 shares of Bay Commercial Bank common stock, or (ii) \$7.25 in cash or (iii) a combination of the two (subject to limitations and adjustments in the merger agreement, including the limitation that no more than approximately 20% of the outstanding shares of Global Bancorp be converted to cash in the merger). The merger consideration is described in detail in the section entitled "JOINT PROPOSAL I – THE MERGERS – Calculation of Merger Consideration to be Paid to Global Bancorp Shareholders" and in the merger agreement attached as <u>Appendix A</u> to this joint proxy statement-offering circular. For Bay Commercial Bank shareholders, the shares held by Bay Commercial Bank shareholders prior to the merger will continue to remain outstanding after the merger.

The merger cannot be completed unless a majority of the shareholders of Global Bancorp and a majority of the shareholders of Bay Commercial Bank approve the principal terms of the merger agreement.

Based on the reasons for the mergers described in this joint proxy statement-offering circular, our respective boards of directors believe that the merger is fair to you and in your best interests. Accordingly, our respective boards of directors recommend that you vote "FOR" approval of the principal terms of the merger agreement.

The accompanying joint proxy statement-offering circular gives you detailed information about the shareholders' meetings, the mergers and related matters and other items being voted upon at the Global Bancorp annual meeting and the Bay Commercial Bank annual meeting. In addition, Bay Commercial Bank and Global Bancorp shareholders are asked to vote on additional proposals that are normally part of the agenda for their annual meeting of shareholders including the election of the board of directors. We urge you to read this entire document carefully, including the considerations discussed under "RISK FACTORS," beginning on page 10, and the appendices thereto, which include the merger agreements.

It is very important that your shares be represented at the annual meetings. Whether or not you plan to attend the annual meeting, please complete, date and sign the enclosed proxy form and return it promptly in the postage-paid envelope provided.

Sincerely,

then 1 then !

Vinod K. Thukral Global Bancorp, Chairman

An<

Lloyd Kendall Bay Commercial Bank, Chairman

Neither the Securities and Exchange Commission, the California Department of Financial Institutions, the Federal Deposit Insurance Corporation, the Board of Governors of the Federal Reserve System, the California Department of Corporations, nor any other state securities commission has approved or disapproved of the securities to be issued in connection with the merger or has determined if this joint proxy statement-offering circular is accurate or complete. Any representation to the contrary is a criminal offense.

This joint proxy statement-offering circular is dated July 22, 2011 and will be mailed to shareholders of Global Bancorp and Bay Commercial Bank on or about July 27, 2011. KYL LB1411535v7

BAY COMMERCIAL BANK

Notice of Annual Meeting of Shareholders

August 16, 2011

To: The Shareholders of Bay Commercial Bank

Notice is hereby given that, under the terms of its bylaws and the call of its board of directors, the annual meeting of shareholders of Bay Commercial Bank will be held at its Headquarters Office, located at 1280 Civic Drive, Suite 100, Walnut Creek, California 94596, on August 16, 2011, at 4:00 p.m. (local time), for the purpose of considering and voting upon the following matters:

- 1. <u>Approval of the Merger Agreement</u>. To approve the principal terms of the Agreement and Plan of Reorganization and Merger dated May 12, 2011, as amended, attached as <u>Appendix A</u> to this joint proxy statement-offering circular ("*Merger Agreement*") for the merger of Global Trust Bank with and into Bay Commercial Bank;
- 2. <u>Adjournment</u>. If necessary to consider and vote upon a proposal to grant discretionary authority to adjourn the annual meeting if necessary to permit further solicitation of proxies if there are not sufficient votes at the time of the annual meeting to approve the principal terms of the Merger Agreement;
- 3. <u>Bay Commercial Bank Proposal A: Election of Directors</u>. To elect seven persons to the Board of Directors, each for a one-year term; and,
- 4. <u>Bay Commercial Bank Proposal B: Ratification of Auditors</u>. To ratify the selection by the board of directors of Moss Adams, LLP, independent public accountants, to audit the financial statements of Bay Commercial Bank for the fiscal year ended December 31, 2011.

The Merger Agreement sets forth the terms of the merger of Global Trust Bank with and into Bay Commercial Bank. As a result of the merger, all current shareholders of Global Bancorp, which holds all of the outstanding shares of Global Trust Bank, will receive either (i) .75678 shares of Bay Commercial Bank common stock, (ii) \$7.25 in cash, or (iii) a combination of the two, per share of Global Bancorp stock owned (subject to limitations and adjustments in the Merger Agreement). The shares held by Bay Commercial Bank shareholders prior to the merger will continue to remain outstanding after the merger. The transaction is more fully described in the enclosed joint proxy statement-offering circular and in the Merger Agreement included as <u>Appendix A</u> thereto.

The board of directors has fixed the close of business on June 17, 2011, as the record date for determination of shareholders entitled to notice of, and the right to vote at, the annual meeting of shareholders and any adjournments or postponements of the annual meeting.

Because the affirmative vote of shareholders holding a majority of the outstanding shares of Bay Commercial Bank common stock is required to approve the principal terms of the Merger Agreement for the merger of Global Trust Bank with and into Bay Commercial Bank, it is essential that all shareholders vote. You are urged to vote in favor of the proposals by signing and returning the enclosed proxy as promptly as possible, whether or not you plan to attend the annual meeting of shareholders in person. If you do attend the meeting you may then withdraw your proxy. The proxy may be revoked at any time prior to its exercise. Whether or not you plan to attend the annual meeting, please complete, date and sign the enclosed proxy form and return it promptly in the postage-paid envelope provided in order to vote.

By Order of the Board of Directors

Klow 4

Keary Colwell, Corporate Secretary

Dated: July 22, 2011

KYL_LB1411535v7

GLOBAL BANCORP

Notice of Annual Meeting of Shareholders

August 18, 2011

To: The Shareholders of Global Bancorp

Notice is hereby given that, under the terms of its bylaws and the call of its board of directors, a annual meeting of shareholders of Global Bancorp will be held at 700 E. El Camino Real, Suite 110, Mountain View, California 94040, on August 18, 2011, at 5:30 p.m. (local time), for the purpose of considering and voting upon the following matters:

- 1. <u>Approval of the Downstream Merger Agreement</u>. To approve the principal terms of the Downstream Merger Agreement ("*Downstream Merger Agreement*"), dated May 12, 2011, as amended, attached as <u>Appendix B</u> to the joint proxy statement-offering circular for the merger of Global Bancorp with and into Global Trust Bank;
- 2. <u>Approval of the Merger Agreement</u>. To approve the principal terms of the Agreement and Plan of Reorganization and Merger, dated May 12, 2011, attached as <u>Appendix A</u> to the joint proxy statement-offering circular ("*Merger Agreement*") for the merger of Global Trust Bank with and into Bay Commercial Bank;
- 3. <u>Adjournment</u>. To consider and vote upon a proposal to grant discretionary authority to adjourn the annual meeting if necessary to permit further solicitation of proxies if there are not sufficient votes at the time of the annual meeting to approve the principal terms of the Merger Agreement; and,
- 4. <u>Global Bancorp Proposal A: Election of Directors</u>. To elect seven persons to the board of directors, for a term of one year.

The downstream merger will not be completed if the merger with Bay Commercial Bank is not approved or all conditions in the Merger Agreement are not satisfied. The Merger Agreement sets forth the terms of the merger of Global Trust Bank with and into Bay Commercial Bank, which will immediately follow the merger of Global Bancorp with and into Global Trust Bank. As a result of the merger with Bay Commercial Bank, all shareholders of Global Bancorp will receive either (i) 0.75678 shares of Bay Commercial Bank common stock, (ii) \$7.25 in cash, or (iii) a combination of the two, per share of Global Bancorp stock owned (subject to limitations and adjustments in the Merger Agreement). The shares held by Bay Commercial Bank shareholders prior to the merger will continue to remain outstanding after the merger. The transactions are more fully described in the enclosed joint proxy statement-offering circular and in the Merger Agreement included as <u>Appendix A</u> thereto.

The board of directors has fixed the close of business on June 30, 2011, as the record date for determination of shareholders entitled to notice of, and the right to vote at, the annual meeting of shareholders and any adjournments or postponements of the annual meeting. You have a right to dissent from the merger and obtain payment of the fair value of your shares by complying with the provisions of the California General Corporation Law contained in <u>Appendix D</u>.

Since the affirmative vote of shareholders holding a majority of the outstanding shares of Global Bancorp common stock is required to approve the principal terms of the Downstream Merger Agreement and the principal terms of the Merger Agreement, it is essential that all shareholders vote. You are urged to vote in favor of the proposals. Whether or not you plan to attend the annual meeting, please complete, date and sign the enclosed proxy form and return it promptly in the postage-paid envelope provided.

By Order of the Board of Directors

2

Bhupen B. Amin, Corporate Secretary

Dated: July 22, 2011

JOINT PROXY STATEMENT-OFFERING CIRCULAR

TABLE OF CONTENTS

	Page
QUESTIONS AND ANSWERS ABOUT THE MERGER AND THE ANNUAL MEETINGS	1
Questions and Answers About This Joint Proxy Material And Voting	1
Questions And Answers About The Merger	2
SUMMARY	4
RISK FACTORS	10
MARKETS; MARKET PRICES AND DIVIDENDS	16
SELECTED FINANCIAL DATA	17
COMPARATIVE PER SHARE DATA	21
THE GLOBAL BANCORP ANNUAL MEETING	22
THE BAY COMMERCIAL BANK ANNUAL MEETING	24
JOINT PROPOSAL I – THE MERGERS	26
UNAUDITED PRO FORMA COMBINED CONDENSED FINANCIAL STATEMENTS	57
UNAUDITED PRO FORMA COMBINED CONDENSED FINANCIAL INFORMATION	58
UNAUDITED PRO FORMA COMBINED CONDENSED BALANCE SHEET AS OF MARCH 31, 2011	59
UNAUDITED PRO FORMA COMBINED CONDENSED STATEMENT OF EARNINGS FOR THE PERIOD JANUARY 1, 2011 THROUGH MARCH 31, 2011	61
INFORMATION ABOUT BAY COMMERCIAL BANK	62
BAY COMMERCIAL BANK: MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS	66
INFORMATION ABOUT GLOBAL TRUST BANCORP AND GLOBAL TRUST BANK	69
GLOBAL BANCORP AND GLOBAL TRUST BANK: MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS	72
COMPARISONS BETWEEN BAY COMMERCIAL BANK AND GLOBAL BANCORP	75
SUPERVISION AND REGULATION	76
BAY COMMERCIAL BANK PROPOSAL A – ELECTION OF DIRECTORS	93
BAY COMMERCIAL BANK PROPOSAL B: RATIFICATION OF SELECTION OF INDEPENDENT AUDITORS FIRM FOR 2011	94
BAY COMMERCIAL BANK OTHER PROPOSED ACTION	94
GLOBAL BANCORP PROPOSAL A – ELECTION OF DIRECTORS	96
GLOBAL BANCORP OTHER PROPOSED ACTION	97

LIST OF APPENDICES

Agreement and Plan of Reorganization and Merger dated May 12, 2011, as amended, by and among Global Bancorp, Global Trust Bank and Bay Commercial Bank	Appendix A
Downstream Merger Agreement between Global Bancorp and Global Trust Bank	Appendix B
Merger Agreement between Bay Commercial Bank and Global Trust Bank	Appendix C
Selected sections of Chapter 13 of the California Corporation Code	Appendix D
Opinion of The Findley Group to Global Bancorp	Appendix E
Opinion of Vining Sparks IBG LP to Bay Commercial Bank	Appendix F
Bay Commercial Bank Financial Statements for the Years Ended December 31, 2010 and 2009 and Independent Auditors Report	Appendix G
Global Bancorp and Subsidiary – Report of Independent Auditors and Consolidated Financial Statements as of and for the two years ended December 31, 2010	Appendix H

QUESTIONS AND ANSWERS ABOUT THE MERGER AND THE ANNUAL MEETINGS

Following are brief answers to certain questions that you may have regarding the proposals being considered at the Global Bancorp annual meeting and the Bay Commercial Bank annual meeting. Global Bancorp and Bay Commercial Bank urge you to read carefully this entire joint proxy statement-offering circular because this section does not provide all the information that might be important to you. All references to the Merger Agreement are to the Agreement and Plan of Reorganization and Merger, dated May 12, 2011, as amended, by and among Bay Commercial Bank, Global Bancorp, and Global Trust Bank, a copy of which is attached as <u>Appendix A</u> to this joint proxy statement-offering circular.

Questions and Answers About This Joint Proxy Material And Voting

Q: Why have I received these materials?

A: This joint proxy statement-offering circular and the enclosed proxy card were sent to you because the boards of directors of Global Bancorp and Bay Commercial Bank are soliciting proxies to vote at their annual meeting of shareholders to be held on August 18, 2011 and August 16, 2011, respectively. You are cordially invited to attend the annual meeting of shareholders for the company in which you hold stock and are requested to vote on the proposals described in this joint proxy statement-offering circular. We intend to mail this joint proxy statement-offering circular and accompanying proxy card on or about July 27, 2011 to all shareholders entitled to vote at the annual meetings.

Q: Who is entitled to vote at the annual meetings?

A: Shareholders of record as of the close of business on June 30, 2011 will be entitled to vote at the annual meeting of shareholders of Global Bancorp. Shareholders of record as of the close of business on June 17, 2011 will be entitled to vote at the annual meeting of shareholders of Bay Commercial Bank.

Q: What am I voting on?

- A: If you are a Global Bancorp shareholder, you are being asked to vote to:
 - approve the principal terms of the Downstream Merger Agreement for the merger of Global Bancorp with and into Global Trust Bank;
 - approve the principal terms of the Merger Agreement for the subsequent merger of Global Trust Bank with and into Bay Commercial Bank;
 - approve, if necessary, any adjournment or postponement of the annual meeting to solicit additional proxies in favor of the merger proposal; and,
 - elect seven persons to the board of directors of Global Bancorp.

If you are a Bay Commercial Bank shareholder, you are being asked to vote to:

- approve the principal terms of the Merger Agreement;
- approve, if necessary, any adjournment or postponement of the annual meeting to solicit additional proxies in favor of the merger proposal;
- elect seven persons to the board of directors of Bay Commercial Bank; and,
- ratify the appointment of Moss Adams LLP as the independent auditors for Bay Commercial Bank for its fiscal year ending December 31, 2011.

Q: How do I vote?

A: For each of the matters to be voted on, you may vote "FOR" or "AGAINST" or abstain from voting.

If you are a shareholder of record, you may vote in person at the annual meeting you attend, or you may vote by proxy using the enclosed proxy card. Whether or not you plan to attend the annual meeting, you are urged to vote by proxy to ensure your vote is counted. You may still attend the annual meeting, and vote in person if you have already voted by proxy.

- To vote in person, come to the annual meeting and you will be given a ballot when you arrive.
- To vote using the proxy card, simply complete, sign and date the enclosed proxy card and return it promptly in the envelope provided. If you return your signed proxy card to us before the annual meeting, we will vote your shares as you direct.

If you are a beneficial owner of shares registered in the name of your broker, bank, or other agent, you should have received a proxy card and voting instructions with these proxy materials from that agent or custodian rather than from Global Bancorp or Bay Commercial Bank. In order to vote, complete and mail the proxy card to ensure that your vote is counted. To vote in person at the annual meeting, you must obtain a valid proxy from your broker, bank, or other agent. Follow the instructions from your broker or bank included with these proxy materials, or contact your broker or bank to request a proxy form.

Q: How many votes do I have?

A: Each share of common stock of Global Bancorp and each share of common stock of Bay Commercial Bank is entitled to one vote with respect to each matter to be voted on at their respective annual meetings.

Q: What constitutes a quorum for purposes of each of the annual meetings?

A: A quorum of shareholders is necessary to hold a valid shareholders meeting. The presence at each of the shareholder meetings in person or by proxy of the holders of a majority of the voting power of all outstanding shares of common stock entitled to vote shall constitute a quorum for the transaction of business. Proxies marked as abstaining (including proxies containing broker non-votes) on any matter to be acted upon by shareholders will be treated as present at the meeting for purposes of determining a quorum but will not be counted as votes cast on such matters. If there is no quorum, a majority of the votes present at the meeting may adjourn the meeting to another date.

Questions And Answers About The Merger

This question and answer summary highlights selected information contained in other sections of this joint proxy statement-offering circular. To understand the merger more fully, you should carefully read this entire joint proxy statement-offering circular, including all appendices and financial statements.

Q: What will happen if Global Bancorp and Bay Commercial Bank shareholders approve the mergers?

A: First, Global Bancorp will be merged with and into Global Trust Bank, and Global Bancorp will cease to exist. Thereafter, Global Trust Bank will merge with and into Bay Commercial Bank, and Global Trust Bank will cease to exist. We expect these mergers to take place during the third quarter of 2011 after receipt of required regulatory approvals. In the merger, each outstanding share of Global Bancorp common stock will be converted into either cash or that number of shares of Bay Commercial Bank common stock, or a combination thereof, determined pursuant to the terms of the Merger Agreement.

Q: If the merger is approved what consideration will Global Bancorp shareholders receive?

A: If you are a holder of Global Bancorp common stock and do not exercise dissenters' rights, you will receive in the merger for each share of Global Bancorp common stock either (i) cash in the amount of \$7.25 per share, (ii) newly-issued shares of Bay Commercial Bank common stock in the exchange ratio of 0.75678 shares of Bay Commercial Bank common stock for each share of your Global Bancorp common stock or (iii) a combination of cash and common stock, subject to the amount of cash consideration paid in the merger not exceeding approximately \$3.067 million and the number of shares of Bay Commercial Bank common stock issued in the merger being not less than 1.692 million.

The limitation on the amount of cash consideration in the merger means that no more than approximately 20% of the outstanding shares of Global Bancorp will be exchanged for cash. Please read the sections entitled "RISK FACTORS – Risks Regarding the Merger" and "JOINT PROPOSAL I – THE MERGERS – Calculation of Consideration to be Paid to Global Bancorp Shareholders," and the Merger Agreement included as <u>Appendix A</u> herein for additional information.

Q: Do Global Bancorp shareholders have dissenters' rights in the merger?

A: Yes. Holders of Global Bancorp common stock who do not vote in favor of the merger and who have fully complied with all applicable provisions of the California Corporations Code, sections 1301, 1302, 1303, and 1304, which sections are attached hereto as <u>Appendix D</u> and incorporated herein by reference, may receive cash in an amount equal to the fair market value, as determined by Bay Commercial Bank, or, if required, by a court of law, of their shares of Global Bancorp common stock, as of May 13, 2011, the business day immediately preceding the announcement of merger. Please read the section entitled "JOINT PROPOSAL I – THE MERGER – Dissenters' Rights of Global Bancorp's Shareholders" and <u>Appendix C</u>, for additional information.

Q: Do Bay Commercial Bank shareholders have dissenters' rights in the merger?

A: No. Under the California Financial Code, provisions of the California Corporations Code providing for dissenters' rights in a merger do not apply to the shareholders of the surviving depository institution.

Q: Why is Global Trust Bank merging with Bay Commercial Bank?

A: Global Bancorp's, Global Trust Bank's and Bay Commercial Bank's respective managements believe that their respective shareholders will benefit from the merger because the business potential for the combined companies exceeds what each company could accomplish individually. They believe that their similar and complementary financial products and services in their respective markets and their combined capital bases will contribute to enhanced future performance, as well as providing a larger shareholder base. Global Bancorp and Bay Commercial Bank believe a larger shareholder base will increase shareholder liquidity and provide for increased shareholder value. Please read the section entitled "JOINT PROPOSAL I – THE MERGER – Bay Commercial Bank Reasons for the Merger" and "– Global Bancorp Reasons for the Merger" for additional information.

Q: Should I send in my certificates now?

A: No. You should not send your Global Bancorp stock certificates in the envelope provided for use in returning your proxy. You will be sent written instructions for exchanging your stock certificates only if the merger is approved and completed.

Q: What happens if I do not return my proxy card?

A: If you fail to execute and return your proxy card or vote in any other way as set forth herein, it will have the same effect as voting against the merger, except for purposes of asserting dissenters rights, discussed above.

Q: What risks should I consider before I vote on the merger?

A: The risks that you should consider in deciding how to vote on the merger are explained in the section of this joint proxy statement-offering circular entitled "RISK FACTORS." You are urged to read this section, as well as the rest of this joint proxy statement-offering circular, before deciding how to vote.

Q: How do I vote?

A: Just indicate on your proxy card how you want to vote. Sign and mail your proxy card in the enclosed envelope as soon as possible so that your shares will be represented at the Bay Commercial Bank annual shareholders meeting or Global Bancorp annual shareholders meeting, as applicable. Alternatively, you may attend the meeting and vote in person.

If you sign and send in your proxy card and do not indicate how you want to vote, your proxy will be voted in favor of the merger. If you do not sign and send in your proxy card or you abstain from voting, it will have the effect of voting against the merger.

You may attend the meeting and vote your shares in person, rather than voting by proxy. In addition, you may withdraw your proxy up to and including the day of the applicable shareholders' meeting by following the directions herein and either change your vote or attend the meeting and vote in person.

Q: If my shares are held in my broker's name, will my broker vote them for me?

A: No. Your broker can only vote your shares if you provide instructions on how to vote them. You should, therefore, instruct your broker on how to vote your shares by following the directions your broker provides when forwarding these proxy materials to you. If you do not provide voting instructions to your broker, your broker will not be able to vote your shares. This will have the effect of not voting in favor of the merger.

Q: How do Global Bancorp's directors plan to vote?

A: All of Global Bancorp's directors have committed to vote their shares in favor of the principal terms of the Merger Agreement. Global Bancorp's directors collectively hold, as of the record date for the annual shareholders' meeting, 205,000 shares which are entitled to vote at the annual shareholders' meeting, or approximately 9.7% of the Global Bancorp common stock eligible to vote. The affirmative vote of a majority of Global Bancorp's issued and outstanding shares of common stock eligible to vote are needed to approve the principal terms of the Merger Agreement.

Q: How do Bay Commercial Bank's directors plan to vote?

A: All of Bay Commercial Bank's directors have committed to vote their shares in favor of the principal terms of the Merger Agreement. Bay Commercial Banks' directors collectively hold, as of the record date for the annual shareholders' meeting, 455,396 shares, or approximately 12.7% of Bay Commercial Bank common stock eligible to vote. The affirmative vote of a majority of Bay Commercial Bank's issued and outstanding shares of common stock eligible to vote is needed to approve the principal terms of the Merger Agreement.

Q: If I own shares in both Global Bancorp and Bay Commercial Bank, should I vote only once?

A: No. If you own shares in both companies, you will receive separate proxy cards for each meeting. It is important that you vote at both meetings, so please complete, sign, date and return your proxy card as instructed by Global Bancorp and Bay Commercial Bank, respectively.

Q: Who can help answer my other questions?

A: If you want to ask any additional questions about the merger, you should contact either Keary Colwell, Bay Commercial Bank, at (925) 476-1800, or Vinod K. Thukral, Global Bancorp, at (650) 810-9400.

SUMMARY

This summary only highlights selected information from this joint proxy statement-offering circular. You should carefully read this entire joint proxy statement-offering circular, including the appendices. These will give you a more complete description of the mergers, the Downstream Merger Agreement, the Merger Agreement, and the transactions proposed. You should also refer to the sections entitled "DESCRIPTION OF GLOBAL BANCORP" and "INFORMATION ABOUT BAY COMMERCIAL BANK."

Forward-Looking Statements

This joint proxy statement may contain various "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Forward-looking statements include projections, statements of the plans and objectives of management for future operations, statements of future economic performance, assumptions underlying these statements, and other statements that are not statements of historical facts. Forward-looking statements are subject to significant business, economic and competitive risks, uncertainties and contingencies, many of which are beyond our control. Generally forward-looking statements can be identified by the use of forward-looking terminology such as "anticipate," "believe," "estimate," "expect," "intend," "project," "should," and similar expressions. Those statements include, among other things, the risk that the merger may not be consummated, whether due to regulatory approval, vote of the shareholders or due to other reason, and the calculation of the final merger consideration payable for Global Bancorp shares. Global Bancorp and Bay Commercial Bank caution you that reliance on any forward-looking statements involves risks and uncertainties, and although Global Bancorp and Bay Commercial Bank believe that the assumptions on which or forward-looking statements are based are reasonable, any of these assumptions could prove incorrect and, as a result, the actual results may vary materially from those anticipated. In light of these and other uncertainties you should not conclude that Global Bancorp and Bay Commercial Bank will necessarily achieve any plans and objectives or projected financial results referred to in any of the forward-looking statements. Global Bancorp and Bay Commercial Bank do not undertake to release the results of any revisions of these forward-looking statements to reflect future events or circumstances.

General

This joint proxy statement-offering circular relates to the proposed merger of Global Bancorp with its subsidiary Global Trust Bank and the subsequent merger of Global Trust Bank with Bay Commercial Bank. Global Bancorp and Bay Commercial Bank believe that the merger will create opportunities to apply their similar community banking philosophies to realize enhanced revenues through asset growth and market penetration.

Parties to the Merger

Global Bancorp/Global Trust Bank 700 E. El Camino Real, Suite 110 Mountain View, California 94040 (650) 810-9400

Global Bancorp is a bank holding company headquartered in Mountain View, California. Global Bancorp has one subsidiary bank, Global Trust Bank, which is a California-chartered commercial bank that serves the greater San Francisco Bay Area of California.

Please read the section entitled "INFORMATION ABOUT GLOBAL BANCORP AND GLOBAL TRUST BANK" for additional information about Global Bancorp and Global Trust Bank.

Bay Commercial Bank 1280 Civic Drive, Suite 100 Walnut Creek, California 94596 (925) 476-1800

Bay Commercial Bank is a California chartered commercial bank which opened for business on July 20, 2004. It serves the greater San Francisco Bay Area through offices located in Walnut Creek, Oakland, and Castro Valley, California.

Please read the section entitled "INFORMATION ABOUT BAY COMMERCIAL BANK" for additional information.

Annual Shareholders' Meeting – Global Bancorp

Global Bancorp will hold its annual shareholders' meeting at 700 E. El Camino Real, Suite 110, Mountain View, California 94040, on August 18, 2011, at 5:30 p.m. (local time). At the annual shareholders' meeting you will be asked to consider and vote on the approval of the principal terms of the Downstream Merger Agreement and the approval of the principal terms of the Merger Agreement, providing for the merger of Global Bancorp with and into Global Trust Bank and Global Trust Bank with and into Bay Commercial Bank, respectively, the election of directors and any other matters that may properly come before the meeting. You may vote at the Global Bancorp special shareholders' meeting if you owned shares of Global Bancorp stock at the close of business on the record date, which is June 30, 2011. On that date, Global Bancorp had 2,115,001 shares of common stock issued and outstanding. Approval of the principal terms of the merger agreements for the mergers requires the affirmative vote of at least a majority of the shares of Global Bancorp common stock outstanding and entitled to vote on the record date. Please read the section entitled "THE GLOBAL BANCORP ANNUAL MEETING" for additional information.

Annual Shareholders' Meeting – Bay Commercial Bank

Bay Commercial Bank will hold its annual shareholders' meeting on August 16, 2011 at 4:00 p.m. (local time), at its headquarters office, located at 1280 Civic Drive, Suite 100, Walnut Creek, California 94596. At the annual shareholders' meeting you will be asked to consider and vote on the approval of the principal terms of the Merger Agreement, the election of directors, the ratification of the board of directors' selection of independent auditors, and any other matters that may properly come before the meeting. You may vote at the Bay Commercial Bank annual shareholders' meeting if you owned shares of Bay Commercial Bank stock at the close of business on the record date, which is June 17, 2011. On that date, Bay Commercial Bank had 3,549,794 shares of common stock issued and outstanding and entitled to be voted. Approval of the principal terms of the Merger Agreement and the merger requires the affirmative vote of at least a majority of the shares of Bay Commercial Bank common stock outstanding on the record date. With respect to the election of directors, the duly-nominated persons who receive the seven highest vote totals will be elected. The ratification of the selection of auditors requires the affirmative vote of a majority of Bay Commercial Bank's shares of common stock represented and voting at a meeting at which a quorum is present.

The Mergers

The mergers will ultimately result in Global Bancorp and Global Trust Bank being merged out of existence and into Bay Commercial Bank. The mergers will not occur without the approval of the shareholders of Global Bancorp and Bay Commercial Bank. There are also other customary conditions which must be met in order for the mergers to be completed. Please read the sections entitled "JOINT PROPOSAL I – THE MERGERS – Structure of the Mergers" and "– Certain Effects of the Mergers" for additional information.

The Merger Agreements

The Downstream Merger Agreement and the Merger Agreement are the legal documents that contain the mergers' terms and govern the merger process, including the issuance of Bay Commercial Bank common stock to Global Bancorp's shareholders in the merger. Please read the entire agreements which are attached to this joint proxy statement-offering circular as <u>Appendix A</u> and <u>Appendix B</u>. Also, please read the section entitled "JOINT PROPOSAL I – THE MERGERS – The Merger Agreements" for additional information.

Consideration to be Paid to Global Bancorp Shareholders

If the mergers are completed, Global Bancorp shareholders will generally have the right to exchange each share of Global Bancorp common stock they own for either (i) 0.75678 shares of Bay Commercial Bank's common stock, (ii) \$7.25 in cash or (iii) a combination of the two, at the election of the holder; provided that Global Bancorp shareholders' election of the type of consideration they wish to receive is subject to the requirements that (i) cash payments made to Global Bancorp shareholders electing to receive cash in whole or in part may not exceed on amount equal to: (a) 20% of the total consideration paid by Bay Commercial Bank in the Merger, less (b) the amount of cash paid to dissenting shareholders and (ii) the number of shares of Bay Commercial Bank common stock issued in the merger shall be not less than approximately 1.692 million.

Regulatory Approvals

Global Bancorp and Global Trust Bank have filed an application with the Federal Deposit Insurance Corporation (the "*FDIC*") for approval of the merger of Global Bancorp with and into Global Trust Bank. Bay Commercial Bank has filed an application with the California Department of Financial Institutions (the "*CDFI*") for approval of the merger of Global Trust Bank with and into Bay Commercial Bank. Bay Commercial Bank has filed an application to become a member bank of the Federal Reserve System and, upon becoming a member bank, Bay Commercial Bank intends to file an application with the Board of Governors of the Federal Reserve System (the "*FRB*") for approval of the merger of Global Trust Bank with and into Bay Commercial Bank. The applications filed with the FDIC for approval of both the merger of Global Bancorp with and into Global Trust Bank and the merger of Global Trust Bank with and into Bay Commercial Bank and the merger of Global Trust Bank with and into Bay Commercial Bank and the merger of Global Trust Bank with and into Bay Commercial Bank and the merger of Global Trust Bank with and into Bay Commercial Bank and the merger of Global Trust Bank with and into Bay Commercial Bank and the merger of Global Trust Bank with and into Bay Commercial Bank are still pending as of the date of this joint proxy statement-offering circular. Similarly, the application filed with the CDFI for approval of the merger of Global Trust Bank with and into Bay Commercial Bank and Global Trust Bank is subject to the parties obtaining all of the state and federal regulatory approvals that may be required by applicable laws and regulations. If all such agencies' approvals are obtained, we expect them to be subject to standard conditions. Global Bancorp and Bay Commercial Bank have no reason to believe that any such conditions cannot be

satisfied. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – Regulatory Approvals" for additional information.

Votes Required; Securities Held by Insiders

Approval of the merger requires the affirmative vote of a majority of the outstanding shares of Global Bancorp's 2,115,001 issued and outstanding shares of common stock eligible to vote, or 1,057,501 shares, and a majority of the 3,549,794 outstanding shares of Bay Commercial Bank issued and outstanding shares, or 1,774,898 shares. Your failure to vote in person or by proxy, or your abstention from voting entirely, will have the same effect as voting against the merger.

The directors and executive officers of Global Bancorp owned 210,000 shares, or approximately 10%, of Global Bancorp's outstanding shares of common stock, all of which are eligible to vote. The directors and executive officers of Bay Commercial Bank owned 474,236 shares, or approximately 13.4% of Bay Commercial Bank's outstanding shares of common stock. Global Bancorp's and Bay Commercial Bank's directors have entered into separate agreements in which they have agreed, among other things, to vote "FOR" approval of the Merger Agreement. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – The Merger Agreement – Director-Shareholder Agreements" for additional information.

Opinion of Global Bancorp's Financial Advisors

In deciding to approve the merger, Global Bancorp's board of directors considered, among other things, the opinion of The Findley Group, Global Bancorp's financial advisors, regarding the fairness, from a financial point of view, of the consideration to be received by Global Bancorp's shareholders as a result of the merger. The financial advisors determined that the Merger Agreement is fair to Global Bancorp's shareholders from a financial standpoint. The advisors' written opinion is attached as <u>Appendix E</u>. You should read it carefully to understand the assumptions made, matters considered and limitations of the review undertaken by the advisors in providing their opinions. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – Opinion of Financial Advisors – Fairness Opinion for Global Bancorp's Board of Directors" for additional information.

Recommendation of Global Bancorp's Board of Directors

On May 12, 2011, Global Bancorp's board of directors approved the merger agreements and the mergers of (i) Global Bancorp with and into Global Trust Bank; and (ii) Global Trust Bank with and into Bay Commercial Bank. Moreover, they believe that the terms of the mergers are fair to you and in your best interests. Accordingly, they recommend a vote "FOR" the proposal to approve the principal terms of the merger agreements and the mergers. The conclusions of Global Bancorp's board of directors regarding the mergers are based upon a number of factors. Please read the sections entitled "JOINT PROPOSAL I – THE MERGERS – "Global Bancorp Reasons for the Merger" and "– Opinion of Financial Advisors – Fairness Opinion for Global Bancorp's Board of Directors" for additional information.

Opinion of Bay Commercial Bank's Financial Advisor

Bay Commercial Bank has received the opinion of Vining Sparks IBG, LP ("*Vining Sparks*"), Bay Commercial Bank's financial advisor, regarding the fairness, from a financial point of view, of the merger consideration to Bay Commercial Bank. The advisor's written opinion is attached as <u>Appendix F</u>. You should read it carefully to understand the assumptions made, matters considered and limitations of the review undertaken by the advisor in providing its opinion. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – Opinion of Financial Advisors – Fairness Opinion for Bay Commercial Bank's Board of Directors" for additional information.

Recommendation of Bay Commercial Bank's Board of Directors

On May 10, 2011, Bay Commercial Bank's board of directors unanimously approved the Merger Agreement and the merger of Global Trust Bank with and into Bay Commercial Bank. Moreover, they unanimously believe that the merger's terms are fair to you and in your best interests. Accordingly, they unanimously recommend a vote "FOR" the proposal to approve the principal terms of the Merger Agreement and the merger. The conclusions of Bay Commercial Bank's board of directors regarding the merger are based upon a number of factors. Please read the sections entitled "JOINT PROPOSAL I – THE MERGERS – Bay Commercial Bank Reasons for the Merger" and "– Opinion of Financial Advisors – Fairness Opinion for Bay Commercial Bank's Board of Directors" for additional information.

Exchange of Global Bancorp Share Certificates

After completing the mergers, holders of Global Bancorp stock certificates will need to exchange those certificates for new certificates of Bay Commercial Bank, or cash, as applicable. Shortly after completing the mergers, the exchange agent for this transaction will send Global Bancorp's shareholders detailed instructions on how to exchange their shares. Please do not send any stock certificates until you receive these instructions. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – The Merger Agreement – Exchange Procedures" for additional information.

Conditions to Closing the Merger

In addition to shareholder approval, Global Bancorp's, Global Trust Bank's, and Bay Commercial Bank's obligations to close the merger depend on other conditions being met prior to the completion of the merger. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – The Merger Agreement – Conditions to the Parties' Obligations" for additional information.

Closing the Merger

If shareholder approval is received as planned, and if the conditions to the merger have either been met or waived, Global Bancorp and Bay Commercial Bank anticipate that the merger will close during the third quarter of 2011. However, neither Global Bancorp nor Bay Commercial Bank can assure you whether or when the merger will actually close. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – The Merger Agreement – The Closing" for additional information.

Termination of the Merger

Global Bancorp, Global Trust Bank, and Bay Commercial Bank can mutually agree to terminate or extend the Merger Agreement. Any of the parties may terminate the Merger Agreement in the event of a material breach or the occurrence of certain other events.

Global Bancorp, Global Trust Bank, and Bay Commercial Bank have agreed that in the event the Merger Agreement is terminated because of a material breach, the non-breaching party will be entitled to receive up to \$400,000 from the breaching party. The parties have also agreed that in the event the Merger Agreement is terminated because of an acquisition event involving one of the parties, the other party will be entitled to receive up to \$400,000 from the party involved in the acquisition event. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – The Merger Agreement – Termination" for additional information.

Federal Income Tax Consequences

The merger is intended to qualify as a reorganization for United States federal income tax purposes under section 368(a) of the Internal Revenue Code. If the merger qualifies as a reorganization, Global Bancorp shareholders will not recognize any gain or loss upon the receipt of Bay Commercial Bank common stock in exchange for Global Bancorp common stock in connection with the merger. Global Bancorp shareholders who exchange their shares for cash will recognize gain or loss to the extent the amount of cash received exceeds or is less than their tax basis in their Global Bancorp common stock. Global Bancorp shareholders who exchange their shares for a combination of stock and cash will recognize gain, but not loss, in an amount equal to the lesser of (1) the amount of cash received in the merger or (2) an amount equal to the excess, if any, of (a) the sum of the amount of cash plus the fair market value of the Bay Commercial Bank shares received, over (b) the tax basis in the Global Bancorp common stock.

For a more complete discussion of the federal income tax consequences of the merger, you should carefully read the discussion in the section entitled "JOINT PROPOSAL I – THE MERGERS – Material United States Federal Income Tax Consequences" of this joint proxy statement-offering circular. Further, you are encouraged to consult your tax advisor because tax matters can be complicated, and the tax consequences of the merger to you will depend upon your own situation. You should also consult your tax advisor concerning all state, local and foreign tax consequences of the merger.

Accounting Treatment

Bay Commercial Bank must account for the merger using the acquisition method of accounting. Under this method of accounting, the assets and liabilities of the company acquired are recorded at their respective fair value as of completion of the merger, and are added to those of the acquiring company. Financial statements of the acquiring company issued after the merger takes place reflect these values, but are not restated retroactively to reflect the historical financial position or results of operations of the company that was acquired. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – Accounting Treatment" for additional information.

Bay Commercial Bank's Management and Operations After the Merger

Effective at the time of the merger of Bay Commercial Bank and Global Trust Bank, the following three directors of Global Bancorp will be appointed to the board of directors of Bay Commercial Bank: Bhupen B. Amin, Pramod Patel, and Harpreet Chaudhary. None of Global Bancorp's executives will be retained after the merger and other than the three directors identified, the other directors of Global Bancorp and Global Trust Bank will no longer be directors. All directors of Bay Commercial Bank will remain as members of the board of directors of Bay Commercial Bank. In addition, the current executive officers of Bay Commercial Bank will remain the executive officers of Bay Commercial Bank. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – Certain Effects of the Mergers" and "– Financial Interests in the Merger of Directors and Executive Officers of Global Bancorp and Global Bancorp and Global Trust Bank" for additional information.

Financial Interests in the Merger of Directors and Officers of Global Bancorp and Global Trust Bank

The directors and executive officers of Global Bancorp and Global Trust Bank have financial interests in the merger over and above those of Global Bancorp shareholders. As previously discussed, as a condition to the merger, three current directors of Global Bancorp will be appointed as directors of Bay Commercial Bank. In addition, the directors and officers of Global Bancorp and Global Trust Bank will also receive continuing insurance protections under the existing directors' and officers' liability insurance policy of Global Bancorp and Global Bancorp Chief Financial Officer, R. Dale McKinney, Chief Credit Officer, Robert C. Navarrete, and Senior Vice President, Akash Brahmbhatt will receive termination benefits of approximately \$170,000, \$170,000, and \$40,000, respectively, pursuant to the terms of their employment contracts, if they are not retained during the one year period after the completion of the merger.

In addition, after the completion of the merger, Bay Commercial Bank will grant contingent non-qualified stock options ("*Contingent Options*") to certain of Global Bancorp's and Global Trust Bank's founding directors, directors emeritus, organizers, founders and founding officers, as well as certain employees who will be hired by Bay Commercial Bank. For a more complete discussion of the terms and circumstances for the granting of Contingent Options, please read "JOINT PROPOSAL I – THE MERGERS – Financial Interests in the Merger of Directors and Officers of Global Bancorp and Global Trust Bank," below. You should consider these interests in deciding how to vote.

Differences in Your Rights as a Shareholder of Global Bancorp

As a Global Bancorp shareholder, your rights are currently governed by its Articles of Incorporation and Bylaws and by California corporations and banking laws. If you do not exercise your dissenters' rights, and depending upon your election, you may receive Bay Commercial Bank common stock in exchange for your Global Trust Bank common stock and become a Bay Commercial Bank shareholder. Consequently, your rights as a Bay Commercial Bank shareholder will be governed by Bay Commercial Bank's Articles of Incorporation and Bylaws and by the California corporations and banking laws. Therefore, the rights of Bay Commercial Bank shareholders differ from the rights of Global Bancorp shareholders in certain respects. Please read the section entitled "COMPARISONS BETWEEN BAY COMMERCIAL BANK AND GLOBAL BANCORP – Comparison of Shareholder Rights" for additional information.

Global Bancorp's Dissenters' Rights

In order to perfect dissenters' rights, a shareholder of Global Bancorp common stock must do the following:

• make a timely written demand upon Bay Commercial Bank for purchase in cash of his or her shares at their fair market value as of May 13, 2011, which demand includes:

- the number and class of the shares held of record by him or her that he or she demand be purchased by Bay Commercial Bank, and
- what he or she claims to be the fair market value of his or her shares as of May 13, 2011;
- have his or her demand received by Bay Commercial Bank within 30 days of the date that the notice of approval of the merger by the shareholders of Global Bancorp was mailed to Global Bancorp shareholders;
- not vote in favor of the principal terms of the merger agreements;
- submit certificates representing his or her shares for endorsement in accordance with section 1302 of the California Corporations Code; and,
- comply with such other procedures as are required by the California Corporations Code.

If dissenter's rights are properly perfected, such dissenter has the right to cash in the amount equal to the fair market value, as determined by Bay Commercial Bank, or, if required, by a court of law, of their shares of Global Bancorp common stock as of May 13, 2011, the business day immediately preceding the announcement of the merger. Please read the section entitled "JOINT PROPOSAL I – THE MERGERS – Dissenters' Rights of Global Bancorp's Shareholders" and <u>Appendix D</u> for additional information.

RISK FACTORS

In addition to the other information included in this joint proxy statement-offering circular or incorporated by reference, you are urged to carefully consider the following factors which contain all known material risks, before making a decision to approve the merger.

Risks Regarding the Merger

The parties may be unable to consummate the proposed mergers.

The mergers of Global Bancorp with and into Global Trust Bank and Global Trust Bank with and into Bay Commercial Bank, described in more detail herein, are anticipated to close during the third quarter of 2011, but must be approved by both Global Bancorp's and Bay Commercial Bank's shareholders before the mergers can be finalized. Consummation of these mergers is also subject to the receipt of required regulatory approvals and the satisfaction of other customary closing conditions. If the mergers are not completed for any reason, Global Bancorp's and/or Bay Commercial Bank's stock price may decline to the extent that the current market price reflects the assumption by investors that the merger will be completed, or because of the costs incurred by the parties in connection with the mergers.

Combining Global Trust Bank and Bay Commercial Bank may be more difficult than expected.

If Global Trust Bank and Bay Commercial Bank are unable to successfully integrate their businesses, operating results may suffer. Both Global Trust Bank and Bay Commercial Bank have operated and, until completion of the merger, will continue to operate independently of one another. It is possible that the integration process could result in the loss of key employees, disruption of one or both of the banks' ongoing business or inconsistencies in standards, controls, policies or procedures. These could negatively affect both Global Bancorp's and Bay Commercial Bank's ability to maintain relationships with customers and employees, or achieve the anticipated benefits of the merger within the time period expected, if at all. As with any merger of financial institutions, there may also be disruptions that cause customers, both deposit and loan, to take their business to competitors. Although none of the parties has been involved in a merger or acquisition involving an entire depository institution prior to the proposed merger, Bay Commercial Bank has managed the process of acquiring the Castro Valley Branch of Community Banks of Northern California, integrating \$16.4 million of loans and \$52 million of deposits into its portfolio, merging the data processing function and information with systems and integrating the branch's staff. However, there can be no assurance that Bay Commercial Bank will successfully integrate the operations of Global Trust Bank into its operations.

If the merger does not qualify as a tax-free reorganization for U.S. Federal Income Tax Purposes, you will recognize gain or loss on the exchange of your shares of Global Bancorp common stock for Bay Commercial Bank shares.

Although the U.S. Internal Revenue Service, referred to in this joint proxy statement-offering circular as the IRS, has not provided a ruling on the merger, Global Bancorp and Bay Commercial Bank will obtain an opinion of Vavrinek, Trine, Day & Co., LLP that, subject to the assumptions and qualifications included in such opinion, the merger will qualify as a tax-free reorganization under section 368(a) of the Internal Revenue Code. This opinion, however, neither binds the IRS nor prevents the IRS from adopting a contrary position. If the merger fails to qualify as a tax-free reorganization, you would generally recognize gain or loss on each share of Global Bancorp common stock surrendered in the merger in exchange for stock of Bay Commercial Bank in the amount of the difference between your basis in such share and the fair market value of the Bay Commercial Bank common stock you receive.

Global Bancorp shareholders may not receive the form of merger consideration that they elect.

The Merger Agreement is designed to ensure that no more than about 20% of the total consideration that will be paid to Global Bancorp shareholders in the merger, whether to shareholders electing to receive cash for their shares in whole or in part or to dissenting shareholders, will be paid in cash and no less than 80% will be paid in Bay Commercial Bank stock. Global Bancorp shareholders may elect to receive cash, Bay Commercial Bank common stock or a combination of cash and stock as their merger consideration, but their elections may not be fully honored. *See* "JOINT PROPOSAL I – THE MERGERS – Election Procedures."

If an election is not fully honored, a Global Bancorp shareholder will incur tax consequences that are likely to differ from those that would have resulted had he or she received the form of consideration elected. *See* "JOINT PROPOSAL I – THE MERGERS – Material United States Federal Income Tax Consequences."

A Global Bancorp shareholder will have less influence as a shareholder of Bay Commercial Bank than as a shareholder of Global Bancorp.

Global Bancorp shareholders currently have the right to vote in the election of the board of directors of Global Bancorp and on other matters affecting Global Bancorp. When the merger occurs, each Global Bancorp shareholder that receives Bay Commercial Bank common stock will become a shareholder of Bay Commercial Bank with a percentage ownership of the combined organization that is smaller than such shareholder's percentage ownership of Global Bancorp. Because of this, Global Bancorp shareholders will have less influence on the management and policies of Bay Commercial Bank than they now have on the management and policies of Global Bancorp.

Uncertainty regarding the merger may result in the loss of the employees and customers of Global Trust Bank and Bay Commercial Bank prior to the completion of the merger.

Employees of Global Trust Bank and Bay Commercial Bank may experience uncertainty about their future role with the combined bank. This may adversely affect the ability of Bay Commercial Bank after the merger to retain and attract key management and other personnel. Similarly, uncertainty regarding the merger may cause customers of Global Trust Bank or Bay Commercial Bank to withdraw their business prior to the completion of the merger. Any loss of either bank's customers could have a material adverse effect on their respective businesses, regardless of whether or not the merger is ultimately completed. There can be no assurance that customers of either Global Trust Bank or Bay Commercial Bank will continue their business without regard to the proposed merger.

Risks Regarding Bay Commercial Bank Common Stock

Bay Commercial Bank may issue common stock and preferred stock without your approval, diluting your proportional ownership interest.

Bay Commercial Bank's Articles of Incorporation authorize it to issue 100,000,000 shares of common stock. As of June 30, 2011, Bay Commercial Bank has 3,549,794 shares of common stock issued and outstanding, with between 1,692,000 to 2,115,001 additional shares to be issued in the merger. Bay Commercial Bank has not issued any shares of preferred stock. Bay Commercial Bank also has 354,979 shares reserved under its stock option plan covering its directors, officers and employees. As of June 30, 2011, there were options outstanding to purchase a total of 341,852 shares at

exercise prices ranging between \$7.05 and \$12 per share. Consequently, any shares of common stock that Bay Commercial Bank issues after the merger with Global Trust Bank will dilute your proportional ownership interest in Bay Commercial Bank.

The price of Bay Commercial Bank common stock may decrease, preventing you from selling your shares at a profit.

The market price of Bay Commercial Bank common stock could decrease and prevent you from selling your shares at a profit. The market price of Bay Commercial Bank common stock has fluctuated in recent years. Fluctuations may occur, among other reasons, due to:

- operating results;
- market demand;
- announcements by competitors;
- economic changes;
- general market conditions; and,
- legislative and regulatory changes.

The trading price of Bay Commercial Bank common stock may continue to fluctuate in response to these factors and others, many of which are beyond Bay Commercial Bank's control. We strongly urge you to consider the likelihood of these market fluctuations before deciding how to vote for the merger. Please read the section entitled "MARKETS, MARKET PRICES AND DIVIDENDS-Bay Commercial Bank Market Information and Dividends" for additional information regarding the trading prices of Bay Commercial Bank common stock.

Bay Commercial Bank's common stock is not listed and is thinly traded which could make it difficult to sell your shares.

Bay Commercial Bank's common stock is quoted on the OTC Bulletin Board under the symbol BCML. Bay Commercial Bank is neither listed on any stock exchange nor on the National Association of Securities Dealers Automated Quotation ("*NASDAQ*"), and Bay Commercial Bank can not guarantee that it will seek or, if it does, that it will be successful in, any such listing in the foreseeable future. There also has been a very limited trading market in the common stock since Bay Commercial Bank's inception and it does not have an established market maker.

An investment in Bay Commercial Bank's common stock is not an insured deposit.

Bay Commercial Bank common stock is not a bank deposit and, therefore, is not insured against loss by the FDIC, any other deposit insurance fund or by any other public or private entity. Investment in Bay Commercial Bank common stock is subject to the same market forces that affect the price of common stock in any company.

Risks Regarding the Businesses of Global Bancorp, Global Trust Bank and Bay Commercial Bank

Global Trust Bank and Bay Commercial Bank face lending risks, especially with respect to their small- and medium-sized business clienteles.

The risk of loan defaults or borrowers' inabilities to make scheduled payments on their loans is inherent in the banking business. Moreover, Global Trust Bank and Bay Commercial Bank focus primarily on lending to small- and medium-sized businesses. These businesses may not have the capital or other resources required to weather significant business downturns or downturns in the markets in which they compete. In addition, as of March 31, 2011, approximately 54.9% and 46.1% of the total loans of Bay Commercial Bank and Global Trust Bank, respectively, were commercial real estate loans. The concentration in commercial real estate loans and the practice of lending to small and medium-sized businesses the risk of loss in the loan portfolio of each bank and thus, in the merger the same risk of loss in the loan portfolio will continue in the surviving institution. Consequently, Global Trust Bank and Bay Commercial Bank may assume greater lending risks than other financial institutions which have a smaller concentration of those types of loans,

and which tend to make loans to larger businesses. Borrower defaults or borrowers' inabilities to make scheduled payments may result in losses which may exceed Global Trust Bank and Bay Commercial Bank's allowances for loan losses. These risks, if they occur, may require higher than expected loan loss provisions which, in turn, could materially impair profitability, capital adequacy and overall financial condition of the resulting bank.

Global Trust Bank and Bay Commercial Bank are limited in the amount they can lend to any individual borrower.

Global Trust Bank and Bay Commercial Bank are limited in the amount that they can lend to a single borrower. Therefore, the size of the loans which they can offer to potential customers is less than the size of loans that their competitors with larger lending limits can offer. Legal lending limits also affect their ability to seek relationships with larger and more established businesses. Through previous experience and relationships with a number of other financial institutions, participations in loans which exceed lending limits are sometimes sold. However, Bay Commercial Bank, as the surviving bank, cannot assure you of any future success that they may have in attracting or retaining customers seeking larger loans or that they can successfully engage in participation transactions for those loans on favorable terms.

Continued declines in Northern California real estate values could further materially impair profitability and financial condition.

As of March 31, 2011, approximately 56.9% and 72.5%, respectively, of Global Trust Bank's and Bay Commercial Bank's loans are secured by real estate collateral. Nearly all of the real estate securing these loans is located in Northern California. Over the past few years, this area has seen significant declines in real estate values, and this trend could continue. Real estate values are generally affected by factors such as:

- the socioeconomic conditions of the area where real estate collateral is located;
- fluctuations in interest rates;
- property and income tax laws;
- local zoning ordinances governing the manner in which real estate may be used; and,
- federal, state, and local environmental regulations

Further declines in real estate values would significantly reduce the value of the real estate collateral securing Global Trust Bank's and Bay Commercial Bank's loans, increasing the likelihood of defaults. Moreover, if the value of real estate collateral declines to a level that is not enough to provide adequate security for the underlying loans, the banks will need to make additional loan loss provisions which, in turn, will reduce their profits. Also, if a borrower defaults on a real estate secured loan, the banks may be forced to foreclose on the property and carry it as a nonearning asset which, in turn, may reduce net interest income. The occurrence of any or all of these events would impact the profitability of Bay Commercial Bank, as the surviving entity. For additional information, please read the sections entitled "INFORMATION ABOUT GLOBAL BANCORP – Management's Discussion and Analysis of Financial Condition and Results of Operations."

Changing interest rates may adversely affect our financial performance.

Global Trust Bank's and Bay Commercial Bank's profitability largely depends on the difference between the rates of interest they earn on their loans and investments, and the interest rates they pay on deposits and other borrowings. This relationship, known as the interest rate margin, is subject to fluctuation and is affected by economic and competitive factors which influence interest rates, the volume and mix of interest-earning assets and interest-bearing liabilities, and the level of non-performing assets. Fluctuations in interest rates will affect the demand of customers for the products and services of the combined bank after the merger. Global Trust Bank and Bay Commercial Bank are subject to interest rate risk to the degree that their interest-bearing liabilities reprice or mature more slowly or more rapidly or on a different basis than their interest-earning assets. Given the respective banks' current volume and mix of interest-bearing liabilities and interest-earning assets, their interest margin could be expected to increase during times of rising interest rates and decline during times of falling interest rates. Therefore, significant fluctuations in interest rates may have an adverse effect on Bay Commercial Bank's results of operations.

Bay Commercial Bank is limited in its ability to pay cash dividends.

Bay Commercial Bank has not declared any dividends in the past, and has no present plans to declare or pay dividends. There are no assurances that Bay Commercial Bank will be able to generate earnings in the future which would permit the declaration of dividends at such time, if ever. Furthermore, Bay Commercial Bank is subject to certain legal and regulatory restrictions on declaring cash dividends, and, even if legally allowed, the amount and timing of any dividends is at the discretion of Bay Commercial Bank's board of directors.

Bay Commercial Bank's future growth may be limited if it is not able to raise additional capital.

Banks are required to conform to regulatory capital adequacy guidelines and maintain their capital at specified percentages of their assets. These guidelines may limit Bay Commercial Bank's ability to grow and could result in banking regulators requiring increased capital levels or reduced loan and other earning asset levels. Therefore, in order to continue to increase its earning assets and net income, Bay Commercial Bank may, from time to time, need to raise additional capital. Bay Commercial Bank cannot assure you that additional sources of capital will be available or, if they are, that the additional capital will be available on economically reasonable terms.

Bay Commercial Bank plans to continue to grow and there are risks associated with growth.

Bay Commercial Bank intends to continue to expand its business and operations to increase deposits and loans. Continued growth may present operating and other problems that could adversely affect the combined business, financial condition and results of operations. Growth may place a strain on administrative, operational, personnel and financial resources and increase demands on systems and controls. The ability to manage growth successfully will depend on the ability to attract qualified personnel and maintain cost controls and asset quality while attracting additional loans and deposits on favorable terms, as well as on factors beyond its control, such as economic conditions and interest rate trends. If it grows too quickly and is not able to attract qualified personnel, control costs and maintain asset quality, this continued rapid growth could materially adversely affect Bay Commercial Bank's, as the combined entity, financial performance.

Global Trust Bank and Bay Commercial Bank compete against larger banks and other institutions.

Global Trust Bank and Bay Commercial Bank compete for loans and deposits with other banks, savings and thrift associations and credit unions located in their service areas, as well as with other financial services organizations such as brokerage firms, insurance companies and money market mutual funds. The combined institution will continue to face such competition. These competitors aggressively solicit customers within their market area by advertising through direct mail, the electronic media and other means. Many of their competitors have been in business longer, have established customer bases and are substantially larger. These competing financial institutions offer services, including international banking services, that Global Trust Bank and Bay Commercial Bank, now and following the merger, may only offer through correspondents, if at all. Additionally, their larger competitors have greater capital resources and, consequently, higher lending limits. Finally, some of their competitors are not subject to the same degree of regulation.

Current banking laws and regulations affect activities.

Global Bancorp, Global Trust Bank and Bay Commercial Bank are subject to extensive regulation. Supervision, regulation and examination of banks and bank holding companies by regulatory agencies are intended primarily to protect depositors rather than shareholders. Bay Commercial Bank will continue to be subject to extensive regulation. These regulatory agencies examine bank holding companies and commercial banks, establish capital and other financial requirements and approve acquisitions or other changes of control of financial institutions. Financial institutions' ability to establish new facilities or make acquisitions requires approvals from applicable regulatory bodies. Changes in legislation and regulatory changes may benefit financial institutions, others may increase their costs of doing business and indirectly assist their non-bank competitors who are not subject to similar regulation.

Recently enacted and potential further financial regulatory reforms could have a significant impact business, financial condition and results of operations.

On July 21, 2010, President Obama signed the Dodd-Frank Wall Street Reform and Consumer Protection Act (the "*Dodd-Frank Act*") into law. The Dodd-Frank Act is expected to have a broad impact on the financial services industry, including significant regulatory and compliance changes. Many of the requirements called for in the Dodd-Frank Act will be implemented over time and most will be subject to implementing regulations over the course of several years. Given

the uncertainty associated with the manner in which the provisions of the Dodd-Frank Act will be implemented by the various regulatory agencies and through regulations, the full extent of the impact such requirements will have on bank operations is unclear. The changes resulting from the Dodd-Frank Act may impact the profitability of business activities, require changes to certain business practices, impose more stringent capital, liquidity and leverage requirements or otherwise adversely affect the business of banking. In particular, the potential impact of the Dodd-Frank Act on operations and activities, both currently and prospectively, include, among others:

- a reduction in the ability to generate or originate revenue-producing assets as a result of compliance with increased capital standards;
- increased competition as a result of interstate de novo branching or acquisition of existing branches by out of state competitors;
- increased cost of operations due to greater regulatory oversight, supervision and examination of banks and bank holding companies, and higher deposit insurance premiums;
- the limitation on the ability to raise new capital through the use of trust preferred securities, as any new issuances of these securities will no longer be included as Tier 1 capital going forward; and,
- the limitation on the ability to expand consumer product and service offerings due to anticipated stricter consumer protection laws and regulations.

Further, banks may be required to invest significant management attention and resources to evaluate and make any changes necessary to comply with new statutory and regulatory requirements under the Dodd-Frank Act, which may negatively impact results of operations and financial condition.

Additionally, it cannot be predicted whether there will be additional proposed laws or reforms that would affect the U.S. financial system or financial institutions, whether or when such changes may be adopted, how such changes may be interpreted and enforced or how such changes may affect the bank business. However, the costs of complying with any additional laws or regulations could have a material adverse effect on banks' financial condition and results of operations.

Economic conditions either nationally or locally in areas in which the entities' operations are concentrated may adversely affect their business.

Deterioration in local, regional, national or global economic conditions could cause Global Trust Bank and/or Bay Commercial Bank to experience a reduction in deposits and new loans, an increase in the number of borrowers who default on their loans and a reduction in the value of the collateral securing their loans, all of which could adversely affect their performance and financial condition, and thus the performance and condition of the combined bank. Unlike larger banks that are more geographically diversified, they both provide banking and financial services locally, specifically, within the greater San Francisco Bay Area. Therefore, they both are particularly vulnerable to adverse local economic conditions and Bay Commercial Bank, as the resulting institution will continue to be vulnerable.

The financial condition and results of operations would be adversely affected if allowances for loan losses are not sufficient to absorb actual losses or if an increase to allowances is required.

Despite its underwriting criteria, a bank may experience loan delinquencies and losses. In order to absorb losses associated with nonperforming loans, a bank maintains an allowance for loan losses based on, among other things, historical experience, an evaluation of economic conditions, and regular reviews of delinquencies and loan portfolio quality. Determination of the allowance inherently involves a high degree of subjectivity and requires it to make significant estimates of current credit risks and future trends, all of which may undergo material changes. Either Global Trust Bank, prior to the completion of the merger, or Bay Commercial Bank, either before or after the merger, may be required to increase its allowance for loan losses for any of several reasons. Changes in economic conditions affecting borrowers, new information regarding existing loans, identification of additional problem loans and other factors, both within and outside of either of their control, may require an increase in allowances. In addition, actual charge-offs in future periods, if not adequately reserved for, will require additional increases in allowances for loan losses will result in a decrease in Global Trust Bank's or Bay Commercial Bank's respective net income and, possibly, respective capital, and may materially affect their respective results of operations in the period in which the allowance is increased.

The parties rely on their management and other key personnel, and the loss of any of them may adversely affect their operations.

Each of Global Bancorp, Global Trust Bank and Bay Commercial Bank is and will continue to be dependent upon the services of its respective executive management up until the consummation of the merger. In addition, assuming the mergers are consummated, Bay Commercial Bank will continue to depend on its ability to retain and recruit key banking officers. The unexpected loss of services of any key management personnel or banking officers could have an adverse effect on the resulting institution, Bay Commercial Bank, or the respective business and financial condition of Global Bancorp, Global Trust Bank or Bay Commercial Bank, because of their skills, knowledge of the market, years of industry experience and the difficulty of promptly finding qualified replacement personnel.

Failure to implement new technologies in operations may adversely affect growth or profits.

Advances in technology increasingly affect the market for financial services, including banking services and consumer finance services. The ability of Global Trust Bank and Bay Commercial Bank to compete successfully in these markets may depend on the extent to which they are able to exploit such technological changes. However, they can provide no assurance that they will be able to properly or timely anticipate or implement such technologies or properly train their respective staffs to use such technologies. Further, the added cost of technology for small banks such as Global Trust Bank and Bay Commercial Bank adversely affects their respective profitability. Thus, any failure to adapt to new technologies could adversely affect their respective business, financial condition or operating results prior to and, as to Bay Commercial Bank, following, the merger.

MARKETS; MARKET PRICES AND DIVIDENDS

Bay Commercial Bank Market Information and Dividends

Bay Commercial Bank has issued and outstanding one class of common stock, of which there were 3,549,794 shares outstanding, held by 120 shareholders of record as of March 31, 2011. None of Bay Commercial Bank's 10,000,000 shares of preferred stock have been issued. In connection with the issuance in 2010 of 2,000,000 shares of Bay Commercial Bank common stock at \$9.00 per share, which raised a total of \$16.83 million of additional capital (after expenses of the offering), Bay Commercial Bank issued warrants to purchase 159,817 shares of common stock to certain shareholders who acquired 1,598,140 shares of common stock in that offering. As of December 31, 2010, there were a total of 159,817 warrants to purchase common stock at \$9.00 per share issued and outstanding.

Trading in Bay Commercial Bank's common stock has not been extensive and such trades cannot be characterized as constituting an active trading market. While Bay Commercial Bank's common stock is not listed on any exchange or quoted by the NASDAQ® Stock Market, some transactions occur on the OTC Bulletin Board. Trades may also occur in unreported private transactions.

The following table sets forth the high and low sale prices for Bay Commercial Bank's common stock for the periods indicated, of which management of Bay Commercial Bank is aware:

Share

			Snare
	Sale Information		Volume
	High	Low	
2011			
Third Quarter (through July 11, 2011)	\$7.00	\$6.75	1,100
Second Quarter	\$7.65	\$6.75	105,200
First Quarter	\$7.90	\$6.99	33,700
2010			
Fourth Quarter	\$8.25	\$6.75	79,000
Third Quarter	\$9.73	\$8.00	20,900
Second Quarter	\$9.95	\$8.90	47,100
First Quarter	\$9.60	\$7.65	17,700

	Sale Information		Share Volume
	High	Low	
<u>2009</u>			
Fourth Quarter	\$8.70	\$7.75	36,700
Third Quarter	\$9.50	\$8.23	59,400
Second Quarter	\$9.70	\$6.85	16,900
First Quarter	\$9.00	\$6.80	83,600

a

To date, Bay Commercial Bank has not paid any cash dividends. Payment of stock or cash dividends in the future will depend upon Bay Commercial Bank's earnings and financial condition and other factors deemed relevant by the board of directors of Bay Commercial Bank, as well as Bay Commercial Bank's legal ability to pay dividends. It is Bay Commercial Bank's current intention to follow its strategic plan of retaining earnings to increase capital and provide additional basis for growth, as well as to buffer the effects of future uncertainty. Accordingly, no assurance can be given that any cash dividends will be declared in the foreseeable future.

Global Bancorp Market Information and Dividends

Global Bancorp equity securities consist of common stock \$.01 par value, of which there were 20,000,000 authorized and 2,115,001 shares issued and outstanding held by approximately 260 holders of record as of March 31, 2011.

There has not been any public trading in Global Bancorp's common stock. Global Bancorp's shares of common stock are not listed on any exchange or quoted by the NASDAQ® Stock Market or the OTC Bulletin Board.

To date, Global Bancorp has not paid any cash dividends and Global Bancorp is restricted from paying dividends through the date of the mergers. If the merger is consummated, payment of stock or cash dividends in the future to holders of Global Bancorp common stock exchanged for Bay Commercial Bank will depend upon Bay Commercial Bank earnings and financial condition and other factors deemed relevant by the board of directors of Bay Commercial Bank, as well as Bay Commercial Bank's legal ability to pay dividends.

SELECTED FINANCIAL DATA

Bay Commercial Bank

The following selected financial data of Bay Commercial Bank has been derived from and should be read in conjunction with Bay Commercial Bank's audited financial statements and notes included therein, attached to this joint proxy statement-offering circular.

The following tables set forth selected historical financial and other data about Bay Commercial Bank at the dates and for the periods shown. The historical financial data for the three months ended March 31, 2011 and March 31, 2010 are derived from unaudited financial statements. However, in the opinion of Bay Commercial Bank management, all adjustments consisting of normal recurring accruals necessary for a fair presentation since audited financial statements as of December 31, 2010 and 2009 and for such three-month unaudited periods have been made. Operating results for the three months ended March 31, 2011 are not necessarily indicative of the results that may be expected for any other interim period or for the entire year ended December 31, 2011. The financial information for the two years ended December 31, 2010 and 2009 of Bay Commercial Bank is based on, and qualified in its entirety by, Bay Commercial Bank financial statements, including the notes thereto. We did not pay any cash or stock dividends to our shareholders during the periods set forth below.

Selected Historical Financial Data (Dollars in thousands, except per share amounts)

	Three Months Ended March 31			Ended mber 31
	2011	2010	2010	2009
Summary of Operations				
Interest Income	\$ 2,195	\$ 2,019	\$ 8,127	\$ 7,225
Interest Expense	385	463	1,782	2,509
Net Interest Income	1,810	1,556	6,345	4,716
Provision for Loan Losses	933	45	1,431	430
Net Interest Income After Provision for Loan				
Losses	877	1,511	4,914	4,286
Non-Interest Income:				
Service Charges on Deposit Accounts	75	73	290	243
Other Non-Interest Income	81	12	75	56
Total Non-Interest Income	156	85	365	299
Non-Interest Expense:				
Salaries and Employee Benefits	774	691	2,551	2,409
Occupancy	154	158	623	498
Other Non-Interest Expense	282	254	1,104	1,048
Total Non-Interest Expense	1,210	1,103	4,278	3,955
Income (Loss) Before Income Taxes	(177)	493	1,001	630
Provision (Benefit) for Income Taxes	(71)	205	380	271
Net Income (Loss)	(106)	288	621	359
Net Income (Loss) Per Share – Basic	(0.03)	0.19	0.26	0.23
Net Income (Loss) Per Share – Diluted	(0.03)	0.19	0.26	0.23

	March 31		December 31	
	2011	2010	2010	2009
Summary of Financial Condition				
Federal Funds Sold	\$ 20,263	\$ 14,489	\$ 33,486	\$ 7,445
Total Loans, Net	135,071	122,953	133,517	129,973
Total Assets	162,889	143,228	173,977	143,408
Total Deposits	128,748	126,339	139,457	126,550
Total Shareholders' Equity	33,392	16,327	33,496	16,035
Book Value Per Share	9.41	10.53	9.44	10.35
Financial Ratios				
Total Risk-Based Capital Ratio	24.60%	13.5%	25.0%	12.6%
Net Charge-Offs/(Recoveries) to Average Net				
Loans	1.00%	0.00%	0.60%	0.00%
Non-Performing Assets to Assets	1.60%	0.00%	2.00%	1.50%
Allowance for Loan Losses to Loans	1.46%	1.40%	1.70%	1.30%
Total Net Loans to Deposits	104.90%	97.3%	95.70%	102.7%
Capital to Total Assets	20.50%	11.40%	19.30%	11.20%
Capital to Total Deposits	25.90%	12.90%	24.00%	12.70%
Allowance for Loan Losses to Non-Performing				
Loans	185.90%	0.00%	122.50%	75.9%
Financial Performance Ratios				
Return on Average Shareholders' Equity	-1.30%	7.20%	2.60%	2.30%
Return on Average Assets	-0.30%	0.80%	0.40%	0.30%
Average Shareholders' Equity to Average Total				
Assets	20.30%	11.50%	15.60%	11.60%

Global Bancorp

The following selected financial data with respect to Global Bancorp has been derived from the audited consolidated financial statements included in this joint proxy statement-offering circular. This information should be read in conjunction with such financial statements and the notes thereto included therein.

The following tables set forth selected consolidated historical financial and other data about Global Bancorp at the dates and for the periods shown. The historical consolidated financial data for the three months ended March 31, 2011 and 2010 are derived from unaudited consolidated financial statements. However, in the opinion of our management, all adjustments consisting of normal recurring accruals necessary for a fair presentation since audited consolidated financial statements as of December 31, 2010 and 2009 and for such three-month unaudited periods have been made. Consolidated operating results for the three months ended March 31, 2011 are not necessarily indicative of the results that may be expected for any other interim period or for the entire year ended December 31, 2011. The consolidated financial information for the two years ended December 31, 2010 and 2009 of Global Bancorp is based on, and qualified in its entirety by, Global Bancorp consolidated financial statements, including the notes thereto. Global Bancorp did not pay any cash or stock dividends during the periods set forth below.

	For the Thre	e Months Ended	For the Ye	ear Ended
	Ma	arch 31	Decem	ber 31
	2011	2010	2010	2009
Summary of Operations				
Interest Income	\$ 694	\$ 504	\$ 2,402	\$ 964
Interest Expense	155	148	681	297
Net Interest Income	539	356	1,721	667
Provision for Loan Losses	151	46	299	475
Net Interest After Provision	388	310	1,422	192
Non-Interest Income:				
Service Charges	3	1	17	2
Other	(12)	6	378	523
Total Non-Interest Income	(9)	7	395	525
Non-Interest Expense:				
Salaries and Employee Benefits	505	499	2,080	1,777
Occupancy	130	121	430	469
Other	263	198	873	845
Total Non-Interest Expense	898	818	3,383	3,092
Loss Before Income Taxes	(519)	(501)	(1,566)	(2,375)
Provision (Benefit) for Income Taxes	(1)		2	2
Net Loss	(518)	(501)	(1,568)	(2,377)
Net Loss Per Share – Basic	(0.25)	(0.24)	(0.74)	(1.12)
Net Loss Per Share – Diluted	(0.25)	(0.24)	(0.74)	(1.13)
Book Value Per Share	7.40	8.10	7.63	8.32
Shares outstanding at the end of the periods	2,115,001	2,115,001	2,115,001	2,115,001

Selected Historical Financial Data (Dollars in thousands, except per share amounts)

. ..

- . .

	For the Three Months Ended March 31		For the Yea Decemb	
	2011	2010	2010	2009
Average Balance Sheet Data				
Interest-Bearing Deposits with Financial				
Institutions	\$14,653	\$27,690	\$26,316	\$15,925
Securities				
Federal Funds Sold	9,350	680	1,439	2,869
Net Loans	44,749	24,370	28,302	11,361
Total Assets	72,984	58,447	60,907	33,738
Total Deposits	57,234	41,269	42,545	33,772
Total Shareholders' Equity	15,528	16,993	16,144	15,135
Book Value Per Share	7.40	8.10	7.63	8.32
Financial Ratios				
Total Risk-Based Capital Ratio	32.47%	52.77%	39.23%	52.43%
Net (Recoveries) Charge-Offs to Average Net				
Loans ⁽¹⁾	N/A	N/A	N/A	N/A
Non-Performing Assets to Assets ⁽¹⁾	N/A	N/A	N/A	N/A
Allowance for Loan Losses to Loans	2.07%	2.17%	2.12%	2.08%
Total Net Loans to Deposits	78.36%	59.24%	74.34%	68.87%
Capital to Total Assets	21.45%	29.30%	24.58%	34.08%
Capital to Total Deposits	27.41%	41.63%	32.79%	52.09%
Allowance for Loan Losses to Non-Performing				
Loans ⁽¹⁾	N/A	N/A	N/A	N/A
Financial Performance Ratios				
Return on Average Shareholders' Equity	-13.14%	-11.63%	-9.34%	-12.84%
Return on Average Assets	-3.00%	-3.64%	-2.59%	-6.85%
Average Shareholders' Equity to Average Total				
Assets	22.77%	31.30%	27.79%	53.43%
Net Interest Margin	3.32%	2.91%	3.13%	2.14%

 $\overline{^{(1)}}$ Global Trust Bank had no non-performing assets at the end of the periods.

COMPARATIVE PER SHARE DATA

We have summarized below historical per share information for Global Bancorp and Bay Commercial Bank and additional information as if the companies had been combined as of March 31, 2011 ("*pro forma*").

The pro forma information is based on the following assumptions, which have been used to calculate the applicable exchange ratio: (i) that no shareholder of Global Bancorp perfects dissenters' rights, and (ii) that about 80% of the total merger consideration will be paid in the form of Bay Commercial Bank common stock with an estimated fair value of \$9.58 per share and the remaining 20% will be paid in cash.

For purposes of the pro forma calculations, we have used the exchange ratio of 0.75678 of Bay Commercial Bank shares for each Global Bancorp common share outstanding.

You should read this information with the historical consolidated financial statements and related notes provided in this joint proxy statement-offering circular for Global Bancorp, Global Trust Bank and Bay Commercial Bank.

You should not rely on the pro forma information as being indicative of the historical results that we would have had or the future results that will occur after the merger. The equivalent pro forma data reflects the acquisition method of accounting and other events directly attributable to the transaction expected to have a continuing impact on Bay Commercial Bank.

The pro forma Bay Commercial Bank basic and diluted earnings per common share have been computed based on the historical number of average outstanding common shares of Bay Commercial Bank, plus the number of shares expected to be issued to Global Bancorp shareholders at about 80% of the total merger consideration, adjusted by the exchange ratio of 0.75678 shares of Bay Commercial Bank common stock for each share of Global Bancorp common stock.

The pro forma equivalent Global Bancorp basic and diluted earnings per share represent the pro forma Bay Commercial Bank basic and diluted earnings per share multiplied by the exchange ratio of 0.75678.

The pro forma Bay Commercial Bank book value per common share amounts are based upon the pro forma total shareholders' equity of Bay Commercial Bank at the balance sheet date, divided by the total pro forma number of Bay Commercial Bank common shares outstanding assuming the conversion of about 80% of Global Bancorp common stock into Bay Commercial Bank common stock at the exchange rate of 0.75678 per share.

The pro forma equivalent Global Bancorp book value per common share represents the pro forma Bay Commercial Bank book value per common share multiplied by the exchange ratio of 0.75678 per share.

	Three Months Ended March 31, 2011
Bay Commercial Bank Historical and Pro Forma	
Basic earnings per common share:	
Historical	(\$0.03)
Pro Forma	(\$0.03)
Diluted earnings per common share:	
Historical	(\$0.03)
Pro Forma	(\$0.03)
Dividends declared on common stock:	
Historical	N/A
Pro Forma	N/A
Book value per common share:	
Historical	\$9.41
Pro Forma	\$9.69
Tangible book value per common share:	
Historical	\$9.38
Pro Forma	\$9.62

	Three Months Ended March 31, 2011
Global Bancorp Historical and Pro Forma Equivalent	
Basic earnings per common share:	
Historical	(\$0.25)
Pro Forma	(\$0.19)
Diluted earnings per common share:	
Historical	(\$0.25)
Pro Forma	(\$0.19)
Dividends declared on common stock:	
Historical	N/A
Pro Forma	N/A
Book value per common share:	
Historical	\$7.40
Pro Forma	\$5.61
Tangible book value per common share:	
Historical	\$7.40
Pro Forma	\$5.61

THE GLOBAL BANCORP ANNUAL MEETING

General

Global Bancorp will hold its annual shareholders' meeting on August 18, 2011 at 5:30 p.m. (local time), at 700 E. El Camino Real, Suite 110, Mountain View, California 94040. At the annual shareholders' meeting you will be asked to consider and vote on the approval of the principal terms of the merger agreements, namely (i) the Downstream Merger Agreement, which relates to the merger of Global Bancorp with and into Global Trust Bank, (ii) the Merger Agreement, which relates to the merger of Global Trust Bank with and into Bay Commercial Bank; and (iii) the election of seven persons to the board of directors; and any other matters that may properly come before the meeting. Discussion of Global Bancorp's Proposal A regarding the election of directors and related information commences on page 96.

Record Date; Stock Entitled to Vote; Quorum

Only holders of record of Global Bancorp common stock at the close of business on June 30, 2011, the record date for Global Bancorp's annual shareholders' meeting, are entitled to receive notice of and to vote at the annual shareholders' meeting. On the record date, Global Bancorp had 2,115,001 shares of its common stock issued and outstanding. A majority of the shares of Global Bancorp common stock issued and outstanding must be represented in person or by proxy at the annual shareholders' meeting in order for a quorum to be present for purposes of transacting business. Abstentions and broker non-votes will be treated as shares present at the annual meeting for purposes of determining the presence of a quorum. A broker non-vote is an unvoted proxy submitted by a broker. In the event that a quorum is not present, it is expected that the annual shareholders' meeting will be adjourned or postponed to solicit additional proxies.

Number of Votes

Each shareholder is entitled to cast one vote for each share of common stock held in that shareholder's name on the books of Global Bancorp as of the record date on any matter submitted to the vote of the shareholders. On the matter of election of directors, votes may be cumulated as explained on page 95.

Votes Required

You may vote in person, or by submitting a properly executed proxy.

To approve the principal terms of the Merger Agreements (i.e., the Downstream Merger Agreement and the Merger Agreement), the holders of a majority of the outstanding shares of Global Bancorp common and preferred stock entitled to vote must vote in favor of the approval of the principal terms of the Merger Agreements. Consequently, a failure to vote, an abstention or a broker non-vote will have the same effect as voting against the Merger Agreements. Under applicable rules, brokers or other nominees who hold shares in street name for customers who are the beneficial owners of such shares may not vote those shares with respect to the Merger Agreements unless they have received specific instructions from their customers. As of the record date, Global Bancorp's directors and executive officers owned 210,000 shares of common stock eligible to vote, representing approximately 10% of Global Bancorp's issued and outstanding shares of common stock entitled to vote.

Adoption of a proposal to adjourn the annual meeting to a later date requires that a majority of the votes cast at the annual meeting must vote in favor of the proposal to adjourn the annual meeting. Accordingly, abstentions will have the same effect as voting against the proposal to adjourn the annual meeting. Broker non-votes will have no effect on this proposal.

Approval of the principal terms of the Merger Agreements by our shareholders is one of the conditions that must be satisfied to complete the mergers. *See* "The Mergers – Conditions to the Mergers."

Voting of Proxies

Submitting Proxies

Global Bancorp shareholders may vote their shares in person by attending the annual shareholders' meeting or they may vote their shares by proxy. In order to vote by proxy, Global Bancorp shareholders must complete the enclosed proxy card, sign and date it and mail it in the enclosed postage pre-paid envelope.

If a written proxy card is signed by a shareholder and returned without instructions, the shares represented by the proxy will be voted "FOR" approval of the mergers. Global Bancorp shareholders whose shares are held in "street name" (i.e., in the name of a broker, bank or other record holder) must either direct the record holder of their shares as to how to vote their shares or obtain a proxy from the record holder to vote at the Global Bancorp annual shareholders' meeting.

In connection with the election of directors, shares may be voted cumulatively if a shareholder present at the meeting gives notice at the meeting, prior to the voting for election of directors, of his or her intention to vote cumulatively. If any shareholder of Global Bancorp gives that notice, then all shareholders eligible to vote will be entitled to cumulate their shares in voting for election of directors. Cumulative voting allows a shareholder to cast a number of votes equal to the number of shares held in his or her name as of the record date, multiplied by the number of directors to be elected. These votes may be cast for any one nominee, or may be distributed among as many nominees as the shareholder sees fit. If cumulative voting is declared at the meeting, votes represented by proxies delivered pursuant to this proxy statement may be cumulated in the discretion of the proxy holders, in accordance with management's recommendation.

It is important that you follow the directions provided by your broker regarding instructions on how to vote your shares. Your failure to instruct your broker on how to vote your shares will have the same effect as voting against the proposal to approve the principal terms of the Merger Agreements and the mergers.

Revoking Proxies

Global Bancorp shareholders of record may revoke their proxies at any time before the time their proxies are voted at the Global Bancorp annual shareholders' meeting. Proxies may be revoked by written notice, including by telegram or telecopy, to the Corporate Secretary of Global Bancorp, by a later-dated proxy signed and returned by mail or by attending the annual shareholders' meeting and voting in person. Attendance at the annual shareholders' meeting will not, in and of itself, constitute a revocation of a proxy. Instead, Global Bancorp shareholders who wish to revoke their proxies must inform Global Bancorp's Corporate Secretary at the annual shareholders' meeting, prior to the vote, that he or she wants to revoke his or her proxy and vote in person. Written notices of proxy revocations must be sent so that they will be received before the taking of the vote at the annual shareholders' meeting as follows:

Global Bancorp 700 E. El Camino Real, Suite 110 Mountain View, California 94040 Attention: Corporate Secretary

Abstentions and broker non-votes will be counted in determining whether a quorum is present. Under the applicable rules of the Financial Industry Regulatory Authority, Inc. ("FINRA"), brokers or members who hold shares in

street name for customers who are the beneficial owners of Global Bancorp common or preferred stock are prohibited from giving a proxy to vote those shares regarding approval of the mergers and the principal terms of the merger agreements, in the absence of specific instructions from beneficial owners. These are referred to as "broker non-votes." Abstentions and broker non-votes will not be counted as a vote "FOR" or "AGAINST" the principal terms of the merger agreements and mergers at the annual shareholders' meeting or any other matter presented at that meeting. However, abstentions and broker non-votes will have the same effect as a vote "AGAINST" the principal terms of the merger agreements and mergers.

Solicitation of Proxies

Global Bancorp's board of directors is soliciting the proxies for the annual shareholders' meeting. Global Bancorp will pay for the cost of solicitation of proxies. In addition to solicitation by mail, Global Bancorp's directors, officers and employees may also solicit proxies from shareholders by telephone, facsimile, telegram, or in person. If management deems it advisable, the services of individuals or companies that are not regularly employed by Global Bancorp may be used in connection with the solicitation of proxies. Arrangements will also be made with brokerage houses and other custodians, nominees, and fiduciaries to send the proxy materials to beneficial owners. Global Bancorp will, upon request, reimburse those brokerage houses and custodians for their reasonable expenses in so doing.

Shareholders who submit proxy cards should <u>not</u> send in any stock certificates with their proxy cards. Instructions for the surrender of stock certificates representing shares of Global Bancorp will be mailed by the exchange agent selected for this transaction, to former Global Bancorp shareholders shortly after the mergers are completed. Please read the section entitled "JOINT PROPOSAL I – THE MERGER – The Merger Agreement – Exchange Procedures" for additional information.

If your shares of Global Bancorp common stock are held in "street name," you will receive instructions from your broker, bank, or other nominee that you must follow to have your shares voted. Please see your instruction form provided by your broker, bank or other nominee that accompanies this proxy statement.

Recommendation of the Global Bancorp Board of Directors

The Global Bancorp board of directors has approved the merger agreements and the transactions contemplated by the merger agreements. Based on their reasons for the mergers described in this document, the Global Bancorp board of directors believes that the mergers are in the best interests of Global Bancorp Bank and its shareholders. Accordingly, the board of directors recommends that the Global Bancorp shareholders vote (i) "FOR" approval of the principal terms of the merger agreements; (ii) "FOR" the election of directors of Global Bancorp as set forth in these proxy materials; and (iii) "FOR" the proposal to adjourn the annual meeting, if necessary, to solicit additional votes in favor of the principal terms of the merger agreements.

THE BAY COMMERCIAL BANK ANNUAL MEETING

General

Bay Commercial Bank will hold its annual shareholders' meeting on August 16, 2011 at 4:00 p.m. (local time), at its headquarters office, located at 1280 Civic Drive, Suite 100, Walnut Creek, California 94596. At the annual shareholders' meeting you will be asked to consider and vote on (i) the approval of the principal terms of the Merger Agreement; (ii) "**FOR**" if necessary, a proposal to adjourn the annual meeting to solicit additional votes in favor of the principal terms of the merger agreement; (iii) to elect seven persons to the board of directors; (iv) to ratify the board's selection of independent auditors, and any other matters that may properly come before the meeting.

Additional Information Regarding Bay Commercial Bank's Proposal A (Election of Directors) and Proposal B (Ratification of Independent Auditors for 2011) for the Bay Commercial Bank Annual Meeting commences on page 93.

Record Date; Stock Entitled to Vote; Quorum

Only holders of record of Bay Commercial Bank common stock at the close of business on June 17, 2011, the record date for the annual shareholders' meeting, are entitled to receive notice of and to vote at the annual shareholders' meeting. On the record date, Bay Commercial Bank had 3,549,794 shares of its common stock issued and outstanding. A majority of the shares of Bay Commercial Bank common stock issued and outstanding and entitled to vote on the record

date must be represented in person or by proxy at the annual shareholders' meeting in order for a quorum to be present for purposes of transacting business. In the event that a quorum is not present, it is expected that the annual shareholders' meeting will be adjourned or postponed to solicit additional proxies.

Number of Votes

Each shareholder of record as of the record date is entitled to cast one vote for each share of common stock held on each matter to come before the meeting. On the matter of the election of directors, votes may be cumulated as explained on page 87.

Votes Required

You may vote in person or by submitting a properly executed proxy.

To approve the principal terms of the Merger Agreement, the holders of a majority of the outstanding shares of Bay Commercial Bank's common stock must vote in favor of the approval of the principal terms of the Merger Agreement. Consequently, a failure to vote, an abstention or a broker non-vote will have the same effect as voting against the Merger Agreement. Under applicable rules, brokers or other nominees who hold shares in street name for customers who are the beneficial owners of such shares may not vote those shares with respect to the Merger Agreement unless they have received specific instructions from their customers. As of the record date, Bay Commercial Bank's directors and executive officers owned 474,236 shares, representing approximately 13.4%, of Bay Commercial Bank's issued and outstanding shares of common stock entitled to vote.

Approval of the principal terms of the Merger Agreement by our shareholders is one of the conditions that must be satisfied to complete the mergers. *See* "The Merger – Conditions to the Merger."

Voting of Proxies

Submitting Proxies

Bay Commercial Bank shareholders may vote their shares in person by attending the annual shareholders' meeting or they may vote their shares by proxy. In order to vote by proxy, shareholders must complete the enclosed proxy card, sign and date it and mail it in the enclosed postage pre-paid envelope.

If a written proxy card is signed by a shareholder and returned without instructions, the shares represented by the proxy will be voted (i) "**FOR**" approval of the principal terms of the Merger Agreement, (ii) "**FOR**" the election of the seven nominees to serve on the board of directors, (iii) "**FOR**" the ratification of the selection of Moss Adams LLP as the independent auditors for Bay Commercial Bank, and (iv) "**FOR**" the adoption of a proposal to adjourn the annual meeting to a later date, if necessary, to solicit further proxies if there are insufficient votes at the time of the annual meeting to approve the principal terms of the Merger Agreement. Shareholders whose shares are held in "street name" (i.e., in the name of a broker, bank or other record holder) must either direct the record holder of their shares as to how to vote their shares or obtain a proxy from the record holder to vote at the Bay Commercial Bank annual shareholders' meeting.

It is important that you follow the directions provided by your broker regarding instructions on how to vote your shares. Your failure to instruct your broker on how to vote your shares will have the same effect as voting against the proposal to approve the principal terms of the Merger Agreement and the merger.

Revoking Proxies

Bay Commercial Bank shareholders of record may revoke their proxies at any time before the time their proxies are voted at the annual shareholders' meeting. Proxies may be revoked by written notice, including by telegram or telecopy, to the Corporate Secretary of Bay Commercial Bank, by a later-dated proxy signed and returned by mail or by attending the annual shareholders' meeting and voting in person. Attendance at the annual shareholders' meeting will not, in and of itself, constitute a revocation of a proxy. Instead, shareholders who wish to revoke their proxies must inform Bay Commercial Bank's Corporate Secretary at the meeting, prior to the vote, that he or she wants to revoke his or her proxy and vote in person. Written notices of proxy revocations must be sent so that they will be received before the taking of the vote at the annual shareholders' meeting as follows:

Bay Commercial Bank 1280 Civic Drive, Suite 100 Walnut Creek, California 94596 Attention: Corporate Secretary

The presence, in person or by properly executed proxy, of the holders of a majority of Bay Commercial Bank's outstanding shares entitled to vote is necessary to constitute a quorum at the annual shareholders' meeting. Abstentions and broker non-votes will be counted in determining whether a quorum is present. Under the applicable rules of FINRA, brokers or members who hold shares in street name for customers who are the beneficial owners of Bay Commercial Bank common stock are prohibited from giving a proxy to vote those shares regarding approval of the merger and the principal terms of the Merger Agreement, or the election of directors, in the absence of specific instructions from beneficial owners. These are referred to as "broker non-votes." Abstentions and broker non-votes will not be counted as a vote "FOR" or "AGAINST" the matters presented at the annual shareholders' meeting. However, abstentions and broker non-votes will have the same effect as a vote "AGAINST" the principal terms of the Merger Agreement and merger.

Solicitation of Proxies

Bay Commercial Bank's board of directors is soliciting the proxies for the annual shareholders' meeting. Bay Commercial Bank will pay for the cost of solicitation of proxies. In addition to solicitation by mail, its directors, officers, and employees may also solicit proxies from shareholders by telephone, facsimile, telegram, or in person. If Bay Commercial Bank's management deems it advisable, the services of individuals or companies that are not regularly employed by it may be used in connection with the solicitation of proxies. Arrangements will also be made with brokerage houses and other custodians, nominees, and fiduciaries to send the proxy materials to beneficial owners. Bay Commercial Bank will, upon request, reimburse those brokerage houses and custodians for their reasonable expenses in so doing.

If your shares of Bay Commercial Bank common stock are held in "street name," you will receive instructions from your broker, bank, or other nominee that you must follow to have your shares voted. Please see your instruction form provided by your broker, bank or other nominee that accompanies this proxy statement.

Recommendation of the Bay Commercial Bank Board of Directors

The Bay Commercial Bank board of directors has unanimously approved the principal terms of the Merger Agreement and the transactions contemplated by the Merger Agreement. Based on Bay Commercial Bank's reasons for the merger described in this document, its board of directors believes that the merger is in the best interests of Bay Commercial Bank and its shareholders. Accordingly, the board of directors unanimously recommends that Bay Commercial Bank shareholders vote (i) "FOR" approval of the principal terms of the Merger Agreement; (ii) "FOR" the proposal to adjourn the annual meeting, if necessary, to solicit additional votes in favor of the principal terms of the Merger Agreement; (iii) "FOR" election of all seven of management's nominees: JAMES L. APPLE, JAMES S. CAMP, GEORGE J. GUARINI, LLOYD W. KENDALL, JR., ROBERT R. LAVERNE, DONALD S. MORROW and DAVID M. SPATZ; and (iv) "FOR" the ratification of the selection of Moss Adams LLP as the independent auditors of Bay Commercial Bank for 2011.

JOINT PROPOSAL I – THE MERGERS

This section describes certain aspects of the proposed mergers of Global Bancorp with and into Global Trust Bank and Global Trust Bank with and into Bay Commercial Bank. Because this is a summary, it does not contain all the information that may be important to you. You should read this entire joint proxy statement-offering circular, including the appendices. A copy of the Agreement and Plan of Reorganization and Merger dated May 12, 2011 ("Merger Agreement") is attached as <u>Appendix A</u> to this joint proxy statement-offering circular and is incorporated by reference herein. The following discussion, and the discussion under the subsection entitled "The Merger Agreement," describes important aspects of the merger and the material terms of the Merger Agreement. These descriptions are qualified in their entirety by reference to <u>Appendix A</u>.

General

As soon as possible after the conditions to consummation of the merger described below have been satisfied or waived, and unless the Merger Agreement has been terminated as discussed below, Global Bancorp will merge with and into Global Trust Bank and a subsequent merger will take place of Global Trust Bank with and into Bay Commercial Bank in accordance with California law.

Upon completion of the merger, each share of Global Bancorp common stock will be converted into the right to receive either (i) 0.75678 shares of Bay Commercial Bank common stock, or (ii) \$7.25 in cash or (iii) a combination of common stock and cash, subject to the amount of cash consideration paid in the merger not exceeding approximately \$3.067 million, including amounts paid to dissenting shareholders, and the number of shares of Bay Commercial Bank common stock issued in the merger transaction being not less than approximately 1.692 million. Cash proceeds payable to Global Bancorp shareholders will be paid promptly following the completion of the merger through a paying agent account.

The Constituent Corporations

Global Bancorp 700 E. El Camino Real, Suite 110 Mountain View, California 94040 (650) 810-9400

Global Bancorp is a bank holding company registered under the Bank Holding Company Act of 1956, as amended.

Global Trust Bank 700 E. El Camino Real, Suite 110 Mountain View, California 94040 (650) 810-9400

Global Trust Bank is a California corporation chartered as a commercial bank by the CDFI. Global Trust Bank commenced business on December 3, 2008 and is headquartered in Mountain View, California. All of Global Trust Bank's deposits are insured by the FDIC to the maximum extent permitted by law.

Global Trust Bank specializes in community-based financial services and operates one full-service branch located in Mountain View, California, from which it conducts a full range of banking services. Global Trust Bank has historically focused on developing banking relationships with small- and medium-sized businesses (generally, businesses whose borrowings range from \$300,000 to \$1.5 million), service firms and individuals in the Greater San Francisco Bay Area of Northern California and especially the Counties of Santa Clara, San Mateo and Alameda. Its banking services include commercial, real estate and consumer lending, a variety of deposit products.

As of March 31, 2011, Global Trust Bank had total assets of approximately \$73 million, including total loans of approximately \$46 million, as well as total deposits of approximately \$57.2 million and total shareholders' equity of approximately \$15.5 million. Additional information and financial statements related to Global Trust Bank are available from the FDIC's website, <u>www.fdic.gov</u>. However, information from the websites is not incorporated by reference into, and does not form a part of, this joint proxy statement-offering circular.

Bay Commercial Bank 1280 Civic Drive, Suite 100 Walnut Creek, California 94596 (925) 476-1800

Bay Commercial Bank is a California corporation chartered as a commercial bank by the CDFI. Bay Commercial Bank commenced business on July 10, 2004 and is headquartered in Walnut Creek, California. Bay Commercial Bank is a member of the FDIC and all of its deposits are insured by the FDIC to the maximum extent permitted by law. Also, Bay Commercial Bank has filed an application to become a member bank of the Federal Reserve System and, if approved, the FRB will become the primary federal regulator of Bay Commercial Bank, but its deposits will continue to be insured by the FDIC.

Bay Commercial Bank specializes in community-based financial services and operates from its headquarters in Walnut Creek and two full service branches located in Oakland and Castro Valley, California, from which Bay Commercial Bank conducts a full range of banking services, including escrow services and section 1031 tax-free exchange services. Bay Commercial Bank has historically focused on developing banking relationships with small and medium-sized businesses, service firms and individuals in the Greater San Francisco Bay Area and especially Contra Costa County and Alameda County.

As of March 31, 2011, Bay Commercial Bank had total assets of approximately \$163 million, including total loans of approximately \$137.3 million, as well as total deposits of approximately \$128.7 million and total shareholders' equity of approximately \$33.4 million. Additional information and financial statements related to Bay Commercial Bank are available from (i) its accounting department, and (ii) the FDIC's website, <u>www.fdic.gov</u>. However, information from such website is *not* incorporated by reference into, and does not form a part of, this joint proxy statement.

Background of the Mergers

Since the formation of Bay Commercial Bank in July 2004, its board of directors and executive management have considered various alternative growth strategies, including geographic expansions, combining with or acquiring other community banks, and participating in government-assisted acquisitions.

The president and chief executive officer of Bay Commercial Bank, George J. Guarini, has been acquainted with some of the directors of Global Trust Bank since that bank was in its early formation stages and he has maintained his acquaintanceship with those directors since Global Trust Bank opened in late 2008. Early in the first quarter of 2011, Mr. Guarini spoke informally with certain board members of Global Trust Bank and discussed the possibility of the two banks joining forces through a merger. He described the advantages for both banks and their stockholders that such a merger could achieve. The discussions centered on a share exchange based upon the respective book values of the two parties and on some representation by Global Trust Bank board members on the resulting entity's board of directors. The Global Trust Bank board members present at that discussion expressed interest in the idea, and expressed an interest in learning more about Bay Commercial Bank's merger plans.

Over the next several weeks, after executing a Confidentiality Agreement, the parties and their representatives reviewed various documents of their counterparty. At such time Bay Commercial Bank was also scheduled for its regular examination by its bank regulatory agencies, and the parties determined to continue discussions but not finalize negotiations until the examination was completed. Such examination was completed in early March, and the parties continued discussions about the proposed merger.

On March 28, 2011, Mr. Guarini presented to the board of directors of Bay Commercial Bank an analysis of a prospective merger with Global Trust Bank and preliminary due diligence findings. The Bay Commercial Bank board of directors discussed with management the prospective terms and effect of a possible merger with Global Trust Bank. After due discussion and consideration, the board of directors of Bay Commercial Bank authorized Mr. Guarini to proceed with further exploratory discussions about the possibility of a merger and to continue preliminary due diligence on Global Trust Bank and Global Bancorp.

Soon after the March 28, 2011 meeting, Mr. Guarini proposed to the board of directors of Global Trust Bank, a merger with Bay Commercial Bank, principally on the following terms:

- Global Bancorp and Global Trust Bank shall merge with and into Bay Commercial Bank with Bay Commercial Bank being the surviving corporation in the merger.
- Global Bancorp's common stock shall be exchanged for shares of Bay Commercial Bank stock. The exchange formula will be calculated based upon the adjusted book value per share of Bay Commercial Bank and Global Bancorp, taking into consideration a fully-funded allowance for loan and lease loss reserve.
- When the merger is effective, the board of directors of Bay Commercial Bank will be comprised of all of its current directors, plus a number of directors from the board of directors of Global Trust Bank, based upon the ownership of each bank as a percentage of the resulting bank.

Over the next few weeks, representatives of Bay Commercial Bank, Global Bancorp and Global Trust Bank continued negotiations over the terms of the merger, including, among other things, the basis for calculating the stock-for-stock exchange

ratio, the valuation of the banks, treatment of unexercised stock options held by officers and directors of the respective banks, employment and staffing issues and other financial and operational issues. The management of both banks continued due diligence on one another and sought to resolve the myriad issues pertaining to the prospective operation and management of Bay Commercial Bank after the merger.

On May 4, 2011, the board members of Global Trust Bank visited Bay Commercial Bank, meeting with several members of the board of directors and management of Bay Commercial Bank. The purpose of the meeting was for representatives from both parties to introduce themselves and to continue discussion of the terms of the merger and of post-merger operational and management issues.

After several weeks of negotiating, completing due diligence on one another's banks, evaluating each other's operations and financial condition, and considering other legal, accounting and regulatory issues, the Bay Commercial Bank board approved making a proposal for a merger exchange ratio based on the adjusted book values of Bay Commercial Bank and Global Bancorp. Such approach would result in the current shareholders of Bay Commercial Bank owning approximately 75% of Bay Commercial Bank after the merger, and the current shareholders of Global Bancorp and Global Trust Bank owing approximately 25% of Bay Commercial Bank after the merger, plus limitations on the amount of cash to be paid in the merger to Global Bancorp shareholders who elect to receive all or a portion of their consideration in cash and other terms as presented herein.

In early May 2011 representatives of Bay Commercial Bank proposed to Global Bancorp and Global Trust Bank revised terms of merger substantially the same as those which are included in the Merger Agreement and set forth in this joint proxy statement-offering circular. After such resolution, the parties concluded their respective due diligence and with advice of counsel completed negotiation and preparation of the definitive Merger Agreement.

The boards of directors of Bay Commercial Bank and Global Bancorp and Global Trust Bank approved, on May 10, 2011 and May 12, 2011, respectively, the Merger Agreement, including the disclosure schedules and the forms of the related agreements, and authorized their respective officers to execute the Merger Agreement and deliver it to the other party. On May 16, 2011, both banks issued a joint press release announcing the signing of the Merger Agreement and its terms.

Global Bancorp and Global Trust Bank Reasons for the Mergers

Global Bancorp, based in Mountain View, California, has conducted general banking operations serving individuals and small- to medium-sized businesses since December 3, 2008 through its wholly-owned banking subsidiary, Global Trust Bank, also headquartered in Mountain View. In serving individuals and small businesses, Global Trust Bank has focused on a community-based approach to banking.

The Global Bancorp and Global Trust Bank boards of directors reviewed the merger agreements and related documents, their strategic alternatives, the competitive banking environment in California, and the prospects for Global Trust Bank if it remained independent. At its meeting on May 12, 2011, The Findley Group discussed with Global Bancorp and Global Trust Bank boards of directors its analysis of the merger with Bay Commercial Bank and delivered to them its opinion that the consideration to be received in the merger was fair to the Global Bancorp shareholders from a financial point of view. Thereafter, the boards of directors approved and authorized the execution of the merger agreements.

The terms of the Merger Agreement, including the consideration to be paid to Global Bancorp shareholders, were the result of arm's-length negotiations. In evaluating the proposal to merge with Bay Commercial Bank, the boards of directors of Global Bancorp and of Global Trust Bank considered a number of factors at their board meetings between March 28, 2011 and May 12, 2011, including the following:

- information concerning the financial performance and condition, business operations, capital levels, asset quality, loan portfolio breakdown, and prospects of Bay Commercial Bank, including the stability of Bay Commercial Bank's management team and the upward performance trend;
- the terms of the Merger Agreement and other documents to be executed in connection with the mergers, including shareholders-director agreement and the Merger Agreement;

- the opinion of The Findley Group that the mergers are fair to the shareholders of Global Bancorp from a financial point of view;
- the prices paid and the terms of other recent comparable combinations of banks;
- the Global Bancorp and Global Trust Bank boards of directors' review with their legal and financial advisors of alternatives to the merger, the range of possible values to Global Bancorp shareholders obtainable through implementation of alternatives and the timing and likelihood of the same;
- the current and prospective economic environment and increasing regulatory and competitive burdens and constraints facing community banks;
- the advantages of being part of a larger entity, including the potential for operating efficiencies, the effect of a higher lending limit with respect to Global Trust Bank's customers, and the generally higher trading multiples of larger financial institutions;
- the business strategies, the strength and depth of management of the combined entity and the extent of their interest in continuing Global Trust Bank's business relationships;
- the ability of a larger institution to compete in the banking environment and to leverage overhead costs;
- the anticipated positive effect of the merger on existing shareholders, employees, officers, and customers of Global Bancorp and Global Trust Bank as Global Trust Bank's offices will remain open, staff elimination will be kept to a minimum, and remaining staff will receive a comparable benefits program;
- information concerning the ability of Global Trust Bank and Bay Commercial Bank to achieve operating efficiencies;
- the anticipated impact on the communities served by Global Trust Bank and Bay Commercial Bank in the merger, and the increased ability to serve the communities through the larger branch network;
- the consolidation that has occurred during the past few years in the banking industry and increased competition from larger banks in California;
- the value of the consideration offered by Bay Commercial Bank compared to the value of the consideration offered in other similar acquisitions of financial institutions in California in the last two years and the prospects for enhanced value of the combined entity in the future;
- the tax-free nature of the Bay Commercial Bank offer relating to the shares being offered;
- the quality of the Bay Commercial Bank common stock to be issued in the merger to the Global Bancorp shareholders and the liquidity of the Bay Commercial Bank common stock; and,
- the prospect for Global Bancorp and Global Trust Bank on a stand-alone basis and on the basis of alternative standalone strategies, such as dividends, share repurchases, restructurings, and growth through acquisitions.

In addition to the advantages, discussed in the previous paragraph, of a merger with a larger financial institution, the boards of directors and management of Global Bancorp and Global Trust Bank also discussed the various risks of combining with Bay Commercial Bank, including:

- the disadvantages of being part of a larger entity, including the potential for decreased customer service;
- the loss of California's only independently-owned Indo-American bank; and,
- the integration of Global Trust Bank and Bay Commercial Bank may divert the combined entities' management from other activities.

However, after weighing the advantages and disadvantages of a merger with Bay Commercial Bank, the Global Bancorp, and Global Trust Bank boards of directors determined that the advantages clearly outweighed the disadvantages. For example:

- the substantially larger lending limits of the combined entity will better serve customers and prospective customers of Global Trust Bank;
- the prospects of the combined entity are substantially greater than the prospects of Global Bancorp and Global Trust Bank on a stand alone basis; and,
- the likelihood that Global Bancorp and Global Trust Bank's financial performance would continue to be challenged by competition in its marketplace and the economic climate.

The foregoing discussion of the information and factors considered by the Global Bancorp and Global Trust Bank boards of directors is not intended to be exhaustive, but constitutes the material factors considered by the boards of directors. In reaching their determination to approve and recommend the principal terms of the merger, the boards of directors did not assign relative or specific weights to the foregoing factors and individual directors may have weighed such factors differently.

For reasons set forth above, the Global Bancorp and Global Trust Bank boards of directors have approved the merger agreements as being in the best interest of Global Bancorp and its shareholders and Global Trust Bank and recommend that the Global Bancorp shareholders approve the principal terms of the mergers.

Bay Commercial Bank Reasons for the Merger

The board of directors of Bay Commercial Bank considered the Merger Agreement and, on May 10, 2011, determined it to be fair, and in the best interests of Bay Commercial Bank and its shareholders, customers, and employees. In reaching its determination, the Bay Commercial Bank board of directors consulted with management, as well as its financial and legal advisors regarding the financial fairness of the consideration to be provided to shareholders of Global Bancorp in connection with the merger.

The Bay Commercial Bank board of directors considered a number of factors, at its meetings and in informal discussions between February 2011 and May 2011. The following are the material factors considered by the board, to which it did not assign any particular weights:

- Bay Commercial Bank's strategic alternatives, the competitive banking environment in California, and the prospects for Bay Commercial Bank if it remained independent without completing the transaction with Global Bancorp and Global Trust Bank;
- The Bay Commercial Bank board of directors' review of Global Trust Bank's business, operations, branch network, financial condition, and earnings on a historical and prospective basis;
- The Bay Commercial Bank board of directors' knowledge and review, based in part on presentations by Bay Commercial Bank's management, of: (i) the business, operations, financial condition, and earnings of Global Bancorp and Global Trust Bank on an historical and prospective basis and of the combined company on a pro forma basis; and (ii) the potential for increased earnings for Bay Commercial Bank shareholders as shareholders of the combined company;
- The stronger capital position of Bay Commercial Bank resulting from the merger with Global Bancorp;
- The terms of the Merger Agreement and general structure of the merger;
- The exchange ratio in cash and stock to be paid to Global Bancorp shareholders in relation to the market value, book value and earnings per share of Global Bancorp common stock;
- The exchange ratio in cash and stock to be paid to Global Bancorp shareholders in relation to the market value, book value and earnings per share of Bay Commercial Bank common stock;

- The general impact that the merger could be expected to have on the constituencies serviced by Bay Commercial Bank, including its customers, employees, and communities, and the skills of the employees of Global Trust Bank that it would intend to retain;
- The added convenience of access to Global Trust Bank's branch system, and the fact that the branch networks of the two banks are complementary and would enhance Bay Commercial Bank's presence in greater San Francisco Bay Area;
- The expectation that the merger will be tax-free for federal income tax purposes for Bay Commercial Bank;
- The business philosophies of Bay Commercial Bank and Global Bancorp;
- The prospect of higher lending limits as a result of the increased size of the combined institution, enhancing the competitiveness of Bay Commercial Bank; and,
- The prospect of financial strength and operating efficiencies of a combined institution.

The board of directors of Bay Commercial Bank also discussed the possibility that the integration of Global Trust Bank and Bay Commercial Bank might divert the combined entities' management from other activities, might not result in the operating efficiencies contemplated, and might result in the loss of key employees that Bay Commercial Bank would prefer to retain.

For reasons set forth above, the Bay Commercial Bank board of directors has unanimously approved the Merger Agreement as being in the best interest of Bay Commercial Bank and its shareholders and unanimously recommends that the Bay Commercial Bank shareholders approve the principal terms of the merger.

Structure of the Merger

The Merger Agreement provides that (i) Global Bancorp will merge with and into Global Trust Bank and, immediately thereafter, (ii) Global Trust Bank will merge with and into Bay Commercial Bank. As a result of the merger, Bay Commercial Bank will be the surviving bank and will operate under the name "Bay Commercial Bank." Each share of Global Bancorp common stock issued and outstanding, other than shares with respect to which dissenters' rights have been perfected, will be converted into either (i) 0.75678 shares of Bay Commercial Bank's common stock (ii) \$7.25 in cash or (iii) a combination of the two, at the election of the holder of such share of Global Bancorp common stock, provided that the resulting mix of consideration is such that at least approximately 80% of the shares of Global Bancorp common stock is exchanged for cash, subject to the prior right of dissenting shareholders to receive cash for their shares. Each share of Bay Commercial Bank's common stock outstanding will remain outstanding after the merger. Please read the sections entitled "JOINT PROPOSAL I – THE MERGERS – Calculation of Consideration to be Paid to Global Bancorp Shareholders," "– Dissenters' Rights of Global Bancorp's Shareholders" for additional information.

Calculation of Consideration to be Paid to Global Bancorp Shareholders

When the merger is completed, Global Bancorp shareholders will have the right to exchange each share of Global Bancorp common stock they own for either (i) 0.75678 shares of Bay Commercial Bank's common stock, (ii) \$7.25 in cash or (iii) a combination of the two, at the election of the holder of such shares of Global Bancorp common stock, provided that the resulting mix of consideration is such that at least approximately 80% of the shares of Global Bancorp common stock are converted into Bay Commercial Bank's common stock and the remainder of the shares of Global Bancorp common stock are exchanged for cash, subject to the prior right of dissenting shareholders to receive cash for their shares. The limitation on the amount of cash consideration in the merger means that no more than approximately 20% of the outstanding shares of Global Bancorp will be exchanged for cash.

Changes to Common Stock Prior to the Effective Date

Both Bay Commercial Bank and Global Bancorp have agreed that during the period of time between the date of the Merger Agreement and the date on which the mergers become effective, neither will (a) declare or pay any dividend or make any other distribution in respect of any of its capital stock; (b) split, combine or reclassify any of its capital stock or

issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock; or (c) repurchase or otherwise acquire any shares of its capital stock.

Fractional Shares

No fractional shares of Bay Commercial Bank's common stock shall be issued in the merger and, in lieu thereof, each holder of Global Bancorp common stock who would otherwise be entitled to receive a fractional share shall receive an amount in cash equal to the product (calculated to eight places) obtained by multiplying such fractional share interest by \$7.25.

Procedures for Surrendering your Stock Certificate(s) and Making an Election of Consideration

In order to make a valid election as to receive cash or Bay Commercial Bank stock in exchange for Global Bancorp stock, a Global Bancorp shareholder must complete a form transmittal letter that will be mailed to the exchange agent appointed by the parties in connection with this transaction at least by 5:00 p.m. (PST) on the 30th day following the day on which the election form and other appropriate and customary transmittal materials are mailed to each record holder of Global Bancorp common stock as of date the mergers become effective. Such transmittal letter will allow holders of Global Bancorp common stock to select either shares of Bay Commercial Bank common stock, cash or a combination of the foregoing. **PLEASE RETAIN THIS JOINT PROXY STATEMENT-OFFERING CIRCULAR, SINCE IT WILL BE OF ASSISTANCE IN MAKING YOUR ELECTION.** If you do not make a valid and timely election, you will receive whatever form of consideration (Bay Commercial Bank common shares or cash) as may be necessary to satisfy the proration provisions discussed below.

A valid election will be properly made and effective only if the exchange agent actually receives a properly completed letter of transmittal by 5:00 p.m. (PST) on or before the 30th day after the letter of transmittal is first mailed. A letter of transmittal will be deemed properly completed only if an election is indicated for each share of Global Bancorp common stock and accompanied by one or more certificates, or customary affidavits and indemnity for lost certificates, representing all shares of Global Bancorp common stock covered by such letter of transmittal. An election may be revoked or changed at any time prior to the election deadline.

GLOBAL BANCORP SHAREHOLDERS SHOULD NOT SEND IN THEIR STOCK CERTIFICATES UNTIL THEY RECEIVE THE ELECTION FORMS AND INSTRUCTIONS FROM THE EXCHANGE AGENT.

If, after taking into account all valid elections, the percentage of Global Bancorp shares of stock being exchanged for Bay Commercial Bank stock or the percentage of Global Bancorp shares of stock being exchanged for cash, do not meet the amounts required by the Merger Agreement, then the exchange agent will deliver the form of consideration (Bay Commercial Bank stock or cash) that is necessary to achieve the requirements of the Merger Agreement, first to shareholders who failed to make a valid election and then, if necessary, to Global Bancorp shareholders who validly elected the other form of consideration. As a result, Global Bancorp shareholders that make a valid election could be subject to a proration process which will result in the holder receiving a different mix of consideration than originally requested. Decisions as to the allocation of consideration will be made in accordance with California Corporations Code section 1101(e).

For details on the proration provisions, please refer to the Merger Agreement, attached as <u>Appendix A</u> to this joint proxy statement-offering circular.

Rights of Holders of Global Bancorp Stock Certificates Prior to Surrender

From the effective date of the merger until each Global Bancorp shareholder surrenders his or her stock certificates, he or she will not be paid dividends or other distributions declared or payable to holders of record of Bay Commercial Bank common stock as of any time subsequent to the effective date. Each Global Bancorp shareholder's rights to dividends or other distributions will be held for the benefit of the shareholder until he or she submits his or her stock certificates. No interest will be paid on dividends or distributions on the Bay Commercial Bank common stock held for the benefit of the shareholder.

Conversion of Global Bancorp Common Stock

Following the effective date of the merger, Bay Commercial Bank will deliver, or cause to be delivered, a letter to the Global Bancorp stockholders, with instructions to tender their Global Bancorp stock certificates to the exchange agent. Such letter will specify that risk of loss and title to the Global Bancorp stock certificates shall pass only upon acceptance of such stock certificates by Bay Commercial Bank or the exchange agent.

Certificates representing shares of Global Bancorp common stock should not be returned at this time. You should return your certificates when you receive transmittal materials from Bay Commercial Bank or the exchange agent. Upon surrender of certificates representing shares of Global Bancorp common stock registered in your name, together with a properly completed letter of transmittal, the exchange agent will mail to you the Bay Commercial Bank common stock to which you are entitled.

Lost Certificates

If your stock certificate has been lost, stolen, or destroyed, before you can receive the merger consideration for the shares represented by such certificate, you must submit an affidavit that the certificate has been lost, stolen, or destroyed and, if required, post a reasonable bond as indemnity against any claim that may be made against Bay Commercial Bank with respect to such certificate.

Material United States Federal Income Tax Consequences

General. The following discussion is a summary of the material United States federal income tax consequences to Global Bancorp shareholders who exchange their stock for Bay Commercial Bank common stock pursuant to the merger. This discussion is based on the Internal Revenue Code of 1986, as amended, United States Treasury Regulations promulgated under the Internal Revenue Code, administrative rulings and pronouncements and judicial decisions as of the date hereof, all of which are subject to change, possibly with retroactive effect. Any such change could alter the tax consequences discussed in this joint proxy statement-offering circular.

As used in this section, an "Global Bancorp shareholder" is a citizen or resident of the United States; a corporation (or other entity treated as a corporation for United States federal income tax purposes) organized under the laws of the United States or any State or the District of Columbia; an estate the income of which is subject to United States federal income taxation regardless of its source; or a trust if (i) a court within the United States is able to exercise primary supervision over the administration of the trust, and (ii) one or more United States persons have the authority to control all substantial decisions of the trust.

This discussion does not address the effects of any state, local, or non-United States tax laws. This discussion does not discuss the tax consequences of transactions effectuated prior or subsequent to, or concurrently with, the merger, whether or not in connection with the merger. Furthermore, this discussion relates only to Global Bancorp shareholders who hold Global Bancorp common stock, and will hold Bay Commercial Bank common stock, as capital assets. The tax treatment of a Global Bancorp shareholder may vary depending upon such shareholder's particular situation, and certain shareholders may be subject to special rules not discussed below. Such shareholders would include, for example, insurance companies, tax-exempt organizations, financial institutions, investment companies, broker-dealers, domestic shareholders whose "functional" currency is not the United States dollar, shareholders who hold Global Bancorp stock as part of a hedge, straddle, constructive sale or conversion transaction, and individuals who receive Bay Commercial Bank stock pursuant to the exercise of employee stock options or otherwise as compensation.

You are strongly urged to consult with your tax advisor with respect to the tax consequences to you of the merger in light of your own particular circumstances, including tax consequences under state, local, foreign and other tax laws and the possible effects of changes in United States federal or other tax laws.

Tax Consequences of the Merger. In connection with the merger, the independent accounting firm of Vavrinek, Trine, Day & Co., LLP will deliver its opinion prior to the effective time of the merger that the merger will constitute a "reorganization" for United States federal income tax purposes within the meaning of section 368(a) of the Internal Revenue Code. Such opinion will be subject to certain assumptions, limitations and qualifications and will be based on the truth and accuracy of certain customary factual representations of Bay Commercial Bank and Global Bancorp. Assuming the merger does qualify as a reorganization within the meaning of section 368(a) of the Internal Revenue Code, then, subject to the limitations and qualifications referred to herein, the following material United States federal income

tax consequences will result from qualification of the merger as a "reorganization" within the meaning of section 368(a) of the Internal Revenue Code:

- Global Bancorp Shareholders Who Receive Only Bay Commercial Bank Shares. If you receive solely Bay Commercial Bank shares, then you will not recognize gain or loss for U.S. federal income tax purposes in the merger, except that any Global Bancorp shareholder who receives cash proceeds in lieu of a fractional Bay Commercial Bank share will recognize capital gain or capital loss equal to the difference between such proceeds and the tax basis allocated to the fractional share. The tax basis of the Bay Commercial Bank shares (including any fractional shares deemed received and exchanged for a cash payment) received by you in exchange for your Global Bancorp common stock will be the same as your tax basis in your Global Bancorp common stock. Your holding period in the Bay Commercial Bank shares (including any fractional shares deemed received by you will include your holding period in your Global Bancorp common stock. Your capital gain or loss on cash proceeds received by you in lieu of a fractional Bank share will be long-term capital gain or loss if you have held your shares of Global Bancorp common stock for more than one year at the effective time of the merger.
- <u>Global Bancorp Shareholders Who Receive Only Cash</u>. If you exchange all of your shares of Global Bancorp common stock for cash in the merger, you will recognize capital gain or capital loss for U.S. federal income tax purposes to the extent the amount of cash received by you in the merger (calculated on a share by share basis) exceeds or is less than your tax basis in your Global Bancorp common stock. Your capital gain or loss will be long-term capital gain or loss if you have held your shares of Global Bancorp common stock for more than one year at the effective time of the merger. Long-term capital gain of a non-corporate U.S. shareholder generally qualifies for a maximum regular U.S. federal income tax rate of 15%.
- Global Bancorp Shareholders Who Receive Both Bay Commercial Bank Shares and Cash. If you receive both Bay Commercial Bank shares and cash in exchange for your Global Bancorp common stock, you will recognize gain, but not loss, for U.S. federal income tax purposes (calculated on a share by share basis) in an amount equal to the lesser of (1) the amount of cash received by you in merger or (2) an amount equal to the excess, if any, of (a) the sum of the amount of cash plus the fair market value of the Bay Commercial Bank shares received by you in the merger, over (b) your tax basis in your Global Bancorp common stock. (The preceding sentence does not apply to any cash you receive in lieu of fractional Bay Commercial Bank shares, the tax consequences of which are discussed above under the subheading "Global Bancorp Shareholders Who Receive Only Bay Commercial Bank Shares.") Your recognized gain will be capital gain unless your receipt of cash has the effect of a distribution of a dividend, in which case your gain will be treated as ordinary dividend income to the extent of your ratable share of Global Bancorp's accumulated earnings and profits as calculated for U.S. federal income tax purposes. For purposes of determining whether your receipt of cash has the effect of a distribution of a dividend, you will be treated as if you first exchanged all of your Global Bancorp common stock solely for Bay Commercial Bank and then Bay Commercial Bank immediately redeemed a portion of the shares for the cash that you actually received in the merger. The IRS has indicated in rulings that any reduction in the interest of a minority shareholder that owns a small number of shares in a publicly and widely held corporation and that exercises no control over corporate affairs would receive capital gain (as opposed to dividend) treatment. In determining whether your receipt of cash has the effect of a distribution of a dividend, certain constructive ownership rules must be taken into account. Your capital gain will be long-term capital gain if your holding period for your Global Bancorp common stock is more than one year. Long-term capital gain and certain dividend income of a non-corporate U.S. shareholder generally qualify for a maximum regular U.S. federal income tax rate of 15%.
- Your aggregate tax basis in the Bay Commercial Bank shares received by you in the merger will equal your aggregate tax basis in your Global Bancorp common stock, (1) reduced by (a) the portion of your tax basis in your Global Bancorp common stock that is allocable to a fractional share of Bay Commercial Bank common stock for which cash is received and (b) the amount of cash received by you in the merger, and (2) increased by the amount of gain (including any portion of such gain that is treated as a dividend as described above), if any, recognized by you in the merger (other than any gain recognized upon your receipt of cash in lieu of a fractional Bay Commercial Bank share). Your holding period for the Bay Commercial Bank shares received by you in the merger will include your holding period for your Global Bancorp common stock.

The foregoing discussion is not intended to be a complete analysis or description of all potential federal income tax consequences of the merger. In addition, the discussion does not address tax consequences which may vary with, or are contingent on, your individual circumstances. Moreover, the discussion does not address any non-income tax or any foreign, state or local tax consequences of the merger. Accordingly, you are strongly urged to consult with your tax

advisor to determine the particular federal, state, local, or foreign income or other tax consequences to you of the merger.

If a Global Bancorp shareholder receives cash pursuant to the exercise of dissenters' rights, that shareholder generally will recognize gain or loss measured by the difference between the cash received and the adjusted tax basis in the shareholder's Global Bancorp common stock. This gain will be long-term capital gain or loss if the shareholder's holding period in the Global Bancorp stock is greater than one year. In certain circumstances, you can be deemed for tax purposes to own shares that are actually owned by a non-dissenter who is related to you, or to own shares of Bay Commercial Bank common stock, with the possible result that the cash received upon the exercise of your rights could be treated as a dividend received in a corporate distribution rather than as an amount received in a sale or exchange of Global Bancorp shareholder that plans to exercise dissenters' rights in connection with the merger is urged to consult a tax advisor to determine the related tax consequences.

Treatment of Bay Commercial Bank, Global Bancorp, and Global Trust Bank. No gain or loss will be recognized by Bay Commercial Bank, Global Bancorp and Global Trust Bank as a result of the merger.

Neither Bay Commercial Bank nor Global Bancorp will request a ruling from the Internal Revenue Service regarding the tax consequences of the merger to Global Bancorp Bank shareholders. The tax opinion provided by Vavrinek, Trine, Day & Co., LLP does not bind the Internal Revenue Service and does not prevent the Internal Revenue Service from successfully asserting a contrary opinion.

A successful Internal Revenue Service challenge to the reorganization status of the merger would result in each Global Bancorp shareholder recognizing taxable gain or loss equal to the difference between the sum of the fair market value of the Global Bancorp stock, as of the closing date of the merger, and the amount of cash received in the merger (including cash received in lieu of fractional shares of Bay Commercial Bank common stock) and the shareholder's tax basis in Global Bancorp stock surrendered in exchange thereof. Bay Commercial Bank common stock so received would equal its fair market value as of the closing date of the merger, and the shareholder's holding period for such stock would begin the day after the merger. Further, if the Internal Revenue Service successfully challenged the reorganization status of the merger, Global Bancorp would be subject to tax on the merger. In addition, if any of the representations or assumptions upon which the tax opinion provided by Vavrinek, Trine, Day & Co., LLP is based are inconsistent with the actual facts, the tax consequences of the merger could be adversely affected.

Backup Withholding. Any cash payments to Global Bancorp shareholders in connection with the merger may be subject to backup withholding at a rate of 28% on a shareholder's receipt of cash, unless such shareholder furnishes a correct taxpayer identification number and certifies that he or she is not subject to backup withholding. Any amount withheld under the backup withholding rules will generally be allowed as a refund or credit against the shareholder's U.S. federal income tax liability, provided the required information is furnished to the Internal Revenue Service.

HOLDERS OF SHARES OF GLOBAL BANCORP COMMON STOCK ARE URGED TO CONSULT THEIR TAX ADVISORS AS TO THE UNITED STATES FEDERAL INCOME TAX CONSEQUENCES OF THE MERGER, AS WELL AS THE EFFECTS OF STATE, LOCAL, AND FOREIGN TAX LAWS.

Regulatory Approvals

Merger of Global Bancorp with and into Global Trust Bank: FDIC

Because the downstream merger is exempt from the California Financial Code governing banks, this merger is governed by the California General Corporations Code which requires approvals by the boards of directors and a majority of the shareholders of Global Bancorp and Global Trust Bank. In addition, the downstream must be approved by the FDIC. The FDIC has the authority to deny an application for approval if it concludes that the combined organization would have an inadequate capital structure, taking into account, among other factors, the nature of the business and operations and plans for expansion.

Global Bancorp submitted its application for FDIC approval of the downstream merger on June 16, 2011, and the FDIC is processing the application in due course.

Merger of Global Trust Bank with and into Bay Commercial Bank: FRB

The application of Bay Commercial Bank for membership in the Federal Reserve System is being processed in due course. If such pending application is approved, the FRB would become the primary federal regulator of Bay Commercial Bank in lieu of the FDIC and Bay Commercial Bank will file its application for approval of the merger with the FRB. Notwithstanding such change in Bay Commercial Bank's primary federal regulator, the deposits of Bay Commercial Bank will continue to be insured by the FDIC to the maximum extent permitted by law.

Merger of Global Trust Bank with and into Bay Commercial Bank: CDFI

Since the survivor of the merger of Bay Commercial Bank and Global Trust Bank will be a California-chartered bank, the approval of the CDFI (along with Bay Commercial Bank's primary federal regulator, as stated above) is required. In determining whether to approve such merger, the CDFI evaluates the application to determine, among other things, that:

- the merger will not result in a monopoly;
- the merger will not have the effect of substantially lessening competition;
- the shareholders' equity of the resulting bank will be adequate and will not jeopardize that bank's financial condition;
- the directors and executive officers of the surviving bank after the merger will be competent;
- the merger will afford a reasonable promise of successful operation and that the resulting bank will be operated in a safe and sound manner, in compliance with all applicable laws; and,
- the merger will be fair, just and equitable.

The application to the CDFI for approval of the merger was received by the CDFI on June 15, 2011 and is being processed in due course. While the CDFI does not review and thus does not approve or disapprove the downstream merger, on June 21, 2011, a request was submitted to the CDFI for an order of exemption for Global Trust Bank to issue its shares to Global Bancorp shareholders in the downstream merger covering a technical aspect of the overall transaction, which request is being processed in due course.

Statutory Waiting Period

Under federal banking laws, a 30-day waiting period must expire following the primary federal regulator's approval of the merger. Within that 30-day waiting period the Department of Justice may file objections to the merger under federal antitrust laws. The agencies may reduce the waiting period to 15 days with the concurrence of the Department of Justice. The Department of Justice could take such action under antitrust laws as it deems necessary or desirable in the public interest, including seeking to enjoin the merger unless divestiture of an acceptable number of branches to a competitively suitable purchaser can be made. If the Department of Justice commences an action challenging the merger on antitrust grounds during either the 30-day or 15-day waiting periods, commencement of that action would stay the effectiveness of the regulatory approvals, unless a court of competent jurisdiction specifically orders otherwise.

The merger cannot proceed in the absence of the regulatory approvals and the expiration of the statutory waiting period. Global Bancorp and Bay Commercial Bank are not aware of any reasons why regulatory approvals will not be received. Global Bancorp and Bay Commercial Bank have agreed to use their reasonable best efforts to obtain all necessary regulatory approvals. *However, there can be no assurance that approvals will be obtained, nor can there be assurance as to the date of any approval. There also can be no assurance that any approvals will not contain unacceptable conditions or requirements.*

Regulatory Matters

Subsequent to a joint examination by the FDIC and the CDFI in 2010, Global Trust Bank agreed to a Board Resolution ("*Resolution*") as requested by the FDIC and CDFI. The Resolution is a board resolution and agreement whereby Global Trust Bank agreed to undertake to take certain corrective actions with respect to its operations. The Resolution requires Global Trust Bank to: (1) maintain qualified management including a chief executive officer, chief financial officer and senior lending officer acceptable to the FDIC and CDFI; (2) notify the FDIC and CDFI in writing when it proposes to add any individual to the Global Trust Bank board of directors, or as a senior executive officer; (3) design, adopt and implement a comprehensive three-year strategic plan and a management succession plan; (4) assess and document the stability of its large depositor concentrations and develop a liquidity contingency plan to replace the loss of large depositors; and (5) shall provide quarterly progress reports to the FDIC and CDFI to secure compliance with the Resolution.

Before entering into the Resolution and since entering into it, Global Trust Bank took and has taken corrective actions that were designed to satisfy the requirements of the Resolution, including filling the position of chief executive officer by the hiring of Mr. W. Robert Sweeney as an independent contractor to be the acting chief executive officer of Global Trust Bank. Mr. Sweeney terminated employment with Global Trust Bank on June 30, 2011, and the board of directors will be providing closer oversight of Global Trust Bank until the closing of the merger, or if the merger is not consummated Global Trust Bank will search for a new chief executive officer. Compliance with the terms of the Resolution is determined by the FDIC and the CDFI during subsequent examinations.

Who Pays For Expenses and Costs of the Merger

All out-of-pocket costs and expenses incurred in connection with the merger (including, but not limited to, counsel fees) shall be paid by the party incurring such costs and expenses.

Certain Effects of the Merger

The Merger Agreement requires Global Bancorp to merge with and into Global Trust Bank and Global Trust Bank to merge with and into Bay Commercial Bank, with Bay Commercial Bank as the only surviving entity. After the mergers, Bay Commercial Bank will continue to have its headquarters at 1280 Civic Drive, Suite 100, Walnut Creek, California 94596. Bay Commercial Bank will continue to operate with its present directors and executive officers; however, three of Global Bancorp's directors will be invited to join the board of directors of Bay Commercial Bank, at the effective time of the merger.

After the merger, there will be no more trading in Global Bancorp common stock.

Financial Interests in the Merger of Directors and Officers of Global Bancorp and Global Trust Bank

Global Bancorp and Global Trust Bank executive officers and directors who are also shareholders of Global Bancorp will participate in the merger in the same manner and to the same extent as all of the other shareholders of Global Bancorp. However, in considering the recommendation of the Global Bancorp board of directors that Global Bancorp shareholders vote in favor of approving the principal terms of the merger agreement and the merger, Global Bancorp shareholders should be aware that Global Bancorp and Global Trust Bank executive officers and directors may have interests in the merger as individuals that are in addition to, or different from, their interests as shareholders of Global Bancorp, generally. The boards of directors of Global Bancorp and Global Trust Bank were aware of these interests and considered them, among other matters, in approving the merger agreement. These interests are described in this joint proxy statement-offering circular in this section and in the "SUMMARY – Financial Interests in the Merger of Directors and Officers of Global Bancorp and Global Trust Bank."

In addition to the financial interests in the merger of the directors and officers of Global Bancorp and Global Trust Bank set forth in the SUMMARY, commencing on page 4, after the completion of the merger, Bay Commercial Bank will grant Contingent Options to certain founding directors, directors emeritus, organizers, founders, and founding officers of Global Bancorp and Global Trust Bank named in the tables below. All Contingent Options granted will be for a five-year term from the effective time of the merger and have an exercise price of \$10 per share. All Contingent Options granted are also subject to termination prior to the end of the option term if the holder of a Contingent Option prior to the end of such option term sells more than 50% of his or her current share ownership in Global Bancorp prior to the merger or its equivalent ownership in Bay Commercial Bank after the merger. In addition, grants of Contingent Options to founding directors and directors emeritus are also subject to termination prior to the end of the option term, if the founding director or director emeritus optionee: (i) fails to maintain a majority of such optionee's current banking relationships with Bay Commercial Bank, or (ii) becomes a director on another board of directors of a community bank that operates offices within the counties in which Bay Commercial Bank has branch offices after the merger. However, with respect to Vinod K. Thukral, Chairman of Global Bancorp and Global Trust Bank, these two conditions lapse one year after the completion of the merger.

Contingent Options granted to organizers and founders are also subject to termination prior to the end of the option term if such organizer or founder optionee also: (i) fails to maintain a significant portion of such optionee's current banking relationships with Bay Commercial Bank, or (ii) becomes a director or officer on another board of directors of a competing bank.

The grant of a Contingent Option to a founding officer optionee is subject to the founding officer-optionee not agreeing to a cash payout in a net amount as yet to be determined and which amount is also acceptable to Bay Commercial Bank for the cancellation of such officer's current Global Bancorp non-qualified stock options. Information as to the identify of the optionees and amount of the Contingent Options of Bay Commercial Bank to be granted after the merger is shown in the following tables.

Options to be Granted to the Founding Directors of Global Trust Bank

Name		Option Share <u>Amount</u>
Bhupen Amin		16,000
Arthur Carmichael		6,666
Harpeet Chaudhay		6,666
Kamleshwar Gunsager	•	6,666
Mahendra Patel		9,657
Pramod Patel		16,000
Vinod Thurkal		20,000
	TOTAL	81,655

Options to be Granted to the Certain Organizers and Founders of Global Trust Bank

Name		Option Share <u>Amount</u>
James Wall		10,000
Mukesh Patel		3,333
R. C. Patel		3,333
Anand Patel		1,667
	TOTAL	18,333

Options that may be Granted to the Founding Officers of Global Trust Bank

		Option Share
Name		Amount
Robert Navarrete		6,666
R. Dale McKinney		6,666
-	TOTAL	13,332

Extensions of Terms of Stock Options Held by Directors of Bay Commercial Bank

Also as provided in the merger agreement, Bay Commercial Bank will extend the terms of certain stock options to its directors to match the expiration date of the Contingent Options which will be the fifth anniversary of the effective date of the merger. Information as to the Bay Commercial Bank options being extended is set forth in the table below.

Name of Director	Option Amount	Original Expiration Date	Extended Expiration Date*
James L. Apple	14,692	7/20/2014	8/29/2016
James S. Camp	14,692	7/20/2014	8/29/2016
George J. Guarini	14,692	7/20/2014	8/29/2016
Lloyd W. Kendall, Jr.	14,692	7/20/2014	8/29/2016
Robert R. Laverne, M.D.	14,692	7/20/2014	8/29/2016
Donald S. Morrow	10,000	6/4/2015	8/29/2016
David M. Spatz	14,692	7/20/2014	8/29/2016

* Assumes the merger is effective August 29, 2011.

Dissenters' Rights of Global Bancorp Shareholders

Dissenters' rights will be available to the Global Bancorp shareholders in accordance with the California Corporations Code. The required procedure set forth in sections 1300 through 1312 of the California Corporations Code must be followed exactly or any dissenters' rights may be lost.

The information set forth below is a general summary of dissenters' rights as they apply to Global Bancorp shareholders and is qualified in its entirety by reference to sections 1300 through 1312 of the California Corporations Code which is attached to this joint proxy statement-offering circular as <u>Appendix B</u>.

For a Global Bancorp shareholder to exercise dissenters' rights, he or she must:

- make a timely written demand upon Bay Commercial Bank for purchase in cash of his or her shares at their fair market value as of May 13, 2011, which demand includes:
 - the number and class of the shares held of record by him or her that he or she demands purchase by Bay Commercial Bank, and
 - what he or she claims to be the fair market value of his or her shares as of May 13, 2011;
- have his or her demand received by Bay Commercial Bank within 30 days of the date that the notice of approval of the merger by the shareholders of Global Bancorp was mailed to Global Bancorp shareholders;
- not vote in favor of the principal terms of the Merger Agreement;
- submit certificates representing his or her shares for endorsement in accordance with section 1302 of the Corporations Code; and,
- comply with such other procedures as are required by the Corporations Code.

Failure to follow the procedures set forth in the Corporations Code will result in a waiver of dissenters' rights. Further, if a Global Bancorp shareholder returns his or her proxy without instructions, which will result in a vote for the approval of the principal terms of the merger, he or she will not be entitled to dissenters' rights. Any demand notices or other documents to be delivered to Global Bancorp sent to Bay Commercial Bank, 1280 Civic Drive, Suite 100, Walnut Creek, California 94596, Attention: Corporate Secretary.

The statement of fair market value by a dissenting Global Bancorp constitutes an offer to sell his or her shares at the fair market value as of May 13, 2011. The fair market value for a share of Global Bancorp common stock on May 13, 2011 was \$6. A demand may not be withdrawn without the consent of Global Bancorp. A proxy or vote against the approval of the merger proposal does not in and of itself constitute a proper demand.

If a Global Bancorp shareholder holds dissenting shares, Global Trust Bank will mail a notice of the approval of the merger by Global Bancorp shareholders within ten days after the date of such approval, accompanied by:

- a copy of sections 1300, 1301, 1302, 1303, and 1304 of Chapter 13 of the Corporations Code;
- a statement of the price determined by Bay Commercial Bank to represent the fair market value as of May 13, 2011 of the dissenting shares; and,
- a brief description of the procedure to be followed if he or she desires to exercise his or her dissenters' rights under such sections.

The statement of price constitutes an offer by Bay Commercial Bank to purchase at the price stated for such dissenting shares.

A Global Bancorp shareholder who wishes to exercise dissenters' rights must submit to Bay Commercial Bank at its principal office or at the office of its transfer agent the certificates representing any shares that he or she is demanding that Bay Commercial Bank purchase for endorsement as dissenting shares, within 30 days after the date on which notice of approval of the merger by Global Bancorp shareholders was mailed to him or her.

If Bay Commercial Bank denies that shares submitted to it as dissenting shares are dissenting shares, or if Bay Commercial Bank and a dissenting shareholder fail to agree on the fair market value of his or her shares, then either the dissenting shareholder or Bay Commercial Bank may file a complaint in the superior court of the proper county in California requesting that the court determine such issue. Such complaint must be filed within six months after the date on which notice of the approval of the merger is mailed to such dissenting shareholder.

On trial of the action, the court will first determine if the shares are dissenting shares, and if so determined, the court will either determine the fair market value or appoint one or more impartial appraisers to do so. If both Bay Commercial Bank and a dissenting shareholder fail to file a complaint within six months after the date on which notice of the approval of the merger was mailed to such dissenting shareholder, his or her shares will cease to be dissenting shares. In addition, if a dissenting shareholder transfers his or her shares prior to their submission for the required endorsement, such shares will lose their status as dissenting shares.

Failure to take any necessary step will result in a termination or waiver of dissenters' rights under Chapter 13 of the Corporations Code. A person having a beneficial interest in Global Bancorp common stock that is held of record in the name of another person, such as a trustee or nominee, must act promptly to cause the record holder to follow the requirements of Chapter 13 of the Corporations Code in a timely manner if such person elects to demand payment of the fair market value of such shares.

Opinion of Financial Advisors

Fairness Opinions for Global Bancorp's Board of Directors

Global Bancorp's board of directors retained The Findley Group to render financial advisory and investment banking services.

Global Bancorp and Global Trust Bank retained The Findley Group to act as their financial advisor in connection with reviewing and considering various strategic alternatives and to provide its opinion of the fairness from a financial viewpoint of the merger consideration that their shareholders will receive in connection with the merger between Global Bancorp, Global Trust Bank and Bay Commercial Bank. As part of its investment banking business, The Findley Group is regularly engaged in the valuation of securities in connection with mergers and acquisitions and valuations for estate, corporate and other purposes. Global Bancorp's board of directors retained The Findley Group based upon its experience as a financial advisor in mergers and acquisitions of financial institutions.

The full text of the fairness opinion, which sets forth, among other things, assumptions made, procedures followed, matters considered, and limitations on the review undertaken, is attached as <u>Appendix E</u> to this joint proxy statement-offering circular. You are urged to read the fairness opinion carefully and in its entirety. The fairness opinion is addressed to Global Bancorp's board of directors and does not constitute a recommendation to any of Global Bancorp's shareholders as to how to vote at the meeting. The description of the fairness opinion set forth in this joint proxy statement-offering circular is qualified in its entirety by reference to the full text of the fairness opinion.

In connection with the fairness opinion, The Findley Group:

- Reviewed the Merger Agreement dated May 12, 2011;
- Reviewed Global Bancorp's consolidated audited financial statements for the period ending December 31, 2010 and 2009;
- Evaluated Global Bancorp's financial results based upon a review of its regulatory reports for the three years ending December 31, 2008 through December 31, 2010;
- Reviewed Global Bancorp's interim financial statement for the quarter ending March 31, 2011;
- Reviewed Bay Commercial Bank's audited financial statements for the period ending December 31, 2010 and 2009;
- Evaluated Bay Commercial Bank's financial results based upon a review of its regulatory reports for a three-year period (from and including the year ending December 31, 2008 through and including the year ending December 31, 2010;
- Reviewed Bay Commercial Bank's interim financial statements for the quarter ending March 31, 2011;
- Reviewed certain internal business and other operating data of Global Bancorp and Bay Commercial Bank;
- Conducted conversations regarding recent and projected financial performance of Global Bancorp and Bay Commercial Bank with respective members of executive management;
- Compared Bay Commercial Bank's recent operating results and pricing multiples with those of certain publiclytraded banks in the United States that The Findley Group deemed relevant;
- Compared Global Bancorp's recent operating results with those of certain other banks in the United States, including applicable California banks, that have recently been acquired;
- Compared the pricing multiples for Global Bancorp in the merger to those of certain other banks in the United States, including applicable California banks, that have recently been acquired;
- Analyzed the present value of the after-tax cash flows Global Bancorp could produce through December 31, 2013 based on projections provided by Global Bancorp's management;
- Reviewed the potential pro forma impact of the merger on the combined company's results and certain financial performance measures of Global Bancorp and Bay Commercial Bank;
- Estimated the non-marketable value of a share of Bay Commercial Bank's common stock;
- Reviewed the historical stock price data and trading volume of Bay Commercial Bank's common stock; and,
- Performed such other analyses as The Findley Group deemed appropriate.

In connection with its review, The Findley Group relied upon and assumed the accuracy and completeness of all of the foregoing information provided to it by Global Bancorp or that otherwise was available to it, and The Findley Group did not assume any responsibility for independent verification of such information. The Findley Group assumed that internal confidential financial projections that Global Bancorp provided were reasonably prepared reflecting the best currently available estimates and judgments of Global Bancorp's future financial performance and did not independently verify the validity of such assumptions. The Findley Group did not make any independent evaluation or appraisal of Global Bancorp's assets or liabilities and was not furnished with any such appraisals. The Findley Group did not examine any of Global Bancorp's individual loan files. The Findley Group is not an expert in the evaluation of loan portfolios for the purposes of assessing the adequacy of our allowance for losses with respect thereto and has assumed that such allowance is, in the aggregate, adequate to cover such losses.

With respect to Bay Commercial Bank, The Findley Group did not conduct any independent evaluation or appraisal of the assets, liabilities or business prospects of Bay Commercial Bank, and was not furnished with any evaluations or appraisals and did not review any individual documents of Bay Commercial Bank.

The fairness opinion is necessarily based on economic, market, and other conditions that were in effect—and on the information that was made available to The Findley Group, as of May 12, 2011. Global Bancorp is a bank holding company and sole shareholders of Global Trust Bank, its principal operating unit. The Findley Group was aware that, as of the date of the fairness opinion, that Global Trust Bank is a relatively young banking entity and as such remain under continued monitoring from the regulatory agencies as a de novo entity. The Findley Group is also aware that while Bay Commercial Bank has acquired a branch office of another banking entity in the past few years, Bay Commercial Bank has never merged another financial entity. Both Global Bancorp and Bay Commercial Bank believe that they have good relationships with the regulatory agencies.

In rendering the fairness opinion, The Findley Group performed a variety of financial analyses. The preparation of an opinion involves various determinations as to the most appropriate and relevant methods of financial analysis and the application of those methods to the particular circumstances. Consequently, the fairness opinion is not readily susceptible to partial analysis or summary description. Moreover, the evaluation of fairness, from a financial point of view, of the merger consideration is to some extent subjective, based on the experience and judgment of The Findley Group, and not merely the result of mathematical analysis of financial data. Accordingly, notwithstanding the separate factors summarized below, The Findley Group believes that its analyses must be considered as a whole and that selecting portions of its analyses and of the factors considered by it, without considering all analyses and factors, could create an incomplete view of the evaluation process underlying its opinion. The ranges of valuations resulting from any particular analysis described below should not be taken to be The Findley Group's view of Global Bancorp's actual value.

In performing its analyses, The Findley Group made numerous assumptions with respect to industry performance, business, and economic conditions and other matters, many of which are beyond Global Bancorp's control. The analyses performed by The Findley Group are not necessarily indicative of actual values or future results, which may be significantly more or less favorable than suggested by such analyses, nor are they appraisals. In addition, you should not view The Findley Group's analyses as determinative of the opinion of Global Bancorp's board of directors or management with respect to the value of Global Bancorp or Bay Commercial Bank or to the fairness of the merger consideration.

Pursuant to the Merger Agreement, each outstanding share of Global Bancorp common stock shall be converted into the right to receive shares of Bay Commercial Bank common stock at a fixed exchange ratio of 0.75678 and/or cash set at \$7.25 per share, at the election of the holder of such shares of Global Bancorp common stock with the total number of Global Bancorp shares electing cash consideration (subject to limitations and adjustments in the Merger Agreement). The aggregate merger consideration to be provided to Global Bancorp shareholders is approximately \$15.334 million. The Findley Group determined that the merger consideration to be delivered to Global Bancorp shareholders is approximately \$15.334 million. The shareholders' equity as of March 31, 2011 but will be equal to the shareholder equity of Global Bancorp on or about September 30, 2011, which is the approximate Closing Date of the Merger. As of March 31, 2011 the shareholder equity of Global Bancorp was approximately \$15.653 million and it is anticipated that Global Bancorp will incur a monthly loss for the remainder of 2011. Under the terms of the Agreement no more than 20% of Global Bancorp's outstanding common stock will be converted into 1,280,472 shares of Bay Commercial Bank common stock pursuant to the exchange ratio of 0.75678.

Analysis of Selected Transactions: The Findley Group performed an analysis of the amount of consideration paid in selected recently announced acquisitions of banking organizations with comparable characteristics to the merger. The Findley Group also reviewed several recapitalizations of banking organizations which resulted in a significant dilution of ownership by existing shareholders of such banking organizations.

The set of comparable transactions consisted of a group of selected transactions for unprofitable or marginally profitable banks in the western United States, which included applicable California transactions, with total assets between \$50 million and \$200 million which were less than seven years old for which pricing data were available. These comparable transactions consisted of 14 mergers and acquisitions of banks or significant recapitalizations of banking organizations with assets between \$50 million and \$200 million and \$200 million that were completed between January 1, 2009 and March 31, 2011. There were several additional transactions announced but a large percentage of the announced transactions were not consummated for a variety of reasons. The analysis yielded multiples of the purchase prices in these transactions as shown below:

	Price/ <u>Book (x)</u>	Price/ <u>Tg Book (x)</u>	Price/LTM <u>Earnings (x)</u>	Price/ <u>Assets (%)</u>	Price/Core <u>Deposits (%)</u>
Maximum	1.49	1.49	63.5	13.5	6.31
Minimum	0.17	0.17	NM	0.8	1.0
Median	0.80	0.80	NM	8.8	10.0
Global Bancorp*	1.00	1.00	NM	21.0	26.87

* Based on Global Bancorp's estimated shareholder equity at the Closing Date and assumes merger consideration of \$15.334 million or \$7.25 per share (80% stock/20% cash).

The price to book and price to tangible book multiples were higher in the merger compared to the medians of the comparable transactions. It was noted that the equity to asset (21.0%) ratio for Company is much higher than the median of the comparable transactions (8.8%) due to the young age of Company. The price to assets and price to deposits multiples in the merger were higher than for the comparable transactions.

Discounted Cash Flow Analysis. Using discounted cash flow analysis, The Findley Group estimated the present value of the future after-tax cash flow streams that Global Bancorp could produce through the year 2013, under various circumstances, assuming that it performed in accordance with the earnings/return projections provided by Global Bancorp. Global Bancorp currently has a tax loss carryover which allows Global Bancorp to pay limited taxes for the next several years. However for purposes of the cash flow analysis The Findley Group chose to evaluate Global Bancorp as if it was paying taxes going forward.

The Findley Group estimated the terminal value for Global Bancorp at the end of December 31, 2013 by considering both a multiple of after tax earnings and a multiple of Global Bancorp's book value per share. The Findley Group determined that the value of Global Bancorp shares would be based upon a multiple of after tax earnings in 2013 (in the range of 10 to 20) and a multiple of Global Bancorp's book value at year end 2013 (between 0.75 and 1.25). The Findley Group discounted the terminal values using discount rates ranging from 6.0% to 8.0%. The discount range was chosen to reflect different assumptions regarding the required rates of return of Global Bancorp and the inherent risk surrounding the underlying projections. This discounted cash flow analysis indicated a range of values per share of \$2.28 to \$7.96. The Findley Group determined that the likelihood of Global Bancorp achieving a higher earnings performance multiple or a higher multiple of book value which would result in a higher terminal value was not likely due to the size of Global Bancorp and the expected cost of compliance and operation of a banking platform the next few years.

The Findley Group then estimated the terminal value for Bay Commercial Bank at the end of December 31, 2013, taking into consideration the Merger, and considering both a multiple of after tax earnings and a multiple of Bay Commercial Bank's book value per share. The Findley Group determined that the value of Bay Commercial Bank shares would be based upon a multiple of after tax earnings in 2013 (in the range of 10 to 20) and a multiple of Bay Commercial Bank's book value at year end 2013 (between 0.75 and 1.25). The Findley Group discounted the terminal values using discount rates ranging from 6.0% to 8.0%. The discount range was chosen to reflect different assumptions regarding the required rates of return of Bay Commercial Bank and the inherent risk surrounding the underlying projections. This discounted cash flow analysis indicated a range of values per share (adjusting for the exchange ratio of 0.75678 of \$4.85 to \$8.56. The Findley Group determined that the likelihood of Bay Commercial Bank achieving a higher earnings performance multiple or a higher multiple of book value which would result in a higher terminal value was more likely than Global Bancorp due to the size of Bay Commercial Bank and the past earnings experience for Bay Commercial Bank over the last few years.

Trading Value of Company Shares. The Findley Group also considered the current trading process of community banks operating in California. For California community banks, the average earnings per share multiple for trailing twelve month earnings was 6.05 and median book value multiple was 0.726 as of December 31, 2010. For the California community banks that are located in Bay Area of California, the most recent earnings per share multiple was negative and the median book value multiple was 0.676 for the financial institutions listed in the report. Most of the trading information reported for the community banks reflects very few trades and limited activity.

Based upon the Agreement and the merger consideration of \$7.25 per share for Global Bancorp common stock, the comparable value would be as follows:

Multiple of Book Value	1.00
Multiple of Earnings per Share	

During 2010 no shares of Global Bancorp common stock were traded. As for Bay Commercial Bank, the most recent trade for its common stock was \$7.40 per share which is almost 0.79 of its March 31, 2011 book value per share. The merger consideration of book value is higher than the most recent trading price of other community banks in the Bay Area that are less than \$200 million in total assets and who have any substantive trading volume.

Comparable Company Analysis. No company or transaction used in the comparable company and comparable transaction analyses is identical to Global Bancorp or Bay Commercial Bank, or the merger. Accordingly, an analysis of the results of the foregoing necessarily involves complex considerations and judgments concerning differences in financial and operating characteristics of Global Bancorp and other factors that could affect the public trading value of the companies to which they are being compared. Mathematical analysis (such as determining the average or median) is not in itself a meaningful method of using comparable transaction data or comparable company data.

Pursuant to an engagement letter dated February 17, 2011, between The Findley Group and Company, Company agreed to pay \$200,000 to The Findley Group and its affiliated company Gary Steven Findley and Associates for all legal, consulting and investment banking services related to the Merger inclusive of the issuance of the fairness opinion.

Other than in connection with its advice and guidance in evaluating strategic alternatives in 2009 and 2010, the facilitation of a strategic planning meetings and recent regulatory matters, Company has not had any material relationship with The Findley Group during the last three years. Neither The Findley Group not its affiliated companies has performed any services for Bank to date. Neither The Findley Group nor any affiliate of The Findley Group has any ownership, either direct or indirect, in Company or Bank.

The fairness opinion is directed only to the question of whether the merger consideration is fair from a financial perspective and does not constitute a recommendation to any of Global Bancorp's shareholders to vote in favor of the merger. Global Bancorp did not impose any limitations on The Findley Group regarding the scope of its investigation or otherwise.

Based on the results of the various analyses described above, The Findley Group concluded that the merger consideration to be received by Global Bancorp shareholders pursuant to the merger is fair from a financial point of view.

The Findley Group will be paid a success fee that includes the work of providing the fairness opinion and other investment banking services in connection with the merger on behalf of Global Bancorp that is equal to \$200,000 less all legal fees paid to Gary Steven Findley and Associates and fees paid to The Findley Group related to the merger from February 17, 2011. Gary Steven Findley is the principal of The Findley Group and Gary Steven Findley &Associates.

Opinion of Bay Commercial Bank's Financial Advisor.

Bay Commercial Bank's board of directors retained Vining Sparks IBG, L.P. ("Vining Sparks") to render financial advisory and investment banking services. Vining Sparks is a nationally recognized investment banking firm with substantial expertise in transactions similar to the proposed transaction and is familiar with Bay Commercial Bank and its business. As part of its investment banking activities, Vining Sparks is regularly engaged in the independent valuation of financial institutions and securities in connection with mergers, acquisitions, underwritings, sales and distributions of listed and unlisted securities, private placements and valuations for estate, corporate and other purposes.

On June 7, 2011, Vining Sparks delivered a written opinion to the Bay Commercial Bank that the terms of the proposed merger of Bay Commercial Bank and Global Bancorp were fair, from a financial point of view, to holders of shares of Bay Commercial Bank common stock. The full text of Vining Sparks' opinion is attached as <u>Appendix F</u> to this proxy statement/prospectus and should be read in its entirety.

For purposes of Vining Sparks' opinion and in connection with its review of the proposed transaction, Vining Sparks has, among other things:

- reviewed the terms of the Agreement;
- reviewed certain publicly available financial statements, both audited (where available) and unaudited, and related financial information of Bay Commercial Bank and Global Bancorp, including those included in their respective annual reports for the past three years and their respective quarterly reports for the past two years;
- reviewed certain internal financial information and financial forecasts relating to the business, earnings, cash flows, assets, and prospects of each company furnished to Vining Sparks by Bay Commercial Bank and Global Bancorp management;
- held discussions with members of executive and senior management of Bay Commercial Bank and Global Bancorp concerning the past and current results of operations of Bay Commercial Bank and Global Bancorp, their respective current financial condition and managements' opinion of their respective future prospects;
- reviewed reported market prices and historical trading activity of Bay Commercial Bank and Global Bancorp common stock;
- compared the proposed financial terms of the merger with the financial terms of certain other transactions that Vining Sparks deemed to be relevant;
- reviewed the potential pro forma impact of the merger; and,
- performed such other financial studies, analyses, and investigations, as Vining Sparks considered appropriate under the circumstances.

Vining Sparks has assumed and relied, without independent verification, upon the accuracy and completeness of all of the financial and other information that has been provided to it by Bay Commercial Bank and Global Bancorp and their respective representatives, and of the publicly available information that was reviewed by it. Vining Sparks is not an expert in the evaluation of allowances for loan losses and has not independently verified such allowances, and has relied on and assumed that the aggregate allowance for loan losses set forth in the balance sheets of Bay Commercial Bank and Global Bancorp is adequate to cover such losses and complied fully with applicable law, regulatory policy and sound banking practice as of the date of such financial statements. Vining Sparks was not retained to and it did not conduct a physical inspection of any of the properties or facilities of Bay Commercial Bank or Global Bancorp, did not make any independent evaluation or appraisal of the assets, liabilities or prospects of Bay Commercial Bank or Global Bancorp, was not furnished with any such evaluation or appraisal, and did not review any individual credit files.

Vining Sparks' opinion is necessarily based on economic, market, and other conditions as in effect on, and the information made available to it as of, the date of its opinion. Events occurring after the date of issuance of the opinion, including, but not limited to, changes affecting the securities markets, the results of operations or material changes in the assets or liabilities of Bay Commercial Bank or Global Bancorp could materially affect the assumptions used in preparing the opinion. Vining Sparks assumed that all of the representations and warranties contained in the Agreement and all related agreements are true and correct, that each party to such agreements will perform all of the covenants required to be performed by such party under such agreements and that the conditions precedent in the Agreement are not waived.

No limitations were imposed by Bay Commercial Bank's board of directors upon Vining Sparks with respect to the investigations made or procedures followed in rendering its opinion. Vining Sparks' opinion as expressed in <u>Appendix F</u> is limited to the fairness, from a financial point of view, of the merger consideration to be paid by Bay Commercial Bank to the holders of shares of Global Bancorp common stock in the merger and does not address Bay Commercial Bank's underlying business decision to proceed with the merger. Vining Sparks has been retained on behalf of the board of directors of Bay Commercial Bank, and its opinion does not constitute a recommendation to any director of Bay Commercial Bank as to how such director should vote with respect to the Agreement.

Vining Sparks relied upon the managements of Bay Commercial Bank and Global Bancorp as to the reasonableness of the financial and operating forecasts, and projections (and the assumptions and bases therefore) provided to or reviewed by Vining Sparks, and Vining Sparks assumed that such forecasts and projections reflect the best currently

available estimates and judgments of Bay Commercial Bank and Global Bancorp management. Bay Commercial Bank and Global Bancorp do not publicly disclose internal management forecasts, projections or estimates of the type furnished to or reviewed by Vining Sparks in connection with its analysis of the financial terms of the proposed transaction, and such forecasts and estimates were not prepared with a view towards public disclosure. These forecasts and estimates were based on numerous variables and assumptions which are inherently uncertain and which may not be within the control of the managements of Bay Commercial Bank and Global Bancorp, including, without limitation to, the general economic, regulatory, and competitive conditions. Accordingly, actual results could vary materially from those set forth in such forecasts and estimates.

In delivering its opinion to the board of directors of Bay Commercial Bank, Vining Sparks prepared and delivered to Bay Commercial Bank's board of directors written materials containing various analyses and other information. The following is a summary of the material financial analyses performed by Vining Sparks in connection with the preparation of its opinion and does not purport to be a complete description of all the analyses performed by Vining Sparks. The summary includes information presented in tabular format, which should be read together with the text that accompanies those tables. The preparation of a fairness opinion is a complex process involving various determinations as to the most appropriate and relevant methods of financial analysis and the application of those methods to the particular circumstances. Therefore, an opinion is not necessarily susceptible to partial analysis or summary description. Vining Sparks believes that its analyses must be considered as a whole and that selecting portions of such analyses and the factors considered therein, without considering all factors and analyses, could create an incomplete view of the analyses and the processes underlying its opinion. In its analyses, Vining Sparks made numerous assumptions with respect to industry performance, business and economic conditions, and other matters, many of which are beyond the control of Bay Commercial Bank, Global Bancorp, and Vining Sparks. Any estimates contained in Vining Sparks' analyses are not necessarily indicative of future results or values, which may be significantly more or less favorable than such estimates. Estimates of values of companies do not purport to be appraisals or necessarily reflect the prices at which companies or their securities may actually be sold.

Summary of Proposal. Vining Sparks reviewed the financial terms of the proposed transaction. Pursuant to the terms of the Agreement, Global Bancorp will merge with and into Global Trust Bank and subsequently, Global Trust Bank will merge with and into Bay Commercial Bank. Each share of Global Bancorp common stock outstanding shall become one share of Global Trust Bank. Subsequently, each share of Global Trust Bank shall be converted into the right to receive for each share of Global Trust Bank common stock (i) one share of Bay Commercial Bank common stock times the Exchange Ratio, and/or (ii) cash in accordance with the Per Share Cash Consideration, at the election of the holder of such share of Global Trust Bank common stock up to the limit set forth in the Agreement. As defined in the Agreement, the Exchange Ratio means a factor of 0.75678 and the Per Share Cash Consideration means \$7.25 per share. All capitalized items used in this paragraph shall have the meanings ascribed to them in the Agreement. The terms of the Merger are more fully set forth in the Agreement.

Based on 2,115,001 shares of Global Bancorp common stock outstanding and Per Share Cash Consideration of \$7.25 per share, the shareholders of Global Bancorp will receive merger consideration of \$15.3 million consisting of up to 20% in cash and not less than 80% in BCB common stock, assuming the shareholders of Global Bancorp elect to receive the maximum amount of cash permitted under the Merger Agreement.

Analysis of Selected Bank Merger Transactions. Vining Sparks reviewed certain publicly available information regarding ten selected merger and acquisition transactions (the "Comparable Transactions") announced from January 1, 2010 to May 18, 2011 involving California banking organizations. The transactions included in the group are shown below and in Section 6. This data was obtained from SNL Financial LC.

<u>Buyer</u>	<u>City</u>	<u>Seller</u>	<u>City</u>
Rabobank Nederland	Utrecht	Napa Community Bank	Napa
Ford Financial Fund, L.P.		Pacific Capital Bancorp	Santa Barbara
Woori Finance Holdings Co.	Seoul	Hanmi Financial Corp.	Los Angeles
Management group	San Francisco	Bank of San Francisco	San Francisco
Grandpoint Bank	Santa Ana	First Commerce Bancorp	Encino
California United Bank	Encino	California Oaks State Bk	Thousand Oaks
Investor group		Bay Cities National Bank	Redondo Beach
First General Bank	Rowland Heights	American Premier Bancorp	Arcadia
Grandpoint Capital, Inc.	Los Angeles	Orange Community Banc	Orange
Embarcadero Bank	San Diego	Coronado First Bank	Coronado

We reviewed the multiples of transaction value to stated book value, transaction value to tangible book, transaction value to last twelve months earnings and tangible book premium to core deposits and calculated high, low, mean, and median multiples for the Comparable Transactions. The median multiples were then applied to Global Bancorp's balance sheet information as of March 31, 2011 and 2010 earnings to derive an imputed range of values of Global Bancorp's common stock. The following table sets forth the median multiples as well as the imputed values based upon those median multiples.

	Comparable	Implied
	Transaction	Value
	Median Multiple	Per Share
Transaction Value/Book Value	1.05x	\$7.77
Transaction Value/Tangible Book Value	1.05x	\$7.77
Transaction Value/2010 Earnings	46.48x	N/M
Tangible Premium/Core Deposits	1.58%	\$7.66

The transaction value of \$7.25 per share falls within the range of values computed in using the Comparable Transactions, which supports the fairness of the transaction.

No company or transaction used as a comparison in the above analysis is identical to Global Bancorp or the Merger. Accordingly, an analysis of these results is not strictly mathematical. An analysis of the results of the foregoing involves complex considerations and judgments concerning differences in financial and operating characteristics of Global Bancorp and the companies included in the Comparable Transactions.

Present Value Analysis. Vining Sparks calculated the present value of theoretical future earnings of Global Bancorp and compared the transaction value to the calculated present value of Global Bancorp's common stock on a standalone basis. Based on projected earnings for Global Bancorp for 2011 through 2015 and a discount rate of 10% and including a residual value, the stand-alone present value of Global Bancorp equaled \$7.29 per share. The transaction value of \$7.25 per share is below this value, which supports the fairness of the transaction.

Pro Forma Merger Analysis. Vining Sparks performed pro forma merger analyses to calculate the financial implications of the merger to Bay Commercial Bank shareholders. This analysis assumes, among other things, the terms of the transaction as indicated above, that the merger closes at September 30, 2011 and cost savings and revenue enhancement opportunities of \$1,000,000 annually in the years 2012 through 2015, which approximates 30% of Global Bancorp's 2010 overhead expenses. This analysis utilized earnings estimates of \$0.40 per share in 2012 and \$0.50 per share in 2013 for Bay Commercial Bank and \$0.21 per share and \$0.52 per share, respectively, for Global Bancorp. This analysis indicated that the merger would be approximately 23% accretive to Bay Commercial Bank's projected earnings per share in 2012 and 40% accretive in 2013.

Vining Sparks has not previously provided investment banking or financial advisory services to Bay Commercial Bank or Global Bancorp. Pursuant to the terms of an engagement letter with Bay Commercial Bank, Vining Sparks will receive a fee upon delivery of its opinion. Vining Sparks' fee is not contingent upon consummation of the Merger. In addition, Bay Commercial Bank has agreed to indemnify Vining Sparks against certain liabilities and expenses arising out of or incurred in connection with its engagement, including liabilities and expenses which may arise under the federal securities laws.

Accounting Treatment

Bay Commercial Bank must account for the merger under the acquisition method of accounting prescribed by accounting principles generally accepted in the United States. Under this method, Bay Commercial Bank's purchase price will be allocated to Global Bancorp's assets acquired and liabilities assumed based upon their estimated fair values as of the completion of the merger. Deferred tax assets and liabilities will be adjusted for the difference between the tax basis of the assets and liabilities and their estimated values. The excess, if any, of the total acquisition cost over the sum of the assigned fair values of the tangible and identifiable intangible assets acquired, less liabilities assumed will be recorded as goodwill and periodically evaluated for impairment. Bay Commercial Bank's financial statements issued after completion of the merger will reflect these values, but historical data are not restated retroactively to reflect the combined historical financial position or results of operations of Bay Commercial Bank and Global Bancorp. For additional information, please read the section entitled "DESCRIPTION OF BAY COMMERCIAL BANK – Management's Discussion and Analysis of Financial Condition and Results of Operations."

The Merger Agreement

The following describes aspects of the mergers, including material terms of the Merger Agreement. The description of the Merger Agreement is subject to, and qualified in its entirety by reference to, the Merger Agreement, which is attached to this document as <u>Appendix A</u>, and is incorporated by reference in this document. We urge you to carefully read the Merger Agreement.

The Mergers

Bay Commercial Bank, Global Bancorp and Global Trust Bank entered into the Merger Agreement on May 12, 2011. Under the Merger Agreement's terms, Global Bancorp will merge with and into Global Trust Bank and Global Trust Bank will merge with and into Bay Commercial Bank. The separate corporate existence of Global Bancorp and Global Trust Bank will cease, and Bay Commercial Bank will be the survivor. Each share of Global Bancorp common stock issued and outstanding, other than shares with respect to which dissenters' rights have been perfected, will be converted into shares of Bay Commercial Bank common stock or cash, or a combination of the two.

Each share of Bay Commercial Bank common stock outstanding immediately before the merger closes will remain outstanding after the mergers close. Please read the sections entitled "JOINT PROPOSAL I – THE MERGERS – Calculation of Consideration to be Paid to Global Bancorp Shareholders," and "– Dissenters' Rights of Global Bancorp's Shareholders" for additional information.

Global Bancorp and Bay Commercial Bank have structured the merger to qualify as a tax-free reorganization from their perspectives. You are urged to read the section entitled "JOINT PROPOSAL I – THE MERGERS – Material United States Federal Income Tax Consequences" for additional information.

The Closing

The Merger Agreement provides that the timing for the closing and the completion of the merger shall be designated by Bay Commercial Bank, with the reasonable concurrence of Global Bancorp, as promptly as practicable following the receipt of all regulatory approvals and expiration of all applicable waiting periods in connection therewith and after the satisfaction of the other conditions precedent set forth in the Merger Agreement. At the closing the parties will exchange various documents, including among other things officers' certificates, as required by the Merger Agreement. The downstream merger of Global Bancorp into Global Trust Bank shall occur first and shall be effective on the date and at the time that the Downstream Merger Agreement is filed with the California Secretary of State. Then immediately thereafter, Global Trust Bank will be merged into Bay Commercial Bank and that merger will be effective on the date and at the time that a copy of the agreement of merger in the form attached to the Merger Agreement as Exhibit 2.2 and as filed with California Secretary of State, is thereafter filed with the CDFI.

Based upon the timing for Bay Commercial Bank's and Global Bancorp's shareholders' meetings and the present and anticipated timing of the regulatory approvals, it is presently anticipated that the merger will be closed during the third quarter of 2011. Neither Bay Commercial Bank nor Global Bancorp can assure you that the merger will close at that time.

Exchange Procedures

The parties will appoint an exchange agent to effect the exchange of shares of Bay Commercial Bank common stock for shares of Global Bancorp common stock. A letter of transmittal will be sent to Global Bancorp shareholders shortly after the merger is completed. If you do not exercise dissenters' rights, you must use the letter of transmittal to receive shares of Bay Commercial Bank common stock or cash in exchange for your shares of Global Bancorp common stock.

In order to promptly receive shares of Bay Commercial Bank common stock, you must deliver to the exchange agent:

- a properly completed letter of transmittal form;
- your certificates representing former shares of Global Bancorp common stock; and,
- any other required documents described in the letter of transmittal.

Do not return your certificates representing shares of Global Bancorp common stock with the enclosed proxy. The certificates should only be forwarded to the exchange agent with the letter of transmittal.

Following the completion of the merger and upon surrender of all of the certificates representing former shares of Global Bancorp common stock registered in your name, or a satisfactory indemnity if any of such certificates are lost, stolen or destroyed, together with a properly completed letter of election and transmittal, the exchange agent will mail to you the Bay Commercial Bank common stock or cash to which you are entitled, less the amount of any required withholding taxes.

Representations and Warranties

The Merger Agreement contains various customary representations and warranties that Bay Commercial Bank and Global Bancorp make for each other's benefit. The representations and warranties relate to, among other things:

- corporate organization and similar corporate matters;
- licenses and permits;
- subsidiaries;
- authorization and enforceability of the Merger Agreement and related matters;
- conflicts under charter documents, required consents or approvals, and violations of any agreements or law;
- capital structure;
- timely filing, accuracy and completeness of documents filed with government agencies and bank regulatory agencies;
- financial statements;
- absence of certain material adverse events, changes, effects or undisclosed liabilities;
- litigation;
- compliance with laws and regulations;
- tax returns and audits;
- agreements with banking authorities;
- insurance;
- brokers and finders;
- the Community Reinvestment Act;
- the Bank Secrecy Act;
- title to assets;
- ownership of real property;
- employees;
- retirement and other employee plans and matters relating to the Employee Retirement Income Security Act of 1974, as amended;

- loans and investments;
- material contracts;
- environmental matters;
- parachute payments;
- risk management instruments;
- Regulation C, Truth in Lending Law;
- fairness and validity of accounting records; and,
- accuracy of corporate records.

The foregoing is an outline of the representations and warranties made by Bay Commercial Bank and Global Bancorp contained in the Merger Agreement attached as <u>Appendix A</u> and is qualified in its entirety by reference to the Merger Agreement. You should carefully review the entire Merger Agreement, and in particular Articles 3 and 4, containing the detailed representations and warranties of the parties.

Additional Agreements

As a condition to the merger, Bay Commercial Bank and Global Bancorp have agreed, among other things, that:

- Each party will grant the other reasonable access to its employees and records and will provide each with meeting materials, minutes of meetings and notices of meetings of its board of directors and loan committee;
- each of them will promptly call its respective meeting of shareholders at the earliest practicable date after approval of meeting materials by the California Department of Financial Institutions and will recommend approval of the merger and Merger Agreement, subject to its fiduciary duties;
- each of them will take all action necessary to complete the merger;
- they will cooperate and jointly prepare and file proxy materials and applications with the appropriate regulatory authorities;
- each of them will provide the other with any correspondence with governmental or regulatory authorities;
- each of them will give notice to the other party of events that may have a material adverse effect on the merger;
- each of them will pay its own costs and expenses incurred in connection with the merger;
- each of them giving notice to the other party of acquisition transactions;
- each of them maintaining assets and properties in good condition and repair;
- each of them providing the other party copies of each credit authorization package for all loans secured by a first trust deed on real property, or liquid collateral consisting of cash, TCD's, stock, bonds or the cash value of life insurance or extensions of credit secured by the foregoing, unsecured loans or unsecured extensions of credit, and renewals of any classified or criticized loans, concurrently with submission to its loan committee;
- each of them charging off loans deemed by each, its outside loan reviewer, the FDIC, the CDFI or its independent auditors to be uncollectible in accordance with GAAP, regulatory accounting principles, and applicable law and regulation which have been classified "loss";

- each of them maintaining reserves in accordance with applicable accounting principles and past practices; including maintaining a reserve for loan losses at an adequate level and charge off all loans deemed uncollectable, as well as changes in asset classifications;
- each of them providing each other with reports of loans classified by bank regulators, renegotiated loans, charged off loans and of other nonperforming assets, participated loans purchased and sold, loans to insiders, and standby letters of credit;
- each of them providing their monthly board reports, copies of all material reports filed with or received from any governmental entity, and monthly unaudited balance sheets, statements of income and changes in shareholders' equity, and information on certain loans and leases;
- each of them timely filing of governmental reports;
- each of them performing all material contractual obligations;
- each of them keeping in effect all material permits and licenses;
- each of them maintaining insurance and bonding coverage;
- filing applications for the merger transactions;
- each of them cooperating with others to satisfy the conditions of the Merger Agreement;
- each of them will notify the other of (i) its violation of promises to carry on its business in the ordinary course until the merger becomes effective, (ii) any event expected to have a material adverse effect on it, or (iii) any awareness of an untrue statement in the proxy materials;
- each of them will deliver to the other, updated disclosure schedules immediately after the merger becomes effective;
- Global Trust Bank and/or Global Bancorp will modify certain financial policies and practices to conform to those of Bay Commercial Bank; and,
- Global Trust Bank and Global Bancorp will, if requested by Bay Commercial Bank, terminate any employee benefit plan or arrangement.

The foregoing is an outline of the additional agreements made by Global Bancorp, Global Trust Bank and Bay Commercial Bank contained in the Merger Agreement attached as <u>Appendix A</u> and is qualified in its entirety by reference to the Merger Agreement. You should carefully review the entire Merger Agreement, and in particular Article 5, containing the detailed additional agreements of the parties.

Conduct of Business Before the Merger

The Merger Agreement places restrictions on and requires commitments by Global Trust Bank, Global Bancorp and Bay Commercial Bank regarding the conduct of their respective businesses between the date of the Merger Agreement and the closing. Global Trust Bank and Global Bancorp have agreed that until the closing they will carry on their business in the ordinary course in substantially the manner in which heretofore conducted and comply with other covenants, which shall include, but not be limited to:

- filing applications for the merger transactions by June 15, 2011;
- terminate all Global Bancorp stock options and commitments to grant stock options and obtain upon such termination a waiver and release of liability; and,
- use commercially reasonable efforts to extend the discovery period of the directors' and officers' liability insurance currently in force for a period of up to 48 months with respect to all matters arising from facts or events which occurred before the Effective Time for which the Global Bancorp or Global Trust Bank would have had an

obligation to indemnify their directors and officers (a "*tail policy*"); provided, however, that the total cost of the premium for such "tail" policy shall not exceed \$80,000, and if Global Bancorp and Global Trust Bank are unable to maintain or obtain a "tail" policy for \$80,000 or less, Global Trust Bancorp shall use commercially reasonable efforts to obtain as much comparable insurance as is available for \$80,000 with respect to acts or omissions occurring prior to the effective time of the Merger by such directors and officers in their capacities as such.

Global Bancorp and Global Trust Bank have also agreed that until the closing they will not take certain action, without Bay Commercial Bank's prior written consent including:

- declare or pay any dividend or make any other distribution in respect of any of its capital stock; split, combine or reclassify any of its capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock; or repurchase or otherwise acquire any shares of its capital stock;
- take any actions which may make their representations and warranties become untrue;
- issue any shares of its capital stock or any securities convertible or exercisable into such capital stock except as specifically provided in the Merger Agreement or grant stock options;
- amend its Articles of Incorporation or Bylaws;
- authorize or encourage another acquisition proposal to acquire them or provide any nonpublic information to any person(s) concerning any such acquisition proposal unless the board of directors of Global Bancorp has determined that the proposal is more favorable to Global Bancorp's shareholders and that the action is necessary for Global Bancorp to comply with its fiduciary duties to shareholders under applicable law;
- acquire all or a substantial portion of the assets of any business or person, other than in the ordinary course of business;
- dispose of any of its assets, except in the ordinary course of business;
- incur any indebtedness, other than in the ordinary course of business;
- make or agree to make any loan with a principal balance or maximum principal balance in excess of \$500,000 for an unsecured or \$1,250,000 for a secured loan, or except in the ordinary course of business pursuant to Global Trust Bank loan policy make or enter into any commitment to make any other loan or extension of credit or renew any criticized or classified loan;
- except as specifically provided in the Merger Agreement and the schedules thereto, grant any general increase in the rates of pay of employees or employee benefits; renew, amend or enter into a new employment agreement; or make any bonus or extraordinary payments, except that bonuses may be paid and arrangements entered into with respect to Global Bancorp or Global Trust Bank employees to encourage their cooperation and assistance in completing the transactions contemplated by this Agreement that are agreed to by Bay Commercial Bank;
- make the credit underwriting policies less stringent than those in effect on December 31, 2010 or reduce the amount of the loan loss reserves;
- acquire any investment security except for federal funds with maturities of one year or less;
- compromise or settle certain tax claims;
- change its fiscal year or methods of accounting in effect at December 31, 2010;
- take any action which would disqualify the mergers as "reorganizations" for tax purposes or make any new, or changes to current, tax elections;
- take commercial property into OREO without an environmental report showing a lack of adverse conditions;

- make any capital expenditures, except those in the ordinary course of business which exceed \$10,000 individually or \$25,000 in the aggregate; and,
- materially change its pricing practices on loans or deposit products.

In addition, Bay Commercial Bank has agreed that until the closing it will carry on its business in the ordinary course in substantially the manner in which heretofore conducted, and to take the following actions until such time or thereafter, as applicable:

- For a period of four years maintain and preserve the rights to indemnification of officers and directors provided in the charter documents of Global Bancorp and Global Trust Bank with respect to liabilities arising out of circumstances occurring or existing prior to the effective date of the merger;
- reserve and make available for issuance such number of shares of Bay Commercial Bank's common stock as necessary to issue to Global Bancorp shareholders in connection with the merger, and have those shares properly qualified or registered, if necessary;
- grant and enter into stock option agreements for such grant of non-qualified stock options for a five year term to
 certain of the holders of Global Bancorp's non-qualified stock options with a date as of the effective time of the
 merger (see section entitled "Financial Interests in the Merger of Directors and Officers of Global Bancorp and
 Global Trust Bank" herein for the names of such holders and other information concerning such options); and,
- as of the effective date of the merger to amend or enter into, as appropriate, stock option agreements with directors of Bay Commercial Bank who currently hold non-qualified stock options in order to extend the expiration of such non-qualified stock options to the fifth anniversary of the effective date of the merger (see section entitled "Financial Interests in the Merger of Directors and Officers of Global Bancorp and Global Trust Bank" herein for the names of such holders and other information concerning such options).

Bay Commercial Bank has also agreed that until the closing it will not, without Global Bancorp's prior written consent:

- Declare or pay any dividend or make any other distribution in respect of any of its capital stock; split, combine or reclassify any of its capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock; or repurchase or otherwise acquire any shares of its capital stock;
- take any action which could make any of its representations and warranties become untrue in a material respect;
- without the prior written consent of Global Bancorp, authorize or knowingly permit any of its representatives, directly or indirectly, to enter into any agreement for the acquisition of any other financial entity by Bay Commercial Bank or acquire or agree to acquire by merging, consolidating with, or by purchasing all or a substantial portion of the assets of, or in any other manner, any business or any person or otherwise acquire or agree to acquire any assets which are material to Bay Commercial Bank, other than in the ordinary course of business consistent with prior practice;
- enter into any agreement or arrangement requiring the payment of an amount in excess of \$200,000 (or in the aggregate \$500,000), except: (i) Bay Commercial Bank may enter into any such agreement without the Global Bancorp's consent if, prior to entering into such agreement, Bay Commercial Bank consults with Global Bancorp; or (ii) for agreements or arrangements in the ordinary course of business consistent with prior practice and after notice to and consultation with Bay Commercial Bank;
- issue any shares of its capital stock or any securities convertible or exercisable into such capital stock except as specifically provided in the Merger Agreement; and,
- take any action which would disqualify the merger as a "reorganization" for tax purposes.

The foregoing is a summary of the material negative and affirmative covenants of the Merger Agreement and is qualified in its entirety by reference to the Merger Agreement. You are encouraged to carefully read the terms of the Merger Agreement attached as <u>Appendix A</u>, including the specific covenants contained in Article 6.

Conditions to the Parties' Obligations

The obligations of Global Bancorp, Global Trust Bank and Bay Commercial Bank to complete the merger are subject to certain mutual conditions, including, but not limited to the following:

- the mergers and the merger agreements being validly approved by the holders of Global Bancorp's shareholders and Bay Commercial Bank's shareholders;
- the receipt of all regulatory approvals;
- the expiration of all applicable waiting periods in connection with the regulatory approvals;
- the qualification or registration of the sale of Bay Commercial Bank's stock resulting from the merger with the appropriate state securities authorities;
- the agreed upon price per share for shares of Global Bancorp common stock which are "dissenting shares" within the meaning of California Corporations Code section 1300(b) shall not exceed the greater of (i) \$6.00 or (ii) an amount which, when combined with other cash amounts payable in connection with the merger, would result in the merger being disqualified from a tax free reorganization pursuant to section 368(a)(1)(A) of the Internal Revenue Code, provided that in no event shall any such combined amount exceed approximately \$3.067 million; and
- the receipt of all consents of third parties required to permit the merger.

Bay Commercial Bank's obligation to complete the merger is also subject to the fulfillment or waiver by it of certain conditions, including but not limited to the following:

- Global Trust Bank's and Global Bancorp's representations and warranties being and remaining true in all material respects;
- Global Trust Bank and Global Bancorp performing, in all material respects, all of their required obligations contained in the Merger Agreement before closing;
- receipt of updated disclosure schedules from Global Trust Bank and Global Bancorp;
- receipt of a director-shareholder agreement executed by each of Global Trust Bank's and Global Bancorp's director-shareholders; and,
- receipt of a fairness opinion that the terms of the merger are fair to the Global Bancorp shareholders from a financial point of view.

Global Trust Bank's and Global Bancorp's obligations to complete the merger are also subject to the fulfillment or waiver by them of certain conditions, including but not limited to the following:

- Bay Commercial Bank's representations and warranties being and remaining true in all material respects;
- Bay Commercial Bank performing, in all material respects, all of its required obligations contained in the Merger Agreement before closing;
- receipt of updated disclosure schedules from Bay Commercial Bank;
- receipt of a director-shareholder agreement executed by each of Bay Commercial Bank's director-shareholders; and,
- receipt of a fairness opinion that the terms of the merger are fair to the Bay Commercial Bank shareholders from a financial point of view.

The foregoing is a summary of the material conditions of the Merger Agreement and is qualified in its entirety by reference to the Merger Agreement. You are encouraged to read the terms of the Merger Agreement attached as <u>Appendix A</u>, including the specific provisions contained in Article 7 of the Merger Agreement.

Termination

Global Bancorp, Global Trust Bank and Bay Commercial Bank can mutually agree to terminate the Merger Agreement and abandon the merger at any time.

Under certain circumstances, either Global Bancorp or Bay Commercial Bank can terminate the Merger Agreement:

- if the conditions to the merger have not been met by October 31, 2011;
- if 20 business days have passed after any governmental entity denies any required approval for the merger;
- if the closing conditions have not been met by the other party by October 31, 2011;
- if the other party materially breaches any representation, warranty, covenant or agreement contained in the Merger Agreement that is not cured by the later of 30 days from the breach or October 31, 2011, which would be reasonably likely to have, a material adverse effect upon the consummation of the merger; or
- if an acquisition event involving Global Bancorp or Bay Commercial Bank has occurred.

If Global Bancorp or Global Trust Bank terminates because of the failure by Bay Commercial Bank's shareholders to approve the mergers following a withdrawal of, or similar change in, the recommendation of approval by Bay Commercial Bank's board of directors; the failure of any conditions of closing by Bay Commercial Bank; a breach and lack of cure of Bay Commercial Bank's representations or warranties; or an acquisition event involving Bay Commercial Bank, then Bay Commercial Bank must pay a total of \$400,000 to Global Trust Bank and/or Global Bancorp.

If Bay Commercial Bank terminates because of the failure by Global Bancorp's shareholders to approve the mergers following a withdrawal of, or similar change in, the recommendation of approval by Global Bancorp's board of directors; conditions to closing were not met by Global Bancorp or Global Trust Bank; representations and warranties by Global Bancorp or Global Trust Bank were breached and not cured; or an acquisition event occurred involving Global Bancorp, then Global Bancorp and Global Trust Bank must pay to Bay Commercial Bank the combined sum of \$400,000.

The foregoing is a summary of the material termination provisions of the Merger Agreement. You are encouraged to read the terms of the Merger Agreement attached as <u>Appendix A</u>, including the specific provisions contained in Article 8 of the Merger Agreement.

Employee Benefits

After completion of the merger, all employees of Global Trust Bank at the date of the completion of the merger that have been offered continued employment, shall be entitled to participate in all of Bay Commercial Bank's employee benefit arrangements on the same basis as other similarly situated employees of Bay Commercial Bank. Each of these employees will be credited for eligibility, participation and vesting purposes with such employee's respective years of past service with Global Trust Bank as though they had been employees of Bay Commercial Bank, except with respect to Bay Commercial Bank's Stock Option Plan, any additional equity compensation plan Bay Commercial Bank may adopt and any 401(k) plan.

All outstanding stock options of Global Bancorp will be automatically cancelled at the effective time of the merger if they are not exercised prior to the effective time of the merger. The non-qualified stock options granted to founding officers R. Dale McKinney (20,000 shares) and Robert Navarrete (20,000 shares) are discussed in the section entitled "SUMMARY – Financial Interests in the Merger of Directors and Officers of Global Bancorp and Global Trust Bank."

Director-Shareholder Agreements

Each of the parties has entered into director-shareholder agreements with each of its directors who hold shares of its stock entitled to vote on the merger and recommend that their shareholders vote in favor of the merger, as applicable. These director-shareholder agreements require each of the directors to vote in favor of the merger. However, these agreements also provide that each director shall not be obligated to recommend to their company's shareholders to vote in favor of the merger if the director is advised in writing by his or her outside legal counsel that, in the exercise of his or her fiduciary duties, a director should not take such action. The director-shareholder agreements also provide that each director will not take any action that will alter the right to vote his or her shares, including the disposition, encumbrance or assignment of the underlying shares or the exercise of stock options.

UNAUDITED PRO FORMA COMBINED CONDENSED FINANCIAL STATEMENTS

As part of the merger, Global Bancorp shareholders will have the option to receive (i) \$7.25 in cash (ii) 0.75678 of a share of Bay Commercial Bank common stock or (iii) a combination of both for each share of Global Bancorp common stock, provided that at least approximately 80% of the merger consideration is paid in common stock. The completion of the merger is subject to customary conditions, including the approval of shareholders and bank regulatory authorities.

An unaudited summary of Bay Commercial Bank's estimated purchase price consideration, assuming an estimated fair value of \$9.58 per share for Bay Commercial Bank stock, which is the determined fair market value, and preliminary purchase price allocations for the merger follows. The purchase price allocation is based on estimates and is subject to change as more information becomes available and after final analysis of the fair value of both tangible and intangible assets acquired and liabilities assumed is completed. Accordingly, the final fair value adjustments may be materially different from those presented in this report.

Scenario with all Global Bancorp shareholders electing to take stock only	(Dollars in thousands)
Total consideration and acquisitions costs (assuming 0% of Global Bancorp shareholder receive cash):	
Common stock consideration 2,115,001 x 0.75678 of Bay Commercial Bank shares x 100% valued at \$9.58 per share)	\$15,334
Cash consideration, 0% of 2,115,001 Global Trust Bank shares at \$7.25 per share Total purchase price consideration	0 \$15,334
Scenario with all Global Bancorp shareholders electing to take cash	(Dollars in thousands)
Total consideration and acquisitions costs (assuming 20% of Global Bancorp shareholders receive	(
Scenario with all Global Bancorp shareholders electing to take cash Total consideration and acquisitions costs (assuming 20% of Global Bancorp shareholders receive cash): Common stock consideration 2,115,001 x 80% x 0.75678 Bay commercial Bank shares valued at \$9.58 per share)	C
Total consideration and acquisitions costs (assuming 20% of Global Bancorp shareholders receive cash):	thousands)

SCHEDULE OF NET ASSET VALUE

	(Dollars in thousands)
Assets acquired or to be acquired:	
Cash and investments	\$27,260
Loans, net	44,749
Intangible assets	219
Other assets	2,840
Total assets acquired	\$75,068
Liabilities assumed or to be assumed:	
Deposits	\$57,110
Borrowings and other liabilities	223
Total liabilities assumed	\$57,333
Net Assets Acquired	\$17,735

UNAUDITED PRO FORMA COMBINED CONDENSED FINANCIAL INFORMATION

The following unaudited pro forma combined condensed financial information presents the impact of the proposed merger on Bay Commercial Bank's historical financial condition and results of operations under the acquisition method of accounting.

The pro forma information is based on the following assumptions, which have been used to calculate the exchange ratio: (i) that no shareholder of Global Bancorp perfects dissenters' rights, and (ii) that 80% of the total merger consideration will be paid in the form of Bay Commercial Bank common stock with a value of \$9.58 per share and the remaining 20% will be paid in cash.

The unaudited pro forma combined condensed balance sheet as of March 31, 2011 has been prepared under the assumption that the merger was completed on that date. The unaudited pro forma combined condensed statement of earnings for the year ended March 31, 2011 has been prepared under the assumption that the merger was completed on January 1, 2011.

The unaudited pro forma combined condensed financial information is presented for illustrative purposes only, and does not indicate either the operating results or financial condition that would have occurred had the acquisition been consummated on the dates indicated, or future results of operations or financial condition. The unaudited pro forma combined condensed financial information is based upon assumptions and adjustments that Bay Commercial Bank believes are reasonable. The unaudited pro forma combined condensed financial information and the accompanying notes should be read in conjunction with the entities' historical financial statements.

As explained further in the accompanying notes to the unaudited pro forma combined condensed financial information, the allocation of the purchase price with respect to the acquisition is based on preliminary estimates and is subject to change as more information becomes available and after final analysis of the fair value of both tangible and intangible assets acquired and liabilities assumed is completed. Accordingly, the final fair value adjustments may be materially different from those presented in this document.

UNAUDITED PRO FORMA COMBINED CONDENSED BALANCE SHEET AS OF MARCH 31, 2011

(Dollars in thousands)	Bay Commercial Bank (Book Value)	Global Trust Bank (Book Value)	Global Bancorp (Book Value)	Merger Adjustments	Pro Forma Company
ASSETS					
Cash & Due From Banks	\$ 2,175	\$ 3,327	\$	\$ (4,367) ^(a)	\$ 1,135
Investment Securities	148				148
Federal Funds Sold	20,263	9,350			29,613
Other Investments	590	14,653		(70) ^(b)	15,173
Total Investments	21,001	24,003		(70)	44,934
Total Loans	137,360	45,694		$(945)^{(c)}$	182,109
Deferred fees, net	(289)				(289)
Loan Loss Allowance	(2,000)	(945)		945	(2,000)
Loans-Net	135,071	44,749			179,820
FF&E	448	412		$(150)^{(d)}$	710
Other Real Estate Owned	1,590				1,590
Deposit Intangible	94			219 ^(e)	313
Deferred Tax Asset	1,120			2,085 ^(f)	3,204
All Other Assets	1,390	494			1,884
TOTAL ASSETS	\$162,889	\$72,985		\$ (2,284)	\$233,590
LIABILITIES Demand Deposits	18,757	5,742	(124)		24,375
NOW	3,647	2,425	(124)		6,072
MMDA & Savings	44,481	24,550			69,031
CD = <100,000	15,661	1,910			17,571
CD=>100,000	46,202	22,607			68,809
Total Deposits	128,748	57,234	(124)		185,858
Borrowings	415	57,251	(121)		415
All Other Liabilities	334	223			557
STOCKHOLDERS' EQUITY					
Stock	32,328	21,000	150	$(12,187)^{(g)}$	41,291
Additional Paid in Capital	265	604		$(604)^{(h)}$	265
Unrealized Gain/Loss	2				2
Retained Earnings (Loss)	903	(5,558)	(23)	5,581 ⁽ⁱ⁾	903
Earnings (Loss)	(106)	(518)	(3)	4,926 ^(j)	4,299
Total Stockholders' Equity	33,392	15,528	124	(2,284)	46,760
TOTAL LIABILITIES & STOCKHOLDERS' EQUITY	\$162,889	\$72,985	\$	\$ (2,284)	\$233,590
Total Shares Outstanding	3,549,794	2,100,001	2,115,001	1,280,480	4,830,274

The following footnotes provide additional explanation of the Merger Adjustments in the foregoing *pro forma*. As used in these footnotes, the term "BCB" refers to Bay Commercial Bank, and "GTB" refers to Global Trust Bank.

^(a) Cash and cash equivalents:	\$ 3,067 1,301 4,367	20% stock purchase (423,000 shares) at \$7.25 per share After tax (34.15% effective tax rate) deal cost (\$2.0 million)
^(b) Investments - discount:	70	Discount on investments in time deposits equal to 90-day penalty.

^(c) Loans portfolio adjustment:	(945)	Assumes the allowance for loan losses is a reasonable estimation of the credit quality discount.			
^(d) Furniture, Fixtures and equipment:	(150)	Discount adjustment for the termination of the 3rd floor lease; write-off of 3rd floor tenant improvements, and write-off other equipment and software related to the discontinued Fiserv system.			
^(e) Core deposit intangible	219	Core deposit premium equal to 1.5% of outstanding non-interest bearing deposits and 0.5% on interest bearing transaction accounts.			
^(f) Deferred tax asset, net	2,452	Estimated total deferred tax asset, net			
	368	Less valuation reserve			
	2,084	Net, deferred tax asset			
		=			
^(g) Common Stock	(21,150)	Elimination of GTB common stock			
	8,963	Issue new BCB common stock at market value			
	(12,187)	1,280,480 shares at BCB market value of \$7.0.			
^(h) Additional paid in capital and retained loss	(604)	Eliminate GTB's additional paid in capital			
	5,581	Eliminate GTB's retained loss			
⁽ⁱ⁾ Earnings	521	Eliminate GTB's current period loss			
	4,405	^(j) Bargain purchase gain, net of expenses			
	4,926	=			
^(j) Bargain purchase gain:	\$ 72,985	Assets acquired			
	(70)	MTM – investments			
	(150) 219	Write-off FF&E			
	2,084	Core deposit premium Deferred tax asset, net			
Net assets acquired	75,069				
Net liabilities assumed	57,333				
Net value of assets acquired	17,736	-			
		- Converted BCB			
Price paid:		<u>GTB shares</u> <u>Shares</u> <u>Market Price</u>			
Stock (BCB market price of \$7.00)	8,963	1,692,001 1,280,480 \$ 7.00			
Cash	3,067				
Total price paid	12,030	-			
Value of assets in excess of price	5,705	-			
After tax deal costs	1,301	_			
Gain net of expenses	\$ 4,405	_			
		-			

UNAUDITED PRO FORMA COMBINED CONDENSED STATEMENT OF EARNINGS FOR THE PERIOD JANUARY 1, 2011 THROUGH MARCH 31, 2011

(Dollars in thousands)	Bay Commercial Bank	Global Trust Bank	Global Bancorp	Merger Adjustment	Pro Forma Company
INTEREST INCOME					
Interest & Fees on Loans	\$2,176	\$ 604			\$2,780
Interest on Securities	2	-			2
Interest on Fed. Funds	17	9			26
Interest on Others	-	81			81
Total Interest Income	\$2,195	\$ 694	-	-	\$2,889
INTEREST EXPENSE					
Interest Paid-NOW	5	5			10
Interest Paid-MMDA & Savings	119	62			181
Interest Paid-TCD =>100,000	185	80			265
Interest Paid-TCD <100,000	49	8			57
Amortization of Deposit Intangible	20			18 ^(k)	38
Other Borrowings	7	-			7
Total Interest Expense	385	155		18	558
NET INTEREST INCOME	\$1,810	\$ 539		\$ (18)	\$2,331
PROVISION FOR LOAN LOSS	933	151		-	1,084
NET INTEREST INCOME AFTER PROVISION FOR LOAN LOSS	906	388		(18)	1,247
NON-INTEREST INCOME					
Service Charges on Deposit Accounts	75	3			78
Other Non-Interest Income	81	(12)		13 ⁽¹⁾	82
TOTAL NON-INTEREST INCOME NON-INTEREST EXPENSE	156	(9)		13	160
Salaries & Employee Benefits	774	505		$(250)^{(m)}$	1,029
Occupancy	154	130		$(23)^{(n)}$	261
Other non-interest Expense	282	262	3	$(150)^{(0)}$	396
Total Non-Interest Expense	\$1,210	\$ 897	\$3	\$(423)	\$1,687
INCOME (LOSS) BEFORE INCOME TAXES	(177)	(518)	(3)	418	(280)
PROVISION (BENEFIT) FOR INCOME TAXES	(71)			(45) ^(p)	(116)
NET OF INCOME (LOSS)	\$ (106)	\$(518)	\$(3)	\$ 463	\$ (164)
NET INCOME (LOSS) PER SHARE – BASIC	\$ (0.03)	· \ - 7	• (*)		\$ (0.03)

^(k) Core deposit premium amortization straight-line over three years.

⁽¹⁾ Adjust loan servicing income.

^(m) Staff reductions; 3 executives; 3 back office support.

⁽ⁿ⁾ Elimination of 3rd floor rent and leasehold costs.

⁽⁰⁾ Reduction in data processing costs, FDIC premiums, professional fees, marketing costs and other miscellaneous expenses.

^(p) Tax impact of cost reductions at 41.5%.

INFORMATION ABOUT BAY COMMERCIAL BANK

Business

Bay Commercial Bank was incorporated under the laws of the State of California on March 24, 2004 and was licensed by the CDFI and commenced operations as a California state-chartered bank on July 20, 2004. The Bank is engaged in substantially all of the business operations customarily conducted by independent community banks in California.

Bay Commercial Bank engages in commercial banking activities with an emphasis on developing a business and professional clientele within its market areas consisting of the greater San Francisco Bay Area with emphasis on the communities surrounding its branch offices in the counties of Contra Costa and Alameda. Bay Commercial Bank focuses on personalized service combined with a full range of traditional core commercial banking products and services for small-to medium-sized businesses and individuals residing in or doing business in the bank's market area. Bay Commercial Bank offers a wide range of commercial banking services, the structure and charges of which are designed to attract the targeted individual and business clientele in its service area.

Bay Commercial Bank offers a wide range of deposit accounts including personal and business checking accounts, money market accounts, and time certificates of deposit. It also offers a full complement of lending products including business-commercial lines of credit, term loans, equipment financing, accounts receivable and inventory financing and other asset-based lending, letters of credit and working capital loans. Bay Commercial Bank also offers an array of real estate loans, including permanent financing and construction loans, loans with fixed and variable rates, and loans secured by a wide variety of properties, including office, multi-family income properties, commercial, land, retail, industrial, hotel, restaurant, mixed use, office condominiums and single family residences. In addition, Bay Commercial Bank may also avail themselves of the bank's on-line banking services and merchants may also use the bank's "scan-a-check" services to make deposits into their accounts from remote locations.

The Bank currently employs 31 full-time employees.

Competition

The banking business in California generally, and in the service area where Bay Commercial Bank is located, is highly competitive with respect to both loans and deposits and is dominated by a relatively small number of major banks which have many offices operating over wide geographic areas. In the bank's primary service area major banks dominate the commercial banking industry. Bay Commercial Bank competes for deposits and loans with other commercial banks, including many of which are much larger than it. Among the advantages which these institutions have over Bay Commercial Bank are their ability to finance extensive advertising campaigns and to allocate their investment assets to regions of highest yield and demand. Many of the major commercial banks operating in the bank's service area offer certain services which are not offered directly by the bank and, by virtue of their greater total capitalization, such banks have substantially higher lending limits than the bank. In addition, in recent years the competition between banks, savings and loan associations and credit unions for the deposit and loan business of individuals has increased. It cannot presently be predicted what effect the increased competition will have on Bay Commercial Bank's continued ability to attract the banking business of the area's businesses, residents, business persons and professionals.

In order to compete with the other financial institutions in its principal marketing area, Bay Commercial Bank uses, to the fullest extent possible, the familiarity of its directors and officers with the bank's primary service area and its residents and businesses and the flexibility which its independent status permits. This includes an emphasis on specialized services, local promotional activity, and personal contacts by the bank's officers, directors, and other employees. Bay Commercial Bank's promotional activities emphasize the advantages of dealing with a locally-owned and headquartered institution more sensitive and responsive to the particular needs of its community. In the event customers require loans in excess of the lending limits permitted to it, the bank attempts to accommodate such loan requirements on a participation basis with its respective correspondent banks. Bay Commercial Bank further seeks to assist its customers in obtaining any required banking or financial services not generally offered by it. In addition, Bay Commercial Bank also is a member of an ATM network and offers, among other services, a night depository, remote deposit capture, courier services, and bank-by-mail services.

Properties

Bay Commercial Bank's head office is located at 1280 Civic Drive, Suite 100, Walnut Creek, California 94596. In addition, Bay Commercial Bank currently operates two additional branch offices, in Oakland and Castro Valley, California. Each office is leased pursuant to an operating lease, the principal terms of which are outlined in the table below.

Date Opened	Address	Approx. Usable Sq. Footage	Current Monthly Obligation	Lease Expiration
2004	1280 Civic Drive, Suite 100 Walnut Creek, California 94596	4,782	\$14,160/mo.	July 2012
2004	155 Grand Avenue, Suite 105 Oakland, California 94612	1,269	\$3,356/mo.	October 2012
2009	3895 E. Castro Valley Boulevard Castro Valley, California 94552	3,520	\$14,106/mo.	May 2013

Management believes that Bay Commercial Bank's present facilities are in good physical condition and are adequately covered by insurance. Certain of the leases noted above contain options to extend the term of the lease. Bay Commercial Bank is confident that, if necessary, it would be able to secure suitable alternative space on similar terms without having a substantial effect on operations.

Securities Authorized for Issuance under Equity Compensation Plan

Bay Commercial Bank's Stock Option Plan was approved by its shareholders. It has no equity compensation plans not previously approved by shareholders. There are currently 341,852 shares of common stock issuable upon exercise of outstanding options and options to purchase 13,127 shares of common stock are still available for future issuances.

Directors and Executive Officers

The following table sets forth certain information regarding the current directors and executive officers of Bay Commercial Bank as of December 31, 2010.

Name	Title/Position	<u>Age (as of</u> <u>December 31,</u> <u>2010)</u>	<u>Year First</u> <u>Elected or</u> <u>Appointed</u>	<u>Principal Occupation During</u> Last Five Years
George J. Guarini	President, CEO and Director	57	2004	Bank Officer
A. Dean Abercrombie	Executive Vice President and Chief Credit Officer	59	2008	Bank Officer
Keary L. Colwell	Executive Vice President and Chief Financial Officer	51	2004	Bank Officer
Janet L. King	Executive Vice President and Chief Operating Officer	48	2004	Bank Officer
Michael J. O'Neill	Executive Vice President and Chief Banking Officer	66	2011	Bank Officer

Name	Title/Position	<u>Age (as of</u> <u>December 31,</u> <u>2010)</u>	<u>Year First</u> <u>Elected or</u> <u>Appointed</u>	Principal Occupation During Last Five Years
James L. Apple	Director	64	2004	Investor/Owner Buckhorn Restaurants
James S. Camp	Director	59	2004	President, SA Camp Co.
Robert G. Laverne, M.D.	Director	62	2004	Doctor
Lloyd W. Kendall, Jr.	Director	64	2004	Retired Attorney
Donald S. Morrow	Director	60	2007	President, Windriver Enterprise
David M. Spatz	Director	63	2004	President, Anyi Lu Real Estate Investor

None of the directors or officers was selected pursuant to any arrangement or understanding other than with the directors and officers of the bank acting within their capacities as such. There are no family relationships between any of the directors and executive officers of the bank. No director or officer of Bay Commercial Bank serves as a director of any company which has a class of securities registered under, or which is subject to the periodic reporting requirements of, the Securities Exchange Act of 1934, or of any company registered as an investment company under the Investment Company Act of 1940.

Compensation of Directors

Bay Commercial Bank's directors are compensated with a retainer in the amount of \$500 per month for attendance at each board meeting and are not separately compensated for committee meetings. See page 88, below, for more detailed information.

Securities Ownership of Officers, Directors, and Principal Shareholders

Bay Commercial Bank is aware of five beneficial owners holding five percent (5%) or more of its common stock as of December 31, 2010 as follows:

Name of <u>Beneficial Owner</u>	Number of Shares of <u>Common Stock</u>	Percentage <u>Owned</u>	Stock Options <u>Granted</u>	<u>Warrants</u>
Black River Ventures	297,385	8.4%	-	-
Starboard Fund for New Bancs	317,167	8.9%	-	29,167
Commerce Street Capital	325,340	9.2%	-	32,534
Ithan Creek Master Investments	322,382	9.1%	-	32,240
JAM Recovery Fund	325,340	9.2%	-	32,534
Total	1,587,614	44.7%	-	126,475

The following table shows the beneficial ownership of directors, executive officers and directors and executive officers as a group as of December 31, 2010:

Name of Beneficial Owner	Number of Shares of Common Stock	Percentage Owned	Stock Options Granted	Warrants
James L. Apple	160,365	4.5%	14,692	<u>vvarrants</u>
Chairman	100,505	1.570	11,092	
George J. Guarini	22,132	0.6%	107,092	-
Director, President, and	,		,	
Chief Executive Officer				
James S. Camp	104,077	2.9%	14,692	-
Director				
Lloyd W. Kendall, Jr.	52,222	1.5%	24,692	-
Director				
Robert G. Laverne, M.D.	65,000	1.8%	14,692	-
Director				
Donald S. Morrow	11,300	0.3%	10,000	-
Director	10.000	1 10/	14 (00	
David M. Spatz	40,300	1.1%	14,692	-
Director	15.040	0.40/	10.000	
A. Dean Abercrombie	15,840	0.4%	10,000	-
Chief Credit Officer and Executive Vice President				
Keary L. Colwell	2,000	0.1%	25,400	
Chief Financial Officer, Secretary,	2,000	0.170	25,400	-
and Executive Vice President				
Janet L. King	1,000	0.0%	25,400	_
Chief Operating Officer and	1,000	0.070	23,400	
Executive Vice President				
Executive vice resident				
Directors and Executive Officers				
as a Group (ten persons)	474,236	13.4%	261,352	-
1 \ 1 /	,	-	,	

⁽¹⁾ In addition to the options to acquire 14,692 shares granted to Mr. Kendall for his services as a director, he was granted an option to acquire 10,000 shares of common stock for legal and consulting services related to the Bank's Bankers Exchange Services Division.

BAY COMMERCIAL BANK: MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion, which analyzes the major elements of Bay Commercial Bank's financial condition and results of operations, should be read in conjunction with the accompanying audited financial statements and notes thereto.

Three Months Ended March 31, 2011

Bay Commercial Bank ended March 31, 2011 with total assets of \$162.9 million, a decrease of \$11.1 million or 6.4% from \$174.0 million as of December 31, 2010. The decline in total assets was primarily funded by a \$13.2 million decrease in cash and cash equivalents due to a decline in total deposits of \$10.7 million.

Total loans increased to \$137.5 million as of March 31, 2011, an increase of \$1.2 million or 0.9% from \$136.2 million as of December 31, 2010. The following table provides the composition of the loan portfolio as of March 31, 2011 and December 31, 2010, respectively.

Composition of Total Loans (amounts in thousands)

Type of Lean	March 31, 2011	% of <u>total</u>	December 31, 2010	% of <u>total</u>
Loans secured by real estate:				
Construction & Development	\$ 13,127	9.6%	\$ 13,281	9.8%
Open-end 1-4 Residential	11,051	8.0	10,492	7.7
Conventional 5+	-	-	-	-
Non-farm, non-residential prop	75,448	54.9	72,939	53.6
Commercial	37,522	27.3	39,252	28.8
Consumer and installment	212	0.2	231	0.2
Total loans	\$137,360	100.0%	\$136,195	100.0%

Non-performing loans totaled \$1.1 million or 0.8% of total loans as of March 31, 2011, which represents two commercial loans and two real estate loans. There were \$1.9 million or 1.4% of total non-performing loans as of December 31, 2010.

Investment securities totaled \$148 thousand as of March 31, 2011, and \$180 thousand as of December 31, 2010.

Total deposits as of March 31, 2011 were \$128.7 million, a decrease of \$10.7 million or 7.7% from \$139.5 million as of December 31, 2010. The decrease in deposits was primarily attributable to deposits related to IRS Code section 1031 Tax Deferred Exchange transactions and maturing time deposits. The following table provides the composition of deposits as of March 31, 2011 and December 31, 2010, respectively.

Composition of Total Deposits (amounts in thousands)

		% of		% of
Type of Deposit	March 31, 2011	total	December 31, 2010	total
Non-interest bearing	\$ 18,757	14.6%	\$ 17,153	12.3%
Interest checking & Savings	4,890	3.8	7,686	5.5
Money market	43,238	33.6	47,260	33.9
Time deposits	61,863	48.0	67,358	48.3
Total deposits	\$128,748	100.0%	\$139,457	100.0%

There are no brokered deposits outstanding as of March 31, 2011 and December 31, 2010.

Shareholders' equity totaled \$33.4 million as of March 31, 2011, a decrease of \$102 thousand or 0.3% from \$33.5 million as of December 31, 2010. This decrease was primarily attributable to the net loss of \$106 thousand.

As of both March 31, 2011 and December 31, 2010, Bay Commercial Bank was considered well-capitalized under the regulatory guidelines. In order for a bank to be considered well-capitalized, it must maintain a Total Capital to Risk-Weighted Asset ratio of 10%, a Tier 1 Capital to Risk-Weighted Asset ratio of 6%, and a Tier 1 Capital to Average Asset ratio (leverage ratio) of 5%. The following table summarizes the capital adequacy for March 31, 2011 and December 31, 2010.

Capital and Ratios (amounts in thousands)

Capital Ratio	March 31, 2011	Ratio	December 31, 2010	Ratio
Total capital (to risk-weighted assets)	\$35,079	24.6%	\$35,155	25.0%
Tier 1 capital (to risk-weighted assets)	\$33,296	23.4%	\$33,379	23.7%
Tier 1 capital (to average assets)	\$33,296	20.1%	\$33,379	20.2%

For the three months ended March 31, 2011, net loss was \$106 thousand as compared to net income of \$288 thousand for the same period of the prior year.

Net interest income was \$1.8 million for the three months ended March 31, 2011, compared to \$1.6 million for the same period of the prior year.

The provision for loan losses was \$933 thousand for the three months ended March 31, 2011, an increase of \$888 thousand or 1,973.5% from \$45 thousand in the same period of the prior year. The additional provision required in the first quarter 2011 was primarily related to the charge-off of two related commercial loans. The allowance for credit losses totaled \$2.0 million or 1.46% of total loans as of March 31, 2011. This compares to \$2.4 million or 1.74% of total loans as of December 31, 2010.

Non-interest income was \$156 thousand for the three months ended March 31, 2011, as compared to \$85 thousand for the same period of the prior year.

Operating expenses totaled \$1.2 million for the three months ended March 31, 2011, an increase of \$106 thousand or 2.7% from the same period of the prior year. The increase is primarily attributable to increased compensation and related expenses. In addition, the Bank experienced a \$20 thousand increase in default and OREO expenses in the three-month period ended March 31, 2011, as compared to the same period of the prior year. Average full-time equivalent staff is at 30 in the three-month period ended March 31, 2011, compared to 26 full-time equivalent staff in the same period of the prior year.

The benefit for income taxes was \$71 thousand in the three months ended March 31, 2011, reflecting an effective tax rate of 40.0%. This compares to a provision of \$205 thousand in the same period of the prior year.

Year Ended December 31, 2010

Bay Commercial Bank had total assets of \$174.0 million as of December 31, 2010, an increase of \$30.7 million or 21.4% from \$143.3 million as of December 31, 2009.

Total loans increased to \$136.2 million as of December 31, 2010, an increase of \$4.4 million or 3.3% from \$131.8 million as of December 31, 2009. This increase was primarily attributable to the origination of commercial real estate loans. The following table provides the composition of the loan portfolio as of December 31, 2010 and December 31, 2009, respectively.

		% of		% of
Type of Loans	December 31, 2010	total	December 31, 2009	total
Loans secured by real estate:				
Construction & Development	\$ 13,281	9.8%	\$ 12,171	9.2%
Open-end 1-4 Residential	10,492	7.7	7,756	5.9
Conventional 5+	-	-	-	-
Non-farm, non-residential prop	72,939	53.6	63,728	48.3
Commercial	39,252	28.8	48,158	36.5
Consumer and installment	231	0.2	7	0.0
Total loans	\$136,195	100.0%	\$131,820	100.0%

Composition of Total Loans (amounts in thousands)

There were \$1.9 million or 1.4% of total loans non-performing as of December 31, 2010. There were \$2.2 million or 1.8% of total loans non-performing assets as of December 31, 2009.

Investment securities totaled \$180 million as of December 31, 2010, a decrease of \$121 thousand or 40.2% from \$301 million as of December 31, 2009. The decrease was primarily attributable to principal repayments during the period.

Total deposits as of December 31, 2010 were \$139.5 million, an increase of \$12.9 million or 10.2% from \$126.6 million as of December 31, 2009. The increase in deposits was primarily attributable to growth in MMDA of \$17.1 million and Interest checking & Savings of \$0.7 million partially offset by declines in DDA of \$2.3 million and Time deposits of \$2.7 million. The following table provides the composition of deposits as of December 31, 2010 and December 31, 2009, respectively.

Composition of Total Deposits (amounts in thousands)

		% of		% of
Type of Deposits	December 31, 2010	total	December 31, 2009	total
Non-interest bearing	\$ 17,153	12.3%	\$ 19,426	15.4%
Interest checking & Savings	7,686	5.5	6,943	5.5
Money market	47,260	33.9	30,165	23.8
Time deposits	67,358	48.3	70,016	55.3
Total deposits	\$139,457	100.0%	\$126,550	100.0%

Shareholders' equity totaled \$33.5 million as of December 31, 2010, an increase of \$17.4 million or 109.3% from \$16.0 million as of December 31, 2009. This increase was primarily attributable to the issuance of 2 million shares of common stock at \$9.00 per share or a total of \$18 million. As of both December 31, 2010 and 2009, Bay Commercial Bank was considered well-capitalized under the regulatory framework for prompt corrective action. In order for a bank to be considered well-capitalized, it must maintain a Total Capital to Risk-Weighted Asset ratio of 10%, a Tier 1 Capital to Risk-Weighted Asset ratio of 5%. The following table summarizes the capital adequacy for December 31, 2010 and 2009.

Capital and Ratios (amounts in thousands)

Capital Ratio	December 31, 2010	<u>Ratio</u>	December 31, 2009	Ratio
Total capital (to risk-weighted assets)	\$35,155	25.0%	\$17,485	12.6%
Tier 1 capital (to risk-weighted assets)	\$33,379	23.7%	\$15,835	11.4%
Tier 1 capital (to average assets)	\$33,379	20.2%	\$15,835	10.4%

For the year ended December 31, 2010, net income was \$621 thousand as compared to \$359 thousand for the year ended December 31, 2009.

Net interest income was \$6.3 million for the year ended December 31, 2010, an increase of \$1.6 million or 34.5% from \$4.7 million for the year ended December 31, 2009. This increase was primarily the result of growth in earning assets and the reduction in deposit interest expense.

The provision for loan losses was \$1.4 million for the year ended December 31, 2010, an increase of \$1.0 million or 232.8% from \$430 thousand for the year ended December 31, 2009. The increase in the provision for loan losses reflects growth within the loan portfolio, the level of non-performing assets and management's assessment of potential losses within the framework of current economic conditions. The allowance for credit losses totaled \$2.4 million or 1.74% of total loans as of December 31, 2010. This compares to \$1.7 million or 1.25% of total loans as of December 31, 2009.

Non-interest income was \$365 thousand for the year ended December 31, 2010, as compared to \$299 thousand for the year ended December 31, 2009. The increase resulted from fees related to Escrow Services and higher deposit-related service charges.

Operating expenses totaled \$4.3 million for the year ended December 31, 2010, an increase of \$323 thousand or 8.2% from \$4.0 million for the year ended December 31, 2009. The increase is primarily attributable to an increase in compensation and related expenses, occupancy and other expenses related to the acquisition of the Castro Valley branch in May 2009.

The provision for income taxes was \$380 thousand for the year ended December 31, 2010 as compared to \$271 thousand for the year ended December 31, 2009 resulting from higher earnings in 2010 compared to 2009.

INFORMATION ABOUT GLOBAL TRUST BANCORP AND GLOBAL TRUST BANK

Business

Global Bancorp and Global Trust Bank were incorporated under the laws of the State of California on May 3, 2007 and July 28, 2008, respectively. Global Trust Bank was licensed by the CDFI and commenced operations as a California state-chartered bank on December 3, 2008 Global Trust Bank is engaged in substantially all of the business operations customarily conducted by independent commercial banks in California.

Global Trust Bank engages in commercial banking activities with an emphasis on developing a business and professional clientele within the Bay Area. It focuses on personalized service combined with a full range of traditional core commercial banking products and services for small to medium sized businesses and individuals residing in or doing business in the immediately surrounding communities. Global Trust Bank offers a wide range of commercial banking services, the structure and charges of which are designed to attract the targeted individual and business clientele in its service area.

Global Trust Bank offers a wide range of deposit accounts which include personal and business checking accounts, money market accounts, and time certificates of deposit. It also offers a full complement of lending products including commercial lines of credit and term loans, SBA loans, credit lines to individuals, professional loans, equipment loans, accounts receivable/inventory financing, and real estate and construction loans.

Global Trust Bank currently employs 15 full-time employees.

Competition

Global Trust Bank differentiates itself from competitors by providing financial products and personal service, convenient, technology-based delivery systems, and providing services to a culturally diverse population. Global Trust Bank is California's only independently owned Indo-American bank and is headquartered in the city of Mountain View.

As of June 2010, there were approximately 50 banks and thrift institutions in the San Jose, Sunnyvale, Cruz, and Santa Clara Metropolitan Statistical Area that Global Trust Bank considers its service area. The majority of deposits in this service area are held by banks that are branches of major nationwide banks. The six largest banks by total deposits are Wells Fargo, Bank of America, Silicon Valley Bank, Comerica, Citibank, and JPMorgan Chase which control over 72% of deposits in this area. Global Trust Bank's share of total deposits in the area as of June 30, 2010 was .07%.

This area is also known for its mixed ethnicity. Several key members of management and the board are of Indian descent. Global Trust Bank offers local convenience and personalized services to the community residents and businesses. The board has a deep commitment to this area and to the concept of a service oriented community bank.

Properties

Global Bancorp and Global Trust Bank's head office and sole branch are located at 700 E. El Camino Real Suite 100, Mountain View, California. This office is leased pursuant to an operating lease, the principal terms of which are outlined in the table below. The lease also contains two options to extend the term of the lease for an additional five years per option.

Date Opened	Address	Approx. Usable Sq. Footage	Current Monthly Obligation	Lease Expiration
12/03/08	700 E. El Camino Real, Suite 110, Mountain View, California	2,885	\$8,293	10/01/13

Management believes that Global Bancorp's and Global Trust Bank's present facilities are in good physical condition and are adequately covered by insurance.

Directors of Global Bancorp

Name and Title Other than Director	Age	Year First Appointed Director	Principal Occupation During the Past Five Years
Vinod K. Thukral	67	2008	Chairman of Global Bancorp and Global Trust Bank, former Professor at Tulane University
Bhupen B. Amin	40	2008	General Counsel and Chief Operating Officer of Lotus Hotels, Inc. in Walnut Creek, CA
Arthur C. Carmichael	70	2008	Managing director CBIZ Insurance Services, San Jose, CA
Harpreet Chaudhary, CPA	49	2008	Certified public accountant, and a Certified financial planner and president of Area Financial Services, Inc. (accounting)
Kamleshwar Gunsagar	68	2008	Executive Vice President and Chief Strategy Officer of Sriya Innovations (greentech space).
Mahendra P. Patel	62	2008	Real Estate Investor and hotelier
Pramod R. Patel	43	2008	President, Raps Hospitality Group (hotelier)

None of the directors was selected pursuant to any arrangement or understanding other than with the directors and officers of Global Bancorp and Global Trust Bank acting within their capacities as such. There are no family relationships between any of the directors and executive officers of Global Bancorp and Global Trust Bank serves as a director of any company which has a class of securities registered under, or which is subject to the periodic reporting requirements of, the Securities Exchange Act of 1934, or of any company registered as an investment company under the Investment Company Act of 1940.

Compensation of Directors

Directors receive no compensation for membership on the boards of directors of Global Bancorp or Global Trust Bank.

Executive Officers

The following table sets forth certain information concerning executive officers of Global Bancorp and Global Trust Bank.

Name	Age	Position and Principal Occupation For the Past Five Years
R. Dale McKinney	65	Senior Vice President & Chief Financial Officer, Global Bancorp and Global Trust Bank.
Robert C. Navarrete	54	Senior Vice President & Chief Credit Officer, Global Trust Bank.

___

Since June 30, 2011, Global Trust Bank has been operating without a president and chief executive officer, but with closer supervision by the board of directors of Global Trust Bank until the consummation of the merger. In the event the merger is not consummated, Global Trust Bank will immediately begin the search for a new president and chief executive officer.

None of the executive officers were selected pursuant to any arrangement or understanding other than with the directors and executive officers of Global Bancorp and Global Trust Bank acting within their capacities as such. There are no family relationships between any of the directors and executive officers of Global Bancorp and Global Trust Bank. There are no material proceedings to which any executive officer of Global Trust Bank or any associate of any executive officer of Global Bancorp and Global Trust Bank is a party or has an interest materially adverse to Global Bancorp and Global Trust Bank.

Securities Ownership of Officers, Directors and Principal Shareholders

Management of Global Bancorp knows of no person who owns, beneficially or of record, either individually or together with associates, five percent or more of the outstanding shares of Global Bancorp's stock, except as set forth in the table below. The following table sets forth, as of May 31, 2011, the number and percentage of shares of outstanding common stock beneficially owned, directly or indirectly, by each of Global Bancorp's and Global Trust Bank's current directors and executive officers and by the directors and executive officers as a group. The shares "beneficially owned" are determined under Securities and Exchange Commission Rules, and do not necessarily indicate ownership for any other purpose. In general, beneficial ownership includes shares over which the director or executive officer has sole or shared voting or investment power and shares which such person has the right to acquire within 60 days of May 31, 2011. Unless otherwise indicated, the persons listed below have sole voting and investment powers of the shares beneficially owned. Management is not aware of any such person having the right to acquire additional shares within 60 days of May 31, 2011, nor of any arrangements which may, at a subsequent date, result in a change of control of Global Bancorp other than the proposed merger.

Name of Owner	Amount and Nature of Beneficial Ownership ^{(1) (2)}	Percent of Class
Bhupen B. Amin	50,000	2.4%
Arthur C. Carmichael	10,000	*
Harpeet Chaudhary, CPA	10,000	*
Kamleshwar Gunsagar	10,000	*
Mahendra P. Patel	25,000	1.2
Pramrod R. Patel	50,000	2.4
Vinod K. Thukral	50,000	2.4
R. Dale McKinney	2,500	*
Robert C. Navarrete	2,500	*
Directors and Executive Officers as a Group (9 Persons)	210,000	9.9

* Less than 1%.

⁽¹⁾ The directors have options to acquire in the aggregate 122,500 shares of Global Bancorp common stock. The directors do not intend to exercise such options prior to the effective time of the merger and if not exercised, such stock options will be cancelled. The exercise prices of such options exceed the current fair market value for Global Bancorp common stock. Due to the expectation that such options will not be exercised, the ownership amounts for each director in the above table exclude the possible beneficial ownership of Global Bancorp common stock for each such director from the exercise of his stock options.

⁽²⁾ Messrs. McKinney and Navarrete hold non-qualified stock options to acquire Global Bancorp common stock in the aggregate amount of 40,000 shares (20,000 shares for each officer). As founding officers, Messrs. McKinney and Navarrete may reach an agreement with Global Bancorp to cancel such options in consideration of a cash payment. If such officers and Global Bancorp do not agree on the terms to cancel such options, then after the effective date of the merger Bay Commercial Bank will grant Messrs. McKinney and Navarrete new non-qualified stock options to acquire Bay Commercial Bank common stock. Therefore, the ownership amounts for these two officers do not include the possible beneficial ownership of Global Bancorp common stock related to their stock options as founding officers. See the sections entitled "SUMMARY" and "JOINT PROPOSAL I – THE MERGERS – Financial Interests in the Merger of Directors and Officers of Global Bancorp and Global Trust Bank" for the details of these option arrangements.

GLOBAL BANCORP AND GLOBAL TRUST BANK: MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Three Months Ended March 31, 2011

Global Bancorp, including Global Trust Bank, ended March 31, 2011 with total assets of \$73.0 million, an increase of \$7.4 million or 11.3% from \$65.6 million as of December 31, 2010. The growth in total assets was primarily funded by a \$7.9 million increase in deposits.

Total loans increased to \$45.7 million as of March 31, 2011, an increase of \$8.3 million or 22.2% from \$37.4 million as of December 31, 2010. The increase was primarily in commercial loans. The following table provides the composition of the loan portfolio as of March 31, 2011 and December 31, 2010, respectively.

Composition of Total Loans (amounts in thousands)						
Type of Loan	March 31, 2011	<u>% of total</u>	December 31, 2010	<u>% of total</u>		
Loans secured by real estate:						
Construction & Development	\$ 3,044	6.7%	\$ 2,476	6.6%		
1–4 Residential	101	.2	100	.3		
Conventional 5+	1,789	3.9	1,794	4.8		
Commercial Real Estate	21,076	46.1	16,394	43.8		
Commercial	18,127	39.7	15,904	42.5		
Consumer and installment	1,557	3.4	730	2.0		
Total loans	\$45,694	100.0%	\$37,398	100.0%		

Global Trust Bank had no non-performing loans as of March 31, 2011 or December 31, 2010. Global Trust Bank also did not have any investment securities as of March 31, 2011 or December 31, 2010.

Total deposits as of March 31, 2001 were \$57.1 million, an increase of \$7.9 million or 16.2% from \$49.2 million as of December 31, 2010. The increase in deposits was primarily attributable to growth in time certificates of deposit of \$4.8 million. The following table provides the composition of deposits as of March 31, 2011 and December 31, 2010, respectively.

Composition of Total Deposit (amounts in thousands)

Type of Deposit	March 31, 2011	<u>% of total</u>	December 31, 2010	<u>% of total</u>
Non-interest bearing	\$ 5,618	9.8%	\$ 4,991	10.2%
Interest checking	2,425	4.3	2,771	5.6
Money market & Savings	24,550	42.9	21,718	44.1
Time deposits	24,517	42.9	19,760	40.1
Total deposits	\$57,110	100.0%	\$49,240	100.0%

Global Trust Bank had no FHLB advances or brokered deposits as of March 31, 2011 or December 31, 2010.

Shareholders' equity totaled \$15.7 million as of March 31, 2011, a decrease of \$.4 million or (2.4%) from \$16.1 million as of December 31, 2010. This decrease was primarily attributable to net operating loss of \$.4 million. As of both March 31, 2011 and December 31, 2010, Global Bancorp and Global Trust Bank were considered well-capitalized under the regulatory guidelines. In order for a bank to be considered well-capitalized, it must maintain a Total Capital to Risk-Weighted Asset ratio of 10%, a Tier 1 Capital to Risk-Weighted Asset ratio of 6%, and a Tier 1 Capital to Average Asset ratio (leverage ratio) of 5%. The following table summarizes the capital adequacy for March 31, 2011 and December 31, 2009.

Capital and Ratios (amounts in thousands)

Capital Ratio	March 31, 2011	<u>Ratio</u>	December 31, 2010	<u>Ratio</u>
Total capital (to risk-weighted assets)	\$16,283	32.5%	\$16,547	39.2%
Tier 1 capital (to risk-weighted assets)	\$15,653	31.2%	\$16,017	38.0%
Tier 1 capital (to average assets)	\$15,653	22.4%	\$16,017	25.2%

For the three months ended March 31, 2011, net operating loss was \$518 thousand as compared to \$501 thousand for the same period of the prior year.

Net interest income was \$539 thousand for the three months ended March 31, 2011, an increase of \$183 thousand or 51.4% from \$356 thousand in the same period of the prior year.

The provision for loan losses was \$151 thousand for the three months ended March 31, 2011, an increase of \$105 thousand or 228.3% from \$46 thousand in the same period of the prior year. The increase in the provision for loan losses reflects management's assessment of potential losses within the framework of current economic conditions. The allowance for credit losses totaled \$945 thousand or 2.07% of total loans at as of March 31, 2011. This compares to \$794 thousand or 2.12% of total loans as of December 31, 2010.

Non-interest income was (\$9) thousand for the three months ended March 31, 2011, as compared to \$7 thousand for the same period of the prior year.

Operating expenses totaled \$900 thousand for the three months ended March 31, 2011, an increase of \$80 thousand or 9.8% from the same period of the prior year. The increase is primarily attributable to increased FDIC insurance and loan administration expenses. Average full-time equivalent staff remained unchanged at 16 in the nine-month period ended March 31, 2011 and in the same period of the prior year.

Due to carry-forward of the net operating loss in previous years, except for the minimum California franchise tax of \$2 thousand, there was no provision for income taxes allocated in the three months ended March 31, 2011. This remained unchanged from the same period of the prior year.

Year Ended December 31, 2010

Global Bancorp, including Global Trust Bank, had total assets of \$65.6 million as of December 31, 2010, an increase of \$14.0 million or 27.1% from \$51.6 million as of December 31, 2009.

Total loans increased to \$37.4 million as of December 31, 2010, an increase of \$13.6 million or 57.1% from \$23.8 million as of December 31, 2009. The increase was primarily in commercial loans. The following table provides the composition of the loan portfolio as of December 31, 2010 and December 31, 2009, respectively.

Type of Loan	December 31, 2010	<u>% of total</u>	December 31, 2009	<u>% of total</u>
Loans secured by real estate:				
Construction & Development	\$ 2,476	6.6%	\$ 0	.0%
1–4 Residential	100	.3	0	.0
Conventional 5+	1,794	4.8	0	.0
Commercial real estate	16,394	43.8	12,805	53.9
Consumer and installment	730	2.0	626	2.6
Commercial	15,904	42.5	10,323	43.5
Total loans	\$37,398	100.0%	\$23,754	100.0%

Composition of Total Loans (amounts in thousands)

Global Trust Bank had no investment securities or non-performing assets as of December 31, 2010 or December 31, 2009.

Total deposits as of December 31, 2010 were \$49.2 million, an increase of \$15.5 million or 45.8% from \$33.8 million as of December 31, 2009. The increase in deposits was primarily attributable to growth in CDs for \$2.7 million; in MMDA & savings for \$8.2 million; in NOW for \$2.4 million and in DDA for \$2.2 million. The following table provides the composition of deposits as of December 31, 2010 and December 31, 2009, respectively.

Composition of Total Deposits (amounts in thousands)

Type of Deposit	December 31, 2010	% of total	December 31, 2009	<u>% of total</u>
Non-interest bearing	\$ 4,991	10.2%	\$ 2,751	8.1%
Interest checking	2,771	5.6	419	1.2
Money market & Savings	21,718	44.1	13,524	40.0
Time deposits	19,760	40.1	17,078	50.6
Total deposits	\$49,240	100.0%	\$33,772	100.0%

Global Trust Bank had no FHLB advances or brokered deposits as of December 31, 2010 and 2009.

Shareholders' equity totaled \$16.1 million as of December 31, 2010, a decrease of \$1.5 million or 8.2% from \$17.6 million as of December 31, 2009. This decrease was primarily attributable to the net operating loss of \$1,570 thousand during 2010. As of both December 31, 2010 and 2009, Global Bancorp and Global Trust Bank were considered well-capitalized under the regulatory framework for prompt corrective action. In order for a bank to be considered well-capitalized, it must maintain a Total Capital to Risk-Weighted Asset ratio of 10%, a Tier 1 Capital to Risk-Weighted Asset ratio of 6%, and a Tier 1 Capital to Average Asset ratio (leverage ratio) of 5%. The following table summarizes the capital adequacy for December 31, 2010 and 2009.

Capital and Ratios (amounts in thousands)

Capital Ratio	December 31, 2010	<u>Ratio</u>	December 31, 2009	<u>Ratio</u>
Total capital (to risk-weighted assets)	\$16,547	39%	\$17,779	54%
Tier 1 capital (to risk-weighted assets)	\$16,017	38%	\$17,457	53%
Tier 1 capital (to average assets)	\$16,017	25%	\$17,457	42%

For the year ended December 31, 2010, net operating loss was \$1,568 thousand as compared to \$2,377 thousand for the year ended December 31, 2009. The variance was primarily due to more provision for loan and lease losses allocated during 2010, and increased operating expenses due to asset growth.

Net interest income was \$1.721 million for the year ended December 31, 2010, an increase of \$1.054 million or 158.1% from \$667 thousand for the year ended December 31, 2009. This increase was primarily the result of growth in earning assets.

The provision for loan losses was \$299 thousand for the year ended December 31, 2010, a decrease of \$176 thousand or 37.1% from \$475 thousand for the year ended December 31, 2009. The decrease in the provision for loan losses reflects management's assessment of potential losses within the framework of current economic conditions. The allowance for credit losses totaled \$794 thousand or 2.12% of total loans as of December 31, 2010. This compares to \$495 thousand or 2.08% of total loans as of December 31, 2009.

Non-interest income was \$395 thousand for the year ended December 31, 2010, as compared to \$525 thousand for the year ended December 31, 2009. The decrease primarily resulted from less fees collected on sale of government guaranteed loans.

Operating expenses totaled \$3.4 million for the year ended December 31, 2010, a slight increase of \$291 thousand or 9.4% from \$3.1 million for the year ended December 31, 2009. Deviations in all major categories were within a small range in between 2009 and 2010.

Due to carry-forward of the net operating loss in previous years, except for the minimum California franchise tax of \$2 thousand, there was no provision for income taxes allocated for the years ended December 31, 2010 and 2009.

COMPARISONS BETWEEN BAY COMMERCIAL BANK AND GLOBAL BANCORP

Comparison of Shareholder Rights

The following discusses some of the similarities and some of the differences in the rights of shareholders of Global Bancorp and Bay Commercial Bank. This discussion is applicable to those shareholders of Global Bancorp who will receive Bay Commercial Bank common stock in the merger and become shareholders of Bay Commercial Bank.

Comparison of Corporate Structure

Global Bancorp has authorized 20,000,000 shares of common stock, \$.01 par value. Bay Commercial Bank has authorized 100,000,000 shares of common stock, no par value and 10,000,000 shares of preferred stock. The shares of common stock of both are fully paid and non-assessable. There are no preemptive or other subscription rights, and there are no conversion rights or redemption or sinking fund provisions with regards to either's common shares. Neither Bay Commercial Bank nor Global Bancorp has any shares of preferred stock outstanding.

After completion of the mergers, Global Bancorp and Global Trust Bank will cease to exist and will be merged with and into Bay Commercial Bank. Global Bancorp's present shareholders who become shareholders of Bay Commercial Bank will be entitled to vote on shareholder matters pertaining to Bay Commercial Bank, such as merger proposals and amendments to the Articles of Incorporation.

Voting Rights

Bay Commercial Bank and Global Bancorp have similar voting rights. As a general rule shareholders of both are entitled to one vote for each share of common stock held. The shareholders of both corporations may cumulate their notes in the election of directors. Neither entity has any supermajority voting provisions.

Dividends

Both Bay Commercial Bank and Global Bancorp are California corporations. Under California law, Global Bancorp would be prohibited from paying dividends unless: (1) its retained earnings immediately prior to the dividend payment equals or exceeds the amount of the dividend; or (2) immediately after giving effect to the dividend (i) the sum of its assets would be at least equal to 125% of its liabilities, or, if the average of its earnings before taxes on income and before interest expense for the two preceding fiscal years was less than the average of its interest expense for the two preceding fiscal years, the current assets of Global Bancorp would be at least equal to 125% of its current liabilities. The primary source of funds with which dividends will be paid to shareholders will come from cash dividends received Global Trust Bank.

Bay Commercial Bank, as a state-chartered bank, is subject to dividend restrictions set forth in California state banking law, as administered by the CDFI. Under such restrictions, it may not pay cash dividends in an amount which exceeds the lesser of the retained earnings of Bay Commercial Bank or its net income for the last three fiscal years (less the amount of distributions to shareholders during that period of time). If the above test is not met, cash dividends may only be paid with the prior approval of the CDFI, in an amount not exceeding the greater of Bay Commercial Bank's retained earnings, net income for its last fiscal year or the amount of its net income for the current fiscal year. Such restrictions do not apply to stock dividends, which generally require neither the satisfaction of any tests nor the approval of the CDFI. Notwithstanding the foregoing, if the CDFI finds that the shareholders' equity is not adequate or that the declarations of a dividend would be unsafe or unsound, the CDFI may order the state bank not to pay any dividends.

Number of Directors

Global Bancorp's bylaws provide for a range of 7 to 13 directors, with the exact number fixed by amendment to the bylaws or by resolution adopted by its board of directors or shareholders. Global Bancorp has currently fixed the number of its directors at seven. Bay Commercial Bank's bylaws provide for a range of 7 to 13 directors, with the exact number fixed by amendment to the bylaws or by resolution adopted by its board of directors or shareholders. Bay Commercial Bank has currently fixed the number of its directors at seven. Following the mergers, three new directors will be added to Bay Commercial Bank's board of directors from the board of directors of Global Bancorp. The board of directors and Bay Commercial Bank have adopted a resolution to fix the number of directors at 10, effective as the effective time of the merger.

Indemnification of Directors and Officers

The articles of incorporation and bylaws of both Global Bancorp/Global Trust Bank and Bay Commercial Bank authorize indemnification of directors, officers and agents to the fullest extent permissible under California law, and authorize the purchase of liability insurance. Both entities have directors' and officers' liability insurance.

SUPERVISION AND REGULATION

Introduction

Banking is a complex, highly regulated industry. The primary goals of the regulatory scheme are to maintain a safe and sound banking system, protect depositors and the Federal Deposit Insurance Corporation's insurance fund, and facilitate the conduct of sound monetary policy. In furtherance of these goals, Congress and the states have created several largely autonomous regulatory agencies and enacted numerous laws that govern banks, bank holding companies and the financial services industry. Consequently, the growth and earnings performance of Global Bancorp, Global Trust Bank and Bay Commercial Bank can be affected not only by management decisions and general economic conditions, but also by the requirements of applicable state and federal statutes, regulations and the policies of various governmental regulatory authorities, including:

- the Federal Reserve Board;
- the Federal Deposit Insurance Corporation; and,
- the California Department of Financial Institutions.

The system of supervision and regulation applicable to Global Bancorp, Global Trust Bank and Bay Commercial Bank governs most aspects of their business, including:

- the scope of permissible business;
- investments;
- reserves that must be maintained against deposits;
- capital levels that must be maintained;
- the nature and amount of collateral that may be taken to secure loans;
- the establishment of new branches;

- mergers and consolidations with other financial institutions; and,
- the payment of dividends.

The following summarizes the material elements of the regulatory framework that apply to these companies. It does not describe all of the statutes, regulations and regulatory policies that are applicable. Also, it does not restate all of the requirements of the statutes, regulations and regulatory policies that are described. Consequently, the following summary is qualified in its entirety by reference to the applicable statutes, regulations and regulatory policies in this joint proxy statement-offering circular. Any change in these applicable laws, regulations or regulatory policies may have a material effect on the business of Global Bancorp, Global Trust Bank and Bay Commercial Bank.

Global Bancorp

Global Bancorp is a bank holding company within the meaning of the Bank Holding Company Act and is registered as such with, and subject to the supervision of, the Federal Reserve Board. It is required to file with the Federal Reserve Board quarterly, semi-annual and annual reports and such additional information as the Federal Reserve Board may require under the Bank Holding Company Act. The Federal Reserve Board may conduct examinations of bank holding companies and their subsidiaries.

Global Bancorp is required to obtain the approval of the Federal Reserve Board before it may acquire all or substantially all of the assets of any bank or ownership or control of the voting shares of any bank if, after giving effect to such acquisition of shares, it would own or control more than 5% of the voting shares of such bank. Prior approval of the Federal Reserve Board is also required for the merger or consolidation of Global Bancorp and another bank holding company.

Global Bancorp is prohibited by the Bank Holding Company Act, except in certain statutorily prescribed instances, from acquiring direct or indirect ownership or control of more than 5% of the outstanding voting shares of any company that is not a bank or bank holding company and from engaging, directly or indirectly, in activities other than those of banking, managing or controlling banks or furnishing services to its subsidiaries. However, it may, subject to the prior approval of the Federal Reserve Board, engage in any, or acquire shares of companies engaged in, activities that are deemed by the Federal Reserve Board to be so closely related to banking or managing or controlling banks as to be acceptable.

The Federal Reserve Board may require that Global Bancorp terminate an activity or terminate control of or liquidate or divest subsidiaries or affiliates when the Federal Reserve Board determines that the activity or the control or the subsidiary of affiliates constitutes a significant risk to the financial safety, soundness or stability of any of its banking subsidiaries. The Federal Reserve Board also has the authority to regulate provisions of certain bank holding company debt, including authority to impose interest ceiling and reserve requirements on such debt. Under certain circumstances, it must file written notice and obtain approval from the Federal Reserve Board prior to purchasing or redeeming its equity securities.

Under the Federal Reserve Board's regulations, a bank holding company is required to serve as a source of financial and managerial strength to its subsidiary banks and may not conduct its operations in an unsafe and unsound manner. In addition, it is the Federal Reserve Board's policy that in serving as a source of strength to its subsidiary banks, a bank holding company should stand ready to use available resources to provide adequate capital funds to its subsidiary banks during periods of financial stress or adversity and should maintain the financial flexibility and capital-raising capacity to obtain additional resources for assisting its subsidiary banks. A bank holding company's failure to meet it obligations to serve as a source of strength to its subsidiary banks will generally be considered by the Federal Reserve Board to be an unsafe and unsound banking practice or a violation of the Federal Reserve Board's regulations or both. Under certain conditions, the Federal Reserve Board may conclude that certain actions of a bank holding company, such as payment of cash dividends, would constitute unsafe and unsound banking practices because they violate the Federal Reserve Board's "source of strength" doctrine.

A bank holding company and its subsidiaries are prohibited from certain tie-in arrangements in connection with any extension of credit, sale or lease of property or furnishing of services. For example, with certain exceptions, a bank may not condition an extension of credit on a promise by its customer to obtain other services by it, its holding company or other subsidiaries, or on a promise by its customer not to obtain services from a competitor. In addition, federal law imposes certain restrictions between Global Bancorp and its subsidiary, Global Trust Bank. As an affiliate of Global Bancorp, Global Trust Bank is subject, with certain exceptions, to provisions of federal law imposing limitations on, and requiring collateral for, extensions of credit by it to its affiliates.

Bay Commercial Bank and Global Trust Bank

As a California state-chartered banks Bay Commercial Bank and Global Trust Bank are subject to primary supervision, examination and regulation by the CDFI and both banks' deposits are insured by the FDIC in accordance with applicable law and regulation. As banks that are not member banks of the Federal Reserve System, the primary federal regulator of Global Trust Bank and Bay Commercial Bank is the FDIC. However as of the date of this joint proxy statement-offering circular Bay Commercial Bank's application to become a member of the Federal Reserve System continues to be considered by the FRB and, if such application for membership is approved, the FRB will become the primary federal regulator of Bay Commercial Bank, but its deposits will continue to be insured by the FDIC. As a consequence of the extensive regulation of commercial banking activities in California and the United States, banks are particularly susceptible to changes in California and federal legislation and regulations, which may have the effect of increasing the cost of doing business, limiting permissible activities or increasing competition.

Various other requirements and restrictions under the laws of the United States and the State of California affect the operations of the banks. Federal and California statutes and regulations relate to many aspects of the banks' operations, including reserves against deposits, interest rates payable on deposits, loans, investments, mergers and acquisitions, borrowings, dividends, branching, capital requirements and disclosure obligations to depositors and borrowers. California law presently permits a bank to locate a branch office in any locality in the state. Additionally, California law exempts banks from California usury laws.

Capital Standards. The FDIC has risk-based capital adequacy guidelines intended to provide a measure of capital adequacy that reflects the degree of risk associated with a banking organization's operations for both transactions reported on the balance sheet as assets, and transactions, such as letters of credit and recourse arrangements, which are reported as off-balance-sheet items. Under these guidelines, nominal dollar amounts of assets and credit equivalent amounts of off-balance-sheet items are multiplied by one of several risk adjustment percentages, which range from 0% for assets with low credit risk, such as certain U.S. government securities, to 100% for assets with relatively higher credit risk, such as business loans.

A banking organization's risk-based capital ratios are obtained by dividing its qualifying capital by its total risk-adjusted assets and off-balance-sheet items. The regulators measure risk-adjusted assets and off-balance-sheet items against both total qualifying capital (the sum of Tier 1 capital and limited amounts of Tier 2 capital) and Tier 1 capital. Tier 1 capital consists of common stock, retained earnings, noncumulative perpetual preferred stock and minority interests in certain subsidiaries, less most other intangible assets. Tier 2 capital may consist of a limited amount of the allowance for loan and lease losses and certain other requirements with some characteristics of equity. The inclusion of elements of Tier 2 capital is subject to certain other requirements and limitations of the federal banking agencies. Since December 31, 1992, the FDIC has required a minimum ratio of qualifying total capital to risk-adjusted assets and off-balance-sheet items of 8%, and a minimum ratio of Tier 1 capital to risk-adjusted assets and off-balance-sheet items of 4%.

In addition to the risk-based guidelines, the FDIC requires banking organizations to maintain a minimum amount of Tier 1 capital to average total assets, referred to as the leverage ratio. For a banking organization rated in the highest of the five categories used by regulators to rate banking organizations, the minimum leverage ratio of Tier 1 capital to total assets is 3%. It is improbable, however, that an institution with a 3% leverage ratio would receive the highest rating by the regulators since a strong capital position is a significant part of the regulators' ratings. For all banking organizations not rated in the highest category, the minimum leverage ratio is at least 100 to 200 basis points above the 3% minimum. Thus, the effective minimum leverage ratio, for all practical purposes, is at least 4% or 5%. In addition to these uniform risk-based capital guidelines and leverage ratios that apply across the industry, the FDIC has the discretion to set individual minimum capital requirements for specific institutions at rates significantly above the minimum guidelines and ratios.

A bank that does not achieve and maintain the required capital levels may be issued a capital directive by the FDIC to ensure the maintenance of required capital levels. As discussed above, Bay Commercial Bank and Global Trust Bank are required to maintain certain levels of capital, as are required of banks generally. The regulatory capital guidelines for the banks follow:

	Requirem	Requirement for the	
	Bank	to be:	
	Adequately	Well	
	Capitalized	Capitalized	
Tier 1 leverage capital ratio	4%	5%	
Tier 1 risk-based capital ratio	4%	6%	
Total risk-based capital ratio	8%	10%	

Prompt Corrective Action. Federal banking agencies possess broad powers to take corrective and other supervisory action to resolve the problems of insured depository institutions, including those institutions that fall below one or more prescribed minimum capital ratios described above. An institution that, based upon its capital levels, is classified as well capitalized, adequately capitalized, or undercapitalized may be treated as though it were in the next lower capital category if the appropriate federal banking agency, after notice and opportunity for hearing, determines that an unsafe or unsound condition or an unsafe or unsound practice warrants such treatment. At each successive lower capital category, an insured depository institution is subject to more restrictions.

In addition to measures taken under the prompt corrective action provisions, commercial banking organizations may be subject to potential enforcement actions by the federal regulators for unsafe or unsound practices in conducting their businesses or for violations of any law, rule, regulation, or any condition imposed in writing by the agency or any written agreement with the agency. Enforcement actions may include the imposition of a conservator or receiver, the issuance of a cease-and-desist order that can be judicially enforced, the termination of insurance of deposits (in the case of a depository institution), the imposition of civil money penalties, the issuance of directives to increase capital, the issuance of formal and informal agreements, the issuance of removal and prohibition orders against institution-affiliated parties and the enforcement of such actions through injunctions or restraining orders based upon a judicial determination that the agency would be harmed if such equitable relief was not granted.

Premiums for Deposit Insurance. The deposit insurance fund of the FDIC insures the banks' customer deposits up to prescribed limits for each depositor. The Federal Deposit Insurance Reform Act of 2005 and the Federal Deposit Insurance Reform Conforming Amendments Act of 2005 amended the insurance of deposits by the FDIC and collection of assessments from insured depository institutions for deposit insurance. The FDIC approved a final rule in 2006 and amended the rule in February 2009 that sets an insured depository institution's assessment rate on different factors that pose a risk of loss to the Deposit Insurance Fund, including the institution's recent financial ratios and supervisory ratings, and level of reliance on a significant amount of secured liabilities or significant amount of brokered deposits (except that the factor of brokered deposits will not be considered for well capitalized institutions that are not accompanied by rapid growth). The FDIC also in February 2009 set the assessment base rates to range between \$0.12 and \$0.16 per \$100.00 of insured deposits on an annual basis. In November 2009, the FDIC approved a final rule to require all insured depository institutions including the banks to prepay three years of FDIC assessments in the fourth quarter of 2009, except in the event such prepayment is waived by the FDIC. While the prepaid assessments are not charged to income for 2009 but rather ratably over three years beginning in 2010, the quarterly amount paid will reduce the cash and liquidity of the banks at year end 2009 and subsequent periods. Due to the significant losses at failed banks and expected losses for banks that will fail, it is likely that FDIC insurance fund assessments on the banks will increase and such assessments may materially adversely affect the profitability of the banks.

Any increase in assessments or the assessment rate could have a material adverse effect on the banks' business, financial condition, results of operations or cash flows, depending on the amount of the increase. Furthermore, the FDIC is authorized to raise insurance premiums under certain circumstances.

The FDIC is authorized to terminate a depository institution's deposit insurance upon a finding by the FDIC that the institution's financial condition is unsafe or unsound or that the institution has engaged in unsafe or unsound practices or has violated any applicable rule, regulation, order or condition enacted or imposed by the institution's regulatory agency. The termination of deposit insurance for the bank would have a material adverse effect on the banks' business, financial condition, results of operations, and/or cash flows.

Federal Home Loan Bank System. The banks are members of the Federal Home Loan Bank of San Francisco (the "*FHLB-SF*"). Among other benefits, each Federal Home Loan Bank ("*FHLB*") serves as a reserve or central bank for its members within its assigned region. Each FHLB is financed primarily from the sale of consolidated obligations of the FHLB system. Each FHLB makes available loans or advances to its members in compliance with the policies and procedures established by the Board of Directors of the individual FHLB. The FHLB-SF utilizes a single class of stock

with a par value of \$100 per share, which may be issued, exchanged, redeemed and repurchased only at par value. As an FHLB member, the banks are required to own FHLB-SF capital stock in an amount equal to the greater of:

- a membership stock requirement with an initial cap of \$25 million (100% of "membership asset value" as defined); or
- an activity based stock requirement (based on percentage of outstanding advances).

Impact of Monetary Policies. The earnings and growth of the banks are subject to the influence of domestic and foreign economic conditions, including inflation, recession and unemployment. The earnings of the banks are affected not only by general economic conditions but also by the monetary and fiscal policies of the United States and federal agencies, particularly the FRB. The FRB can and does implement national monetary policy, such as seeking to curb inflation and combat recession, by its open market operations in United States Government securities and by its control of the discount rates applicable to borrowings by banks from the FRB. The actions of the FRB in these areas influence the growth of bank loans and leases, investments and deposits and affect the interest rates charged on loans and leases and paid on deposits. The FRB's policies have had a significant effect on the operating results of commercial banks and are expected to continue to do so in the future. The nature and timing of any future changes in monetary policies are not predictable.

Extensions of Credit to Insiders and Transactions with Affiliates. The *Federal Reserve Act* and FRB *Regulation O* place limitations and conditions on loans or extensions of credit to:

- a bank's executive officers, directors, and principal shareholders (i.e., in most cases, those persons who own, control or have power to vote more than 10% of any class of voting securities);
- any company controlled by any such executive officer, director, or shareholder; or
- any political or campaign committee controlled by such executive officer, director, or principal shareholder.

Loans and leases extended to any of the above persons must comply with loan-to-one-borrower limits, require prior full board approval when aggregate extensions of credit to the person exceed specified amounts, must be made on substantially the same terms (including interest rates and collateral) as, and follow credit-underwriting procedures that are not less stringent than, those prevailing at the time for comparable transactions with non-insiders, and must not involve more than the normal risk of repayment or present other unfavorable features. In addition, Regulation O provides that the aggregate limit on extensions of credit to all insiders of a bank as a group cannot exceed the bank's unimpaired capital and unimpaired surplus. Regulation O also prohibits a bank from paying an overdraft on an account of an executive officer or director, except pursuant to a written pre-authorized interest-bearing extension of credit plan that specifies a method of repayment or a written pre-authorized transfer of funds from another account of the officer or director at the bank.

Consumer Protection Laws and Regulations. The banking regulatory agencies are focusing greater attention on compliance with consumer protection laws and their implementing regulations. Examination and enforcement have become more intense in nature, and insured institutions have been advised to monitor carefully compliance with such laws and regulations. The banks are subject to many federal and state consumer protection and privacy statutes and regulations, some of which are discussed below.

The Community Reinvestment Act (the "CRA") is intended to encourage insured depository institutions, while operating safely and soundly, to help meet the credit needs of their communities. The CRA specifically directs the federal regulatory agencies, in examining insured depository institutions, to assess a bank's record of helping meet the credit needs of its entire community, including low- and moderate-income neighborhoods, consistent with safe and sound banking practices. The CRA further requires the agencies to take a financial institution's record of meeting its community credit needs into account when evaluating applications for, among other things, domestic branches, mergers or acquisitions, or holding company formations. The agencies use the CRA assessment factors in order to provide a rating to the financial institution. The ratings range from a high of "outstanding" to a low of "substantial noncompliance." In its last examination for CRA compliance, both banks were rated "satisfactory."

The *Equal Credit Opportunity Act* (the "*ECOA*") generally prohibits discrimination in any credit transaction, whether for consumer or business purposes, on the basis of race, color, religion, national origin, sex, marital

status, age (except in limited circumstances), receipt of income from public assistance programs, or good faith exercise of any rights under the Consumer Credit Protection Act.

The *Truth in Lending Act* (the "*TILA*") is designed to ensure that credit terms are disclosed in a meaningful way so that consumers may compare credit terms more readily and knowledgeably. As a result of the TILA, all creditors must use the same credit terminology to express rates and payments, including the annual percentage rate, the finance charge, the amount financed, the total of payments and the payment schedule, among other things.

The *Fair Housing Act* (the "*FH Act*") regulates many practices, including making it unlawful for any lender to discriminate in its housing-related lending activities against any person because of race, color, religion, national origin, sex, handicap or familial status. A number of lending practices have been found by the courts to be, or may be considered, illegal under the FH Act, including some that are not specifically mentioned in the FH Act itself.

The Home Mortgage Disclosure Act (the "HMDA"), in response to public concern over credit shortages in certain urban neighborhoods, requires public disclosure of information that shows whether financial institutions are serving the housing credit needs of the neighborhoods and communities in which they are located. The HMDA also includes a "fair lending" aspect that requires the collection and disclosure of data about applicant and borrower characteristics as a way of identifying possible discriminatory lending patterns and enforcing anti-discrimination statutes.

The *Right to Financial Privacy Act* (the "*RFPA*") imposes a new requirement for financial institutions to provide new privacy protections to consumers. Financial institutions must provide disclosures to consumers of its privacy policy, and state the rights of consumers to direct their financial institution not to share their nonpublic personal information with third parties.

Finally, the *Real Estate Settlement Procedures Act* (the "**RESPA**") requires lenders to provide noncommercial borrowers with disclosures regarding the nature and cost of real estate settlements. Also, RESPA prohibits certain abusive practices, such as kickbacks, and places limitations on the amount of escrow accounts.

Penalties for noncompliance or violations under the above laws may include fines, reimbursement and other penalties. Due to heightened regulatory concern related to compliance with CRA, ECOA, TILA, FH Act, HMDA, RFPA, and RESPA generally, Bay Commercial Bank may incur additional compliance costs or be required to expend additional funds for investments in its local communities.

Recent Legislation and Other Changes. Federal and state laws affecting banking are enacted from time to time, and similarly federal and state regulations affecting banking are also adopted from time to time. The following include some of the recent laws and regulations affecting banking.

The Dodd-Frank Act, signed into law in July 2010 will significantly affect the current regulatory structure and many of the operating activities of financial institutions and their holding companies. The Dodd-Frank Act creates a new interagency council, the Financial System Oversight Council that is charged with identifying and monitoring the systemic risk to the U.S. economy posed by systemically significant, large financial companies, including bank holding companies and non-bank financial companies. The Office of Thrift Supervision will be eliminated, and its powers distributed among the Federal Reserve, the Office of the Comptroller of the Currency, and the FDIC.

Title XIV, known as the Mortgage Reform and Anti-Predatory Lending Act imposes significant new requirements and limitations on the terms of residential mortgage loans and on mortgage loan originators, including brokers and lenders. The legislation also establishes a floor for capital of insured depository institutions that cannot be lower than the standards in effect today, and directs the federal banking regulators to implement new leverage and capital requirements within 18 months that take into account off-balance sheet activities and other risks, including risks relating to securitized products and derivatives.

The Dodd-Frank Act also creates a new Consumer Financial Protection Bureau with broad powers to supervise and enforce consumer protection laws. The Consumer Financial Protection Bureau has broad rulemaking authority for a wide range of consumer protection laws that apply to all banks and savings institutions, including the authority to prohibit "unfair, deceptive or abusive" acts and practices. The Consumer Financial Protection Bureau has examination and enforcement authority over all banks and savings institutions with more than \$10 billion in assets. Banks and savings institutions with \$10 billion or less in assets, which currently includes the Bank, will continue to be examined by their applicable bank regulators.

The legislation also broadens the base for FDIC insurance assessments. Assessments will now be based on the average consolidated total assets less tangible equity capital of a financial institution. The Dodd-Frank Act also permanently increases the maximum amount of deposit insurance for banks, savings institutions and credit unions to \$250,000 per depositor, retroactive to January 1, 2008, and non-interest bearing transaction accounts have unlimited deposit insurance through December 31, 2013. The Dodd-Frank Act also repeals the prohibition on payment of interest on demand deposits.

Although the substance and scope of these regulations cannot be determined at this time, it is expected that the legislation and implementing regulations, will increase the banks' operating and compliance costs.

The Electronic Funds Transfer Act (the "*EFTA*") provides a basic framework for establishing the rights, liabilities, and responsibilities of consumers who use electronic funds transfer ("*EFT*") systems. The EFTA is implemented by the Federal Reserve's Regulation E, which governs transfers initiated through ATMs, point-of-sale terminals, payroll cards, automated clearinghouse (ACH) transactions, telephone bill-payment plans, or remote banking services. Regulation E was amended in January 2010 to require consumers to opt in (affirmatively consent) to participation in the Bank's overdraft service program for ATM and one-time debit card transactions before overdraft fees may be assessed on the consumer's account. Notice of the opt-in right must be provided to all existing and new customers who are consumers, and the customer's affirmative consent must be obtained, before charges may be assessed on the consumer's account for paying such overdrafts.

The new rule provides bank customers with an ongoing right to revoke consent to participation in an overdraft service program for ATM and one-time debit card transactions, as opposed to being automatically enrolled in such a program. The new rule also prohibits banks from conditioning the payment of overdrafts for checks, ACH transactions, or other types of transactions that overdraw the consumer's account on the consumer's opting into an overdraft service for ATM and one-time debit card transactions. For customers who do not affirmatively consent to overdraft service for ATM and one-time debit card transactions, a bank must provide those customers with the same account terms, conditions, and features that it provides to consumers who do affirmatively consent, except for the overdraft service for ATM and one-time debit card transactions.

The mandatory compliance date for the Regulation E amendments is July 1, 2010 provided that the banks may continue to assess overdraft service fees or charges on existing customer accounts prior to August 15, 2010, without obtaining the consumer's affirmative consent. The banks' compliance with the new Regulation E amendments has minimal impact on their revenue from overdraft service fees and non-sufficient funds ("*NSF*") charges, as both banks do not provide automatic overdraft services for ATM and one-time debit card transactions.

On October 22, 2009, the Federal Reserve Board issued a comprehensive proposal on incentive compensation policies intended to ensure that the incentive compensation policies of banking organizations do not undermine the safety and soundness of such organizations by encouraging excessive risk-taking. The proposal, which covers all employees that have the ability to materially affect the risk profile of an organization, either individually or as part of a group, is based upon the key principles that a banking organization's incentive compensation arrangements should (i) provide incentives that do not encourage risk-taking beyond the organization's ability to effectively identify and manage risks, (ii) be compatible with effective internal controls and risk management, and (iii) be supported by strong corporate governance, including active and effective oversight by the organization's board of directors. The proposal also contemplates a detailed review by the Federal Reserve Board of the incentive compensation policies and practices of a number of "large, complex banking organizations." Any deficiencies in compensation practices that are identified may be incorporated into the organization's supervisory ratings, which can affect its ability to make acquisitions or perform other actions. In addition, the proposal provides that enforcement actions may be taken against a banking organization if its incentive compensation arrangements or related risk-management control or governance processes pose a risk to the organization's safety and soundness and the organization is not taking prompt and effective measures to correct the deficiencies. Similarly, on January 12, 2010, the FDIC announced that it would seek public comment through advance notice of rule making on whether banks with compensation plans that encourage risky behavior should be charged at higher deposit assessment rates than such banks would otherwise be charged.

In May 2009 the Helping Families Save Their Homes Act of 2009 was enacted to help consumers avoid mortgage foreclosures on their homes through certain loss mitigation actions including special forbearance, loan modification, pre-foreclosure sale, deed in lieu of foreclosure, support for borrower housing counseling, subordinate lien resolution, and borrower relocation. The new law permits the Secretary of Housing and Urban Development (HUD), for mortgages either in default or facing imminent default, to: (1) authorize the modification of such mortgages; and (2) establish a program for payment of a partial claim to a mortgagee who agrees to apply the claim amount to payment of a mortgage on a 1- to 4-family residence. In implementing the law, the Secretary of HUD is authorized to (1) provide compensation to the mortgage for lost income on monthly mortgage payments due to interest rate reduction; (2) reimburse the mortgage from a guaranty fund in connection with activities that the mortgage is required to undertake concerning repayment by the mortgagor of the amount owed to HUD; (3) make payments to the mortgage on behalf of the borrower, under terms defined by HUD; and (4) make mortgage modification with terms extended up to 40 years from the modification date. The new law also authorizes the Secretary of HUD to: (1) reassign the mortgage to the mortgage; (2) act as a Government National Mortgage Association (GNMA, or Ginnie Mae) issuer, or contract with an entity for such purpose, in order to pool the mortgage into a Ginnie Mae security; or (3) resell the mortgage in accordance with any program established for purchase by the federal government of insured mortgages. The new law also amends the Foreclosure Prevention Act of 2008, with respect to emergency assistance for the redevelopment of abandoned and foreclosed homes (neighborhood stabilization), to authorize each state that has received certain minimum allocations and has fulfilled certain requirements, to distribute any remaining amounts to areas with homeowners at risk of foreclosure or in foreclosure without regard to the percentage of home foreclosures in such areas.

Also in May 2009, the Credit Card Act of 2009 was enacted to help consumers and ban certain practices of credit card issuers. The new law allows interest rate hikes on existing balances only under limited conditions, such as when a promotional rate ends, there is a variable rate or if the cardholder makes a late payment. Interest rates on new transactions can increase only after the first year. Significant changes in terms on accounts cannot occur without 45 days' advance notice of the change. The new law bans raising interest rates on customers based on their payment records with other unrelated credit issuers (such as utility companies and other creditors) for existing credit card balances, though card issuers would still be allowed to use universal default on future credit card balances if they give at least 45 days' advance notice of the change. The new law allows consumers to opt out of certain significant changes in terms on their accounts. Opting out means cardholders agree to close their accounts and pay off the balance under the old terms. They have at least five years to pay the balance. Credit card issuers will be banned from issuing credit cards to anyone under 21; unless they have adult co-signers on the accounts or can show proof they have enough income to repay the card debt. Credit card companies must stay at least 1,000 feet from college campuses if they are offering free pizza or other gifts to entice students to apply for credit cards.

The new law requires card issuers to give card account holders "a reasonable amount of time" to make payments on monthly bills. That means payments would be due at least 21 days after they are mailed or delivered. Credit card issuers also may not set early morning or other arbitrary deadlines for payments. When consumers have accounts that carry different interest rates for different types of purchases payments in excess of the minimum amount due must go to balances with higher interest rates first. Consumers must "opt in" to over-limit fees. Those who opt out would have their transactions rejected if they exceed their credit limits, thus avoiding over-limit fees. Fees charged for going over the limit must be reasonable. Finance charges on outstanding credit card balances are computed based on purchases made in the current cycle rather than going back to the previous billing cycle to calculate interest charges. Fees on credit cards cannot exceed 25% of the available credit limit in the first year of the card. Credit card issuers must disclose to cardholders the consequences of making only minimum payments each month, namely how long it would take to pay off the entire balance if users only made the minimum monthly payment. Issuers must also provide information on how much users must pay each month if they want to pay off their balances in 36 months, including the amount of interest.

On February 17, 2009, the American Recovery and Reinvestment Act of 2009 ("ARRA") was enacted to provide stimulus to the struggling U.S. economy. ARRA authorizes spending of \$787 billion, including about \$288 billion for tax relief, \$144 billion for state and local relief aid, and \$111 billion for infrastructure and science. In addition, ARRA includes additional executive compensation restrictions for recipients of funds from the U.S. Treasury under the Troubled Assets Relief Program of the Emergency Economic Stimulus Act of 2008 ("EESA"). The provisions of EESA amended by the ARRA include (i) expanding the coverage of the executive compensation limits to as many as the 25 most highly compensated employees of a Troubled Assets Relief Program ("TARP") funds recipient and its affiliates for certain aspects of executive compensation limits and (ii) specifically limiting incentive compensation of covered executives to one-third of their annual compensation which is required to be paid in restricted stock that does not vest until all of the TARP funds are no longer outstanding (note that if TARP warrants remain outstanding and no other TARP instruments are outstanding, then such warrants would not be considered outstanding for purposes of this incentive compensation restriction. In addition, the board of directors of any TARP recipient is required under EESA, as amended to have a company-wide policy regarding excessive or luxury expenditures, as identified by the U.S. Treasury, which may include excessive expenditures on entertainment or events; office and facility renovations; aviation or other transportation services; or other activities or events that are not reasonable expenditures for staff development, reasonable performance incentives, or other similar measures conducted in the normal course of the business operations of the TARP recipient.

EESA, as amended by ARRA, provides for a new incentive compensation restriction for financial institutions receiving TARP funds. The number of executives and employees covered by this new incentive compensation restriction depends on the amount of TARP funds received by such entity. For community banks that have or will receive less than \$25 million, the new incentive compensation restriction applies only to the highest paid employee. This new incentive compensation restriction prohibits a TARP recipient from paying or accruing any bonus, retention award, or incentive compensation during the period in which any TARP obligation remains outstanding, except that such prohibition shall not apply to the payment of qualified long-term restricted stock by such TARP recipient and grandfathered bonuses.

In addition, EESA as amended by ARRA provides that for any TARP recipient which is an SEC reporting company, must, in its annual meeting materials, include a nonbinding shareholder approval proposal of executive compensation. In addition, shareholders are allowed to present other nonbinding proposals with respect to executive compensation.

ARRA also provides \$730 million to the SBA and makes changes to the agency's lending and investment programs so that they can reach more small businesses that need help. The funding includes:

- \$375 million for temporarily eliminating fees on SBA-backed loans and raising SBA's guarantee percentage on some loans to 90%;
- \$255 million for a new loan program to help small businesses meet existing debt payments; and,
- \$30 million for expanding SBA's Microloan program, enough to finance up to \$50 million in new lending and \$24 million in technical assistance grants to micro-lenders.

On February 10, 2009, the U.S. Treasury, the Federal Reserve Board, the FDIC, the Office of the Comptroller of the Currency, and the Office of Thrift Supervision all announced a comprehensive set of measures to restore confidence in the strength of U.S. financial institutions and restart the critical flow of credit to households and businesses. This program is intended to restore the flows of credit necessary to support recovery.

The core program elements include:

- A new Capital Assistance Program to help ensure that our banking institutions have sufficient capital to withstand the challenges ahead, paired with a supervisory process to produce a more consistent and forward-looking assessment of the risks on banks' balance sheets and their potential capital needs;
- A new Public-Private Investment Fund on an initial scale of up to \$500 billion, with the potential to expand up to \$1 trillion, to catalyze the removal of legacy assets from the balance sheets of financial institutions. This fund combines public and private capital with government financing to help free up capital to support new lending;
- A new Treasury and Federal Reserve initiative to dramatically expand—up to \$1 trillion—the existing Term Asset-Backed Securities Lending Facility (TALF) in order to reduce credit spreads and restart the securitized credit markets that in recent years supported a substantial portion of lending to households, students, small businesses, and others; and,
- A new framework of governance and oversight to help ensure that banks receiving funds are held responsible for appropriate use of those funds through stronger conditions on lending, dividends and executive compensation along with enhanced reporting to the public.

In October 2008, the President signed the *EESA*, in response to the global financial crisis of 2008 authorizing the United States Secretary of the Treasury with authority to spend up to \$700 billion to purchase distressed assets, especially mortgage-backed securities, under TARP and make capital injections into banks under the Capital Purchase Program. EESA gives the government the unprecedented authority to buy troubled assets on balance sheets of financial institutions under the Troubled Assets Relief Program and increased the limit on insured deposits. Some of the other provisions of EESA are as follows:

• accelerated from 2011 to 2008 the date that the Federal Reserve Bank could pay interest on deposits of banks held with the Federal Reserve to meet reserve requirements;

- to the extent that the U.S. Treasury purchases mortgage securities as part of TARP, the Treasury shall implement a
 plan to minimize foreclosures including using guarantees and credit enhancements to support reasonable loan
 modifications, and to the extent loans are owned by the government to consent to the reasonable modification of
 such loans;
- limits executive compensation for executives for TARP participating financial institutions including a maximum corporate tax deduction limit of \$500,000 for each of the top five highest paid executives of such institution, requiring claw backs of incentive compensation that were paid based on inaccurate or false information, limiting golden parachutes for involuntary and certain voluntary terminations to 2.99x their average annual salary and bonus for the last five years, and prohibiting the payment of incentive compensation that encourages management to take unnecessary and excessive risks with respect to the institution;
- extends the mortgage debt forgiveness provision of the Mortgage Forgiveness Debt Relief Act of 2007 by three years (2012) to ease the income tax burden on those involved with certain foreclosures; and,
- qualified financial institutions may count losses on FNMA and FHLMC preferred stock against ordinary income, rather than capital gain income.

In California, the enactment of AB329 in 2009, the Reverse Mortgage Elder Protection Act of 2009 prohibits a lender or any other person who participates in the origination of the mortgage from participation in, being associated with, or employing any party that participates in or is associated with any other financial or insurance activity or referring a prospective borrower to anyone for the purchase of other financial or insurance products; and imposes certain disclosure requirements on the lender.

The enactment of AB1160 in 2009, requires a supervised financial institution in California that negotiates primarily in any of a number of specified languages in the course of entering into a contract or agreement for a loan or extension of credit secured by residential real property, to deliver, prior to the execution of the contract or agreement, and no later than three business days after receiving the written application, a specified form in that language summarizing the terms of the contract or agreement; provides for administrative penalties for violations; and requires the California Department of Corporations and the Department of Financial Institutions to create a form for providing translations and make it available in Spanish, Chinese, Tagalog, Vietnamese, and Korean.

The enactment of SB306 makes specified changes to clarify existing law related to filing a notice of default on residential real property in California, including (among other things): clarifying that the provisions apply to mortgages and deeds of trust recorded from January 1, 2003 through December 31, 2007, secured by owner-occupied residential real property containing no more than four dwelling units; revising the declaration to be filed with the notice of default; specifying how the loan servicers have to maximize net present value under their pooling and servicing agreements applies to certain investors; specifying how and when the notice to residents of property subject to foreclosure is to be mailed; and extending the time during which the notice of sale must be recorded from 14 to 20 days. The bill also makes certain changes related to short-pay agreements and short-pay demand statements.

On February 20, 2009, Governor Schwarzenegger signed ABX2 7 and SBX2 7, which established the California Foreclosure Prevention Act. The California Foreclosure Prevention Act modifies the foreclosure process to provide additional time for borrowers to work out loan modifications while providing an exemption for mortgage loan servicers that have implemented a comprehensive loan modification program. Civil Code section 2923.52 requires an additional 90-day period beyond the period already provided before a Notice of Sale can be given in order to allow all parties to pursue a loan modification to prevent foreclosure of loans meeting certain criteria identified in that section.

It is impossible to predict with any degree of accuracy the competitive impact the laws and regulations described above will have on commercial banking in general and on the business of the parties in particular, or to predict whether or when any of the proposed legislation and regulations will be adopted. It is anticipated that banking will continue to be a highly regulated industry. Additionally, if experience is any indication, there appears to be a continued lessening of the historical distinction between the services offered by financial institutions and other businesses offering financial services. Finally, the trend toward nationwide interstate banking is expected to continue as a result of the enactment of interstate banking and branching laws. As a result of these factors, it is anticipated that banks will experience increased competition for deposits and loans and, possibly, further increases in their cost of doing business.

FOR SHAREHOLDERS OF BAY COMMERCIAL BANK ONLY

The following two proposals are items to be voted upon at the Annual Meeting of Shareholders of Bay Commercial Bank and are to be voted upon only by shareholders of Bay Commercial Bank

Voting Procedures

Please refer to information previously set forth above with respect to eligibility to vote, the Record Date, and revocation of your proxy. Each holder of record of Bay Commercial Bank common stock is entitled to one vote, in person or by proxy, for each share of common stock of Bay Commercial Bank held as of the Record Date, except that shareholders may have cumulative voting rights with respect to the election of directors. See "Cumulative Voting," below.

Unless otherwise instructed, each valid proxy returned which is not revoked will be voted FOR the election as directors of the nominees named in this joint proxy statement-offering circular and FOR the ratification of the selection of the independent accounting firm, and at the proxy holders' discretion, on such other matters, if any, that may come before the annual meeting (including any proposal to adjourn the annual meeting). At this time, the Board of Directors is not aware of any other matters to come before the annual meeting other than those matters set forth in this joint proxy statement-offering circular.

With respect to the election of persons to the board of directors of Bay Commercial Bank, the validly-nominated nominees for election as directors who rank first, second, third, fourth, fifth, sixth, and seventh in number of votes received, will be elected as directors, even if some or all of such nominees receive less than a majority of the total votes.

Ratification of the selection of Moss Adams LLP as independent auditors for Bay Commercial Bank for the year ending December 31, 2011 by the board of directors requires a vote "FOR" such proposal by a majority of the votes cast.

Adoption of a proposal to adjourn the annual meeting to a later date requires that a majority of the votes cast at the annual meeting must vote in favor of the proposal to adjourn the annual meeting. Accordingly, abstentions will have the same effect as voting against the proposal to adjourn the annual meeting. Broker non-votes will have no effect on this proposal.

Votes Required

The following paragraphs explain the vote required for each proposal. In each case, a quorum must be present for the vote to be valid. A majority of the shares entitled to vote, represented in person or by proxy, shall constitute a quorum.

PROPOSAL A: ELECTION OF DIRECTORS. The validly-nominated nominees for election as directors who rank first, second, third, fourth, fifth, sixth, and seventh in number of votes received, will be elected as directors, even if some or all of such nominees receive less than a majority of the total votes.

PROPOSAL B: RATIFICATION OF SELECTION OF INDEPENDENT ACCOUNTING FIRM FOR 2011. Approval of this proposal requires an affirmative vote of the holders of a majority of the shares of the common stock represented and voting.

Such other matters, if any, as may properly come before the annual meeting will generally require the affirmative vote of the holders of a majority of the shares of the common stock represented.

With regard to the election of directors, votes may be cast in favor or withheld; votes that are withheld will be excluded from the vote and will have no effect. Abstentions may be specified on all proposals other than the election of directors and will be counted as shares that are present or represented at the annual meeting for purposes of determining a quorum on the proposal on which the abstention is specified. However, because ratification of the selection of the independent accounting firm requires the affirmative vote of the holders of a majority of the shares of the common stock represented and voting, abstentions will have the effect of a negative vote.

Under applicable California law, broker non-votes are counted for the purpose of determining the presence or absence of a quorum for the transaction of business but are not otherwise counted. Therefore, broker non-votes will have no effect on the outcome of the election of directors or the ratification of the selection of the independent accounting firm.

Cumulative Voting

Cumulative voting allows a shareholder to cast for any candidate a number of votes greater than the number of votes that the shareholder normally is entitled to cast. A shareholder may cumulate votes either (i) by giving one candidate a number of votes equal to the number of directors to be elected multiplied by the number of votes to which that shareholder's shares are normally entitled or (ii) by distributing the shareholder's votes on the same principle among as many candidates as the shareholder sees fit No shareholder can cumulate votes unless, prior to the annual meeting, the shareholder has given notice of the intent to cumulate. If any shareholder has given notice to cumulate, then all shareholders may cumulate their votes for candidates in nomination. The seven candidates receiving the highest number of votes shall be elected. The board of directors does not, at this time, intend to give such notice or to cumulate the votes it may hold pursuant to the proxies solicited herein unless the required notice by a shareholder is given. Therefore, discretionary authority to cumulate votes in such event is solicited in this joint proxy statement-offering circular.

Annual Report

A copy of the 2010 Annual Audited Financial Report as of December 31, 2010 for Bay Commercial Bank (the "*Bay Commercial Bank 2010 Annual Report*") is attached as <u>Appendix G</u>. Additional copies of the Bay Commercial Bank 2010 Annual Report are available without cost upon request by writing to Keary L. Colwell, Chief Financial Officer, Bay Commercial Bank, 1280 Civic Drive, Suite 100, Walnut Creek, California 94596.

Market Makers

Bay Commercial Bank's Common Stock is traded on the OTC Bulletin Board under the symbol BCML.OB. A list of the brokerage firms making a market in our common stock are listed on the Bank's website at <u>www.baycommercialbank.com</u>.

Security Ownership of Certain Beneficial Owners and Management

Please refer to the information regarding the security ownership of certain beneficial owners and management of Bay Commercial Bank commencing at page 64.

Please also refer to the table showing the beneficial ownership of directors, executive officers and directors and executive officers as a group as of December 31, 2010, set forth above at page 65.

Board of Directors

The bylaws of Bay Commercial Bank provide that the number of directors of the Bank shall be not less than 7 nor more than 13 and that the board of directors may fix the number of directors within that range by resolution. The number of directors is currently fixed at seven. Each director holds office until the next annual meeting of shareholders and until such director's successor is duly elected and qualified, or until his death, resignation, or removal.

No director, nominee for director or executive officer of Bay Commercial Bank has any family relationship with any other director or executive officer or director or executive officer of Bay Commercial Bank.

The board of directors has nominated each of the currently serving directors of Bay Commercial Bank for reelection at the annual meeting. See "Proposal A: Election of Directors." Accordingly, no vacancy on the Board of Directors will exist upon the election of directors pursuant to Proposal A.

Committees of the Board of Directors

The board of directors of Bay Commercial Bank held 11 regularly scheduled and seven annual meetings in 2010. During 2010, Director Apple attended 71% of all board meetings. All other directors attended at least 75% of (i) all board meetings and (ii) all meetings of committees of the board on which the director served during the period in which he served.

The Executive Committee of the board of directors presently includes Directors Apple, Guarini, Spatz, and Camp. The Executive Committee is responsible for considering the qualifications of individuals to serve as directors and recommending a slate of directors for election at the annual meeting. The Executive Committee also generally has the power to act for the full board of directors between board meetings. The Executive Committee met once in 2010.

The Human Resources/Compensation Committee presently includes Directors Spatz and Camp. This committee has responsibility for all personnel and compensation policy matters pertaining to bank employees, officers, and directors. It also monitors the Bay Commercial Bank's compliance with laws and regulations applicable uniquely to the protection of employees and officers. This committee met four times in 2010.

The Audit Committee, comprised of Directors Kendall, Apple, and Laverne, provides assistance to the Board of Directors in fulfilling its responsibilities with respect to the oversight of the integrity of the financial statements and systems of internal controls, the independent auditors' qualifications, independence and performance and the performance of the internal audit function. This committee also is responsible for monitoring compliance with regulations, and monitoring the relationship with the primary regulators of Bay Commercial Bank, the FDIC and the CDFI. This committee met 11 times in 2010.

The Loan Committee presently includes Directors Apple, Spatz, Morrow, and Guarini. This committee examines and approves loans above a specified size and monitors regular reviews of the entire loan portfolio. It is also responsible for lending, credit policies, and monitors compliance with such policies. This committee met 39 times in 2010.

The Asset Liability Committee, comprised of Directors Apple, Kendall, Laverne, and Camp, is responsible for asset liability management and investment policies. The committee also monitors liquidity, interest rate risk, and securities activities. This committee met 11 times in 2010.

Board of Directors' Compensation

Bay Commercial Bank pays each director other than Mr. Guarini (referred to below as "outside directors") a retainer of \$500 per month for attendance at board meetings and committee meetings. Director fees paid in 2010 totaled \$36,000, or \$6,000 for each outside director. Directors Kendall and Apple are the only outside directors who currently participate in our health insurance plan, and they do so at entirely their own cost. We no longer allow other outside directors to join our health insurance plan. Directors also may be granted stock options under our 2004 Stock Option Plan.

Executive Officers

Each executive officer is appointed annually by the board of directors in accordance with the bylaws of Bay Commercial Bank. The following are all of the current executive officers of the Bank, their occupations for the previous five years, ages and the lengths of service as an officer.

(See description of Mr. Guarini's position with Bay Commercial Bank, and his background under the heading "BAY COMMERCIAL BANK PROPOSAL A – ELECTION OF DIRECTORS" below.)

A. DEAN ABERCROMBIE.

Mr. Abercrombie has served as Executive Vice President and Chief Credit Officer of Bay Commercial Bank since October 2008. Prior to joining Bay Commercial Bank he served as Chief Credit Officer for Community Bank of the Bay from 2002 to 2005, and from 2005 to October 2008 he held senior lending positions at Bank of Petaluma and Wells Fargo Bank. At December 31, 2010, Mr. Abercrombie was 59 years of age.

KEARY L. COLWELL

Ms. Colwell has served as Executive Vice President, Chief Financial Officer and Secretary of Bay Commercial Bank since its inception in March 2004. Prior to joining Bay Commercial Bank she served as Executive Vice President, Chief Financial Officer and Secretary of Bank of San Francisco. At December 31, 2010, Ms. Colwell was 51 years of age.

JANET L. KING

Ms. King has served as Executive Vice President and Chief Operating Officer of Bay Commercial Bank since March 2004. Prior to joining Bay Commercial Bank, she served as Chief Branch Administrative Officer of Circle Bank. At December 31, 2010, Ms. King was 48 years of age.

MICHAEL J. O'NEILL

Mr. O'Neill joined Bay Commercial Bank as Executive Vice President and Chief Banking Officer in February 2011. Prior to joining Bay Commercial Bank, he was a Team Leader and Private Client Advisor for Bank of America from 2006–2010. At December 31, 2010, Mr. O'Neill was 66 years of age.

Executive Compensation

Bay Commercial Bank's compensation philosophy is to pay for performance as an important way to encourage the actions necessary to achieve the Bank's strategic objectives of increasing profitability and maximizing shareholder value.

Bay Commercial Bank's compensation philosophy is implemented through its compensation program, which is structured to:

- promote annual operating objectives,
- promote long-term strategic plans,
- ensure continuity of management,
- recognize the level of management expertise,
- be competitive within the industry and community, and
- provide internal equity.

Bay Commercial Bank's compensation program includes base salary, annual bonus, a stock option plan, a severance plan, and other benefits. Bay Commercial Bank has entered into employment agreements with each of Mr. Guarani, Mr. Abercrombie, Ms. Colwell, and Ms. King. See "Employment Agreements."

Base Salary. Generally, Bay Commercial Bank targets base salary at median to high competitive levels. The competitive levels are based on comparable positions in other banks. In addition, Bay Commercial Bank takes other factors into consideration including an individual's specialized expertise, level of experience, broad range of expertise allowing the executive to assume multiple responsibilities, historical performance and salary requirements, leadership and the community, and contract terms.

For 2010, Mr. Guarini, Mr. Abercrombie, Ms. Colwell, and Ms. King earned salaries of \$206,553, \$136,484, \$160,991, and \$136,484, respectively. For 2011, the base annual salaries payable to Mr. Guarini, Mr. Abercrombie, Ms. Colwell, and Ms. King are \$250,000, \$165,000, \$165,000, and \$165,000, respectively.

Annual Bonus. The purpose of the annual bonus is to provide incentive for achieving defined target performance levels based on Bay Commercial Bank's annual business and profit plan. The annual goals typically include objectives regarding earnings, loan and deposit growth, asset quality, operating efficiency and regulatory examinations. Annual bonus awards are determined based on performance and the performance of the individual executive.

For 2010, Mr. Guarini, Mr. Abercrombie, Ms. Colwell, and Ms. King earned annual incentive bonuses of \$49,573, \$21,837, \$25,222, and \$21,837, respectively.

Stock Option Plan. The purpose of the stock option plan is to serve as a long-term incentive program by directly tying rewards to the long-term success and increases in shareholder value. Many of the options granted to the executive officers were granted as an inducement to attract and retain executives with the required talent and experience to manage Bay Commercial Bank. All stock option grants are approved by the full board of directors. See "Other Employee Benefit Plans-Stock Option Plan."

Information regarding the number of other stock options held by our executive officers is provided under "Other Employee Benefit Plans-Stock Option Plan."

Severance Benefits. The purpose of the severance benefits is to promote continuity of management. Pursuant to his or her employment agreement, each executive is eligible for severance if terminated without cause, including following a change in control. See "Employment Agreements."

Other Benefits. The executive officers are entitled to participate in all employee benefit plans including the Bank's vacation, 401(k), and welfare and other benefit plans. Each executive officer is entitled by contract to four weeks of annual vacation. The welfare and benefits plans include workers' compensation benefits, medical and dental, life insurance, and long-term disability insurance. Pursuant to his or her employment agreement, each executive is entitled to a term life insurance benefit, payable to his or her designated beneficiary, in an amount equal to the executive's thencurrent base annual salary.

Decisions on the compensation of Bay Commercial Bank's executive officers are generally made by the Human Resources/Compensation Committee, the members of which are outside directors of Bay Commercial Bank. All decisions by this committee relating to the compensation of executive officers are reviewed by Bay Commercial Bank's full Board of Directors.

Employment Agreements

Employment Agreement with Mr. Guarini. Bay Commercial Bank has entered into a three-year employment agreement with Mr. Guarini dated April 21, 2010. The agreement provides for, among other things, an annual base salary of at least \$206,553, incentive bonuses, a monthly automobile allowance of \$800 and group insurance benefits, as well as a term life insurance benefit payable to Mr. Guarini's designated beneficiary in an amount equal to Mr. Guarini's thencurrent annual base salary and participation in any retirement, profit-sharing, salary deferral, medical expense reimbursement, and other similar plans Bay Commercial Bank may establish for its employees. The agreement generally provides for indemnification of Mr. Guarini to the maximum extent permitted by law and applicable regulations for any expenses incurred by him, and for any judgments, awards, fines, or penalties imposed against him, in any proceeding relating to his actions (or the Bank's actions) while an agent of Bay Commercial Bank. The agreement also provides for the advancement of expenses to Mr. Guarini and coverage under a director and officer liability insurance policy.

The agreement provides that if Bay Commercial Bank terminates the agreement without cause, it must, for the remainder of the contract term (but in no event for less than 12 months), continue to pay Mr. Guarini one-twelfth of his then-current base annual salary plus one-twelfth of the amount of the incentive payment made to him during the preceding year. Bay Commercial Bank must also, for the remainder of the contract term (but not more than 24 months), continue to provide him with health insurance benefits on the same terms as when he was employed. In addition, if, within one year of a change in control, Mr. Guarini's employment is terminated without cause or if Mr. Guarini terminates his employment for "good reason," then he will be entitled to a severance payment equal to two times his then-current base annual salary plus two times any incentive payment made to him in the preceding year. The term "good reason" means any of the following: (1) a material permanent reduction in Mr. Guarini's total compensation or benefits; (2) material permanent reduction in Mr. Guarini's principal office so that his commute distance is increased by more than 40 miles from Walnut Creek, California.

Employment Agreements with Mr. Abercrombie, Ms. Colwell and Ms. King. Bay Commercial Bank has entered into employment agreements with Mr. Abercrombie, Ms. Colwell and Ms. King, dated April 21, 2010. Each is for a term of three-year terms. Each agreement provides for, among other things, a minimum annual base salary (\$136,848 in the case of Mr. Abercrombie, \$160,991 in the case of Ms. Colwell and \$136,484 in the case of Ms. King), incentive bonuses, a monthly automobile allowance (\$375 in the case of Mr. Abercrombie and Ms. King and \$500 in the case of Ms. Colwell) and group insurance benefits, as well as a term life insurance benefit payable to the executive's designated beneficiary in an amount equal to the executive's then-current annual base salary and participation in any retirement, profit-sharing, salary deferral, medical expense reimbursement, and other similar plans Bay Commercial Bank may establish for its

employees. Each agreement generally provides for indemnification of the executive to the maximum extent permitted by law and applicable regulations for any expenses incurred by him or her, and for any judgments, awards, fines, or penalties imposed against him or her, in any proceeding relating to his or her actions (or Bay Commercial Bank's actions) while an agent of Bay Commercial Bank. The agreement also provides for the advancement of expenses to the executive and coverage under a director and officer liability insurance policy.

Each agreement provides that if Bay Commercial Bank terminates the agreement without cause, Bay Commercial Bank must, for 12 months thereafter, continue to pay the executive one-twelfth of his or her then-current base annual salary plus one-twelfth of the amount of the incentive payment made to him or her during the preceding year. Bay Commercial Bank must also, for the remainder of the contract term (but not more than 12 months), continue to provide the executive with health insurance benefits on the same terms as when he or she was employed by Bay Commercial Bank. In addition, if, within one year of a change in control, the executive's employment is terminated without cause or if the executive terminates his or her employment for "good reason," then he or she will be entitled to a severance payment equal to one times his or her then-current base annual salary plus one times any incentive payment made to him or her in the preceding year. The term "good reason" means any of the following: (1) a material permanent reduction in the executive's title or responsibilities; or (3) a relocation of the executive's principal office so that his or her commute distance is increased by more than 40 miles from Walnut Creek, California.

Other Employee Benefit Plans

401(k) Profit Sharing Plan. In 2005, Bay Commercial Bank established a 401(k) Profit Sharing Plan (the "401(k) Plan") which permits each participating employee with a minimum service requirement to contribute to the Plan through payroll deductions (the "salary deferral contributions") of up to the maximum amount allowable by law, thereby deferring taxes on all or a portion of these amounts. Under the 401(k) Plan, Bay Commercial Bank presently does not match a participant's tax deferred contributions.

Bay Commercial Bank may make matching and employer contributions to the 401(k) Plan in such amounts as may be determined by the board of directors. Any such contribution vests 100% after a participant has completed three years of service, provided that any such contribution which has not yet vested will vest upon the participant's attainment of age 65 or upon the participant's death or permanent disability. Bay Commercial Bank may also make additional, special contributions to the 401(k) Plan, which vest immediately. Participants are entitled to receive their salary deferral contributions and vested benefits under the 401(k) Plan upon termination of employment, retirement, death, or disability. Participants have the right to self-direct all of their salary deferral contributions among all funds sold by Charles Schwab and Company.

2004 Stock Option Plan. In 2004, Bay Commercial Bank's Board of Directors adopted, and its shareholders approved, the 2004 Stock Option Plan (the "Stock Option Plan"). The Stock Option Plan provides for the granting by the board of directors of incentive stock options (within the meaning of Internal Revenue Code section 422) to employees and non-qualified stock options to employees, non-employee directors and, in connection with the formation, organizing consultants or other incorporators. Factors considered by the board of directors in granting options to officers and employees include the performance of Bay Commercial Bank, the employee's or officer's job performance, the importance of his or her position, and his or her contribution to the organization's goals for the award period. Generally, stock options must be granted at an exercise price of not less than 100% of the fair market value of the shares on the date of grant and have an exercise period of not longer than ten years. No options were granted in 2010. At December 31, 2010, total options outstanding under the Stock Option Plan were 341,852.

The following table sets forth the unexercised stock options held by the executive officers as of December 31, 2010. No options were exercised during 2010 by the executive officers.

Name	Number of Securities Underlying <u>Options</u>	Value of Unexercised <u>Options</u> ⁽²⁾
President/CEO George J. Guarini ⁽¹⁾	107,092	\$ -
EVP/CCO A. Dean Abercrombie	10,000	6,700
EVP/CFO Kearly L. Colwell	25,400	-
EVP/COO Janet L. King	25,400	-

⁽¹⁾ Mr. Guarini was granted non-qualifying stock options to acquire 14,692 shares of Common Stock for his role as an organizer and director.

Certain Relationships and Related Transactions

Bay Commercial Bank has had and expects to continue to have banking transactions with many of its directors and executive officers (and their associates). Loans to any director or executive officer of Bay Commercial Bank or any of its subsidiaries (or any associate of such persons) have been made in the ordinary course of business on substantially the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with other persons, and were judged not to involve more than the normal risk of collection or present other unfavorable features. Loans by Bay Commercial Bank to any director, executive officer or principal shareholder of Bay Commercial Bank or any of its subsidiaries (as such persons are defined by regulation) are subject to limitations under California and federal law. Among other things, a loan to a director, executive officer or principal shareholder of Bay Commercial Bank or any of its subsidiaries must be on non-preferential terms and, if all loans to a given person exceed \$25,000, such loans must be approved in advance by the board of directors. Bay Commercial Bank had nine such loans totaling \$3.5 million outstanding, \$3.4 million in undisbursed loan commitments, and \$14,500 in letters of credit as of December 31, 2010, all of which were performing in accordance with their terms as of that date.

In 2006, Bay Commercial Bank opened its Bankers Exchange Services Division. Through this division, Bay Commercial Bank acts as a qualified intermediary in transactions qualifying under the Internal Revenue Code section 1031, Tax Deferred Exchanges. In connection with the establishment of this division, Mr. Kendall agreed to assist the Bank with setting it up and to provide it with referral and legal services support. In return, Bay Commercial Bank agreed to pay him a nominal commission-based fee for referrals, granted him a stock option and pay half of the costs of his health insurance coverage. As a result of a substantial decline in transaction volume, Bay Commercial Bank discontinued paying Mr. Kendall referral fees during 2008 and discontinued paying for any of the costs of his health insurance coverage in September 2009. For additional information regarding the compensation provided to Mr. Kendall, see "Board of Directors' Compensation."

Since 2007, Bay Commercial Bank has maintained a separate referral arrangement with a company controlled by Mr. Kendall that specializes in assisting clients with tax deferred exchanges. Under this arrangement, Bay Commercial Bank has agreed to refer to Mr. Kendall's company certain types of exchange transactions that Bay Commercial Bank is unable to transact itself, for which Mr. Kendall's company pays Bay Commercial Bank a fee. No transactions were referred by Bay Commercial Bank, and no fees were paid to Bay Commercial Bank, during 2010. The board of directors of Bay Commercial Bank (excluding Mr. Kendall) believes that this arrangement is on terms that are fair and reasonable to Bay Commercial Bank and that are at least as favorable to Bay Commercial Bank as comparable transactions with unrelated parties.

⁽²⁾ Fair market value is calculated based on a bid price of \$6.99 per share as of December 31, 2010 less the underlying exercise price per share of in-the-money options. None of the options held were in-the-money as of December 31, 2010.

BAY COMMERCIAL BANK PROPOSAL A – ELECTION OF DIRECTORS

Seven directors are to be elected at the annual meeting, each to hold office until the 2012 annual meeting of shareholders and until his respective successor is duly elected and qualified, or until his death, resignation, or removal. The nominees proposed by the board of directors for election are: Messrs. Apple, Camp, Guarini, Laverne, Kendall, Morrow, and Spatz. Each is presently serving as a director. In the event that any of the nominees for election as director become unavailable, which is not expected, it is intended that, pursuant to the accompanying proxy, votes will be cast for such substitute nominee or nominees as may be designated by the board of directors, unless the board of directors reduces the number of directors.

The following sets forth, as to each nominee for election as a director of Bay Commercial Bank, such person's age, principal occupation during at least the last five years, and the period during which each person has served as a director of Bay Commercial Bank.

JAMES L. APPLE

Mr. Apple has been the Managing General Partner of Aspen Plaza Ltd., Napa Town Center, Ltd., and Valley Center Towers, Ltd., all of which are real estate partnerships engaging in retail and office development, for more than five years. At December 31, 2010, Mr. Apple was 64 years of age. He has served as a director since 2004.

JAMES S. CAMP

Mr. Camp has been the President of the S. A. Camp Companies for more than five years. At December 31, 2010, Mr. Camp was 59 years of age. He has served as a director since 2004.

GEORGE J. GUARINI

Mr. Guarini has been President and Chief Executive Officer of Bay Commercial Bank since its incorporation in March 2004. Prior to joining Bay Commercial Bank, Mr. Guarini held several senior lending positions at financial institutions in California. At December 31, 2010, Mr. Guarini was 57 years of age. He has served as a director since 2004.

LLOYD W. KENDALL, JR.

Mr. Kendall is a practicing attorney, specializing in income tax and real estate law. Since 1982, he has been the President and Chief Executive Officer of Exchange Support Services, Inc., which assists clients in completing tax-free exchanges. Until 2007, Mr. Kendall also served as the President and Chief Executive Officer of Lawyers Asset Management, Inc., which also assisted clients in completing tax-free exchanges. At December 31, 2010, Mr. Kendall was 64 years of age. He has served as a director since 2004.

ROBERT R. LAVERNE, M.D.

Dr. Laverne has been an anesthesiologist at John Muir Medial Center in Walnut Creek, California for more than five years. At December 31, 2010, Dr. Laverne was 62 years of age. He has served as a director since 2004.

DONALD S. MORROW

Mr. Morrow is the Chief Executive Officer and owner of Wind River Enterprises, Inc., DBA North Bay Auto Auction, an auto auction house in Fairfield, California. Mr. Morrow was appointed to fill a vacant director seat in December 2007. At December 31, 2010, Mr. Morrow was 60 years of age.

DAVID M. SPATZ

Mr. Spatz retired as a senior executive from Chevron Corporation in 2000. Presently, he is a real estate investor, and owns and manages several real estate properties. He has been the President of Anyi Lu International, Inc., a manufacturer of designer shoes, since 2005. At December 31, 2010, Mr. Spatz was 63 years of age. He has served as a director since 2004.

The Board of Directors recommends a vote FOR the election of each of the nominees named above.

BAY COMMERCIAL BANK PROPOSAL B: RATIFICATION OF SELECTION OF INDEPENDENT AUDITORS FIRM FOR 2011

The firm of Moss Adams LLP, independent public accountants, was appointed by the board of directors in 2011 to audit the financial statements of the Bank for 2011.

The selection of an independent accounting firm to provide audit services for Bay Commercial Bank has been approved annually by the board of directors and ratified by the shareholders. Shareholders are again being asked to act upon a proposal to ratify the board's selection of Moss Adams LLP to audit the financial statements of Bay Commercial Bank for 2011.

Moss Adams LLP has advised Bay Commercial Bank that one or more of its representatives will be present at the annual meeting to make a statement if they so desire and to respond to appropriate questions.

The Board of Directors recommends a vote FOR this proposal.

BAY COMMERCIAL BANK OTHER PROPOSED ACTION

The board of directors is not aware of other business which will come before the annual meeting, other than as set forth in this joint proxy statement-offering circular, but if any such matters are properly presented, proxies solicited hereby will be voted on such matters in the proxy holder's discretion.

BAY COMMERCIAL BANK

Klow 4

Keary L. Colwell Chief Financial Officer and Secretary

Walnut Creek, California July 22, 2011

FOR SHAREHOLDERS OF GLOBAL BANCORP ONLY

The following proposal is an item to be voted upon at the Annual Meeting of Shareholders of Global Bancorp and are to be voted upon only by shareholders of Global Bancorp

Voting Procedures

Please refer to information previously set forth above with respect to eligibility to vote, the record date of June 30, 2011, and revocation of your proxy. Each holder of record of Global Bancorp common stock is entitled to one vote, in person or by proxy, for each share of common stock of Global Bancorp held as of June 30, 2011, except that shareholders may have cumulative voting rights with respect to the election of directors. See "Cumulative Voting," below.

Unless otherwise instructed, each Global Bancorp valid proxy returned which is not revoked will be voted FOR the election as directors of the nominees named in this joint proxy statement-offering circular and at the proxy holders' discretion, on such other matters, if any, that may come before the annual meeting (including any proposal to adjourn the annual meeting). At this time, the Global Bancorp Board of Directors is not aware of any other matters to come before the annual meeting other than those matters set forth in this joint proxy statement-offering circular.

Adoption of a proposal to adjourn the annual meeting to a later date requires that a majority of the votes cast at the annual meeting must vote in favor of the proposal to adjourn the annual meeting. Accordingly, abstentions will have the same effect as voting against the proposal to adjourn the annual meeting. Broker non-votes will have no effect on this proposal.

Votes Required

The following paragraphs explain the vote required for the election of directors. A quorum must be present for the vote on the election of directors to be valid. A majority of the shares of Global Bancorp common stock entitled to vote, represented in person or by proxy, shall constitute a quorum.

Such other matters, if any, as may properly come before the annual meeting will generally require the affirmative vote of the holders of a majority of the shares of the common stock represented.

With regard to the election of directors, votes may be cast in favor or withheld; votes that are withheld will be excluded from the vote and will have no effect. With respect to the election of persons to the board of directors of Global Bancorp, the seven validly-nominated nominees for election as directors who receive the most votes will be elected as directors.

Abstentions may be specified on all proposals other than the election of directors and will be counted as shares that are present or represented at the annual meeting for purposes of determining a quorum on the proposal on which the abstention is specified.

Under applicable California law, broker non-votes are counted for the purpose of determining the presence or absence of a quorum for the transaction of business but are not otherwise counted. Therefore, broker non-votes will have no effect on the outcome of the election of directors or the ratification of the selection of the independent accounting firm.

Cumulative Voting

Cumulative voting allows a shareholder to cast for any candidate a number of votes greater than the number of votes that the shareholder normally is entitled to cast. A shareholder may cumulate votes either (i) by giving one candidate a number of votes equal to the number of directors to be elected multiplied by the number of votes to which that shareholder's shares are normally entitled or (ii) by distributing the shareholder's votes on the same principle among as many candidates as the shareholder sees fit. No shareholder can cumulate votes unless, prior to the annual meeting, the shareholder has given notice of the intent to cumulate. If any shareholder has given notice to cumulate, then all shareholders may cumulate their votes for candidates in nomination. The seven candidates receiving the highest number of votes shall be elected. The board of directors does not, at this time, intend to give such notice or to cumulate the votes

it may hold pursuant to the proxies solicited herein unless the required notice by a shareholder is given. Therefore, discretionary authority to cumulate votes in such event is solicited in this joint proxy statement-offering circular.

Annual Report

A copy of the 2010 Annual Audited Financial Report as of December 31, 2010 for Global Bancorp (the "*Global Bancorp 2010 Annual Report*") is attached as <u>Appendix H</u>. Additional copies of the Global Bancorp 2010 Annual Report are available without cost upon request by writing to Vinod Thukral, Chairman, Global Bancorp, 700 E. El Camino Real, Suite 100, Mountain View, California 94040.

Security Ownership of Certain Beneficial Owners and Management

Please refer to the information regarding the security ownership of certain beneficial owners and management of Global Bancorp set forth at page 71.

Please also refer to the table showing the beneficial ownership of directors, executive officers and directors and executive officers as a group as of May 31, 2011, set forth above at page 71.

Board of Directors

The bylaws of Global Bancorp provide that the number of directors of the Bank shall be not less than 7 nor more than 13 and that the board of directors may fix the number of directors within that range by resolution. The number of directors is currently fixed at seven. Each director holds office until the earliest of the next annual meeting of shareholders and until such director's successor is duly elected and qualified, the effective date of the merger or until his death, resignation, or removal.

No director, nominee for director or executive officer of Global Bancorp has any family relationship with any other director or executive officer or director or executive officer of Global Bancorp.

The board of directors has nominated each of the currently serving directors of Global Bancorp, except for Mr. Jim Wall for re-election at the annual meeting. See "Global Bancorp Proposal A: Election of Directors."

GLOBAL BANCORP PROPOSAL A – ELECTION OF DIRECTORS

Seven directors are to be elected at the annual meeting, each to hold office until the 2012 annual meeting of shareholders and until his respective successor is duly elected and qualified, or until his death, resignation, or removal or until the effective date of the merger. The nominees proposed by the board of directors for election are: Messrs. Bhupen Amin, Arthur Carmichael, Harpeet Chaudhary, Kameleshwar Gusagar, Mahendra Patel, Pramod Patel, and Vinod Thukral. Each is presently serving as a director. In the event that any of the nominees for election as director become unavailable, which is not expected, it is intended that, pursuant to the accompanying proxy, votes will be cast for such substitute nominee or nominees as may be designated by the board of directors, unless the board of directors reduces the number of directors.

The following sets forth, as to each nominee for election as a director of Global Bancorp, such person's age, principal occupation during at least the last five years, and the period during which each person has served as a director of Global Bancorp.

Name and Title Other than Director	Age	Year First Appointed Director	Principal Occupation During the Past Five Years
Vinod K. Thukral	67	2008	Chairman of Global Bancorp and Global Trust Bank, former Professor at Tulane University
Bhupen B. Amin	40	2008	General Counsel and Chief Operating Officer of Lotus Hotels, Inc. in Walnut Creek, CA
Arthur C. Carmichael	70	2008	Managing director CBIZ Insurance Services, San Jose, CA

Name and Title Other than Director	Age	Year First Appointed Director	Principal Occupation During the Past Five Years
Harpreet Chaudhary, CPA	49	2008	Certified public accountant, and a Certified financial planner and president of Area Financial Services, Inc. (accounting)
Kamleshwar Gunsagar	68	2008	Executive Vice President and Chief Strategy Officer of Sriya Innovations (greentech space).
Mahendra P. Patel	62	2008	Real Estate Investor and hotelier
Pramod R. Patel	43	2008	Hotelier

The Global Bancorp Board of Directors recommends a vote FOR the election of each of the nominees named above.

GLOBAL BANCORP OTHER PROPOSED ACTION

The board of directors is not aware of other business which will come before the annual meeting, other than as set forth in this joint proxy statement-offering circular, but if any such matters are properly presented, proxies solicited hereby will be voted on such matters in the proxy holder's discretion.

GLOBAL BANCORP

Muth Thent

Vinod Thukral Chairman

July 22, 2011

APPENDIX A

Agreement and Plan of Reorganization and Merger dated May 12, 2011, by and among Global Bancorp, Global Trust Bank and Bay Commercial Bank

AGREEMENT AND PLAN

OF REORGANIZATION AND MERGER

BY AND AMONG

BAY COMMERCIAL BANK,

GLOBAL BANCORP

AND

GLOBAL TRUST BANK

GWH-Agreenates and Plan EXECUTION COPY 5-11-11.

Lable of Contents

ARTRUE I.	DEPINITIONS
ARTICLE 2.	THE MERGER
2.1	The Merger
2.2	Effect of Downstream Merger and Merger
23	Articles of Inconjoration and Bylaws
24	Bank and Company Common Stock
2.5	Conversion of GT Bank Common Stork
2.6	Election and Protating Procedures
2.7	Company Stock Options
2.8	Dissenters' Rights
2,9	Board of Directors of GT Bank and Bank toRowing the Dowastream
	Morger Effective Time and Effective Time
1.10	Executive Officers of GT Bank and the Bank following the Downstream
	Merger Effective Time and Effective Time
ARTICLED.	REPRESENTATIONS AND WARRANTES OF BANK
1,1	Organization, Corporate Power, Etc
12	Scenses and Permits
3.3	Subsaharas
20.5	Authorization of Agreement, No Conflicts
3.5	Capital Structure
3.6	Bank Filings
3.7	Accuracy of information Supplies
38	Compliance with Applicable Laws
39	Litigation
510	Agreements with Banking Authorities
311	Insusance
3.40	Title to Assets other than Real Property
3.13	Real Property
5.14	Jaxes
515	Performance of Obligations
3.16	Brokers and Finders
3.17	Absence of Material Adverse Effect
346	Dedisclosed Liabilities
319	 Liability Linde: Regulation C. Truth in Lending Law and HMDA
3.20	Bank Secrety Act
3.2)	Community Reinvestment Act
3.22	Loant and I westments
3.23	 Employees, Employee Berefit Plans; ERISA
5.24	Hazardovs Materials
3 25	Risk Management Instruments
3.26	Aurounting Readers

Page

Table of Contents

	Page
3.27	Material Contracts
3.28	Performance of Obligations in a manufacture in a manufact
3,29	Powers of Attorney
3.30	Parachute Payments
331	Corporate Records
3.52	Accounting and Tax Matters
3 13	Conor mity Reinvestment Actuaction and a construction and 31
3.34	Community Reinvosment Act
	REPRESENTATIONS AND WARRANTIES OF COMPANY AND GT BANK
4.	Organization, Corporate Power; Etc.,
4.2	Lipenses and Permity
43	Subsidiaries,
4.4	Authorization of Agreement: No Conflicts
4,5	Capital Supervised and a statistic statistic of the statistic statistics of the statistic statistics and and and a statistic statistics and a statistic statistic statistics and a statistic statistics and a statistics an
4.6	Company Filings
4.7	Accuracy of Information Supplied
4.8	Compliance, with Applie able Laws
4,9	Litigator
410	Agreements with Bonking Authorities
4.1	Insurance
4.82	Title to Assets other than Real Property
4.63	Real Property
4.14	[axcs
4.15	Loans and Investments
4.16	Brokers and Finders
417	Material Contracts
a 18	Absence of Material Adverse Effect
4.19	Performance of Cb igations
4.30	 Employees; Employee Benefit Plans; ERJSA annual annual annual annual annual annual 39
4.21	Powers of Attomes and a contract of the contract of the second seco
4.22	Hezardous Matemaia
4.23	Parachole Payments
4.24	 Risk Management Instruments
4.25	Lizbidity Unite: Regulation C, Trath in Londing Law and rIMDA
4 26	Condisclosed Elapidaties
4 27	Bank Secrecy Act
4.78	Accounting Records
4.29	Corporate Records
4.30	Accounting and Tax Matters
4.31	Columnately Reinvestment Act
431	Community Reinvestment Act.,
431	Community Reinvestment Action and a community Reinvestment Action and a community Reinvestment Action and a community and a commun

Table of Contents

Page

.

	ADDITIONAL AGREEMENTS	
5.5		.45
\$.2	Sharehoider Approval	
5.3	Falong of Necessary Actions and a summing and a summing and	
5.4	Joint Proxy Statement and Offering Circular and Applications	46
5.5	Expenses	
5.6	Sufficiation of Cestain Faculty	48
5.7	Updated Schedules	. 48
58	Additional Accusate/Appraisals	- 44
5.9	Employee Plans.	49
10310384	CONTRACT OF DUICTURE O	40
¢i.!	Afformative Concret of the Company and Bank	
6.2	Negative Covenants of the Company and GT Bank	
6.3	Atternative Conduct of Bank	
6.4	Negalive Covenants of Bank	
6.5	Access to Operations	DT
ARTICLE 3.	CONDITIONS PRECEDENT TO CLOSING	62
7.1	Conditions to the Parties' Obligations	62
7.2	Conditions to Company's and GT Bank's Obligations	
73	Conditions to Bank's Oldigations	
	THE REPORT OF A DESCRIPTION OF A STREET AS A DESCRIPTION	
ANG (G.).2-8.	TERMENATION, AMENDMENTS AND WRIVERS	
\$.2	Terraination of Agreement	
8.2 8.3	Effect of Terminalion	
6.) 84	Weiver of Craditions.	
	Funce Majone	
8.5	Moneyary Consequences of Tengaration	
8.5	Effect of Termination; Survival	67
ARTICLE 9.	EMPLOYEE BENEFITS	68
91	Suppoyee Benefits	68

Table of Contents

ARTICLE IO.	GENERAL PROVISIONS	З
10.3	Nonsurvival of Representations and Warranties	3
10.2	Notices	i8
10.3	Counterparts	9
10.4	Entire Agreement/No Third Party Rights/Assignment	ß
10.5	Nondisclosure of Agreement	0
	Confidentiality	Ň
0.7	, · · · · · · · · · · · · · · · · · · ·	80
10.8	Headings/Table of Contents	
10-9	Enforcement of Agreement	
	Severability	
10.11	Attorneys' Fees	10

<u>Pag</u>e

.

.

List of Schedgles

29	List of GT Bank Directors forning Bank Board of Directors
3.3	Bard: Subsidiary
3.4.2	Conflict with, Violation of Default, or Loss of Material Britefit under Articles of Incorporation or Bylaws
3.5	Options
3.6	Bank Filmgs
38	Investigation or Review by Governmental Enfity
3.9	Litigation
3.11	Invarian
3 12	Title to Assets other than Real Property
0.13	Real Property
3,14,1	Filing of Tax Returns
3 14.2	Payment of Taxes
3.14.3	Audit History
3.14.4	Statute of Limitations
3.15	Performance of Obligations
336	Successed Finders
3.18	Undisclosed Liabilities
3,19	Liability order Regulation C, YIL, and HMDA
3.20	Back Secrecy Act
3.22	Leans and Investments
3.23	binp:0yee, Employee Benefit Plans; ERISA
3.23.1	Compensation Obligations
0.23/2	List of Plans
3.23 3	Employee Searcht Plan subject to ERISA
3.23.4	Cortain Benefit Arrangements
3.23.5	Plans and Benefits Compliance
3.23.6	Plan Liabilities and Contingent Liabilities
3.23.8	Certain Other Plan Benefit Accorgenizats
3.24	Hazardaus Methors
4.4.2	Conflict with, Violation of Default, of Loss of Material Bench, under
	Articles of Incorporation or Bylaws
4.5	Options Preferred Stock and Womatile
4.6	Company Filings
4 8	lavestigation or Review by Covernationtal Entity
4.9	Litigation
4.)0	Agreements with Banking Authoritors
401	Insurance
412	Title to Assets other than Real Property
4.13	Real Property
4 (4.)	Filong of Tax, Returns
4 (4 2	Payment of Taxes
4 14.3	Audit History
414.4	Statute of Limitations

List of Schedules

- 4.14.5 Withholding Obligations
- 4.15 Loans and Investments
- 4.15 Brokers and Finders
- 4.17 Material Conton is
- 4.19 Performance of Obligations
- 4.14.9 Basis in Assets, Net Operating Loss, Net Capital Loss, Unused Credit,
 - Unused Tax, Excess Charitable Contributions, and Deferred Gain or Loss
- 4.20.1 Compensation Obligations
- 4.20.2 List of Mans
- 4.20.3 Employee Benefit Plan subject to ERISA
- 4 20 4 Certain Hought Ansangements
- 4.20.1 Plans and Benefits Compliance
- 4.20.6 Pian Lisbilities and Contingent Liabilities
- 4.20.8 Certain Other Plan Benefit Arrangements
- 4.22 Hazardous Materials
- 4.25 L shility under Reputation C, TH, and HMDA
- 4.26 Undisclosed Erabilities
- 6.2.11 Approved Payments
- 6.2.14 Nongualified Stock Option Grants to be Mede by Bank

List of Fahib is

- 2.1 Downstream Merger Agreement
- 2.2 Marger Agreement
- 7.2.3 Director Shareholder Agreement
- 7.3.3 Director-Sourcholder Agreement

AGREEMENT AND PLAN OF REORGANIZATION AND MERGER

This Agreement and Plan of Reorganization and Metger (this "Agreement") is entered into as of May E2, 2011, by and among Boy Commercial Bank, a California state chartered backing corporation ("Company") and Global Trust Bank, a California Cancelanceed bank and a wholly-owned subsidiary of Company ("OT Bank.").

RECITALS

WHEREAS, the respective Boards of Directors of Bank, Company and GT Bank have each determined that it is in the best interests of Bank, Company and GT Bank and then requestive shareholder(s) for the merger of the Company with and into GT Bank ("Dowostream Merger") and the vibio-paratimetrics of GT Bank with and into the Bank ("Morger"), upon the terms and subject to the conditions set forth in this Agreement and is accordance with federal and state laws;

WHEREAS, each of the Boards of Directors of Bank, Company and GT Bank has approved this Agreement and the transactious contempiated hereby;

WITTREAS. Bank's and Company's Boards of Directors have resolved to recommend approval of the transactions to their respective shareholders:

WHEREAS, upon the consummations of the marger of GT Bank with and into the Bank, the Bank shall be the samiving corporation; and

WHEREAS, it is the intention of the parties to this Agreement that the bosinest combinations contemplated hereby he treated as "subrgarizations" under Section 368 of the Internal Revenue Code of 1986, as amended.

NOW THEREFORE, in consideration of these premises and the representations, wavraptics and agreements hencie contained, Back, Company and GT Back hereby agree as follows:

ARTICLE 1. DEFINITIONS

As used in this Agreement, the following terms shall have the meanings set forth below:

"Acquisition Event" shall mean any of the following

- ía) Either Bank, Company or GT Bank resulted vely, shall have authorized. recommended, publicly purposed or publicly approaced an intention to authorize, recommend or propose, or shall have entered or appounced an intention to enter into a letter of intent, an agreement-in-principle or a definitive agreement with any Person (other than Bank, Company or GT Sault), to effect an Acquisition Transaction or failed to publicly oppose a Tender Offec or an Exchange Offer (as defined below's. As used herein, the term "Acquisition Transaction" shall mean a merger, consolidation or schular transaction involving Bank, Company or GT. Bank, (ii) the disposition, by sale, lease, exchange, dissolution or liquidation, or otherwise, of all or substantially d) of the assets of 2 aok, Company or GT Bank, or any avector assets of Bank, Company or GT Saok the disposition or lease of which would result in a material change in the business or business operations of Bank, Company or GT Bank, or (in) the issuance, other than pursuant to outstanding stock options, sale or other disposition by Bank, Company or GT Bank functioning, without Unitation, by way of merger, consolidation, share exchange or only similar transaction) of shares of Bank Common Societ, Company, Common Stock, GT Bank Common Stock or other Equity Securities, or the grant of any option, warrant or reheatinght to acquire shares of Sank Common Stock or Company Common Stock, GT Common Study or other Equity Securities, representing directly, or on an as exercised, as-exchanged or as-converted basis (in the case of options, warrants, rights or exchangeable or convertible Equity Securities), 15% or more of the Voting Securities of the Bank, Company or GT Bank; or
- Prior to regulitation of this Agreement (i) any Person (other than a person who is (b) a party to a Bank Director-Shareholder Agreement, GT Back Director-Shareholder Agreement or a Company Director-Shareholder Agreement) sholt have increased the number of shares of Bank Common Stock, GT 8ar4 Common Stock or Contomy Contoms Stock over which such Person has beneficial assumption (as such term is defined to Rule 13d-3 promulgated unger the Exchange Act) by a number that is greater than 5% of the then outstanding shares. of Bank Common Stock, GT Bank Common Stock or Commany Common Stack. as applicable, if, after giving effect to such expresse, such Person gwns beneficially, above than 10% of the outstanding shares of Bank Common Stock, GT Bank Common Stock or Company Common Stock, or (ii) any "group? (as such term is defined under the Exchange Act) would have been formed which benefic ally owns, or has fan right to propure beneficial ownership of, more that 10% of the fixe outstanding shares of Bank Common Stock, GT Bank Common, Stock or Company Common Stock.

"Acquisition Proposal" singl have the meaning given such term in Section 6.2.5.

"Albliace" or "affiliate" shall areas, with respect to any other Person, suy Person that, directly or indirectly, controls or is controlled by or is under common control with such Person.

"Aggregate Cash Amount" means the product obtained by multiplying (i) the number of snares of Company Common Stock outstanding at the Effective Time of the Downstream Minger by (ii) \$7.25, and multiplying that product by 20%

"Aggregate Bank Share Annual" means the amount equal to the product obtained by multiplying (i) the number of shares of Company Common Stock outstanding at the Effective Time of the Downstream Merger by (ii) the Exchange Ratio and multiplying that product by 80%

"Bapk" shall mean Bay Commercial Bonk.

"Brack Common Stock" shall mean the conclusion stock, on particular per spare, of Barik,

"Bank Director-Shareholder Agreement" shall have the meaning given such term in Section 7.2.3

"Ban's Fairness Opinion" shall have the meaning given to such term in Section 7.2.4.

"Bank Filongs" shall have the meaning given such term in Section 3.6.

"Bank Financial Statements" means the ambient financial statements (balance choese, statements of inclure, statements of each flow and statements of changes in financial position) and notes therein of Bank and the related opinions thereon for the years ended December 31, 2010, 2009 and 2008.

"Bonk Stock Options" shall mean the options to purchase shares of Bank Common Stock, purtuant to the Bank Stock Option Plans as described in Schedule 3.5 hereto.

"Bank Stock Option Plans' shall mean Bonk's written stock option plans, predecessor stock option plan and predecessor warrants as described in Schedule 3.5 herety.

"Benefit Arrangenters" shall have the innaning given such true in Section 4.20.4.

"RHC4," shoft mean the Bank Holding Company Act of 1956, as amended.

¹ Business Day" shall mean any day, other than a Saharday. Sondoy or any other day, such as a legal holiday, on which Colifornia state banks in Liabforma are not open for substantially all their banking business.

"Colifornia Corporations Code" shall mean the General Corporation Law of the State of California

"California Financial Code" shall mean the Financial Code of the State of California.

"Cash Election" shall have the meaning given to such term in Section 2.6.1.

"Cash Proration Factor" shall have the meaning given to such term in Section 2.6.3(b)(m).

"CDFP" shall mean the California Department of Financial Justitutions.

"Certificates" shall have the meaning given to such term in Section 2.5.3.

"Classified Assets" shall have the meaning given to such term in Soci on 5.1.14

" Closing" sholl have the meaning given to such term in Section 2.4.

"Closing Date" shall have the resaring given to such term in Section 2.1.

"Combination Cash Election" shall have the meaning given such turn in Section 2.6.1.

"Combination Stock Rection" shall have the meaning given such term in Section 2.6.1.

"Company" shall mean Globel Baircorp.

"Company Conneuer Stock" shall mean for common stock, \$ 01 per value per share, of Company.

"Company Fairness Opinion" shall have the meaning given to such term in Section 2.3.1.

"Company Filings" shall have the meanings given such term in Section 4.6.

"Company Financial Statements" means the audited consolidated financial statements (balance sheets, statements of income, statements of each flow and statements of changes in financial position) and notes therete of the Company and GT Bank and the related opinions therein for the years called December 31, 2010, 2009, and 2008.

"Company or GT Bank Superior Proposal 1 shall have the meaning given to such term in Section 6.2.5.

"Company Stock Option Plan" shall mean the Global Sancorp 2007 Stock Incontive Plan.

"Confidential Information" process all information exchanged heretofere or hereafter between Bank and its affiliates and agents, on the one band, and Company and GT Bank, thris affiliates and agents, on the other band, which is information related to the business, fipancial position or operations of the Person responsible for furnishing the information or an Altibate of such Person (such information to include, by way of example only and not of limitation, client hists, company mercuric, mercuric memoranda, strategic plans, budgets, forecasts/projections, computer models, marketing plans, files relating to have originated by such Person, loans and least participations, purchased by such Person from others, investments, deposits, leases, contracts, employment records, minutes of board of directors meetings (and contrillers thereof) and stockholder meetings, legal proceedings, reports of examination by any Governmental Entry, and such other records or documents such Person may supply in the other party pursuant to the terms of this Agreement or as contemplated hereby). Nowithstanding the foregoing, "Confidential Information," shall not include any information that (i) at the time of disclosure of thereafter is generally available to and known by the public (other than as a result of a disclosure directly or indirectly by the recipients or any of their officers, directors, employees or other representatives or agents), (b) was available to the recipients on a nonconfidential basis from a source other than Persons responsible for farmishing the information, provided that such source is not and was not bound by a confidentiality agreement with respect to the information, or (iii) has been independently acquired or developed by the recipients without violating any obligations under this Agreement.

"Default" shall mean, as to any party to this Agreement, a failure by such party to perform, in any material respect, any of the agreements or coversatio of such party contained in Arrocles 3 or 6.

"Derivatives Contract" shall have the meaning given such term in Section 4.24.

"Descenting Shares" Disserting Shares" shall mean chares, the boldow of which have lawfully disserted from the Merger in accordance with Section 1300 er seq. of the California Corporations Code

"Downstream Merger" shall have the meaning set forth in Section 2.1.

"Downstream Merger Effective Time" shall have the memory set forth in Section 2.1.

"Effective Tyme" shall have the meaning given such term in Section 2.1,

"Election Deadlore" shall have the meaning given such termion Section 2.6.2.

"Election Form" shall have the assaying given such term in Section 2.5.1.

"Floring Form Record Date" shall have the meaning given such term in Section 2.5.1.

"Employee Plan" shall have the meaning given such term to Sociton 4.20.3.

"Environmental Laws" shall mean and include any and all laws, statutes, onlinances, roles, regulations, orders, or determinations of any finerminents) Entry pertaining to health or to the environment, including, without limitation, the Clean Air Act, as smonded, the Competensive Environmental Response, Compensation and Lability Act of 1980, as amended, the Federal Water Pollation Control Act Ameniments, the Occupational Sately and Health Act of 1970, as amended, the Resource Conservation and Recovery Act of 1976 as amended ("RCRA"), the Hazardous Materials Transportation Act of 1975, as amended, the Safe Drinking Water Act, as amended, and the Toxic Substances Control Act, as amended.

"Equity Securities" shall have the meaning given to such term in the Exchange Act.

"ERISA" shall mean the Employee Residement Income Security Act of 1974, as amended.

"Exchange Act" shall mean the Secorities Exchange Act of (934, as amended.

"Exchange Agent" shall mean such Person as the Company and Bank shall have agreed upon and appointed to perform the duties set forth in Section 2.5.3.

"Exchange Offer" shall mean the commencement (as such term is defined in Role 14d-2 under the Exchange Act) of an exchange offer or the filing by any Person of a registration statement order the Secondairs. Act with respect to an exchange offer to purchase any shares of Back Common Stock or Company Common Stock such that, upon consummation of soch offer, such Person would own or control 15% or more of the thep outstanding shares of Back Common Stock or Company Common Stock, as applicable

"Exchange Ratio" means a factor of 0, 75678.

"Expenses" shell have the meaning given study to on in Soution 8-5-4.

"FDIC" shell mean the Federal Deposit Insurance Corporation.

"FDI Act" shall mean the Federal Deposit Insurance Act.

"Federal Reserve" shall mean the Board of Governors of the Federal Reserve System.

"GAAP" shall mean generally accepted accounting principles accepted in the United States of America.

"Governmental Entity" shall mean any entity, federal, wate local or foreign government or any administrative agency or commission or other governmental authority or instrumentality whatsoever

GT Bank" shall mean Global Trust Bank.

"GT Benk Collateralizing Xeal Estate" shall have the meaning given such terro in: Section 4.22.1.

"GT Bank Commun Stock" shall mean the common stock, no par value, of GT Bank,

"GT Bank and Company Director-Sharebolder Agreement" shall have the meaning given such term in Section 7.3.3

"GT Bank Properties" shall have the meaning given such term in Section 4.72.1.

"Bazardous Substances" shall have the meaning given such term in Section 4 22.4.

"IPMDA" shall mean florer Mortgage Disclosure Act.

"IRC" shall mean the internal Revenue Code of 1986, as amended.

"Joint proxy statement and offering circular" shall have the meaning given to such term in Section 5.4.

"Knowledge" shall mean, with respect to any representation or warranty contained in this Agreement, the actual knowledge, after trasonable inquiry, including inquiry of direct reports, of any diarotic in exercisive officer of Bank, GT Bank or Company, as the case may be.

"Last Regulatory Approval" sital) mean the final Requisite Regulatory Approval required, from any Governmental Entity under applicable federal laws of the United States and laws of any state having jurisdiction over the Morger or the Downstream Merger, to permit the parties to consummate the Merger and the Downstream Merger.

"Loan Loss Reserve? shall have the meaning set forth in Section 6 U.10(a).

"Mailing Date" shall have the meaning given such term in Section 2.6 %

"Material Adverse Effect" shall mean a material adverse effect: (i) on the business, assets, results of operations, financial condition or prospects of a Person and its subsidiaries, if any, taken as a whole (unless specifically indicated otherwise); or (ii) on the ability of a Person that is a party to this Agreement to perform its obligations under this Agreement or to consummate the transactions contemplated by this Agreement; provided, however, that a Material Adverse Effect shall not be defined to include the impact of (a) changes in banking and similar laws of general applicability or interpretations thereof by the Regulatory Authority, (b) changes in GAAP in regulatory accounting reprirements applicable to banks and their holding companies generally and (d) changes identified or expressive contemplated in this Agreement or otherwise agreed to in writing by a Person.

"Morger" shall have the meaning set forth to Section 2.4.

"Morger Agreement" shall have the meaning given to such terms in Section 2.1.

"Monyer Consideration" shall mean the Aggregate Cash Amount plus the Aggregate. Share Amount.

"OREO" shall have the meaning given such term in Section 3-13.

"Perfected Descenting Shares" shall mean Descenting Shares, the holders of which have not effectively withdown or lost their divisioning rights under Section 1300 et seq. of the California Corporations Code. Dissenting Shares which have lost their status as dissenting shares under Section 1302 of the California Corporations Code shall not be deemed Perfected. Dissecting Shares.

"Pre Share Cash Consideration" means \$7.25 per share of Company Common Stock.

"Persons" or "persons' shall mean an individual, corporation, partnership, limited lizbility company, joint vecture, trust or unincorporated organization, Governmentel limity or any other legal entity whetsoever,

"Regulatory Authority" shall mean any Governmental Euroy, the approval of which is legally required for tensionmation of the Merger of the Downstream Merger.

"Requisite Regulatory Approvals" shall have the meaning set forth in Section 7.1.2.

"Returns" shall mean all returns, declarations, reports, statements, and other documents required to be filled with respect to tederal, state, local and foreign Taxes, and the term "Return" means any one of the toregoing Returns.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Stock Flex tion" shall have the meaning given to such term in Section 2.6.1.

"Stock Proration Factor" shall have the meaning given to such term in Section 2.0.3(a)(iii)

"Subsidiary" shall mean, with respect to any corporation (the "parent"), wy other corporation, association or other business entity of which more than 50% of the shares of the Voting Stock are owned or controlled, derectly or indirectly, by the parent or by one or more Subsidiaries of the parent, or by the parent and one or more of its Subsidiaries.

"Surviving Corporation" shall have the meaning given to such term in Section 2.1.

"Taxes" shall mean all federal, state, local and foreign not income, gross income, gross receipts, sales, use, ad valorem, transfer, franchise, profits, homed, lease, service, so vice use, withholding, payroll, employment, excise, severance, stamp, occupation, premium property, windfall profits, customs, duties, or other taxes, logether with any interest and any proalities, additions to tax, or odditional emounts with respect therein, and the train "Too" means any one of the foregoing Taxes.

"Tay Filings" shall mean any applications, reports, statements or other Returns related to any Person's mass required to be filed with any locat, state or federal Governmental Entity before the Morger may become effective

"Tenger Offer" shall mean the communication (as such term is defined in Rule 14d-2 under the Exchange Act) of a forder offer or the filing by any person of a registration statement under the Scientifies Act with respect to a tender offer to putchase any shares of Bank Common Stock of Company Common Sinck such that, upon consummation of such offer, such personmented own or control 15% or more of the face outstanding. Voting Securities of Back of Company.

"Understanding" shall have the meaning set forth in Section 6 (15).

"Undesignated Shares" shall have the meaning set forth in Section 2.6 ()

"Updated Schedules" shall mean the schedules provided by Bank to GT Bank and Company, and the schedules provided by GT Bank and Company to Bank, immediately prim to the Closing as set forth in Section 5.7.

"Voting Securities" or "Voting Sinch" shall mean the stock or other securities or any other interest entitling the nolders thereof to vote in the election of the directors, intsless or Persons performing similar functions of the Person in question, including, without limitation, activoting securities that are convertible or exchangeable into voting securities, but shall not include any stock or other interest so entitling the holders thereof to vote only upon the happening of a contingency (other than a conversion or exchange thereof into voting securities), whether or not such contingency has accurred.

ARTICLE 2. THE MERGER

2.1 The Metger, Closing and Effective Time

Subject to the terms and conditions of this Agreement, as promptly as practicable following the receipt of the Last Regulatory Approval and the expiration of all applicable waiting periods, (a) the Company shall be interged with and isto GT Bank, with GT Bank being the surviving corporation of the merger (the "Downstream Merger Servicing Corporation"), all parsitant on the Agreement of Merger attached to this Agreement as Exhibit 2.1 (the "Downstream Merger Agreement") and to accordance with the applicable provisions of the Cilifornia Financial Code are the California Corporations Code (the "Downstream Merger") and (ii) after the Downstream Merger, GT Bank shall be merged with and into the Bank with the Bank being the surviving corporation of the merger (the "Sorviving Corporation"), all pursuant to the Agreement of Merger attached to this Agreement as Exhibit 2.2 (the "Merger Agreement") and in zecordance with the applicable provision (the California Corporation of the California Financial Code and the California Corporation of the merger (the "Sorviving Corporation"), all pursuant to the Agreement of Merger attached to this Agreement as Exhibit 2.2 (the "Merger Agreement") and in zecordance with the applicable provisions of the California Financial Code and the California Corporations Code (the "Merger")

The closing of the Downstream Merger and Merger (the "Closing") shall take place at a totation and time and Boxiness Day to be designated by Bank, with the reasonable concurrence of the Company (the "Closing Date"). The Merger shall be effective when the Merger Acatement (logether with say other documents required by law to effective when the Merger shall have been fold with the Secretary of State of the State of California and the CDFL. When used in this Agreement, the terms "Effective Time" shall mean the time of filling of the Merger Agreement with the Secretary of State and the CDFL. When used in this Agreement with the Secretary of State and the CDFL. When used in this Agreement with the Secretary of State and the CDFL. When used in the time of filling of the Merger

"Downstream Merger Effective Time" shall mean the time of filling of the Merger Agreement for the Downstream Merger with the Secretary of State and the COPI

7.2 Effrit of Downstream Meiger and Merger

Downstream Merger. At the Effective Time of the Downstream Merger, by operation of taw, ell of the oights, privileges, powers and franchises and all property and assets of every kind and description of the Company and GT Bank shall be vested an and be held and enjoyed by the Downstream Merger Surviving Corporation, without further act or deed, and all the estates and interests of every kind of the Company and GT Bank, methoding all debts due to either of them, shall be as effectively the property of the Downstream Merger Surviving Corporation as they were of the Company and GT Bank immediately prior to the Effective Time, and the ticle to any real estate vested by deed or otherwise in either the Company of GT Bank shall not revert or be in any way impaired by reason of the Downstream Merger, and all rights of creditors and liens apon ally property of the Company and GT Bank shall be proserved onimpaired and all idels, tiabilities and duties of the Company and GT Bank shall be debts, leabilities and duties of the Downstream Merger Surviving Corporation and may be enforced against it to the same retent as it gues debts, leabilities and duties had hear inclusivel in contracted by it, and none of such debts, leabilities or duties shall be expanded), increased, broadened or collarged by reason of the Downstream Merger

Merger. At the Biffective Tines of the Merger, by operation of law, all of the rights, privileges, powers and francinees and all property and assets of every kind and description of Bank and GT Bank shall be vested in and be held and enjoyed by the Surriving Corporation, without further act or deed, and all the estates and interests of every kind of Bank and GT Bank, including all debts due to either of them, shall be as effectively the property of the Surviving Corporation the title to any real estate vested by doed or otherwise in either Bank or GT Back shall not even or be in any way impaired by reason of the Merger, and all rights of rections and bens upon any property of Bank shall be debts, heldsities and duties of the Surviving Corporation and may be rediment in the same extent as if such debts, liabilities and duties had been incurred or contracted by in, and none of such debts, liabilities or duties shall be expanded, increased, broadened or enlarged by reason of the Merger.

3.3 Articles of Incorporation and Dylaws

The Articles of Incorporation and Bylaws of GT Bank in effect itemediately prior to the Drovestream Menger Effective Time shall be the Articles of Incorporation and Bylaws of the Downstream Merger Surviving Corporation until amended and the same of the Surviving Corporation shall be "Global Trust Bank."

The Articles of Incorporation and Bylaws of the Bank it effort immediately prior to the Effective Time shall be the Articles of Incorporation and Bylaws of the Surviving Corporation until encaded and the name of the Surviving Corporation shall be "Bay Confidenceal Back."

2.4 Bank and Company Common Stock

The authorized and issued capital stock of the Bank introdiately prior to the Effective Time, on and after the Effective Time, pursuant to the Merger Agreen out and without ony further action on the past of Bank, shall remain unchanged, except with respect to those shareholders of GT Bank who receive Bank Common Stock, in which case those shares shall become issued and outstanding shares of Bank.

The authorized and issued capital stock of the Company immediately prior to the Downstream Merger Effective Time, on and after the Downstream Merger Effective Time, pursuant to the Merger Agreement for the Downstream Merger and without any dother action on the part of Company, shell broncan capital work of GT Bank, as described in the Downstream Merger Agreement. Fresh share of Company Common Stock outstanding shares immediately prior to the Downstream Merger and without any further detion on the part of Company, shell become for the Downstream Merger and without any further detion on the part of Company, shell become one share of GT Bank.

2.5 Conversion of GJ Bank Common Stock.

At the Effective Time and pursuant to the Mergee Agreement

2.5.1 Conversion of GT Bank Common Stock

(a) Subject to the exceptions and limitations in Section 2.5.1(b) and 2.6, each obstanding share of GT Bank Common Stock shall, without any further action on the pair of GT Bank or the hoklers of any such chares, be converted into the right to receive for each share of GT Bank Common stock (a) one share of Bank Common Stock varies the Evchange Ratio, and/or (a) each in accordance with Per Share Cash Consideration, at the direction of the holder of such share of UT Bank Common Stock up to the limit set forth herein.

(b) Certain Exceptions and Finitutions

(A) <u>Perfected Dissenting Shares</u>. GT Bank Perfected Dissenting Shares shall not be converted into shares of Bank Common Stock, but shall, after the Effective Time, be entitled only to such rights as are granted them by Section 1300 et seq. of the California Corporations Code (each dissenting shareholder who is contiled to payment for his or her shares of GT Bank Common Stock shall receive such payment in an amount as determined pursuant to Section 1300 et seq. of the California Corporations Code); and

(B) <u>Fractional Shares</u> No fractional shares of Bank Contribut Stock shall be asseed in the Merger and, in lieu thereof, each holder of GT Bank Contribut Stock who would otherwise be entitled to receive a fractional share shall receive an amount in cash equal to the product (calculated to eight places) obtained by multiplying such backonal share interest by the Per Share Cash Consideration. (c) The holders of certificates formerally representing shares of GT Bank Common Steck shall cease to have any rights an shareholders of GT Bank, except such rights, of any, as they may have pursuant to law. Except as provided above, until certificates representing shares of GT Bank Common Stock are successively on the Euclideage Agent pursuant to Section 2.5.3 hereof, the certificates of each indicer shall, after the Effective Time, represent for all purposes only the right to receive shares of Bank Common Stock or cash in accordance with the Exchange Ratio on Pro Share Cash Consideration, respectively.

(d) Notwithstanding the calculations and allocations of consideration in this Agreement. (i) the maximum amount of cash to be paid in the aggregate by Bank in the Merger for Perfected Dissenting Shares and for shares of GT Bank Common Stock whose holders receive each in whole or in part, shall not exceed the Aggregate Cash Amount, and (ii) the minimum number of Bank shares to be issued in the Merger shall equal the Aggregate Bank Share Amount.

2.5.2 <u>Reservation of Shares</u>. Prior to the Effective Line, the Moord of Directors of the Bank shall receive for issuance a sufficient number of shares of Bank Common Stock for the purpose of issuing as chares to the shareholders of GT Hank in accordance between

2.5.3 Exchange of GT Bank Comman Stock.

(a) As some as reasonably practicable after the Effective Time, holders of record of contificates foursely representing shares of OF Bank Common Stock ("Certificates") shall be instructed to reader such Certificates to the Exchange Agent pursuant to a letter of transmittal that the Bank shall deliver or cause to be delivered to such holders. Such letter of transmittal shall specify that risk of loss and title to Certificates and pass only upon acceptance of such Certificates by the Bank of the Exchange Agent.

(b) After the Effective Time, each holder of a Comficate that summaders such Certificate to the Bank of the Exchange Agent will, upon acceptance thereof by Bank of the Exchange Agent, be entitled to receive shares of Bank Common Stock is each at the electron of the holder of shares of GT Bank Common Stock in accordance with Section 2.5.1 hereof.

(c) The Bank of the Exchange Agent shall accept Certificates upon compliance with such reasonable terms and conditions as the Bank of the Exchange Agent may impore to effect an orderly exchange thereof in accordance with customary exchange practices. Certificates shall be appropriately endersed or accompanied by such instruments of tracefor as the Bank of the Exchange Agent may reasonably require.

(d) At the Pifective Time, GT Bank shall deliver a certified copy of a list of is chareholders to the Bank or the Exchange Agent. After the Effective Time, there shall be no further transfer of Certificates on the stock transfer records of GF Bank, and if such Certificates are presented to GT Bank for transfer, they shall be canceled against delivery of chares of Bank Common Stock or the Per Share Cesh Consideration in accordance with Sections 2.5.1 and 2.6 hereof. Bank shall not be chigated to deliver any shares of Bank.

Compose Stock to any holder of GT Common Stock until such bolder surrenders the Crutofeates. as provided herein. No dividends declared will be remitted, nor any voting rights granted, to any person entitled to receive Bank Common Stock under this Agreement until such person surrenders the Centificates representing the right to receive such Bank Common Stock, at which sine such dividends on whole shares of Bank Common Stock with a record date on or after the Effective Time shall be remined to such person, without interest and less any taxes that may have been imposed thereon, and voting rights will be restored. Neither the Exchange Agont por any party to this Agreement nor any atfibate (hereof shall be liable to any holder of GT Sank-Common Stock represented by any Certificate for any consideration paid to a public official pursuant to applicable abandoned property, escheat or similar laws. The Bank and the lixehange Agont shall be entitled to rely upon the stock transfer books of GT Bank to establish the identity of those persons coulded to receive consideration specified in this Agreement, which books shall he mechanics with respect thereto. In the event of a dispute with respect to ownership of stock represented by any Certificate, the Bank or the Exchange Agent shall be entitled to deposit any consideration in respect thereof to escreve with an independent third party and thereafter he relieved with respect to any claims thereto.

(c) If any shares of Bank Common Stock are to be issued and/or Per-Share Cash Consideration paid to a person other than a person in whose name a surrendered Certificate is registered, it shall be a condition of issuence and/or payment that the surrendered Certificate shall be properly endorsed or otherwise presented in proper form for transfer and that the person requesting such includes and/or payment shall pay to the Bank or the Exchange Agent any required transfer or other taxes or establish to the satisfaction of the Bank or the Exchange Agent that such tax has been poid or is not applicable.

(1) In the event any Certificate shall have been lost, stolen or destroyed, the owner of such lost, stolen or destroyed Certificate shall defiver to the Bank or the Eschange Agent an attituated stoting such fact, in form satisfactory to Bank and, at Bank's discretion, a bond in such reasonable sum as the Bank or the Fachange Agent may direct as indemnity against any claim that may be made against the Bank or OT Bank or any other party with respect to the Certificate alleged to have been lost, stolen or destroyed. Upon such delivery, the rowner shall have the right to receive the shares of Bank Common Stock and/or Per Share Cash Consideration in accordance with Sections 2.5.1 and 2.5 hereof.

(g) Any holder of GT Bank Common Stock who has not surrendered his/hei/ito phares to the Exchange Agent within six months after the Effective Time, shall look only to Bank for payment of consideration payable is respect of any GT Bank Communic Stock Certificates not earlier exchanged by Exchange Agent. No interest will be paid or accrued on any amounts supports to payable.

2.6 <u>Election and Progubor</u>, Procedures

2.5.1 An election form and other appropriate and costantiary transmittel materials in such form as the Bank and GF Bank shall instructed y spree ("Election Fourt") shall be masked an more than thirty (30) days after the Effective Tome or on such other date as Bank and GT Bank shall proceedly agree ("Mailing Date") to each holder of record of GT Bank Common

Stock as of the Effective Time ("Election Form Record Date"). The Bank and GT Bank shall provide to the Exchange Agent all information reasonably necessary for it to perform its obligations as spinified herein. Each Election Form shall permit the holder (or the benefic at owner through appropriate and customary documentation and instructions) to elect (at 'Election') to receive either (i) Bank Common Stock with respect to all of such holder's GT Bank Common Stock (a 'Stock Election'), (ii) each with respect to all of such holder's GT Bank Common Stock (a 'Cash Election'), or (iii) for a specified number of shares of GT Bank Common Stock. Bank Common Stock (a 'Cash Election'), or (iii) for a specified number of shares of GT Bank Common Stock (a 'Cash Election'), and five a specified number of shares of GT Bank Common Stock (a 'Combination Cash Election'), subject to the provisions contained in this Agreement. Any GT Bank Common Stock (other than GT Bank Perfected Discoming Shares) with respect to the bolder (or the beneficial owner, as the case may be) shall not have submitted to the Exchange Agent an effective, properly completed Election Form received prior to the Election Deadline shall be deemed to be 'Undesignated Shares' hereunder.

2.6.2 Any Election shall have been properly made and effective only if the Excharge Apent shall have actually received a uniperly completed Election Form by \$100 P.M. California time on or before the Milestay following the Mailing Date, or such other time and date as Bank and GT Bank may mountably agree (the "Election Deadline"). An Election Form shall be documel preparely completed only if an Election is indicated for each share of GT Bank Common. Shock revered by such Election Form and if accompanied by one or more Certificates for customary affidavits and indemotification regarding the loss or destruction of such fidales or the guaranteed delivery of such Cerniteates) representing all shares of GT Bank Common Study covered by such blection Form, together with the daily executed transmittal materials included in or required by the Election Form. Any Election Form may be revolved or dweight by she personsubmitting such Election Torm at or prior to the Floation Deadline . In the event an Election Form is revoked prior to the Election Deadline, the shares of GT Bank Conution Stock. represented by such Election Form shall amomatically become Undesignated Shares unters and until a new Election from is purperly completed with respect to such shares on or before the Election Dearline and the Bank shall cause the Certificates representing such Undesignated Shares to be promptly returned without charge to the person submitting the revoked Electron-Four spon written request to that effect from the holder who submitted such Election Form. Subject to the terms of this Agreement and of the Election Form, the Exclusive Agent shall have reasonable discretion to determine whether any election, revucation or change has been proposily or timely made and to disregard immaterial defects in the Election Forms, and any devisions of Bank and GT Bank required by the Exchange Agent and made in good faith in determining suchmatters shall be binding and cosclosive. The Exclosing Agent shall putify as soon as reasonably possible any person of any costenial defect in his or her Election Form.

2.6.3 The Bank shall use its best efforts to ease the Exchange Agent to effect the allocation among the holders of GT Bank Common Stock of rights to receive Bank Common Stock or eash in the Merger as follows:

(a) If the conversion of shares of GT Bank Contation Stock for which Cash Elections and Combination Cash Elections shall have effectively been made would result at a number of shares of Bank Common Stock being issued that is less then the Aggregate Bank. Share Amount (which shall be determined for this purpose on the assumption that all shares of GT Bank Common Stock (other than those for which Cash Elections or Combination Cash Elections have been made) would be cutified to reavive Bank Connector Stock) thru, to the extent accessary so that the sources of shares of Bank Common Stock to be issued in the Merger shall be equal to the Aggregate Bank Shore Amount, the Exchange Agent shall moke the following allocations and adjustments in the following order:

(i) shares of GT Bank Common Stock for which effective Cash Elections or Combination Cash Elections have been made shall be converted into the right to receive cash in an amount equal to the Per Share Cash Consideration.

(ii) the Exchange Agenc shall select by lot such murber of holders of Undesignated Shares to receive the Per Share Cash Consideration as shall be necessary so that the shares of Bank Common Stock to be received by other bolders of Undesignated Shares, when combined with the number of shares of Bank Common Stock for which Stock Electrons or Combination Stock Electrons have been made, shall be equal to the Aggregate Bank Share Amount. If all Undesignated Shares are converted into the right to receive the Per Stare Cash Consideration and the shares for which Stock Election and Combination Stock Electrons are still less them the Aggregate Bank Share Amount, then,

(ii) a stock protation factor (the "Stock Protation Factor") shall be determined by dividing (x) the Aggregate Bank Share Amount by (y) the product of (i) the total number of shares of GT Bank Common Stock with respect to which effective Stock Elections and Combination Stock Elections were made multiplied by (ii) the Exchange Katio. Each holder of GT Bank Common Stock who inside an effective Stock Election or Combination Stock Election shall be entitled to:

(1) the number of shares of Stark Common Stock equal to the preduct of (x) the Exchange Ratio, multiplied by (y) the number of shares of GT Bank Common Stock concred by such Stock Election, exultiplied by (z) the Stock Protation Factor; and

 $\{2\}$ cash in an amount equal to the product of (x) the Per Share Cash Consideration, multiplied by (y) the number of shares GT Bank Common Stock covered by such Stock Election or Combination Stock Election, multiplied by (z) one mious the Stock Provation Factor.

(b) If the conversion of the shares of GT Back Common Stock for which Cash Floctions and Combination Cash Elections shall have effectively been made (based upon the Exchange Radio) and for which Ocseenting Stares of GT Back Common Stock are Perfected would result in a number of shares of GT Back Common Stock being converted to eash that is more than the Aggregate Cash Amount, then, to the extent necessary so that the aggregate amount of each does not exceed the Aggregate Cash Amount, the Exchange Agent shall make the following allocations and adjustments in the following order. (i) each holder of GT Bank Common Stock who made an effective Stock Election or Combination Stock Election shall receive the number of shares of Bank Common Stock could to the product of the Exchange Ratio (ault-plied by the number of shares of Bank Common Stock covered by such Stock Election or Combination Stock Flection;

(ii) the Exchange Agent shall select by lot such humber of holders of Undesignated Shares to receive Back Common Stock as shall be necessary so that the shares of Back Common Stock to be received by those holders, when combined with the number of shares of GT Common Stock for which a Stock Election of Combination Stock Election has been made shall be equal to at tenst the Aggregate Back Share Amount. If all Findesignated Shares plus all shares as to which Stock Elections and Combination Stock Elections have been made together are still total less than Aggregate Dank Share Amount shares, them.

(iii) a cash protation factor (the "Cash Protation Factor") shall be determined by dividing (x) the Aggregate Bank Share Account (test the product of (i) the sum of the shares for which an effective Stock Election and Combination Stock Election has been made plot all the Undergrated Shares multiplied by (ii) the Exchange Ratio by (y) the product of (i) the sum of the total number of chares of G. Bank Common Stock with respect to which effective Cash Elections and Combination Cash Elections were made multiplied by (ii) the Exchange Ratio. Each holder of GT Bank Common Stock who made an effective Cash Election of Combination Cash Election shall be entitled to:

(1) cash equal to the product of (x) the Per-Sitare Cash Consideration, multiplied by (y) the number of shares of G7 Bank Common Stock covered by such Cash Fluctum or Combination Cash Election, multiplied by (z) one minus the Cash Program Factor, and

(2) the number of shares of Rack Common Steck equal to the product of (x) the Exchange Ratio, anthiphed by (y) the number of shares of GT Bank Common Stock covered by such Cash Flection or Combination Cash Election, multiplied by (z) the Cash Protation Sactor

(c) If the aggregate number of shares of GT Hank Common Stock fra which Stock Elections and Combination Stock Blockons shall have effectively been made would result in a number of shares of Bank Common Stock being issued that is equal to the Aggregate Bank Share Amount:

(i) the shares of GT Bank Common Stock for which effective Stock Electrons and Combination Stock Elections have been made shall be converted into the right to receive Sank Common Stock equal to the product of the Evaluance Rano multiplied by the number of shares of GT Bank Common Stock covered by such Stock Electrons and Combination of Stock Elections;

(ii) the shares of GT Bank Common Stock for which effective Cash Elections and Commination Cash Elections have been made shall be converted into the right to receive the Per Share Cash Consideration: and (jii) the Undesignated Shares shall be converted into the right to receive the Per Share Cash Consideration.

(d) Nounthstanding any other provision of this Agreement, if, after applying the allocation rates set forth in the preceding subsections of this Section 2.6, the number of shares of Bank Common Stock teat would be issued pursuant in the Merger is less than the Aggregate Bank Share Amount, the Bank shall be authorized to realfocate shares of Bank Common Stock and eash among the holders of the GT Bank Common Stock in good faith and in such a manner as Bank reasonably determines to be fair and equitable, or to vary the number of shares of Bank Common Stock to be resent in the Merger, in a manner such that the mumber of shares of Bank Common Stock to be issued in the Merger, in a manner such that the mumber of shares of Bank Common Stock to be issued in the Merger shall be as close as possible to, but shall not be less than, the Aggregate Bank Share Amount.

(c) Notwithstanding any other provision of this Agreement, if any Dissenting Shares fail to become Perfected Dissenting Shares, such Dissenting Shares shall automatically be converted into and represent the right to receive the equisideration for shares of GT Bank Common Stock as provided in this Agreement, without interest thereon. The consideration payable for any such Dissenting Shares shall be payable in eash, in shares of Bank Common Stock, or in such combination of eash and Bank Common Stock, or in such combination of eash and Bank Common Stock, or in such combination of eash and Bank Common Stock as shall be determined by Bank as being receivery in appropriate to pressure the states of the Marger as a "peorganization" within the meaning of section 368(s) of the Code

2.6.4 A draft of the raisolations required by Section 2.6(c) shall be prepared by Bank prior to the Giosing and functioned to GT Bank at least two Business Days prior to the Effective Time showing the manner of calculation in reasonable detail. Any calculation of a portion of a share of Bank Common Stock shall be rounded to the nearest ten-thousandth of a share, and any cash payment shall be rounded to the nearest cent. The final calculations shall be agreed open by both parties and shall be set finith an a cartificate executed by the Chief Final calculation of Bank as of the Chief Final calculations.

3.7 Company Stock Options.

(a) Subject to the terms of the Company Stock Option Plan, each person who holds one or more options to purchase Company Common Stock shall be permitted to exercise any options granted under the Company Stock Option Plan, prior to the Effective Trate, in accordance with the terms of the Company Stock Option Plan.

(b) For any options to purchase Company Common Slock not exercised prior to the Effective Time, such Company stock options that are then outstanding and unexercised will be cancelled. Nothing in this Agreement shall be distinct to proclede builders of Company Strick Options from exercising those Company Stock Options at any time in accordance with their terms, including prior to the Effective Time.

2.8 <u>Tesservers' Rights</u>

Shares of GT Bank Common Stock, the holders of which have lawfully dissented from the Merger in accordance with Section 1300 *et seq.* of the California Corporations Code are herein called "Dissenting Shares." Dissenting Shares, the holders of which have not effectively withdrawn or lost their dissenters' rights under Section 1300 *et seq.* of the California Corporations Code ("Perfected Dissenting Shares"), shall not remain constanding pursuant to Section 2.4 or converted pursuant to Section 2.5, out the holders thereof shall be entitled only to such rights as are granted by Section 1300 *et seq.* of the California Corporations Code. Each holder of Perfected Dissenting Shares who is entitled to payment for his or her GT Bank Common Stock, as applicable, pursuant to the provisions of Section 1300 *et seq.* of the California Corporations Code shall receive payment threefor from CT Bank (but only after the amount thereof shall have been agreed upon or finally determined pursuant to such provisions).

2.9 <u>Board of Directors of GT Bank and Bank following</u> the Bownstream Margar Effective Time and <u>Effective Time</u>.

At the Downstream Merger Effective Time, all of the existing members of the heard of directors of GT Bank just prior to the Downstream Merger Effective Time shall remain and constitute all of the members of the board of directors of GT Bank immediately after the Downstream Merger Effective Time.

At the Effective Time, all of the existing members of the board of directors of Bank just prior to the Effective Time shall remain and constitute members of the board of directors of the Bank immediately after the Effective Time. In addition, the three members of the board of directors of GT Bank identified in Schedule 2.9 to this Agreement shall serve as directors of the Bank after the Effective Time. Prior to the Effective Time, the Bank agrees to appoint such three members to its board with their appointments to be effective members directore the Effective Time.

7.10 Executive Officers of GT Back and the Back following the Downstream Merger-Effective Time and Effective Time

At the Downstream Merger Effective Type, the theorewisting executive officers of GC Bank shall remain the executive officers of GT Bank.

At the Effective Tion, the three existing executive officers of the Back shall remain the executive officers of the Back.

ARTICLE J. REPRESENTATIONS AND WARRANTIES OF BANK

Bank represents and warrants to Company and GT Bank as follows:

3.1 Organization, Corporate Power Etc.

Back is a California banking corporation duty organized, validly existing and in good standing under the laws of California and bas all requisite corporate power and corporate authomy to own, tease and operate its properties and assets and to carry on its business substantially as it is being conducted on the dele of this Agreement. Back has all requisite corporate power and corporate swherity to own, tease and operate due of this Agreement, except where the failers to have such power or authomy would and have a Material Adverse Effect on Back or the ability of Back to concentrate the transactions contemplated by this Agreement. Back has all requisite corporate power and corporate sucharity to enter into this Agreement. Back has all requisite corporate power and corporate sucharity to enter into this Agreement. Back has all requisite corporate power and corporate sucharity to enter into this Agreement. Back has all requisite corporate power and corporate sucharity to enter into this Agreement and, subject to obtaining all requisite Regulatory Approvals. Back will have the requisite corporate power and corporate power and corporate obligations bereated with respect to the consummation of the transactions contemplated hereby. Back as of the date of this Agreement is not a member of the bederal Reserve System and anticipates becoming a member of the Federal Reserve System and anticipates becoming a member of the Federal Reserve System.

3.2 <u>Logenses and Ferniss</u>

Bank has all material because, constitutes, franchises, pigns and permits that are necessary for the conduct of its business as presently conducted, and such because are in full force and effect, except for any failure to be in full force and offect that would not, individually or in the aggregate, have a Materiol Adverse Effect on Bank or on the ability of Bank to consummate the transactions contemplated by this Agreement. The properties, assets, operations and businesses of Bank are and have been maintained and conducted, in all material respects, in compliance with all applicable locenses, certificates, franchises, rights and permits

3.3 Subsidiarity

Except as disclosed in Schedule 3.3, there is no corporation, partnership, joint venture or other entity in which Bank owns, directly or indirectly (except as pledgee pursuant to loans or stock or other interest held as the result of or in heu of fereclosure pursuant to pledge or other security arrangement), any equity of other voting interest or position.

3.4 Authorization of Agreement, No Conflicts

3.4.1 The execution and delivery of this Agreement and the Merger Agreement by Back, and the consummation of the transactions contemplated hereby and thereby, have been duly authorized by all recessary corporate action on the part of Back, subject only to the approval of this Agreement, the Merger Agreement and the Merger by Back's shareholdes. This Agreement has been duly executed and delive of by Back and constitutes a legal, radial and binding obligation of Bank, enforceable in peopridance with its terms, except as the enforceability thereof may be limited by bankraptey, liquidation, receivership, conservatorship, insulvency, moratorium or other similar laws affecting the rights of creditors generally and by general equivable principles and by Section S(t)(G)(D) of the FOM Act, 12 U.S.C. § 1818(b)(G)(D). The Morger Agreement, upon the receipt of all Requisite Regulatory Approvals and the due execution and filling of such Merger Agreement in accordance with the applicable provisions of the California Conjunctions Code and the California Sinancial Code, will constitute a legal, valid and binding obligation of Bank, enforceable in accordance with its terms, except as the enforceability thereof may be limited by bankraptey, liquidation, recovership, conservatorship insolvency, moratorium or other similar laws affecting the rights of creditors generally and by general equitable principles and by Section S(t)(G)(D) of the Federal Doposit foreaction by the limited by bankraptey, liquidation, recovership, conservatorship insolvency, moratorium or other similar laws affecting the rights of creditors generally and by general equitable principles and by Section S(t)(G)(D) of the Federal Doposit foreaction. Act, 12 G S C § 1818(b)(6)(D)

3.4.2 Except as disclosed on Schedule 3.4.2 the execution and delivery of this Agreeman and the Merger Agreement, and the consummation of the transactions contemplated hereby and thereby, do not and will not conflict with, or result in any violation of or default or loss of a material benefit under, any provision of the Arheles of Incorporation or Bylaws of Bank, any material mortgage, indentare, lesse, agreement or other material instrument or any permit, concession, grant, franchise, license, judgment, order, decree statute, law, ordinance, tule or regulation applicable to Bank or any of its assets or properties, often that any such conflict, violation, default or loss which (i) will not have a Material Arbeirse Effect on the Bank; or (ii) will be check or weived prior to the Effective Time.

3.5 Capital Structure

The authorized capital stock of Bank consists of 100,000,000 shares of Bank Common Stock, no par value per share, and 10,000,000 shares of preferred stock, so par value. On the date of this Agreement, 3,549,794 shares of Bank Common Stock were outstanding, 354,979 shares of Bank Common Stock were reserved for issuance pursuant to Bank Stock Options granted under the Bank Stock Option Plans, and no sbares of preferred stock were outstanding All cutstanding shares of Bank Common Stock are validly issued, fully paid and unmassessable and do not possess any preemptive rights and were not issued in violation of any preemptive rights or any shoclar rights of any Person. Except for the Bank Stock Options described on Scienable 3.5 to this Agreement, Bank does not have outstanding any options, warrants, calls, rights, commitments, securities or agreements of any character to which Bank to a party or by which it is bound obligating Bank to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock of Bank or obligating Bank to grant, calls, right, commitment or agreement.

3.6 Bank Filings

Except as disclosed on Schedule 3.6, since January 1, 2008, Bank has filled all reports, registrations and statements, together with any amendments required to be made with exspect linerete, that were required to be filed with (2) the FDIC, or (9) any other applicable federal, state or local governmental or regulatory authority. All such reports, registrations and filings, and all reports sent to Dank's shoreholders choing the three-year period ended Determiner 31, 2010 and

the three months ended March 31, 2011 (whether or not filed with any Regulatory Authoniy), are collectively referred to as the "Bank Filings." Except to the extent prohibited by law, copies of the Bank Filings have been made available to Company. As of their respective filing or mailing dates, each of the past Bank Filings (2) was true and complete in all material respects (or was amended so as to be so promptly following discovery of any discrepancy); and (b) complied in all material respects with all of the statutes, refes and regulations enforced or promulgated by the Regulatory Authority with which it was filed (or was amended so as to be so promptly following discovery of any such noncompliance) and none contained any untrue statement of a material fact or omotion to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, nor unsteading.

3.7 Accuracy of Information Supplied

Bank has delivered or will deliver to Company codies of the Rank Financial Statements, and Bank will hereafter until the Closing Date deliver to Company copies of additional linaneral statements of Bank as provided in Sections 5.3.4 (in). The Bank Financial Statements have been propered (and all of self additional financial statements will be prepared) in accordance with GAAP, or applicable regulatory accounting principles applied on a consistent basis during the penieds involved (merge as may be indepated on the mass thereta) consistently followed farmighout the periods covered by such statements, and present (and, when prepared, will present) fairly the financial position of Bank as of the respective dates indicated and the results. of operations, each flows and changes in shareholders' equaty as the respective dates and for the respective periods covered by such financial statements (subject, in the case of the unaudited statements, to recurring edjustments nermal in nature and empound) inclusing, but not limited to the colleptation for loan and lease losses. In addition, Bank has derivered or made available to Company copies of all management or other letters delivered to Bark by its independent accountants in connection with any of the Bank Financial Statements or by such accountants or any consultant regarding the internal controls or internal compliance procedures and systems of Bank issued at any tang since langary 1, 2008, and will make available for respecting by Company or its representatives, at such tunes and places as Company may reasonably request, reports and working papers produced or developed by such accountants or consultants.

5.8 Compliance with Applicable Laws

Except as disclosed on Schedulo 5.8, the business of Baak is not being conducted in violation of any law ordinance or regulation except for violations which individually or in the eggentgate would not have a Material Adverse Effect on Back. Except as set firth op Schedule 3.8, so investigation or review by any Governmental Patity with respect to Bank, other than regular bank examinations, is pending or threatened, yor has any Governmental Energy indicated to Bank an intention to conduct the same.

3.9 <u>Littlestion</u>

Except as set finitions Schedule 3.9, three is on soit, action in proceeding or investigation pending or threatened against or affecting Bank which, if adversaly determined, would have a

Material Advesse Street on Bank; nor is there any judgment, decree, inputcion, rule or order of any flowenumental Entity or arbitrator outstanding against Bank that has, or which, insofar as reasonably can be foreseen, in the fatore would have, any such Material Adverse Effect. Schedule 3.9 contains a true, correct and complete list, including identification of the applicable insurance policy covering such litigation, if any, subject to reservation of rights, if any, the applicable deductible and the amount of any reserve therefor, of all pending litigation in which Bank is a named party, and except as disclosed on Schedule 3.9, all of the litigation shown on such Scholule is adequately covered by insurance in force, except for applicable deductibles, or has been adequately reserved for in accordance with Bank's pror business practices.

3.10 Agreements with Banking Authorities

Bank is not a party to any written agreement or commutatulum of understanding with, or order or directive from, any Governmental Entry.

3.11 <u>Texaster</u>

Bank has in full force and effect policies of insurance with respect to its assess and business against such cashelines and contingencies and in such amounts, types and forms as are customarily appropriate for its business, operations, projecties and exacts. Schedule 3.11 contains a last of all policies of insurance and bonds carried and owned by Eask. Eask is not up default under any such policy of insurance to bond such that it can be canceled and all material current claims outstanding thermoder have been filed in timely fashion. Bank has filed claims with, or gates notice of claim to, its insurers or bonding companies in timely fashion with respect to all material matters and occurrences for which it helieves it has coverage.

3.12 Title to Assets other than Real Property.

Except as disclosed on Schedule 3.12. Back has marketable sith to or a valid leasehold interest to all material properties and assets (other than real property which is the subject of Section 3.11) used in its basiness, free and clear of all mortgages, covenants, conditions, instructions, essenties, thus, sectority interests, charges, claims, assessments and aneurobrances, except for: (a) rights of lessons, besides or subjected in such matters as are reflected in a written lease; (b) encumbrances as set forth its the Bank Fatencial Statements; (c) current Taxes (including assessments collected with Taxes) not yet due which have been fully reserved for; (d) encumbrances, if any, that are not substantial in character, should or extent and do not detract materially from the value, or interface with greenet use or the aniloty of Bank or its Subsidiary to sell or otherwise dispose of the property subject thereto or affected thereby; and (c) other matters as described in Schedule 3.12. All such properties and assets are, and require only roughe matters as described to keep them, in good working condition, dotinal webr and lear excepted.

3.13 Real Property

Schedule 3.1% is an accurate list and general description of all real property owned or lossed by Bark, including Other Real Estate Ovaped ("OREO"). Bank has marketable (stip to the

real properties that it owns, as described in such Schoole, froe and clear of all mortgages, covenants, conditions, restrictsons, casements, heas, seconty interests, charges, claims, assessments and encumbrances, except for (a) rights of lessons, lessons or sublessees in such matters as are reflected in a written lease, (b) liens for concest Taxes booluding researches collected with Taxes) on yet due and payable; (c) encombrances, if any, that are not subsignifal in character, amount or obtain and do one materially deteast from the value, or incritere with present non- on the ability of Rank to dispose, of Bank's interest in the property subject thereto or affected thereby; and (d) tabel matters as described in Schedule 3.13. Bank has valid legschold interests in the leaseholds it holds, files and clear of all mortgages, liens, security interests, pharges, claims, assessments and encombrances, except to: (a) claims of lessors, co-lesses or sublessees in such motions as are reflected in a written lease; (b) title exceptions affecting the forestate of the lessor under such leases, and (c) other metters as described in Schedule 3.13. The activities of Bank with respect to all read property uwined to feased by it for use in conjection with its operations are in all material respects permitted and authorized by applicable zonerahows, ordinatives and regulations and all laws and regulations of any Governmental Entry. Except as set forth on Schedule 2.15. Bank onjoys quiet possession under all material leases to which it is the lesses and all of such leaves are valid and in bull force and chicel, except as the enforceability thereof may be lamited by bankraptey, liquidation, receivership, conservatorship, insolvency, moratorium or other similar laws affecting the rights of creditors generally and by general equitable principles. Materially all huildings and improvements on real properties owned or leased by Dask are in good condition and repair, and do not require more than normal and rootine maintenance to keep them in such regulitory, gurnal wear and tear excepted

3-14 <u>Taries</u>

3.14.1 <u>Filing</u> of Returns. Except as set forth on Schedule 3.14.1, Bank has duly prepared and filed or caused to be duly prepared and filed all inderal, state, and local Returns (for Tax or informational purposes) which were required to be filed by or in respect of Bank or any of their properties, income and/or operations on or prior fo the Closing Date. As of the time they were filed, the foregoing Returns accurately reflected the material facts regarding the memory business, asset, operations, activities, states, and any other information required to be shown thereon. Except as set forth on Schedule 3.14.1, no extension of time within which Bank may file any Return is emiretly in from. Within five business duys following a written request by the Company Rank visib deliver to the Company a true and correct copy of Back's 2010 federal and California income tax returns, as filed with the respective taxing authorities.

3.14.2 Poyntent of Taxes. Except as disclosed on Schedule 3.14.2 with respect to all ontourts in respect of Taxes imposed on Bank or for which Bank is or could be liable, whether to taxing authorities (as, for example, under law) or to other Poissons (as for example, under Tax allocation agreements), with respect to all taxable periods or portions of periods ending on or before the Closing Date, all applicable tax laws and agreements have been or will be fully complied with in all material respects, and all such amounts required to be paid by or on before the date hereot have been paid.

3.14.3 <u>Audit History</u>. Except as disclosed on Schedule 3.14.3, there is no review or such by any taking authority of any Tax liability of Bank currently in progress of which Bank has Knowledge. Except as disclosed on Schedule 3.14.3, the Bank has not received any written notices since its inception of any proding or threateneoi audit, by the Internal Revenue Service or any state, local or foreign agency, for any Returns on Tax hability of the Bank for any period. The partently have no unpaid deficiencies assessed by the Internal Revenue Service or any state, local or foreign taxing authority arising out of any examination of any of the Returns of Bank field for fixed years ended on or after December 31, 2004 through the Closing Date, nor to the Knowledge of the Bank is there reason to believe that any material deficiency will be assessed

3.14.4 Statute of <u>Limitations</u>. Except as disclosed on Schedule 3.14.4, an agreements are in force or are currently being regotiated by or on behalf of the Back for any wayver or for the extension of any stabile of finitations governing the time of sourcements or collection of any Tax. No closing agreements or compromises exist concerning Taxes of the Back

3.14.5 <u>IRC Section 382 Applicability</u>. The Bank, including any party joining in any consolidated return to which the Bank is a member, has not undergone an "ewnership change" as defined in (RC Section 382(g) within the "testing period" (as defined in IRC Section 382) ending immediately before the Effective Time, and not taking into account any transactions contemplated by this Agreement.

3.14.6 <u>Withholding Obligations</u> Except as set forth on Schedule 3.14.6, the Back has withheld from each payment made to any of its officers, directors and employees, the amount of all applicable Taxes, including, but not limited to, income tax, social security contributions, incruployment contributions, backup withholding and other deductions required to be withhold therefore by any Tax law and have paid the same to the proper taxing authorities within the time required under any applicable Tax law.

3.19.7 Tax Liens. There are no fax hens, whether imposed by any federal, statelocal or foreign taxing authority, outstanding against any assols owned by the Back except for (.) hers for Taxes that are not yet due and payable and (ii) Tax hens that have been account in for-Bank's Financial Statements.

3.14.8 <u>Tax Reserves</u> Back has made full and adequate provision and reserve for all federal, state, local or foreign Taxes for the current period for which Tax and information returns are not yet required to be filled. Bank Financial Statements contain fair and authorital actuals for the payment of all Laxes for the periods covered by the Bank Financial Statements and all periods prior thereto.

3.15 Performance of Obligguous

Except as disclosed on Schedule 3.15, Bank has performed all material oblightons required to be performed by it to ease and Bank is not in material default under or in breach of any term or provision of any covenant, contract, lease, indentified or any other agreement, written or oral, to which it is a party, is subject or is otherwise lineard, and no even has encourted that.

with the giving of notice or the passage of time of both, would constitute such a default or breach, where such default or breach or failure to perform would have a Material Adverse Effect on Rank. Except as disclosed on Schedule 3.15 or in the portion of Schedule 3.16 that identifies 90 day past due of classified or nonaccrual loans, no party with when Bank has an egreement that is of material importance to the businesses of Dank is in default thereander.

3.16 Brokets and Finders

Except as disclosed on Schedule 3.16, Bank is not a party to or obligated under any agreement with any broker or finder relating to the transactions contemplated bareby, and on the transactions contemplated bareby, and on the transactions provided for herein or therein, will result in any liability to any broker or finder Bank agrees to indomnify and hold harmbers Company and its affiliates, and in definit with coursel selected by Company and reasonably selected to contemplated by broker of this Section 3.16.

3.17 Absence of Material Adverse Effect

Since January 1, 2008, the business of Back has been conducted only in the ordinary course in the same memories thurstofore conducted, and no event or electrostance has occurred or is expected to occur which has had or which, with the passage of time or otherwise, could reasonably be expected to have a Material Adverse Effect on Back.

3.18 Undesciosed Leahitities

Except as disclosed on Schedule 3.18, Bank has no trabilities or obligations, either ancient, contingent or otherwise, that are material to Bank and that have not been: (a) reflected or disclosed in the Bank Financial Statements, or (b) incurred subsection (to December 31, 2000 in the ordinary course of business. There is no basis for the assertion against Bank of any tiability, onlication or claim (including without limitation that of ony Governmenta) Entry) that well have or course, or could reasonabily be expected to have or cause, a Material Adverse Effect on Bank that is not fully and fairly reflected and disclosed in the Bank Financial Statements or on Schedule 3.20

3.19 Leahilyty Uniter Regulation C, Truth in Lending Law and HMDA

To Bank's Knowledge, and except as disclosed on Schedule 5.19, Bank has no habilities, or obligations either accrued, contingent or otherwise, that have a Material Adverse Effect on Bank with respect to Regulation C. Truth in Lending Low and (MMDA disclosures.

3.20 Bonk Scorecy Act

Barlt is in compliance with the Bank Secrecy Act (3) U.S.C. § 5322, et any) and related state and federal anti-accept functions laws, regulations and guidelines, including without lamination there provisions of federal regulations requiring (a) the filing of reports, such as Correctly Transaction Reports and Suspicious Activity Reports, (b) the maintenance of records and [c) the exercise of due dilicence in joint(lying evolumers.

3.2) Community Reinvestment Act

Bank received a rating of "varia/actory" or better in its most recent examination or interim, review with respect to the Community Reinvestment Act. Bank has not been advised in writing of any concerns regarding compliance with the Community Reinvestment Act by any Governmental Entity or by any other Person.

3.22 Loans and Investments

Except as set forth on Solicidule 3.22, all loans, leases and other extensions of credit, anormatical security agreements in other agreements supporting any loans or extensions of credit, and investments of Bank are, and consocute, on all material respects, the legal, Valid and binding oblegations of the parties thereto and are enforceable against such parties in accordance with their terms, except as the enforceability thereof may be (inited by applicable law and otherwise by bashrupicy, liquidation, receivership, conservatorship, insolvency, motatorion of other singlalaws affecting the rights of creditors generally and by general equitable principles. Except as described on Schedule 3.22, as of March 33, 2011, no leans or investments held by Back are: (i) more than mostly (90) days past due with respect to any subsoluted payment of principal or interest, other than loans on a minimerical status; (ii) classified as "loss", doubtful." "substandard" or "specially mentioned" by Bank or any banking regulators; or (iii) on a measured status in accordance with Bank's loan review procedures. Except as set forth on Schedule 3.22, none of such assets (other than toans) are subject to any restrictions, comractual, steautory or other, that would materially impair the ability of the entity bolding such (avestment) to dispose freely of any such assets at any time, except restrictions on the public distribution or transfer of any such favesiments under the Securities Act and the regulations therounder or state. securities lowe and pledges or security interests given in connection with government deposits. All loars, leases or other extensions of credit outstanding, or commitments to make any loans, leases or other extensions of prodit made by Bank to any Affiliates of Bank are disclosed on For outstanding loans or extensions of cubile where the original principal Schedole 3.22 amounts are individually in excess of \$100,000 and which by their trans are either secured by collaters) or supported ity a guaranty or similar ubligation, the security interests have been duly prefected to all material respects and have the priority they perpert to have in all material pryone is, other that 'we operation of law, and, in the case of each guaranty or somilar obligation, each has been duly executed and delivered to Bank and is still in full force and effect.

3/23 Employees, <u>Employee Benefit Plans: ERJSA</u>.

3.23.1 All material obligations of Bark for payment to trusts or other finds or to any Governmental Funty or to any individual, director, officer, employee or agent (or his or herheids, legators or legal representatives) with respect to unemployment compensation benefits, profit-sharing, pension or retirement benefits or potal security benefits, whether arising by operation of fails, by contract or by past custom, have been properly accrued on the Bark's Financial Statements for the periods povered thereby and paid when due. Except as disclosed on Schedule 3.23, all material obligations of the Bank, whether arising by operation of faw, by contract or by past custom for vacation or holiday pay, bonuses and other forms of compensation which are payable to vis directors, officers, employees or agents have been properly accrued on she Bank's Financial Statements for the periods covered thereby and paid when due. There face

no unfair labor practice complaints, strikes, slowdowns, stoppages as other controversies pending or attempts to prionize or controversics the stened by Bank or say of its Affiliates and/or relating to any of its employees that me likely to baye a Material Adverse Effect on Bank. Bank is not a party to any collective barganing agreement with respect to any of its employees and, except as set forth on Schedule 3.23.1. Except as set forth on Schedule 3.23.4. Bank is not a party to a written enaplestrant contrart with any of its enaployees, and there are an understandings with respect to the employment of any officer or resplaym of Bank which are not terminable by Bank withow Hability on not more than thirty (30) days' notice. Eccept as disclosed in the Bank's Financial Statements for the periods covered thereby, all material sums due for employee compensation have been paid and all employee combibutions for employee benefits, preloding deferred compensation obligations, and all insterial benefit obligations under any Employee Can (as defined in Section 3.23.3 hereof) or any Benchi Arrangement (as defined in Solition 3.73.4 hereof) have been duly adequately paid or provided for in accordance with plandemounts. Except as yet (reth on Schedule 3.23.1, on director, officer or employee of the Bark. as entitled to receive any gayment of any amount under any existing agreement, severance planor other benefit plan as a resnu of the consumpation of any transaction contemplated by this Agreement or the Merger Agreement. The Bank has materially complice with all applicable federal and state statutes and regulations which govern workers' compensation, equal employment opportunity and equal pay, indicating, but not limited to, all tivil rights laws, Presidential Executive Order 11246, the Four Labor Standards Act of 1938, as smended, and the Americana with Disabilities Act.

7.23.7 Bank has delivered as Schedule 1.23.2 a complete fist of all Employee Plans and Benefit Attangements (as defined in Socilium 3.23.3 and 3.23.4 bereat), including all plans or practices providing for corrent compensation or arrivaly for active employees, including, but not limited in, all employee benefit plans, all pension, profit-shering, retirement, bonus, stock option, locentive, deferred compensation, severance, long-term disability, medical, dental, bealth, hospitationtion, sife insurance or other insurance plans or related benefits.

3,23.3 Except as disclosed on Schodule 3 23.3, the Bank maintains, administers or otherwise contributes to any "employee benefit plan," as defined in Section 3(3) of ERISA. which is subject to any provisions of FRISA and covers any employee, whether active or extired. of the Bank of any Subscharies (any such plan bring lighting referred to as an "Employee Plan"). True and complete copies of each such Employee Plan, including anomhorids threate, have been providually delivered or made available to the Bank, to entite with $(i_1 * i_2)$ systematics regarding plan assets with respect to such Employee Plans. (ii) a true and complete copy of the annual reports for the most receive three years (Form \$500 Somes including if applicable, Schedules A and B therein) propared in connection with any such Employee Plan, (iii) a true and complete copy of the actuarial valuation reports for fire most recent three years. If any, prepared in connection with any such Euglinyee Plan covering any active employee of the Bank or its Submitiaties, (iv) a copy of the most recent amountry plan description of each such Employee. Plan, together with any modifications thereto, and (v) a copy of the most exact favorable determination letter (c) applicable) from the Internal Revenue Service for each Employce Plan. None of the Employee Plans is a "multiemployer plan" as defined in Section 3(57) of ERISA or a "multiple ensployer plan" as covered in Section 412(c) of the IRC, and the Bank has not been obligated to make a costruction to any such multicreptoyer or shultable employer plan written the

past five years has been, subject to Tide IV of ERISA. Fach Employee Plan that is intended to be qualified under Section 401(a) of the JRC is so qualified and each trust instituted pursuant thereto is exempt from income tax under Section 501(a) of the IRC, and Bank is not aware of any fact which has occurred that would existe the loss of such qualification of exemption.

5.23.4 Except as disclosed on Schedule 5.23.4, Bank has not maintained (other then bute salary and base wages) any form of current or deferred compensation, house, stock option, stock appreciation right, severance pay, salary continuation, retirement or scientive plan or anangement for the hought of any director, efficer or employee, which is active or retired, of the Bank or for any news or classes of such directors, officers or employees. Except as disclosed on Schedule 3.23.4, the Bank does not maintain any group or individual health insurance, welfare or similar plan or arrangement for the benefit of any director, officer or employee of the Bank, whether active or retired, or for any class or classes of such directors, officers or employees. Any such plan or arrangement described in this Section 3.23.4, copies of which have been delivered or made available to Bank, shall be toreits reterred to as a "Benefit Arrangement."

3.23.5. To Bank's Knowledge, a'3 Employed Plans and BancSt Arrangements are operated in material compliance with the requirements prescribed by any and all statutes. governmental or coun orders, or governmental rules or regulations currently in effect, including but not limited to ERISA and the IRC, applicable to study plays or arrangements, and plan documents relating to any such plans or atrangement materially comply with or will be attended to materially comply with applicable legal technicments. Neither the Bank, not any Employee Flan not any trust created thereunder, nor any trustee, administrator not any other fiduciary. thereof jus engaged in a "prohibited transaction," as defined in Section 406 of ERISA and Section 4575 of the IRC, that could subject the Bank to liability under Section 409 or 502(1) of FRISA or Sections 4975 of the IRC or that would adversely affect the qualified states of such plans, each "plan official" within the meaning of Section 412, with respect to each Employee Plan, to each of the Bank's Knowledge, no employee of the Bank, nor any fiduciary of any Employee Plan, has engaged in any breach of fiduciary duty as defined in Part 4 of Subtitle B of Title I of ERISA which could subject either Bank or any of its Subsidiaries to liability if the Bank, Bank or any such Subsidiary is oblighted to indemnify such Person against Jabibity. Except as unsclosed on Schedule 3.23.5, the Bank has not failed to make any material contribution of pay my amount due and owing as required by law or the terms of any Employee. Plan of Bencht Attangement.

3.23.6 Except as set forth on Scheidele 3.23.6 to Employer Pica or Benefit Arrangement has any material liability of any nature, accred or contingent, including, without limitation, liabilities for federal, state, local or foreign taxes, interest or penalty other than hability for claims arising in the course of the administration of each such Employee Plan. Except as set forth an Schodule 3.23.6, to each of the Bank's Knowledge there is no pending or threatened legal action, proceeding or investigation against any Employee Plan that could result or interestal liability to such Employee Plan, other form routine claims for benefits, and there is no basis for any such legal action, proceeding or investigation. 3.23.7 To each of the Bank's or Bank's Knowledge, each Benefit Attanguation which is a group health plan (within the meaning of such term under IRC Section 4980B(g)(2)) materially complies and has materially complied with the requirements of Section 601 through 608 of ERISA or Section 4980B of the IRC governing continuation coverage sequirements for employee-provided group nextth plans.

3.23.8 Except as disclosed on Schedule 3.23.8, Bank has not instituted any Employee Plus or Benefit Atrangement pursuant to which any benefit or other payment will be required to be made by the Bank or its Affiliates or pursuant to which any other bouefit will accrose or vest in any director, officer or employee of the Bank or its Affiliates, in other case as a world of the construction of the transactions contemplated by this Agreement or the Merger Agreement.

3.24 Razardous Materials

Except as set for on Schedule 3.24:

3.24.1 Erroge for ordinary and monssary quantities of choosing, post control and office supplies, and other small quantities of Hazardous Substances that are used in the ordinary course of hosionss and in compliance with applicable Environmental Laws, or ordinary rubbieh, debris and nonhazardous solid waste stored in gamage cans or hins for regular disposal off cite, or petroletam conversed in, and de minutus quarkities discharged from, motor vehicles in they ordinary operation on any of the Bank Projectics (as defined below), Buok has not engoged in the generation, use manufacture, treatment, transportation, storage (in tanks or otherwise), or disposal of Hazardous Substances other than as permitted by and only to compliance with applicable law. Since the earlier of the date that the Bank was formed as a legal entity, or material amount of Hazardous Substances has been achesed, circuital or disposed of or otherwise deposited, on, in or from any real property which is now or has been previously owned since the date the Bank, respectively opened for business, or which is currently or during the past three years was leased, by Bank, including OREO (collectively, the "Bank Properties"), and the Bank has not received written period that any of the same has docurred with respect to any real property in which the Sank now holds any security interest, mortgage or other lies or interest. ("Bank Collateralizing Real Estate"), except for (i) matters disclosed on Schedule 3.24; and (ii) no activity has been undertaken on any of the Bank Properties since the variation of the date One either the Bank opened for business, and the Bank has not received any written notice that activities have been or are being undertaken on any of the Baok Collateralizing Real Estate, that would cause or courrilitate to ...

(a) any of the Bank Properties or Bank Collatorulizing Real Estate becoming a treatment, storage or disposal facility within the meaning of RCRA or any similar state low or local ordinance;

(b) a release or threatened release of any Hazardous Substances under circumstances which would violate any Environmental Laws: or (c) the discharge of Hazardous Substances into any soil, subsurface water or ground water or into the sir, or the discigne or filing of any waters, that would require a permit or any other approval under the i-ederal Water Pollution Control Act, 33 U.S.C. §1251 cr arg., the Gean Air Act, as amended, 42 U.S.C. § 7401 et seq., or any similar federal or state law or local ordinance: the cumulative effect of which would have a Material Adverse Effect on any Bank Property or any Bank Collactativing Real Estate involved.

3.24.2 Except as disclosed on Schedule 3.24, there are not, and never have been, any underground storage tanks located in or upder any of the Bank Properties or the Bank Collateralizing Real Estate.

3.24.3 Bank has not received any written notice of and to the Knowledge of Bank, nor has received any verbal nonce of, any pending of themstened claims, investigations, administrative proceedings, hitigation, mychatory hearings or requests or demands for remedial or responsive actions of for compensation, with respect to any of the Bank properties or Bank Collateralizing Real Estate, alleging concomphance with or workston of any Environmental Law or seeking celief under any Environmental Law, and none of the Bank Properties or Hank Cuttateralizing Real Estate is listed on the United State Environmental Protection Agenes's National Priorities List of Hazardous Waste Sites, or, to the Xnowledge of Bank, any other list, schedule, log, investory or record of hazardous waste sites manifement by any federal, state or local agency.

1.24.4 "Hazanāms Sobstance" sind mosn any bazandnus, toxic or information substance, material, gas or costrois regulated by any local state or federal Governmentel Entry, in any officer agencies.

3.25 Risk Management Instruments

Neither the Bank nor any Subsidiary is a party or has agreed to enter into an exchange traded or over-the-counter equity, interest rate, foreign exchange or other swap. forward, future, option, eap, theor or collar or any other contract that is not included on the balance sheet and is a derivatives contract (including various confinations thereof) (each a "Derivative's Contract") or owns accurates that (i) are referred to generally as "structured cours," "high risk mongage derivatives," "copper' floating rate notes" or "capped ricentity or exchange derivatives" or (ii) are likely to have changes in value as a result of interest or exchange rate changes that significantly exceed normal changes in value as a result of interest or exchange rate changes that significantly exceed normal changes in value attributable to interest or exchange rate changes, except for those Derivatives Contracts and other instruments legally purchased or entered into in the orderary course of business consistent with safe and sound hanking practices and regulatory guidance and previously disclosed to the Ravk

3/26 Accounting Reports

The Basis maintains accounting records which fairly and validly reflect, in all material respects, its transactions and accounting controls sufficient to provide costonable assurances that such transactions are (i) exceeded in accordance with its management's general or specific.

authorization, and (5) recorded as necessary to permit the preparation of financial statements in conformity with GAAP. Such records, to the estend they contain material information pertaining to Bank which is not easily and read by available elsewhere, how there dophicated, and such cuplicates are stored safely and securely.

3.27 Matarial Contracts

Schedule 3.27 to this Agreement contains a complete and accurate written list of all agreements, obligations or anderstandings a) involving payment or recently of consideration encoding \$100,000 per year, and/or b) which are not terminable within six (6) months, written and one), to which the Bank is a party as of the date of this Agreement, except for loans and other extensions of credit made by the Bank in Ate ordinary course of its business and those items specifically disclosed in the Bank Financial Speciments.

5.28 Performance of Obligations

by cept es disclosed on Schedule 3.28, the Bank has performed all material obligations required to be performed by it to date and is not in material default under or in breach of any term or provision of any covenant, contract, lease, indentute or any other agreement, written or mal, to which it is a parsy, is solghed to is indertwise bound, and an event has occurred that, with the giving of entire or the passage of time or both, would constitute value a default or breach or failure to perform would have a Materia. Adverse Effect on the Bank. Except as disclosed on Schedule 3.28 and except with respect to lease by the Bank, no party with when the Bank has an agreement that is of material importance to the pusheses of the Bank is in default thereupder.

3.29 Powers of Altomey

No power of attorney or surilar authorization given by the Bank is presently in effect or outstanding other than powers of attorney given in the redinary course of bas areas with respect to rembine matters.

3.30 Parachete Payments

The consummation of the Merger will not entitle any director, officer or employee of the Bank to any payment that would constitute a paracluite payment under IRC Section 280G.

3.31 Comorate Records

The minute books of the Bank accurately reflect all material schons taken by its shareholders, locard of directors and committees of their board of directors.

5.32 Accounting and Tax Matters

The Bork has not through the core hereof taken or agreed to take any action that would provent the Bank from qualifying the Merger as a conganization within the meaning of Nection 505 of the IRC.

133 Intellectual Property

To the nation that the Brink maps on preservors becauses and other log as to use tradespecets, made names, andemarks, service marks, internet domain name registrations, inventions, and processes used in its businesses, the Bank have not received any notice of conflict with respect thereto that asserts the rights of others. To the Back's Knowledge tim Back lies in all material respects performed all of the colligations required to be performed by it and is not in default in any material respect order any contract, agreement, arrangement or commitment teleting to any of the foregoing.

3.34 Material Interest of Cottain Persons

None of the current officers or directors of the Bank has any material interest in any material contract or property (real or personal), tangible or intangable, used in or pertaining to the business of the Bank (except his her ownership of the Bank's Shares).

ARTICUE 4 REPRESENTATIONS AND WARRANTIES OF COMPANY AND GT BANK

Company and GT Bank, jointly and severally, represent and warrant to Bank as follows.

4.1 Creanization: Corporate Jower, Etc.

ERCh of Company and GT Back is a California componation dely organized, validly existing and in good standing under the laws of the State of Califerria and has all requisite cosporate power and authority to own, lease and operate its properties and assets and to carry on its hasiness substantially as it is being conducted on the date of this Agreement. Company is a bonk holding company registered under the BHCA. Each of Company's Subridiaries has ailrequisite corporate power and authority to own, lease and operate its properties and assets and to carry on its business substantiality as it is being conducted on the date of this Agreement, except where the failure to have such power or authority would not have a Material Adverse Effect on Company takes as a whole or the ability of Company to consummate the transmission contemplated by this Agreement. Company has all requisite conporate power and automity to enter into this Agreement and, subject to obtaining all Requisite Regulatory Approxys. Company well have the requisite corporate power and authority to perform its inspective obligations bereastles with respect to the communication of the transactions contemplated hereby. Company is the sole shareholder of GT Bards. GT Bark is a state chartered barking corporation liacound to constart backing business in California. GT Bank is not a member of the Federal Reserve System. G1 Bank's deposits are insured by the FDIC in the manner and to the justextent provided by law.

4.2 Licenses and Permits

Company and GT Bank have all material licenses, certificates, frenchisos, nights and permus that are necessary for the conduct of their business as presently conducted, and such licenses are in full force and effect, except for any failure to be in fall force and effect dist would not, individually or in the aggregate, have a Material Adverse Effect on Company and GT Bank taken as a whole or on the ability of Company and GT Bank, to consummate the transactions contemplated by this Agreement. The properties, assets, operations and businesses of Company and GT Bank are and have been maintained and conducted, in all material respects, in complance with all applicable licenses, certificates, franchises, rights seat presents. The Company has no Knowledge of any facts or encounstance with respect to GT Bank's SBA lending programs that would probable the Bank upon the consummation of the Merger, from being able to make guzganteed toats under the programs administered by the U.S. Small Business Administration as a successor entity; provided, that the Bank has complied with the U.S. Small Business Administration regulations on successor entities...

4.3 Subsidiance

There is no corporation, partnership, joint venture or other entity in which Company or GT Bank own, directly or indirectly (except as pleatgre pursuant to loans or stock or other interest held as the result of or in ties of foreclessive pursuant to pleage or other security arrangement) any equity or other voting interest or position, other than Company's ownership of GT Bank or any entity facilitating the issuances of trust preferred securities.

4.4 Authorization of Acreement: No Conflicts

The execution and delivery of this Agreement and the Merger Agreement. 441 by Company and G1 Bank, and the consummation of the nansactions contemplated hereby and thereby, have been duly authorized by all pressury emporate action on the part of Company and GT Bank, subject only to the approval of this Agreement, the Merger Agreement and the Merger by Company's shareholders. This Agreement has been du'y executed and delivered by Company. and GT Back are constitutes a legal, valid and binding obligation of Company and GT Bank, enforceable in accordance with its terms, except as the enforceability thereof may be limited by bankroptey, liquidation, roceivership, conservatorship, insolveney, moratorium or other similar laws afteeting the rights of creditors generally and by general coultable principles and by Section 8(b)(6)(D) of the FDI Act, 12 U.S.C. § (318(b)(6)(D). The Merger Agreement, apon the receipt of all Requisite Regulatory Approvals and the due execution and filing of such Merger Agreement in accordance with the applicable provisions of the California Corporations Coric and the California Fiennial Cutle, will constitute a legal, valid a stim sling obligation of Company. and GT Bank, enforceable in accordance with its terms, except as the enforceability thereof may be limited by backruptcy, signification, receivership, conservatorship, insolvency, moratorium, or other similar laws affecting the rights of creditors generally and hy ceneral equitable principles and by Section X(b)(6)(D) of the FDI Act, 12 U.S.C. § 1818(b)(6)(D).

4.4.2 Except as disclosed on Schedule 4.4.2, the exception and delivery of this Agreement and the Morger Agreement, and the consummation of the transactions contemplated hereby and thereby, do not and will not conflict with, or result in any violation of or defails to have of a context in any provision of the Activities of formpoission or Bylaws of Company or GT Bank, any material mongage indennire lease, agreement or other material instrument or any permit, concession, grant, franchise, beense, judgment, order, decree, statue, law, ordinance, rule or regulation applicable to Company or GT Bank or any of their respective assets or properties, other than any such conflict, violation, default or loss which (i) will not have a Material Adverse Effect on Company and GT Bank token as a whole, or (ii) will be cared or waved prior to the Effective Time.

4.5 Capital Structure

Two automized tapital stock of Company consists of 20,000,000 shares of Company Common Stock, \$.01 pa value per share. On the date of this Agreement, 2,115,001 shares of Company Common Stock were obtaineding, and 495,000 shares of Company Common Stock were reserved for issuence pursuant to outstanding Company Stock Options under the Company Stock Option Plan. The authorited capital stock of GT Bank consists of 20,000,000 shares of GG Bank Common Stock All outstancing shares of Company Common Stock and GT Bank Common Stock are validly issued, fully paid and nonassessable and do not possess any preemptive rights and overe not issued in violation of any preemptive rights or any similar rights of any Person. Except for the Company Stock Options described on Schedule 4.5 to this Agreement, neither the Company nor GT Bank has outstanding any options, warrants, calls, rights, communents, securities or agreements of any character to which Company is a party or by which it is bound obligating Company to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock of Company is obligating Company to grant, extend or catter interary stock option, warrant, call, right, commitment or agreement.

4.6 Company Pilings

Except as disclosed on Schedule 4.6, since January 1, 2008, Company and OT Bank have such filed all reports, registrations and statements, together with any anomimums required to be mode with respect torreto, that were required to be filed with (a) the Federal Reserve or any Federal Reserve flank; (b) the CDFE (c) the SFF, (d) the FDIC, and (e) any other applicable federal, state or local governmental or regulatory anthonity. All such reports, registrations and filings are collectively refining to as the "Company Filings." Except to the extert prohibited by law, replies of the Company Filings have been made available to Bank. As of their respective filing or mailing dates, each of the past Company Filings (a) was true and complete in all material respects (or was amended so as to be su promptly following discovery of any discrepancy), and (b) complied in all moterial respects with all of the statutes, rules and regulations enforced or promotigated by the Regulatory Authority with which it was filed for was amended so as to be so promptly following discovery of any such noncomplance) and none costained end on the statement of a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances male which they were made, not misleading.

4.7 Accuracy of Information Samplied.

Company has delivered of will deliver to Bank capies of the Company Financial Statements, and Company will beceafter unfit the Closing Oate deliver to Bank copies of additional financial statements of Company as provided in Sections 5.1.1(iii). The Company Encantral Statements have been prepared (and all of said additional lineactal statements will be prepared) in accordance with GAAP, or applicable regulatory accounting principles applied or a consistent basis during the periods involved (except as may be unlikated in the order thereto). consistently followed throughout the periods revented by with statements, and present (and, when prepared, will present; fairly the Business) cosition of Company or GT Bank, as appropriate, as of the respective detes indicated and the results of operations, each flows and changes in shareholders' equity at the respective dates and for the respective periods covered by such financial statements (subject, in the case of the unaudited statements, to recurring adjustments nomial in nature and amount) including, but not limited to the calculation for loan and lease. Josses In addition, Company has delivered or made evailable to Bank popies of all management. or other letters delivered to Company or GT Bank by their indepresent accomptants in connection with any of the Company Financial Statements or by such accountance or any consultant ceparding the internal controls or internal compliance procedures and systems of Company or GT Bank (syned at any time smoot lanuary 1, 2007, and will make available for

inspectice by Bank on its representatives, at such times and places as Bank may reasonably request reports and working papers produced or developed by such accountants or consolitants.

4.8 Compliance with Applicable Laws

Except as disclosed on Schedule 4.8, the respective businesses of Company and GT Bank are not being conducted in violution of any law, ordinance or regulation, except for violations which individually or in the aggregate would not have a Material Adverse Effect on Company and GT Bank, taken as a whole. Except as set forth on Schedule 4.8, an investigation or review by any Environmental Energy with respect to Company to GT Back, other theo regular back examinations, is prinding or threatened, nor has any Governmental Energy indicated to Company or GT Back an intension to conduct the same.

4.9 Lititation

Except as set forth on Nobedule 4.9, there is no suit, bettom or proceeding or investigation pending or threatened against or affecting Company which, it adversely determined, would have a Material Adverse Effect on Company and GT Bark taken as a whole; nor is there any judgment decree, injunction, rule or order of any Governmental Entity or arbitrator outstanding against Company or GT Bark that bay, in which, insofar as reasonably can be forescen, in the bittore would have, any such Material Adverse Effect. Schedule 4.9 contains a true, correct and complete list, including identification of the applicable insurance policy covering such highlion, if any, subject to reservation of rights, if any, the applicable deductible and the anomal of any reserve therefor, of all pending litigation in which Company or GT Bank is a named party of which Company has knowledge, and except as disclosed on Schedule 4.9, all of the litigation shown on such Network litigation generated by insurance in force except for applicable deductibles, or has been adequately reserved for in accordance with Company's prior besiness oractices.

4.10 Agreenses s with Banking Authorates.

Except as yet forth on Schedule 4.10, neither Company nor GT Bank is a party to any written agreement or memoratedum of understanding with, or order or directive from, any Governmental Entity.

4.31 te sugance

Company and G1 Dark each has in full force and effect policies of theorance with respect to its assets and business against such casualties and contingene es and in such a nature, types and formes as are customarily appropriate for its bacterist, operations, projectins and assets School de 4.11 repetations a first of all polities of insurance and hends repetations projectins and assets School de 4.11 repetations a first of all polities of insurance and hends repetations outstanding thereunder have been filed in the ely fashion. Company and GT Bank have filed claims outstanding thereunder have been filed in the ely fashion. Company and GT Bank have filed claims with or given notice of claim to, their insurances or bonding companies in timely liashion with respect to all insurance and eccurrences for which they helicys they have coverage.

4.2 Title to Assets other than Rest Property.

Everya as disclosed on Schedule 4.12, Company and GT Eark have marketable title to or a valid leasehold interest in a limiterial properties and assets (other than respectly which is

the subject of Section 4.13) used in their business, free and clear of all mortgages, covenants, conditions, restrictions, easoments, liens, security interests, charges, claims, assessments and cheminbranees, except for: (a) rights of tessors, lessees or sublessees to such matters as are reflected in a written tense, (b) exemptranees as set forth in the Company Financial Statements, (c) current Texes (including assessments collected with Texes) not yet due which there fully reserved for: (d) encumbranees, if any, that are not solutionized use, or the ability of Company or GT Bank or its Subsidiary to sell or otherwise dispose of the property subject thereto or affected thereby; and (e) other maters as described in Schedule 4.12. All such properties and assess are, and require only routine maintenance to keep them, in good working condition, normal wear and tear cycepted.

4.13 <u>Real Property</u>

Schedule 4.33 is an accentre list and general description of all real property owned or leased by Company and GT Bard, including Other Real Estate Owned ("OREO"). Company and GT Bank have marketable title to the real properties that they own, as described in such Schedule, free and clear of all mortgoges, covenants, conditions, restrictions, catements, hens, scentity microsis, charges, claims, assessments and encumbrances, except for (u) news of lessons, lessers or subjestees in such matters as are reflected in a written lease; (b) heris for current layes including assessments collected with Taxes) not yet due and pavable. (c) encumbriances, if any, that mus not substantial in character, amount or extent and do not materially detract from the value, or interface with present use, or the ability of Company or GT. Dank to dispuse, of Company or OT Bank's interest in the property subject formes or affected thereby, and (d) other metrics as described in Schedule 4.13. Company and GT Bank have valid. insertiold interests in the leaseholds it holds, free and clear of all mortgages, liens, security interests, charges, claims, essentiments and englimbrances, except for (a) claims of lessons, cotensions or sublessees in such matters as are reflected in a written lease; (b) title exceptions affecting the fee entrie of the lessor under such leases; and (c) other matters as described in Schoolle 4.13. The activities of Company and GT Bank with respect to all real property owned. or leased by it for use in connection with its operations are in all material respects pointiled and authorized by applicable zoning laws, ordinances and regulations and nli Jaws and regulations of any Governmental Littity - Except as set forth on Schedule 4.1.3. Company and GT Bank copyabid possession under all material leases to which it is the lessee and all of such leases are valid and in fall force and effect, except as the colorizability thereof may be 'invited by bankrupicy, injustation, incoversing, conversionship, assolvency, moraurium or other similar laws affecting the nights of creditors generally and by general equitable principles. Materially all buildings and improvements on real properties owned or leased by Company and GT Roak are in good condition and repair, and do not require more than corrant and costine multiconnec to keep them in such condition, normal wear and tear excepted.

4.14 <u>Taxes</u>

4.14.1 Filing of Returns. Except as set forth on Schedule 4.14.1, Company and GT Bank bave duty prepared and filed or caused to be duly prepared and filed all federal, stole, and local Returns (for Tax or informational purposes) which were required to be filed by or in respect of Company and GT Bank or any of its properties, menne and/or operations on or pivor to the (Testing Date - As of the time they were filed, the facegoing Returns aroundedly reflected.

the material facts regarding the meaner, business, asset, optrations, activities, states, and any other information required to be shown thereon. Except as set forth on Schedule 4.14 i, no extension of time within which Company or GT Baak may file any Return is currently in force. Within five business days following a written request by the Bank, the Company shall deliver to the Bank a true and correct copy of Company's 2010 federal and California income tas returns, as filed with the respective taking authorities.

4.14.2 Payment of Taxes - Except as discloard on Schedule 4.14.2 with respect to all amounts in respect of Taxes imposed on Company of GT Bank or for which Company of GT Bank is or could be liable, whether to taxing authorities (as, for example, under low) or to other Persone (as, for example, under Tax allocation agreements), with respect to all taxable periods on periods ending on or before the Closing Date all applicable tax laws and agreements have been or will be fully completed with to all material respects, and all such amounts required to be bate by or on behalf of Company or GT Bank to taxing authorities or such other Persons on or before the date hereof have been paid.

4.14.3 Audit History. There is no review or audit by any faxing authority of any Tax liability of the Company or QT Bank currently in progress of which the Company or GT Bank has knowledge. Except as disclosed on Schedwie 4.14.3, neither Company nor GT Bank has received any written notices since its inception of any pending or threatened audit, by the Internal Revenue Service or one state, local or foreign agency, for any Returns or Tax hability of either the Company or GT Bank for any penol. Nother the Company nor GT Bank currently has any experied deliveration assessed by the Internal Revenue Service or any state, local or foreign taxing authority arising out of any examination of any of the Returns of either the Company or GT Bank for any Subsidiaries tiled for fiscal years ended on or after Deter ber 51, 2000 through the Cherling Date, not es there means to believe that any material deficiency on any such entities with be assessed.

4.4.4 <u>Statute of Limitations</u>. Except as disclosed on Schedule 4.14.4, no agreements are inforce or are currently being negotiated by or on behalf of either the Company or G1 Bank for any wriver or for the extension of any statule of limitations governing the trace of essessments or collection of any Tax. No closing agreements or compromises exist concorning Taxes of either the Company or GY Bank.

4.14.5. <u>Witheriding Obligators</u> — Surept as set finih on Scanlah 4.14.5, the Company action GT Bank, as applicable has withheld from each payment made to any of its officers, directors and employees, the amount of all applicable Taxes, including, but not limited to, income tax, social security contributions, unemployment contributions, backap withholding and other dadhelions required to be withheld therefrom by any itax invit and have paid the same to the proper taking anthomics without the intrinequired onder any applicable. Tax is invit to the same to the proper taking anthomics without the intrinequired onder any applicable.

4.14.6 <u>Taylings</u>. There are on Tox lices, whether imposed by any federal, state, local or foreign taking authority, outstanding against any assets owned by either the Company or GT Back except for lices for Taxes that are not yet due and payable.

4.14.7 Tax Reserves. The Company and GT Bank each has made full and adequate provision and reserve for all federal, size, bocal or foreign Pakes for the current period for which Tax and information returns are not yet required to be filed. The Company Financial Statements contain fair and sufficient accruals for the payment of all Taxes for the periods covered by the Company Financial Statements and all periods prior thereto.

4.14.8 EQC Section 382 Applicability. Neither the Company nor Bank, including any party joining in any consolidated return to which Company is a member, has undergone at "ownership change" as defined in IRC Section 382(g) within the "testing period" (as defined in IRC Section 382) anding immediately before the Effective Time, and not taking into account any transactions contemplated by this Agreement.

4.15 Loans and investments

Except as set forth on Schedule 4.15, all loans, leases and other extensions of circla, guaranties, security agreements or other agreements supporting any locus or extensions of credit, and investments of Contrary and GT Bank are, and constitute, in all custorial respects, the legal, valid and binding obligations of the parties thereto and are enforceable against such parties to accordance with their terms, recept as the enforceability thereof may be limited by applicable law and otherwise by bankruptcy, liquidation, receivership, conservatorship, insolvency, minimizion or other similar laws affecting the rights of creddors generally and by general equitable principles. Except as described on Schedule 4.15, as of March 31, 2011, no loans or investments held by Company or GL Bank are (a) more than minety (90) days past due with respect to any scheduled payment of principal or interest, other than loans on a non-nervial scalar, (ii) classified up "less," "doubtfol," "substandard" to "spin isity mentioned" by Company, GT Bank or any banking cognitions, or (iii) on a renearing status in accordance with Company or GT Bank's loan review procedures. Follept as set forth on Schedule 4.15, nove of such assets (other than loaves) are subject to any restrictions, contractual, statutory or other, that would matrically impair the ability of the entity holding such investment to dispose freely of any such assets at any time, except restrictions on the public distribution or transfer of any sixth investments under the Securities Act and the regulations thereunder or state securities laws and pledges of security interests gives in connection with government depends. All loads, leases or other extensions of credit datatanding, or cramminum in mole any loans, leases or other extensions of credy made by Cramony or GT Bank to any Affiliates of Company or GT Bank are disclosed on Submitting 4.15. For ourstanding loans or extensions of credit where the original principal amounts are individually in excess of \$100,000 and which by their terms are other serviced by collateral or supported by a guaranty or similar obligation, the security interests have been duly perfected in all material respects and have the property they purport to have might material accession other than by operation of law, and, in the case of each presenty or similar obligation, each has been duly executed and deliverent to Company or GT Bank and is still in full force and effect.

4.16 <u>Brokers and Finders</u>

Except as set forth on Senedule 4.16, norther Company nor GT Bank as a party to un obligated under any agreement with any broker or funder relating to the transactions contemplated hereby, and norther the execution of this Agreement, on the Mesgie Agreement, for the consummation of the transactions provided for been or deterin, will result in any liability to any broker or finder. Company agrees to indemnify and hold harmless Bank, and to defend with convect selected by Back and reasonably satisfactory to Company, from and against any liability, cost or expresse, including acorneys' fees, occurred in convection with a locach of this Section 4.16

4.17 Material Contracts.

Schedule 4.17 to this Agreement contains a complete and accurate written list of all agreements, obligations or understandings a) involving payment or receipt of consideration exceeding \$10,000 per year and/or b) which are not remainable within six (6) months, written and ord, to which either the Company or GT Bark is a party as of the date of this Agreement, except for loans and other extensions of credit made by GT Bank in the ordinary course of each business and those steins specifically disclosed in the Company F naticial Statements.

4.18 Absence of Malerial Adverse Effect.

Since January 1, 2008, the businesses of Company and GT Bank (or GT Bank alone, prior to formation of Company) have been conducted ordy in the oromaty course, in the same manner as theretofore conducted, and no event or c roumstance has occurred or is expected to event which to Company's or GT Bank's Norwhedge has had or which with the passage of tene or otherwise, could reasonably be expected to have a Material Adverse Effect on Company and/or GT Bank taken as a whole.

4.19 <u>Performance of Obligations</u>

Except as disclosed on Schedule 3.19, the Company and GT Bank have each performed all material obligations required to be performed by them to date and neither is an material detail: under or in breach of any term or provision of any covenant, contract, lease, indenture or any other agreement, written or orall to which either is a party, is subject or is otherwise bound, and no event has occurrent that, with the giving of notice or the passage of time or both, would creating eached a detault or breach, when such default in breach or failure to perform would have a Material Adverse Effect on Company and/or GT Sank taken as a whole. Eacept as disclosed on Schedule 4.19 and except with respect to boars by GT Bark, no party with whom Company or GT Bank has an agreement that is of intervital importance to the businesses of either the Company or GT Bank is in default therounder.

4 20 [Imployees Employee Benefit Plans; FRISA]

4.20.1 Except as set forth in Schedule 4.20.1 all matched oblegations of tabler the Company or GT Bank to payments to trusts or other funds or to any Governmental Entity or to any individual, director, officer, coupleyee or agent (or hos or her here, legatees or legal representatives) with respect to unemployment compensation benefits profit-sharing, pension or retirement benefits or social security benefits, whether arising by operation of 'aw, by contract or by post englors, have been property accured on the Company Pinancial Statements for the periods covered thereby and past when due. Except as disclosed on Schedule 1.20.1, all material obliganous of other the Company or GT Bark, whether arising by operation of law, by contract or by past custom for vacation or boliday pay, benuses and other forms of compensation which are payable to its directors, officers, employees or agends taxe been properly accured on the Company Pinancial Statements for the periods covered thereby and paid when due. There are no onfair labor gradite complaints, strikes, slowdowns, stoppages or other controversies pending or onfair labor gradite complaints, strikes, slowdowns, stoppages or other controversies pending or attempts to philomize or controversies threatened between Company or UT Bank or any of their Affiliates and/or relating to any of their employees that are likely to have a Material Adverse. Effect on Company or GT Baas. Neither the Company use GT Baak is a party to any collective bargaining agreement with respect to any of its employees and, recent as set both on Schedule 4.20.1. Neither the Company por GT Bank is a party to a written employment contract with any of its employees, and there are no understandings with respect to the employment of any officer or employee of the Company or GT Bank which are not terminable by either the Company or G1 Back without isobility on no; more than thirty (30) days' notice. Except as declosed in the Company Financial Statements for the periods covered thereby, all material sums due for employee compensation have been paid and all employer contributions for employer bracfits, including deferred compensation oblightions, and all material benefit. obligations under any Employee Plan (as defined in Section 4.20.3 Intereof) or any Benefic Arrangement (as defined in Section 4.21.4 hereof) have been only and adequately paid or provided for in accordance with plan documents. Except as set forth on Schedule 4.201, no d rector, officer or employee of the Company or G1 Bark is entitled to receive any poyment of any amount under my existing agreement, severance alon or other bonefit plan as a result of the consummation of any transaction contemplated by this Agreement or the Merger Agreement. The Company and GT Bank have each statenally complied with all applicable federal and state variases and regulations which govern workers? compensation, equal employment opportances and equal pay, inducting, but not initial to, all civil rights layes, Prosticutial Execution Order 11246, the Fast Labor Standards Act of 1938, as amended, and the Americans with Disabilities A.ct.

6.20.2 Company and GT Bank have delivered as Schedule 4.20.2 a complete list of all Employee Plans and Benefit Amangements tas defined in Sections 4.20.3 and 4.20.4 hereof), including all plans or practices providing for current compensation or accruals for active employees, including, but not limited to, all employee benefit plans, all pension, profit-sharing, remement, bound, stock option, incentive, deferred compensation, severance, long-torm disability, aexists', dental, leatth, hospitalization, life incurance of other assurance plans or related innefits.

4.20.3 Except as disclosed on Schedele 4.20.3, namer the Company nor GT Bank maintains, administers or otherwise contributes to any Temployee benefit plan," as defined in Section 3(3) of ERISA, which to subject to any provisions of ERISA and covers any employee, whether active or retired, of the Company or GT Bank or any 61 its Subinthanos (any such plan bring herein referred to as an "Employee Plan"). Thus and complete copies of each such Employee Plan, including according to thereto, have been providely delivered or made accelable to the Bank, logistics with (i) all agriculturity regarding plan assets with respect to such Employee Plans, (ii) a true and complete copy of the annual reports for the most recent three years (Form 3500 Sectes including, if applicable, Schedieles A and B thereto) prepared in connection with any such Employee Plan, (iii) a true and complete copy of the actuarial valuation reports for the most recent three years, if any, prepared in connection with our such Employee Plan covering any active employee of the Company or GT Bank or its Subadianes, (ii) a copy of the asist recent summary plan description of each such Employee Plan, together with any andifications thereto, and (iv) a recy of the next recent three (if lephcable), from the Internal Revenue Service for each Employee Plan. None of the imployee Plans is a "multiemployer plan" as defined in Section 3(37) of ERISA et a "multiple employer plan" as covered in Section 412(c) of the IRC, and neither the Company for GT Bank has been obligated to make a contrelation to any such institionaployer or multiple employer plan within the past five years. None of the Employee Plans of the Company on GT Bank is, or for the last five years has been, subject to Thile IV of ERISA. Each Employee Plan that is imended to be qualified under Section 401(a) of the IRC is so qualified and each trust maintained pursuant thereto is exempt from income tax under Section 501(a) of the IRC, and neither Company nor GT Bank 15 not aware of any fact which has occurred that would ecuase the loss of such gualification or exemption.

4.20.4 Except as disclosed on Schedule 4.20.4, better the Company on GT Back maintains (other than base salary and base wages) any form of current or deferred compensation, borne, stock option, stock appreciation tight, severance pay, salary continuation, retirement or incentive plan or arrangement for the benefit of any director, officer or employee, whether active or retired, of the Company on GT Bank or for any class or classes of such directors, others or employees. Except as disclosed on Schedule 4.20.4, neither the Company not GT Bapk trainians any group or individual health insurance, welfare or sum at plan or arrangement for the benefit of any director, officer or employee, whether active or retired or for any class or classes of such whether active or retired or for any class or classes of such directors, officers or employees. Whether active or retired or for any class or classes of such directors, officers or employees Any such plan or arrangement described in this Section 4.20.4, represent whether have mendelivered or mark available to Bank, shall be herein referred to as a Benefit Arrangement."

4 20.5 To each of the Company's and GT Bank's Knowledge, all Employee Plans and Benefit. Arrangements are operated in material compliance with the requirements prescribed by any one all statules, governmental or court orders, or governmental rules or regulations currently in effect, including but not kniwled to ERISA and the IRC, applicable to such plans or amangements, and plan documents relating to any sock plans or arrangement. materially comply with or will be anonaed to materially comply with applicable logal reganzers - Neither the Concerny, OT Benk, not zoy Etaployte Plan for any trusts created durrander, an any trastre, administrator can any other follows y thereof has engaged in a "prohibited transection," as defined in Section 406 of ER/SA and Section 4975 of the IBC, that could subject the Company of GT Back to Easility under Sect on 409 or 502(t) of ERISA or Section 4975 of the IRC or that would adversely affect the quaitified status of such plans; each "plan official" within the meaning of Section 412 of ERISA of each Employee Plan is bonded to the extent required by such Section 442; with respect to each Employee Plan, to each of the Company's of GT Bank's Knowledge, no employee of the Company of GT Bank, nor any fiduciary of av y limployee Play, has engaged in any breach of fiduciary doty as defined in Part 4. of Subility Blot Tath Lof FRISA which could selve the the Company, GT Bank or any of its Subsidiation to fieldity if the Company GT Bank or any such Subsidiary is obligated to ademnify such leason against hability. Except as displayed on Schedule 4.20.5, no they he Cempany for GT Bank has failed to make any material contribution or pay any amount due and owing os required by law or the terms of any Employed Plan or Benefit Attemptnetter.

4.20.6 Except as set forto on Schedule 4.20.6, no Employee Plac or Benefit Areasygement has any material liability of any nature, accuracity contingent, including, without heritation, liamatters for feneral, state, local or foreign taxes, interest or penalty other fram liability for closms arising in the course of the administration of each such Employee Plan Except as set forth on Schedule 4.20.6, to each of the Courpany's or GT Bank's Knowledge there is no pending or threatened legal a new, proceeding or investigation against any Employee Plan that could result in masserial Stability to each Employee Plan, other than routine clatent for brineSis, and there is no basis for any seen legal action, proceeding or investigation.

4.20.7 To each of the Company's or GT Benk's Knowledge, each Benetit Arrangement which is a group health plan (within the meaning of such term under IRC Section 4980B(g)(2)) materially complies and has materially complied with the requirements of Section 601 through 608 of ERISA or Section 4980B of the IRC governing continuation deverage requirements for employee-provided group health plans.

4.20.3 Except as disclosed on Schedule 4.20.8, neither the Company nor Gt Bank maintains any Employee Plan or Benefit Arrangement pursuant to which any benefit or other payment will be required to be made by the Company or GT Bank or its Affiliates or parsmant to which any other benefit will accrue to vost at any director, officer or employee of the Company or GT Bank or its Affiliates, in other case as a result of the consummation of the transactions contemplated by this Agreement or the Marge: Agreement

4.21 Powers of Alloung

No power of submey or similar authorization given by either the Company or GT Bank is presently in effect or outstanding other than powers of attorney given in the ordinary course of butiness with respect to routice matters.

4.22 Hozardous Materials Except as set forth on Schedule 4 22:

4 22.1 Except for ted mary and neuclasary quantities of cleaning, pest control and office supplies, and other small quantities of Hazardows Substances that are used in the online ycourse of business and in compliance with applicable. Festion neutral flaws, or endorsely rubbishdenois and annhazaidous solid waste socied in gathage cans or bins for regular disposal off-site. in prisolners costained in, and de merimus augnoties discharged from, motor vehicles in their ordinary operation on any of the GT Bank Properties (as defined below), neither the Company nor OT Bank has estgaged in Use generation, use, manufacture, treatment, transportation, storage, (in tasks or otherwise), or disposal of Hazardova Substances other than as permitted by and only in compliance with applicable law. Since the carlier of the date that either the Commany of the Bonk was formed as a legal entity, no material amount of Hazardous Substances has been coleased, consted or disposed of, or otherwise deposited, on, in or from any real property which is now he has been previously unward since the date the Company or GT Bank, respectively opened for business, or which is currently or during the dost three years was teased, by either the Company of GT Bank, including OREO (collectively, the "GT Bank Properties"), and nother the Company nor GT Bank has received written notice that any of the same has occurred with respect to any real property to which either the Company or GT Gank now holds any security. microst, mortgage or other lien or interest ("OT Back Collaterativing Real Estate"), except for (\hat{a} matters disclosure on Schedule 4.72, and (\hat{a}) are activity has been numerication on any of the GY. Bank Properties since the earlier of the date that either the Company of the Bank opened for business, and neither the Company ner GL Bank has received any written notice that activities have been or are being undertaken on any of the GT Bank Collateralizing Real Estate, that would cause or contribute to:

(a) any of the GT Bank Properties or GT Bank Collateralizing Real Estate becoming a treatment, storage or disposed facility within the obtaining of RCRA or any simple state law or local codimance;

(b) a release or threatened release of any Habardous Substances under circumstances which would violate any Environmental Laws; or

(c) the discharge of Hazardrees Substances into any sub, subsorface water or ground water or into the air, or the dredging or fibling of any waters, that would require a promitive any other opprival under the Federal Water Polletion Control Act, 35 U.S C. § 1251 wireq., the Clean Air Act, as oriented, 42 U.S C. § 7401 et seq., or any similar federal or state law or local ordinance; the comulative effect of which would have a Material Adverse Ellipot on any GT Bank Property or any GT Bank Collateralizing Real Estate mode ed.

4.32.2 Except as disclosed on Schedule 4.22, there are not, and never have here, any underground storage tanks located in or under any of the GU Bank Proporties in the GU Bank Collatoralizing Real Estate.

4.27.3 Neither the Company ner GT Bank has received any written durice of, and to the Knowledge of either the Company and GT Bank, neither has received any vorbal notice of, any pending or threatened claims, investigations, administrative proceedings, litigation, regulatory bearings or requests of demands for ventedial or responsive actions or for compensation, with respect to any of the GT Bank Properties or GT Bank Collisteralizing Real Estate, alleging noncompliance with crivitation of any Environmental Caw or socking telef under any Environmental Law, and none of the GT Bank Properties or GT Bank Collisteralizing Real Estate is holed on the United States Environmental Protection Agency's National Priorities itest et Hazetdous Waste Sites, or, to the Knowledge of the Company or GT Bank, any other list, schedule, log, inventory or record of hazerdous waste sites maintained by any federal, state or local agency.

4 22.4 "Hagardous Substances" shall mean any legatelous, toxic or infectious substance, material, gas or waste which is regulated by any local, state or federal Governmental Entity, or any of their agencies.

+23 Printinte Payments

The entrumentation of the Merger will not entitle any director, officer or employee of the Company or GT Book to any payment that would constitute a parachuse payment under (RC Section 280G)

4.24 Rick Management Jostnaneals

Nother the Company, GT Bank nor any Subsidiary is a party or has agreed to coast into an exchange traded or over-the-conner reprity, interest care, foreign exchange or other swap,

forward, finure, option, cap, floor or collar or any other contract that is not included on the balance sheet and is a derivatives contract (including various combinations thereof) (each, a "Derivatives Contract") or owns scentries that (i) are referred to generally as "structured notes," "high risk mortgage derivatives," "capped floating rate notes" or "capped floating isternoutlyage derivatives" or (ii) are likely to have changes in value as a result of interest or exchange rate changes that significantly exceed normal changes in value attributed to interest or exchange rate changes, except for three Derivatives Contracts and other instruments legally purchased or enterest into in the ordinary course of business consistent with safe and sound banking practices and regulatory guidance and previously disclosed to the Bank.

9.25 Liability Under Regulation C. Truth in Londing Law and HMDA.

To either the Company's or GT Bank's Knowledge, and eatent as disclosed on Selectule 4.25. UP Bank has no liabilities or obligations online account, coordingent or otherwise, that have a Material Adverse Effect on GT Bank with respect to Regulation C, Treth in Londing Law and HMDA disclosures

a.25 Undisclosed Liabilities

Except as disclosed on Schedule 4.36, neither Company for GT Bank has any habilities or obligations, either accrued, conjungent or otherwise, that are material to Company and GT Bank taken as a whole and that have not been (a) reflected or disclosed in the Company Establishes Statements: (b) meaned subsequent in December 31, 2008 in the ordinary mouse of business; or (c) interred in connection with the isolation of first preferred seconds of debt affer the date of bis Agreement. These is no basis for the assertion against Company or GT Basis of any liability, obligation or oblin (including without limitation that of any Governmental Endity) that will have or cause, or could reasonably be expected to have or cause, a Material Adverse Effect on Company and GT Basis taken as a whole that is not fully and fairly reflected and disclosed in the Company Financial Statements or on Schodule 4.26.

4 27 Bank Secrecy Act

GT Bank is in compliance with the G1 Bank Speccey Act (31 U S.C. § 5322, et eq.) and related state and federal anti-money heatsforing laws, regulations and guidelines, including vertices broatshow these provisions of federal regulations requiring (a) the filing of reports, such as Convertey Transaction Reports and Suspicious Activity Reports, (b) the maintenance of records and (c) the contribution disignee in identifying customers.

4.28 Accounting Records

The Company and GT Bank each maintains accounting records which fairly and validly reflect, in all material respects, its transactions and accounting controls sufficient to provide reasonable assurances that such transactions are (i) executed in accordance with its management's general or specific authorization, and (ii) menular as personable operation of financial statements in conformity with GAAP. Such records, to the extent they resiste material information pertaining to Company and/or GT Baak which is not easily and readily available class here, have been deplicated, and such duplicates are slored safely and securely.

4 29 Corporate Records

The mante books of Company and GT Bank each accurately reflect all material actions taken by theirs shareholders, board of directors and contributees of their board of directors.

4.20 Accounting and Tax Matters

The Company and GT Bark each has not through the date hereof taken or agreed to take any action that would preven either the Company or GT Bank from qualifying the Morger as a reorganization without the meaning of Section 368 of the URC.

4.31 Community Reinvestment Act

GT Bank received a rating of "softsfactory" or better in its most recent examination or interim review with respect to the Community Reinvestment Act. GT Bank has not been advised in writing of any concerns regarding compliance with the Community Reinvestment Act by any Governmental Entity or by any other Person.

4.37 Jule Herball Property

To the extent that Company and GT Bank owns or possesses howeves and other rights to use made secrets, trade names, trademarks, service marks, Internet domain name registrations, inventions and processes used in its businesses, Company and GT Bank have not received any notice of conflict with respect thereto that asserts the rights of others. To the Company's and GT Bank's Knowledge the Company and GT Bank have in all material respects performed all of the obligations required to be performed by it and is not in default in any insterial respect under any contract, oprement, attrangement or commutment relating to any of the foregoing.

4.33 Material Interest of Cristani Persons

None of the concent officers or directors of the Company and GT Back has any observal interest in any material contract or property (real or personal), tangible or intangible, used in or pertaining to the pusiness of the Company and/or GT Bank (except his/her ownership of the Company's Shares).

ARTICLE 5. ADDITIONAL AGREEMENTS

5.1 Accessive Information

3.1.1. Upon resonable notice, each party shall pertoit the other party and its accountants, coursel and other representatives reasonable access to its officers, employees, properties, books, compacts, commitments and records from the date hereof through the Riffeetive Time, and thall furnish or provide such access as soon as practicable, (i) a copy of each of the Bank Filangs or Cumpany Friengs, as appropriate, filed cobsequent to the date of this Acreement promptly after such document has been filed with the appropriate Covernmental briefly, (ii) unless otherwise prohibited by law, a copy of tach report, schedule and other document filed microcover by either party during such period with or from any Segulatory Authority in the linemal Revenue Service, as to documents other than related to employees or costonners and other than these distributed to beaks generally; (iii) as promptly as practicable following the end of each calle and the tile date hereof, a halance shoel of the respective party's business.

properties, assets, financial condition, results of operations, liabilities, personnel and otherwise as the other party may reasonably request.

5.1.2 Until the Effective Time, the Bank's and the Company's Chief Executive Officer for Charman, if a party has no Chief Executive Offices) shall be extitled to receive a copy of board packages prepared for meetings of the Board of Directors and the Learn Committee of the other party, and a copy of the monotes of, and revolutions adopted at, such meetings. Each party shall provide the other party's Chief Executive Officer (or Chairman, if a party bas no Chief Executive Officer) with notice of the dates, times and places of such meetings at the same time notice is given to the mombers of the Board of Directors of the other party. Notwithstanding the foregoing, neither party shall be obligated to disclose any information to the other party that is entitled to be protected from disclosure under an attorney-clicot providege which would be lost due to such disclosure, or to which such party would not have access pursuant to Section 5.1, above.

3.2 Shareholder Approval

Company and Bank shall promptly call their respective meetings of chareholders to be hold at the carliest practicable date after the date on which the joint proxy statement and oftening circular is approved by the COFI, but in no event later than 30 days following such approval by the CDFI. for the purpose of approving the principal terms of this Agreement and authorizing the Merger Agreements, the Downstream Merger and the Morger. Subject to its respective fiduciary duries, each of Company's. GT Bank's and Bank's Bread of Directors will recommend to its respective shareholders approval of this Agreement, the Morger Agreement and the Merger.

§3 Taking of Necessary Action

5.3.1 Subject to the terms and conditions of this Agreement, each of the parties berein agrees, subject to applicable laws and the fiduciary duties of Bank's, GT Bank's or Company's Board of Directors, as advised in writing by its respective counsel, to use all reasonable offens promptly to take or cause to be taken all action and promptly to do or cause to be cone all things accessary, proper or advisable under applicable laws and regulations to consummate and inske effective the transactions contemplated by this Agreement and the Merger Agreement, including, without limitation, the derivery of any continents or other gocument reasonably readdsted by counsel to a party to this Agreement. Without limiting the foregoing. Company, GT Back and Back will use facir measurable efforts to obtain all consents. of third parties a 4 Government But ties necessary or, in the reasonable opinion of Company, GT Bank or Bank advisable for the concumpation of the transactions contemplated by this Agreement. In case at any time after the Effective Time any further action is necessary or desizable to curry out the purposes of this Agreement, the Merger Agreement, or to vost the Spryiving Corporation with full title to all properties, assets, rights, approvals, immunities and teancluses of 6T Bank, the proper afficers or directors of Company, GT Bank or Back, as the case may be shall take all such necessary action.

5.4 Joint Proxy Statement and Offening Circular and Applications

5.4.1 Company and Bank will cooperate and jointly prepare and file as promptly as procheable the joint proxy statement and offering circular, the statements, applications, correspondence or forms to be filed with appropriate state securities law regulatory authorities,

and the violements, correspondence or applications to be filed to obtain the Requisite Regulatory Approvals to consummate the transactions contemplated by this Agreement. Each of Company and Bank shall use all reasonable efforts to obtain the grant of a permit from the CDFI to use the joint proxy statement and offering circular as promptly as practicable after such lifting, and Company and Back shall thereafter that the joint proxy statement and offering circular to their respective shareholders. Each party will funcish all financial or other information, emitification, consects and opinions of counsel concurring it and its Schwidcarry as propint by reasonably increasing the Requisite Regulatory Approvals and to prepare other documents necessary to the consummation of the Merger.

5.4.2 Each party shall provide to the other at the request of the other party, (i) immediately prior to the filling thereof, copies of all material statements, applications, correspondence or forms to be filled with state securities law regulatory authorities, and other appropriate regulatory authorities to obtain the Requisite Regulatory Approvals, (ii) printighly after delivery to a mempi form, such regulatory authorities all written communications, letters, regimes in other documents relating to the transactions contemplated by this Agreement, and (iii) written summaries of all oral communications with any state securities law regulatory authorities or other appropriate bank regulatory authorities regarding the Requisite Regulatory Approvals.

2.4.3 None of the information supplied or to be supplied by any party and approved by such party which is included in the joint proxy statement and offering circular ("joint proxy statement and affering circular") to he filed with the CD71 by GT Dash and Bask in connection with the Downstream Merger and Merger in any with More Securities law regulatory authorities, and other appropriate regulatory authorities to obtain the Requisite Regularity Approvals (c) with respect to the party providing soch information, will at the time it. becomes effective, contain any uprue statement of a material fact or emit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not pusleading; and (ii) with respect to both parties as to the foint provy statement and othering encular and any amendment or supplement thereto. will, at all times from the first date of mailing to shareholders through the date of the last meeting of shareholders to be hold in connection with the Morger, contain any untrue statement. of a material factor or orbit to state a material fact required to be stated therein or processing to make the statements Country in Tight of the circumstances under which they were mode, put mixleading, and (iii) in any applications and forms to be filted with securities or "blue sky" authorities, self regulatory authorities, or any Governmental Entity in connection with the averget, the issuance of any shares of Bank Common Stock in connection with the Merger, or any Requisite Regulatory Approvals will, at the time filed or at the time liney become effective, contain any antrue slatement of a material fact or omit to state any material last required to be stated therein or necessary to make 212 statements therein. In light of the excumstances under which they were stade not misleading. The joint proxy statement and offering childlar will comply in all motorial respects with the applicable federal and state securities laws, rules and regulations.

5.5 Expenses

5.5.1 Whether or not the Morger is consummated, all casts and expresses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party including the same except as otherwise growided in Article S.

5.5.2 The Company and GT Bank shall each use its best efforts to ensure that its attorneys, accountants, financial advisors, investment bankers and other consultants engaged by them to connection with the transaction contemplated by this Agreement submit full and (maj bills on or before the Closing Date and that such expenses are property reflected on the books of Company and GT Bank.

5.6 <u>Notefication of Cortain Ferrats</u>

5.6.4 Book shell provide to Company, as soon as practicable, written notice (sent via facsingle and overnight mail or courser) of the occurrence or failure to occur of any of the events, circumstances or conditions that are the subject of Sections 6.1 and 6.3, which notice shall provide reasonable detail as to the subject matter thereof.

5.5.2 Company shall provide to Bark, as soon as practicable, written notice (sent via latstrate and overnight mail or courier) of the occurrence or failure to occur of any of the events, circumstances or conditions that are the subject of Section 6.3 and 6.4, which nonce shall provide reasonable detail as to the subject matter thereof.

5.6.3 Each party shall promptly advise the others in writing of any change or event which could reasonably be expected to have a Material Adverse Effect on such party or on its ability to consummate the transactions contemplated by this Agreement or the Mergee Agreement.

5.6.4 Bank and Company shall immediately notify the other in writing in the event that such party becomes aware that the joint proxy statement and offering circular at any time contains any untrue statement of a material fact or omits to state a material fact original in be stated therein or necessary in order to make the statement therein, in tight of the circulastances under which they were made, out misleading or that the joint proxy statement and affering circular otherwise is required to be amended and supplemented, which notice shall specify, in reasonable detail, the circumstances thereof. The Bank and/or the Company shall promptly amend and supplement such materials and disseminate the new or modified information so as to fully comply with the Scenetics Act and California Emercial Code. The call of occket costs and expresses of preparing, filling and dissemulating any amendment or supplement to the joint proxy statement and offering circular shall be home by the party of parties (or an equitable hearis) whole comparing the view domination in supplement back and offering circular shall be home by the party of parties (or an equitable hearis) whole comparing in the view domination requires amendment.

3.7 Updated Schedules;

Bank has delivered to GT Bank and Company on or before the date of this Agreement all of the Schedules to this Agreement which Bank is required to deliver to GT Bank and Company hereunder (the "Bank Schedules"). Company and GT Bank have delivered to Bank on an before the date of this Agreement all of the Schedules to this Agreement which Company and GT Bank are required to deliver to Bank becomder (the "Company Schedules"). Emmediately prior to the Choosing Date, Bank shall have prepared updates of the Bank Subralules provided for in this Agreement and shall deliver to GT Bank and Company reveard schedules containing the updated information (or a confinition signed by Bank's Chief Executive Officer stating that there have here no changes on the applicable schedules); and GT Bank and Company shall have prepared updates of the Company Schedules provided for in this Agreement and shall deliver to Bank revised Schedules containing updated information (or a certificate signed by Company's Chief Executive Officer stating that there has been no change on the applicable schedules).

5.8 Addrivenet Accounts/Appraisals.

lownediately prior to the Closing Date, at Bank's request, the Company and/or GT Bank shall, consistent with GAAP and applicable banking regulations, modify or change its loss, accord, reverse, tax, lingation and teal estate valuation policies and practices (including logclassifications and fevels of reserves) as may be necessary to conform GT Bank's practices and recthods to those of the Bank, provided, however, that no account or reserve made by the Company or GT Bank pursuant to this Section 5.8, or any hitigation or regulatory processing arising out of any such account to this Section 5.8, or any hitigation or regulatory processing arising out of any such account to this Section 5.8, or any hitigation of 8, shell constitute ar be deemed to be a breach, violation of an failure to valisfy any representation, warranty, covenant, coordinant to their provision of this Agreement or otherwise be considered in determining whether any such breach, violation or failure to satisfy shall have occurred.

5.9 <u>Freployee Plans</u>

Immediately prior to the Closing Date, of Bank's request, the Company and/or GT Bank shall terminate any temployee Plan or Benefit Amangement, provided, however, that no second or reserve nade by the Company of GT Bank as a result of a termination requested by the Bank ourseant to the Section 5.9, or any fitigation or regulatory protocolong anying on of any such accord or reserve, or any other effect on the Company or GT Bank resulting from the Company's or GT Bank's compliance with this Section 5.9, shall construct to be deemed to be a breach, violation of or failure to satisfy any representation, warrancy, company, condition or other provision of this Agreement to informative be newsidered in determining oberbar any such breach, violation or follow to variably shall have occurred.

ARTICLE 6. CONDUCT OF BUSINESS.

6.1 Affirmative Conduct of the Company and GT Sank.

During the period from the date of eventiation of this Agreement through the Effective Time, the Company and GT Bank shall every on their business in the ordinary course in substantiality the continent is which heretofore conducted, subject to changes in applicable law and directives from regulators, and use all commercially reasonable efforts to preserve interface business organizations and preserve G1 Bank's relationships with customers, depositions, suppliers and others having business dealings with it; and, to these cods, shall fulfill each of the following

6.1.1 Use their commercially reasonable efforts, or cooperate with others, to expeditionary bring about the satisfaction of the conditions specified in Article - hereot; 6.5.2 Advise frank promptly in writing of any charge that would have a Material Adverse Effect on the Company of GT Bank, or of any matter which would make the representations and warrantier set forth in Acticle 3 hercef not true and correct on any material respect as of the mailing date of the joint proxy statement and offering circular and at the Effective Time;

6.1.3 Keep in bill force and affect all of their existing material primits and licenses:

6.1.4 Insurance

(a) Use commercially seasonable efforts to maintain insurance or bonding coverage on all material properties for which they are responsible and on their business operations, and carry colless than the same coverage for fidelity, public liability, personal upuy, property damage and other tasks equal to that which is in effect as of the date of this Agreement; and nobity the Bonk in writing promptly of any facts or commissioners which rould affect their ability to maintain such insurance or handling coverage, and

(b) Bank shall permit Company and GT Bank to use continercally inaxonable efforts in extend the discovery period of the directors' and officers' liability insurance correctly inforce for a period of up to 48 months with tespect to all matters arising true facts or events which occurred before the Effective function for which the Company or GT Bank would have had an obligation to informatly their directors and officers (a "tail" policy"); provided, however, that the total cost of the premium for such "tail" policy shall not extend \$80,000. If Company and GT Bank are usable to maintain or obtain a "tail" policy for \$80,000, the Company shall use commercially reasonable efforts to obtain as much comparable insurance as is available for \$80,000 with respect to acts or omissions occurring prior to the Effective Time of the Merger by such directors and officers in their capacities as such

6.1.5. Perform their regional contractual obligations and not breach or eauge a default on any of such obligations, and not amend, modify, or, except as they need be terminated is accordance with their terms, terminate any material contract, agreement, understanding, commitment, or offer, whether written or oral (collectively referred to as an "Understanding") or materially default to the performance of any of their obligations under any Understanding where such default would have a Material Adverse Effect on Company and GT Bank, as a whole.

6.1.6 Doly observe and conform to all legal requirements applicable to their bosiness, except for any legal requirement where the failure to so observe and conform to such requirement would not, individually or in the aggregate, and, in the future will not, have a Material Adverse Effect on the Company or G1 Banx,

6.1.7 Duly and finitely file as and when due all reports and Returns required to be filled with any Governmental Soury. 6.1.8 Maintain their tangible assets and properties in good condition and separt, normal wear and test excepted in accordance with prior precises;

6.1.9 Principly advect the Bank to writing of any event of any other banks that within the Knowledge of Company or GT Bank, whereby any Person or related group of Persons acquires, or proposes to acquire, after the date of this Agreement, directly or indirectly, record or bracficial ownership (as defined in Rule 13d-3 premulgated by the SEC pursuant to the Exchange Act) or control of 5% or more of the outstanding shares of the Company's Compton Stock either prior to or Eller the record date fixed for the Company's shareholders' meeting or any adjourned meeting thereof to approve the transactions contemplated norein.

6 ! 10 (a) Maintain a reserve for loan and lease losses ("Loan Loss Reserve"), at a level that is consistent with GT Bank's past practices and methodology as a effect on the date of the execution of this Agreement and in accordance with GAAP and applicable regulatory accounting principles and banking laws and regulations and there shat, be no change in the calculation and reporting of the Loan Loss Reserve;

(b) Charge off all wars, receiveb as and other assets, or porturus thereof, deemed by GT Bank, its outside in an portfolio reviewer, the FDIC, the CDFI, or its independent address, to be uncellectuble in accordance with GAAP, regulatory accounting generiples, and applicable law or regulation, or which have been classified as "lass" or as directed by any regulatory authority, on ess such class feature or direction has been desregarded in good faith by GT Bank, GT Bank has submined in ording to such regulatory authority the basis upon which it has so disregarded such classification or direction, and such regulatory authority such such classification or direction, and such regulatory authority retracts its direct on requiring such charge off;

5.1.1] Firmish to Bank, as soon as practicable, and in any event within fireen (15) days after it is prepared: (i) a copy of any separt submitted to the Board of Directors of the Company or GT Bank and access to the work an oppers related thereto, provided, however, that the Company and/or GT Bank and not furnish Bank any matchais relating to deliberations of Company's or GT Back's Board of Directors with respect to its approval of this Agreement. communications of Company's or GT Bank's legal counsel with the Buard of Directors or officers of Company or GT Bank regarding the Company's or GT Eark's rights against or obligations to Bank under this Agroement, or books, records and documents covered by the attorney-sheat privilege or which are attorneys' work product; (i.) copies of all utsterial reports, renewals. If mes, conficientes, statements, correspondence and other documents specific to the Company or GT Bank or filed with or received from the FDIC, CDFI or any Covenimental Fromy to the extent not prohibited ity applicable law, right ation or order; (iti) manifely phaudited convolutated balance sheets, statements of income and observes in shareholders' eventy for the Company and GT Bank and quarterly insidened consolidated balance sheets, statements of uncome and charges in shareholders' equity for the Company and OT Bank, in each case prepared on a basis consistent with past practice; and (iv) such other reports as Basic stay. reasonably request (which are otherwise detiverable under this Section 6.1.1.1) relating to the Cormany and GI Back:

61.1.2 The Company and GT Bank agree that through the Effective Time, as of their respective detes, (i) each Company Filing will be true and complete in all material respects, and (ii) each Company Filing will comply in all material respects with all of the stature, roles and explaining enforced or proposigned by the Covenance of a rotterial fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they will be made, not nacleading. Any financial statement contained in any of such Company Filings that is pitended to present the incareal position of the Company and GT Bank during the periods involved to which it colates will factly present in all material respects the financial position of the Company and GT Bank and will be proposed in accordance with GAAP on convertant with applicable regulatory accounting principles and banking law and banking regulations, except as stated therein;

6.1.13 Maintain reserves for contingent liabilities in accordance with GAAP or applicable regulatory accounting principles and consistent with past directices;

6.1.14 Intern the Bark of the anounts and categories of any loans, leases or other extensions of credit, or other assets of the Company or GT Dank, this have been classified by any bark regulatory authority as "specially mentioned," "tenegoticted," "substandard," "dominful," "loss" or any comparable classification ("Classified Assets"). GF Bark will femish to Hark, as some as practicable, and in any event within filtern (15) days after the end of each classification category, (a) sometened eredits by type, (iii) screepotiated leans by type (leans on which interest has been renegotiated to lower than market rates because of the financial condition of the borrowere); (iv) delinquent credits by type, including an aging into 30.89 and 904 day categories; (v) loans or leases or other assets charged off, in whole or in part, dating the previous month by type; and lyn) OREO or essets owned stating with respect to each its type;

6.1.15 Fourish to Back, upon Back's request, schedules with respect to the following. (d) participating heaves and heaves, stating, with respect to each, whether it is perchased or sold and the loan or lease type: (ii) loans or leases (including any commitments) by the Company of GT Bank to any director or officer (at or above the Vice President level) of the Company of GT Bank or to any Pérsen holding 5% or more of the capital stock of the Company, including, with respect to anoth such such lease or lease, the identity and, to the Knowledge of the Company or G1 Bank, the relation of the horrower to the Company of G1 Bank, the relation of the horrower to the Company of G1 Bank, the relation of the horrower to the Company of G1 Bank, the relation of the horrower to the Company of G1 Bank, the relation of the horrower to the Company of G1 Bank, the loan or lease type and the outstanding and andrawn amounts: and (at) standby letters of credules by type.

6.1.16 Make available to for Bank copies of rach could authorization package, convexing of all applications for and financial information regarding loans, renewals of bans or other extensions of credic secured by real property, or liquid collimetal consisting of cash. TCD is stock, bonds or the cash value of bic insurance or extensions of credit secured by the foregoing, ansecured loans or unsecured extensions of credit, and renewals of any Cassified or criticized loans which are considered by GT Bank after the date of this Agreement, concurrently with submission to GT Bank's loap committee; 5.1.17 Prior to the Effective Time terminate all Company Stock Options and commitments to great stock options by motual agreement with each and all of its holders of outstanding Company Stock Options and others that have a boria fide contractual right for the grant of a Company Stock Option. As part of such agreement, there shall be a waive and release of the Company and its successors for any liability associated with such stock option matter;

6.1.18 [Reserved];

5.1.19 Comply with the requirements of all applicable rules and regulations, the noncompliance with which would materially and adversely affect the assets, liabilities, husiness, financial condition or results of operations or prosperts of the Company or GT Bask, and

6-1.25 File all necessary applications with the FDIC, Federal Reserve and CDF, for the masserious contemplated by the Downstream Merger Agreement as soon as possible, but no later than June 15, 2011 and furnish to the Back, as soon as practicable, and in environt willow forces down atter it has prepared all applications to be submitted to the FDIC, Federa Reserve and CDFI for approval of the Downstream Merger.

6.2 Negetive Covenants of the Company and <u>GT Bank</u>.

Between the date hereof and the Effective Time, except as contemplated by this Agreement, and subject to requirements of law and eignistion generally applicable to banks, rethricture the Company nor OT Bank shall, without the prior written consent of the Back (which consent shall not be unreasonably withheld and which consent shall be defined grapted if a thin three (3) Business Days of Back's receipt of written notice of a request for prior written consent, together with such information in writing and appropriate documents as will enable flank to reasonably evaluate whether to consent to such request, written notice of objection is not received by the Company and Gi Back):

6.2 % (a) Dec are or pay any devideed or make any other distribution in respect of any of its capital stock, (b) split, torobles or inclusivity any of its capital stock or issue or authorize the issuence of any other second ties to respect of the two of or in substitution for shares of at separal stock; or (c) reputchase or otherwise acquire any shares of as capital stock;

6.2.2 Take any action that would or nught result in any of the representations and warronties of Company and GT Bank set forth in this Agreement becoming cutrue in any material respect or any of the conditions to the Merger set forth in Article 7 not bring satisfied, except to the extent such actions are required to be undertaken by applicable law, regulation or at the direction of any Regulatory Automity.

6.2.3 Issue, deliver, sell, or grant, or authorize the issuence, delivery, sale or grant of, or purchase, any shares of the capital stock of the Company or GT Bank or any securities convertible or exercisable into or exchangeable for such cepital stock, or any rights, warrants or options, including options under any stock option plups or enter into any agreements to do any of the foregoing:

6.3.4 Amend its Ameles of Incorporation or Bylaws, except as required by applicable law or by the terms of this Agreement.

6.3.5 Analogize or knowingly permit any of its representatives, directly or instructive, to solicit or encourage any Acquisition Proposal (as hereinafter defined) or participate in any discussions or neperiations with, or provide any nonpublic information to, any Person or some of persons (other than the Back, and their representatives) concerning any such solicities or encouraged Acquisition Proposal. The Company shall potify Bank within 24 hours if any inquiry regarding an Acquisition Proposal is received by the Company or GT Bask, isoloding the terms thereof. For purposes of this Section 6.2.5, "Asymptotical Proposal" shall mean any (a) proposal pursuant to which any Person other than the Bank would acquire or participate in a merger or other business combination or reorganization involving the Company or GT Bank; (h) mapped by which any Person or group, other than the Bank, would acquire the right to vote ies percent (10%) or more of the capital stuck of the Company or Gill Bank entitled to vote for the election of directory; (et acquisition of the acoust of the Company or CT liank other than on the ordinary course of busicess, or (d) acquisition in excess of ten percent (10%) of the outstanding capital stock of the Company or UT Bank, other than as concemplated by this Agreement. Notwithstanding the foregoing, withing contained in this Agreement shall prevent the Company's or GT Bank's Beard of Directory from (i) furnishing neep ublic information to, or entering into discussions or negotiations with, any person in connection with an unsolicited bata 5dd written Anape vitime Proposal by such person, or recommending an unsolucited bona fide written Ammisition Proposal to the shareholders of the Company, if and only to the extent that (A) the Board of Directors of the Company or GT Basik, an appropriate has determined and believes in good faith (after consultation, with and the concurrence of its (maneral advisor) that such Acquisition Proposal would, it consummated, result in a transaction materially much favorable, from a financial point of view, to the Company's shareholders or the Company as appropriate toars the transaction contemplated by this Agreement fairly such more favorable. Acquisition Proposal being referred to 10 that Agreement as a "Company or GU Bank Superior Proposal 1 and the Company's or GT Bard's Board of Directors as againmatic has determined. in good taph, after consultation with and based on women advice from its partside legal courset, that such action is necessary for the Company of GT Bank, as appropriate to comply with its fiduciary duties in shoreholders under applicable law, and (B) prior to furnishing sock temptolic information to, or cutering into discussions or negotiations with, such person, the Company's or OT Bank's Board of Directory, as appropriate has received from such person on executed. confidentiality accornent, or (i) complying with Rule 14e-2 promulgated under the Exchange Act with report to an Acquisition Proposal, if such Rule is applicable thereto.

6.2.6 Acquite or spect to acquite by morging, consolidating with, or by purchasing all or a substantial portion of the assets of, or in any other manner, any business or any Porton or otherwise acquire or agree to acquire any astets which are material to the Company or GT Bank, other than in the ordinary course of business consistent with prior produce;

6.2.7 Sell, lease or otherwise dispose of any of its assets (other than lease which shall be governed by Section 6.7.9, below) which are material, individually or in the aggregate,

to Company or GT Baak, except in the ordinary course of business consistent with prior practice, and after police to and consultation with Bank;

6.2.8 Inclusion way indefinitions for borrowed money or guarantee any such indebtations to use or will any debt securities of Company or GT Bank or guarantee any debt securities of others of others of minimum the originary course of business consistent with prior practice;

6.2.9 Erner into any Understanding except: (a) for federally-bosined deposits and for thort-term dobt securities (obligations maluring within one year) issued, in the ordinary course of husiness consistent with prior practice, and liabilities ansing out of, momed in connection with, or related to the consummation of this Agreement, (b) for commitments to make loans or to otherwise extend coefficient in the ordinary course of business consistent with prior practice and the loan policy of GF Bank, provided that GT Bank shall provide the Bank with each information as the Dank may reasonably request regarding any proposed loans or other extension of credit, prior to the Bank conserving to the making of such loan; and (c) with respect to the sale of loans, any such sale shall be subject to the following conditions: (n) any proposed with or without recourse, shall be presented to the Bank for review pilor to the consummation of any such sale and (n) the Bank shall have the right of first reford to the Sale products any loan wheth GT Bank proposes to sell to a third party, provided however shot the Sale purchase any loan wheth GT Bank proposes to sell to a third party, provided however shot the Sale purchases such loan of the same material terms on which with from its proposed to be sold to such third party,

6.2.20 Make, or other into a commitment to make, any load with a principal balance or maximum principal balance in excess of \$200,000 unsecured and \$1,250,000 scored, or, except in the ordinary course of bismess pursuant to G1 Back's Lean Policy, make or enter into a commitment to pake any other least or other extension of crudit or ready any criticized in classified lean;

6.2.14 Except on the ordinary course of business consistent with price plantice or as required by an existing contract, and provided prior disclosure thereof has been made in Schedule 6.2.11, grant any general or on-form increase in the rates of pay of employees or temployee benefits or any bornes in salary or employee benefits of any officer, employee or agent or pay any bornes to any Person, recain any person (including a former employee or manager) as a consultant to provide services at any time prior to the Effective Time or make any promose or commitment to make any payment to any such person, provided that bondses may be paid and arrangements enforce into with respect to current Company or GT Back complayers to encourage their cooperation and assistance in completing the transactions concemptated by this Agreement that are agreed to by the Bank:

6.3.12 Self mansfer, morigage, encumber or otherwise dispose of any assets or liabilities except in the ordinary course of bosiness consistent with prior practice or as required by any existing contract;

6.2.33 Make the credit underwriting policies, standards or positions relating to sne making of loans and other extensions of credit or communications to make loans and other extensions of credit, or the loans Loss Reserve policies, less stringent than those in effect on

December 51, 2010 or reduce the amount of the Loan Exist Residues or any other reserves for potential losses or consingencies

6.2.14 Make any capital expenditures, or commitments with respect thereto, except those in the otdinary course of husiness which do not exceed \$10,000 individually or \$25,000 in the aggregate;

6.2.15 Renew, extend or an and any existing employment contract or agreement, cater tate any new colp opported contract or agreement or make any bonus or any opposition extraordinary payments to any Person;

6.2.16 Acquire any investment security except for toderal funds or obligations of the United States Treasury or agency obligations with materials of one year or least

6.2.17 Except as otherwise required to correct a prime filling, compromise or otherwise softle or adjust any assertion or claim of a definition y in Taxes (or interest thereon or penalties in connection therewish) on file any appeal from an asserted deficiency everyt in a form previously approved by the Hank in writing, which approval will not be unreasonably withhold, or file is around any federal, foreign, state or local Tax Rutum or report or moke any tax election or change any nucleod or period of accounting unless tequired by GAAP or applicable law and, then, only after submitting such Tax return or report or proposed Tax election or change in any nictbod or period of accounting, to Bank for its approval, which it shall not unreasonably withhold or delay;

6.2.18 Except as contemplated in this Agreement, terminate any Employee Plan or Benefit Atrangement,

6.2.19 Coange its fiscal year or methods of accounting motifact at December-M, 2010, except as required by changes in GAAP or regulatory accounting principles as construct to by Bank's independent public accountings or by Section 5.8 of this Agreement;

6.2.20 Take or cause to be seven any actival which would dispealify the Merger as a "reorganization" within the strating of Section 368(a) of the IRC as a tax free storeganization,

6.2.21 Take or cause to be taken into OREO any commercial property without an environmental report reporting no adverse environmental condition on such property, with a copy of such report delivered to the Back any prior to taking such property into OREO;

6.2.22. Make any only relections with respect to Taxes of any changes to current cleasings with respect to Taxes affecting the assets owned by the Company or GT Bank; or

6.2.23 Materially change its pricing practices on loans or deposit products.

6.3 Albimative Conduct of Bank.

During the period from the date of execution of this Agreement through the Effective. Time, the Bank shall carry on its husiness in the architery course in substantially the manner in which heretoform conducted, subject to changes in law applicable to all California state-changeed banks and directives from regulators, and use all commercially reasonable efforts to preserve intact its business organization and preserve its relationships with enclosers, and, to these ends, shall fulfill each of the following

6.3.1 Use its conductially reasonable offorts, or cooperate with others, to expeditiously bring about the satisfaction of the conditions specified in Article 7 hereof,

6.3.2 Advise the Company promptly in writing of any change that would have a Material Adverse Effect on the Bank taken as a whole, or of any matter which would make the representations and warranties set forth as Article 4 hereof not true and correct in any material respect as of the effective date of the Registration Statement and of the Effective Taxie;

6.3.3 Keep in full force and effect all of its existing material permits and hourses:

6.3.4 Dury observe and conform to all legal requirements applicable to its lensings, except for any legal requirement where the failure to so observe and conform to such requirement would not, individually of in the aggregate, and, in the future will not, have a Material Adverse Effect on the Bank;

6.3.5 Doly and timely file as and when due all reports and Reforms required to as filed with any Governmental Entity:

6.3.6 The Back agrees that through the Effective Time, as of him expertive dates, (i) each Back Filling will be true and complete in all material respects; and (ii) each Back Filling will ecceptly in all material respects with all of the statutes, rules and regulations enforced in principle date by the Gevernmental Ensity with which it will be filed and none will contain any unified systemetry of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances oncer which they will be made, not misleading. Any financial statement contained in any of such Back Filings that is intended to present the trume all position of the Back during the periods involved to which it relates will fairly present in all material respects the financial position of the Back and will be proposed in accordance with GAAP or conversional with applicable regulating according principles and backing law and backing regulations, mempions stated therein.

6.3.7 File at necessary applications with the PDIC, Federal Reserve and CDFI for the transactions contemp ated by this Agreement as soon as possible, but no later than June 15, 2051 and furnish to the Company, as soon as practicable, and in any event within linken days after it has prepared all applications to be submitted to the FDIC. Federal Reserve and CDFI for approval of the Merger;

6.3.8 Use commercially reasonable efforts to maintain inscrime in bonding coverage on all material properties for which it is responsible and on its business operations, and carry not kess than the same coverage for fidelity, public liability, personal injury, property damage and other risks equal to that which is in effect as of the date of this Agreement; and samily the Company and GT Bark in writing promptly of any facts or circumstances which could affect its ability to maintain such insurance or bonding coverage;

6.5.9 Use enromercially reasonable efforts to have the shares of Bank Commun. Stock to be issued in connection with the Merger qualified or registered for office and sale, to the extent required, under the securities laws of each for which the which shareholders of the Company reside,

6.3.10 Reserve and make available for issuance in connection with the Merger and in accordance with the terms and conditions of this Agreement such number of shares of Bank Common Stock to be issued to the shareholders of UT Bank in the Merger pursuant to Article 3 hereof;

6.3.11 Comply with the requirements of all applicable roles over regulations, the noncompliance with which would have a Vaterial Adverse Effect on the Bank.

63.17 Perform its maximal contractual obligations and not breach or default or say of sixh obligations, and con amend, modely, or, except as they may be terminated in an orderate with their terms, terminate any material Understanding, or materially default in the performance of any of its obligations under any Understanding where such default would have a Material Adverse Effect on Company and UT Bank taken as a whole:

6.3.12 For a period of four years after the Effective Time, maintain and pressove the rights to indemnification of officers and directors provided for in the Acticles of Interpretation and Bylaws of GI Dark, or any indemnification agreecess) of GT Bank, as in effect on the date hereof with respect to indemnification for highlities and claims insting out of acts, emissions, events, matters or encounstances occurring or evising prior to the Effective Time, including, without limitation, the Merger and the other transactions contemplated by this Agreement, to the extent such rights to indemnification are not in excess of that permitted by applicable state or fode(a) laws or regulatory authorities, and

6.3.19 Bank shall at sill times have available for insuance a consister of shares of Bank Common Stock sufficient to satisfy the exercise of converted compatified stock options to be issued to certain holders of compatified stock options under the Company's stock incentive plan, and shall obtain, to the victorit genessary, a stock permit or other approval for the issuance of such shares of Back Common Stock. Bank shall gratt and enter into stock option agreements for such grant of nongoalified stock options to certain of the holders of the Company's nonqualified stock options, in the amounts and on the terms set forth in Schedule 6.3.14 to this Agreement. Such grants shall be effective at the Effective Time. Also effective as of the Effective Time, Bank shall anend or enter into, as appropriate, stock option agreements with directors of Bank who currently hold immunalified stock options in order to extend the expiration of such amounthful stock options to the same expression date specified in Schedule 6.3.14 for i conjustified stock options to be guarded in extendance with Section 6.5., 4. For purposes of this Agreement, negotiating or entering into any such manufacturation agreement of such stock options shall not breach any representation or warranty of Bask to this Agreement, not shall it be violate any of Bask's affirmative or negative covenants herein.

6.3.15 Maintain its tangible assets and properties in good condition and repair, normal wear and tear excepted in scendance with prior practices;

6.3.16 Primptly advice the Company in writing of any event or any other transaction within the Knowledge of the Bask, whereby any Person or related group of Persons acquires, or proposes to acquire, after the date of this Agreement, directly or indirectly, record on beneficial ownership (as defined in Roic 13d-3 promulgated by the SEC parsuant to the Exchange Act) or control of 5% or more of the outstanding shares of the Bask's Common Stock other prior to or effect the record date fixed for the Bank's shareholders' meeting or may adjourned nucling the record to approve the transactions contemplated betein:

6.3.17 (a) Maintain a Loss Reserve at a level that is consistent with the Bank's gast gractices and methodology as in effect on the data of the execution of this Agreement and in accordance with GAAP and applicable regulatory according principles and banking laws and regulations and there shall be no change in the calculation and reporting of the Loan Loss Reserve;

(a) Charge off all ioans, receivables and other assets, or portions thereof, deemed by the Bank, its eatside loan portfolio reviewer, the PDIC, the CDPI, or its independent additions, to be encollectible in accordance with GAAP, regulatory accounting principles, and applicable law or regulation, or which have been classified as "loss" or as directed by any regulatory automaty, onless such classification or direction has been disregarded to good fastinely Bank, the Bank has submitted in writing to such regulatory automaty the basis upon which it has su disregarded such classification or direction, and some regulatory automaty retricts its direction requiring each charge-off,

6.3 i.7 Function to the Company, as soon as practicable, and in any event within fifteen (15) days effect it is prepared: (a) a copy of any report submitted to the Board of Directors of the Dank and access to the working papers related therete, provided, however, that the Bank and access to the working papers related therete, provided, however, that the Bank and attends the Company any materials relating to detiberations of the Bank's Board of Directors with respect to its approval of this Agreement, companying the Bank's rights against or obligations to the Board of Directors or officers of the Bank regarding the Bank's rights against or obligations to the Company under this Agreement, or books, records and decomposes covered by the storney-client privilege or which are atternets, correspondence and other documents specific to the Bank or filed with or received from the FDIC, CDF1 or any Governmental Banky to the extent not prohibited by applicable law, regulation or order, (ar) monthly transfited consoliented balance shores, statements of consolients of the Bank and quarterly unaudited consolidated balance shores, statements of consolients of the Bank and quarterly unaudited consolidated balance shores, statements of consolients of specifies and changes in shareholders' equity for the Bank, in each case prepared on a basis consistem with past practice;

and (iv) such other reports as Company may reasonably request (which are otherwise deliverable under this Section 6.3.17) relating to the Back;

6.3.1.8 Maintain reserves for contragent liabilities in accordance with GAAP or applicable regulatory accounting principles and consistent with past practices;

6.4.19 Inform the Company of the amounts and categories of any loans, leases or ether extensions of credit, or other assets of the Back, that have been classified by any back regulatory authority as "specially mentioned," "monegoriated," "substandard," "dombiful," "loss" is any comparable classification ("Classified Assets"). Back well formship to the Company, as soon as practicable, and in any event within fifteen (15) days after the end of each catedar oranth, schedules including the following – (i) Classified Assets by type and its classification category; (ii) nonseculat credits by type; (iii) receiptuated forms by type and its classification category; (ii) nonseculat credits by type; (iii) receiptuated forms by type (loans on which interest has been renegolisted to lower them market rates because of the financial conducts of the borrowers). (iv) delenquent credits by type, including an aging into 30-59 and 901 day categories; (v) loans or leases or other assets charged off, in which or in part, during the previous month by type, and (vi) OREO or assets around stating with respect to each is type,

6.3.20 Furnish to the Company, upon the Company's request, schedules with respect to the following: (i) participating lines and leases, stating, with respect to each, whether it is prochased or sold and the toan or lease type, (ii) loans of leases (including any commitments) by the Bank to any director or officer (at or above the Vice President level) of the Bank or to any Person hold ag 5% or more us the capital stock of the Bank, including, with respect to each such loan or lease, the identity and, to the Knowledge of the Bank, the relation of the borrower to the Bank, the loan or lease type and the outstanding and undrawn amounts, and (jii) standby latters of credit, by type; and

6.3.24 Make available to the Company copies of each credit authorization package, convisting of all applications for and financial information regarding loans, renewals of loans or other extensions of credit secured by a first trast deed on real property, or liquid colluteral consisting of cash. TCD's, stock, honds or the each value of life interance or extensions of credit secured by the foregoing, unsecured loans or unsecured contents of credit, and renewals of matching with authorized loans which are considered by the Baak after the date of this Agreement, concurrently with submission to the Baak's loan committee.

6.4 Negative Conversion of Bank

Between the date hereof and the Effective Time, except as concemptated by this Agreement and subject to requirements of law and regulation generally applicable to banks, Bank shall not, without Company's prior written consent (which consent shall not be pricesonably withheld and which consent shall be deemed granted if within three [3] Business Usys of Company's seee pl of written notice of a request for prior written consent, logether with such information in writing and appropriate documents as will enable the Company and GT Bank to reasonably evaluate whether to consent to such request, written contor of objection is out received by the Bank): 6.4.1 (a) Declare or pay any dividead or make any other distribution in respect of any of its capital stock; (b) aphi, combine or reclassify any of its capital stock or issue or although the issuence of any other scounties in respect of, in tien of or in subsettedion for shores of its capital stock; or (c) repurchase in otherwise any price any shares of its capital stock;

6.4.3. Take any action that would or might result in any of the representations and wascenties of Bank set forth in this Agreement becoming unnue in any material respect or any of the condutions to the Merger set forth in Article 7 not being satisfied, except to the evicuasuch actions are recursed to be undertaken by applicable law, regulation or at the direction of any Repuistory Authority;

6.4.3 Issue, deliver, sell, is grant, or authorize the issuance, delivery, sale or grant of, or purchase, any shores of the capital stock of Bank or any securities convertible or eventicable into or exchangeable for such capital stock, or any rights, warrants or optices, including oppool under any stock option plans or enter two any agreements to do any of the foregoing, except in connection with the esponde of Bank stock options or each equily compensation instruments or rights under the Bank's stock option plans or Hank Common Stock paysons to the exercise of Hank stock options, none of which shall be deemed to cause the representation in Section 4.5 to be untrue in any material respect; and

6.4.4 Take or cause to be taken any action which would disqualify the Merger as a "reorganization" within the meaning of Section 368(a) of the IRC as a tax-free treorganization.

6.4.5 Without the prior written coupent of the Company, authorize or knowingly permit any of its representatives, directly or indirectly, to enter into any agreement for the acquisition of any effect limanetal entity by Bank. Without the prior written consent of the Company, acquire or agree to acquire by merging, consolidating with, or by purchasing all or a substantial portion of the essets of, or in any other manner, any business or any Person of otherwise acquire or agree to acquire any essets which are material to the Bank, other date in the ordinary course of business consistent with prior practice.

6.4.6 Filter into any agreement or attangement requiring the payment of an isonomial excess of \$200,000 (or in the aggregate \$500,000), except: (i) the Bank may enterised any such agreement without the Company's consent if, prior to entering tale such acceleration, the Bank consults with the Company's or [ii) for agreements or arrangements to the ordinary course of business consistent with prior practice and after notion to and consultation with Bank:

6.5 Access to Operations.

Which there (30) Business Days of the Choring Date, Company and GT Bank shall afford to Bank and its authorized agents and representatives, access, during normal business hours, to the operations, books, and other information relating to the Company and GT Bank for the sole purpose of assuring an orderly transition of operations, including the data processing conversion, in the Morger. Bank shall give reasonable notice for access to the Company or GT Bank (or both, as appropriate), and the date and time of sish access will thre be motivally agreed to by the Company or GT Bank (or both, as appropriate) and the Bank. Bank's access shall be conducted in a manner which does not unreasonably interfere with the Company's or GT Bank's normal operations, customers and employee relations and which does not interfere with the ability of the Company and GT Bank to consummate the transactions contemplated by this Agreement.

ARTICLE 7. CONDITIONS PRECEDENT TO CLOSING

7.1 Conditions to the Parties' Obligations.

The obligations of all the parties to this Agreement to effect the Morger shall be subject to the fulfillment of the following conditions.

7.1.1 The principal terms of this Agreement, the Downstream Merger Agreement, and the Merger Agreement shall have been validly approved by the holders of a supporty of the outstanding shares of Bank Common Stock and by the holders of a majority of the outstanding shares of Company Common Stock entitled to vote;

7.1.2 All permits, approvals, nondisapprovals and consents required to be obtained, and all warting periods required to expire, prior to the consummation of the Merger and Downstream Merger under applicable federal laws of the United States or applicable laws of any state basing jorisdict on over the transactions contemplated by this Agreement, the Downstream Merger Agreement and the Merger Agreement shall have been obtained or expired, as the case may be fall such permits, approvals and consents and the lapse of all such waiting periods heing referred to as the "Requisite Regulatory Approvals"), without the imposition of any condition would reduce the benefits of the transaction contemplated hereby to such a degree that the affected party would not have entered into this Agreement had such conditions, testrictions or requirements been known at the date hereof;

7.1.3 [Reserved]

6.1.4 The issuance of Bank Common Stock in order to effect the Merger shall have been qualified or registered with the appropriate state securities law or "blue sky" regulatory authorities of all states in which qualification or registration is required under the State securities laws, and such qualifications or registration shall not have been suspended or revoked;

7.1.5 The agreed upon price per share for shares of GT Bank Common Stock which are "dissenting shares" within the meaning of California Corporations Code Section 1300(e) shall not exceed the greater of (i) \$6,00 nr (ii) an amount which, when combined with other cash amounts payable in connection with the Merger, would result in the Merger being diaqualified fram a tax free reorganization pursuant to Section 368(a)(1)(A) of the IRC, provided that in the vent shall any such combined amount exceed the Aggregate Cash Anzonic,

i.1.6 The agreed upon price per share for shares of Bank Common Stock which are engible to be Back Perfected Dissenting Shares persoant to California Corporations Code. Section 1300(b) shall not exceed the greater of 57.50 or an amount which, when combined with other cash almounts payable in connection with the Merger, would result in the Merger being disqualified from a tax free reorganization parabut to Section 368(a)(1)(A) of the IRU;

7.1.7 Bank shall have obtained all consents of third parties required to germit the Merger to proceed in accordance with material contracts, including without haristion the leased premises of Bank, and excluding Bank's data processing contract, and

7.2 Conditions to Company's and GT Bank's Obligations.

The obligations of Company and GT Bask to effect the Merger shall be subject to the fulfilment (or waver, in writing, by Company) of each of the following conditions:

7.2.1 Except as otherwise provided to this Section 7.2. (a) the representations and warrantee of Bank contained in Article 3 shall be true and correct in all matrixel respects as of the Closing as abough made at the Closing, except to the extent they expressly refer to an carlie: true and except where the failure to be true and correct, individually or in the aggregate, would not have or would not be reasonably lokely to fusive, a Moternal Adverse Effect on the Surviving Corporation or upon the consummation of the transactions contemplated hereby; (b) Bank shall have doly performed and complied in all material respects with all agreements and coverants required by this Agreement to be performed or complied with by it prior to or at the Closine, except where the failure to so perform and comply, individually or in the aggregate, would not have or would not be reasonably likely to have a Material Adverse Effect on the Surviving Corporation of the failure to so perform and complied with by it prior to or at the Closine, except where the failure to so perform and complied builty or in the aggregate, would not have or would not be reasonably likely to have a Material Adverse Effect on the Bank, or upon the consortination of the transactions contracted tracky, (c) none of the events or conditions emitting Company to reminant this Agreement under Article 8 shall have accurred and in continuing, and (d) Bank shall have delivered to Company certificates dated the date of the Closing and signed by the Chief Executive Officer and Corporate Secretary to the effect set forth in Subsections 7.2.1(a), (b) and (c);

7.2.2 Bank shall have delivered its Updated Schedules to Compony two days immediately preceding the Closing Date and none of such Updated Schedules shall reflect any nem that was not on the Bank Schedules for in the Bank Financial Statements) delivered on the Gate of execution of this Agreement that has had, would have, or could be reasonably likely to have, a Material Adverse Effect on the business, conditions, properties or capitalization of Bank taken as a whole, at or after the Closing.

7.2.3 All of Bark's director-shareholders shall have delivered to Commany dated the date of this Agreement the Bark Director-Shareholder Agreements in the form attached hereto as Pahibil 7.2.3, which shall provide, among other things, that the director will (j) vote his or her shares in favor of the transactions contemplated by this Agreement and (ii) recommend that Bank shareholders approve this Agreement and the transactions contemplated hereby, and such Bank Director Shareholder Agreements shall innerities in full effect and not modified, rescinded of backed as of the Closing; and

7.2.4 The fairness opinion (the "Bonk Fairness Opinion") commissioned by Bank's Board of Directors shall provide as of the date of maring the joint graxy Matemant and offering circular to Bank's shareholders that the terms of the Merger, from a financial standpoint,

are fair to the shareholders of the Bank, and shall rai have been revisited, at any time prior to the meeting of Bank's shareholders at which the Merger is to be voted on. The Bank shall immediately notify Company of the Bank Pairness Opinion is revealed.

7.3 <u>Conditions to Bask's Obligations</u>.

The obligations of Bank to effect the Merger shall be subject to the fulfilment (or waiver, in writing, by Bank) of each of the following conditions:

7.3.1 Except as otherwise provided in this Section 7.3, (a) the representations and warranties of Considery and GT Back compared in Article 4 shall be true and correct in all material respects as of the Closing as though made at the Closing, except to the extent they expressly refer to an earlier one or a covenant of Company or GT Sault that specifically permits. a change in the facts represented in a representation or warranty, and except where the failure an be true and correct, individually or in the ageregate, would not have or would not be meanably likely to have, a Material Adverse Effect on the business, conditions, capitalization or preperies. of Company and GT Bark taken as a whole or upin consummation of the transactions contemplated hereby: (5) each of Company and GT Bank shall have duly performed and complied to all material a sparts with all agreements and coverants required by this Agreement to be performed or complied with it prior to or at the Closing, except where the fathere to so perform and comply, individually or in the apprepate, would not have an would not be reasonably likely to have a Material Adverse Effect on the business, conditions, capitalization or properties of Company and GJ Ragly taken as a whole, or upor, the consummation of the transactions contemplated hereby; [e) none of the events or conditions enabling Bank to terminate this Agreement under Article 8 shall have occurred and be continuing; and (d) Company shall have delivered to Bank continuous duted the date of the Closing and signed by the President and Chief Executive Officer and Chief Signatial Officer of Company in the offect set forth in Subarctions 7.3.1(a), (b) and (c);

7.3.2 Company shall have delivered its Updated Schedules to Bank two days immediately preceding the Closing Date and none of such typdated Schedules shall write 1 any item that was not on the Company Schedules (or in the Company Financial Statements) delivered on the date of execution of this Agreement that has had, in would have, or coold be reasonably likely to have, a Material Adverse fifter) on the biologies, conditions, properties or capitalization of Company and its Subsidiaries, token as a whole, at or after the Effective Time, or on the consummation of the transactions contemplated hereby;

7.3.3 All of GT Boak's and Company's Greetor-shareholders shall have delowered to Bank and dated the date of this Agreement the GT Bank and Company Director-Shareholder Agreements in the form attached hences as Exhibit 7.3.3 which shall provide, among office things, that the director will (i) vote his or her shares as favor of the transactions contemplated my flies Agreement, and (ii) recommend that Company and/or G1 Bank shareholders (as appropriate) approve this Agreement and the transactions contemplated hereby; and

7.5.4 The factness opinion (the "Company Faitness Opinio") complianed by the Company's Board of Directors shall provide as of the date of matting the joint proxy.

statement and offering circular to the Company's shareholders that the terms of the Merger, from a financial standpoint, are fair to the shareholders of the Company, and shall not have been revoked, at any time prior to the meeting of the Company's shareholders at which the Merger is to be voted on. The Company shall immediately notify the Bank if the Company Fairness Optiments revoked.

ANTICLE 8. TERMINATION, AMENDMENTS AND WAIVERS.

8.1 Termination of Agreement.

Anything herein to the contrary notwithstending, this Agreement and the transactions contemplated hereby moluting the Merger may be terminated at any time before the Effective Time, whether before or after approval by the shareholders of Company and Bank as follows, and in no other manner

8.1.1 By meters convent of Company and GT Book, on the one hard, and Baok on the other;

8.1.2 By Company or Bank, (i) if any conditions set forth in Section 7.1 shall not have been met by October 31, 2011, or (ii) upon the expiration of 30 But nees Days after any Universimital Entrity denses or refuses to grant any approval, consort or authorization required to be obtained to order to consumptive the transaction contemplated by this Agreement unless, within said 30 Business Day period affer such denial or telesal, all parties hereto agree to resubmit the application to the Governmental Entity that has denied, or refused to grant the 1 approval, consent or authorization requested.

E.1.3 By the Company, if any conditions set forth in Section 7.2 shall not have been mot, or by the Bank, if any conditions set forth in Section 7.3 shall not have been met, in either case by Ontober 34, 20, ϵ , or such earlier time as it becomes apparent that such condition cannot be met;

8.1.4 By Company of there shall have been (i) a breach of any of the representations or warranties of Bank set forth in this Agreement, (ii) a default in the observance of in the doe and timely performance of easy of Bank's covenants and agreements beach contained, which breach or default, in the reasonable opnion of Company, by its nation contot be cared or is not cared prior to the carlier of October 31, 2011 or thruy (20) days from the date the Company gives the Bank notice of such breach or default, and which breach or default would, in the reasonable opinion of Company, individually or in the aggregate, have, or he reasonable to have, a Material Adverse Effect on Bank or upon the consumption of the transactions contemplated hereby:

81.5 By Boak if there shall (nove been (i) a breach of any of the representations or warranues of Company or GT Bank set forth to this Agreement, (ii) a detault in the observance or in the due and timely performance of any of covenants and agreements of Company or GT Bank herein contained, which preach or default, in the reasonable opinion of Bank, by its nature cannot be cared or is not cured prior to the earlier of October 31, 2013 or shirty (30) days from the date the Bark gives the Company notice of such breach or default, and which breach or default would, in the reasonable opinion of Bank, individually or in the aggregate, have, or be reasonably likely to have, a Material Adverse Effect on Company or Gf Bank or upon the consummation of the transactions contemplated hereby; or

8.1.6 By the Company or Bank, if an Acquisition Front involving the Company or Dank shall have accounted

8.2 <u>Effect of Tempination</u>

In the event that this Agreement shall be terminated pursuant to Section 8.1 hereof, all further obligations of the parties hereto upder lios Agreement shall terminate without further hability of any party to another, provided, however, that no termination of this Agreement each becttor 8.1 for any reason or in any manner shall release, or be construed as so intraving, any party from its obligations under Sections 8.5, 10.5 or 10.6, beyond and notwellistanding the foregoing if such termination shall result from the willful failure of a party to failfill a condition to the performance of the obligations of any other party or to perform a covenant of such party in this Agreement, such party shall, subject to the provision of Soction 8.5, be fully liable for any and all damages, costs and expenses (including, but not (initial to, reasonable attorneys) joes sustained or incurred by the other party or porties in connection with negotisting and implementing the transactions contemplated in this Agreement).

8.3 Waiver of Candilipos

If any of the conditions specified in Section 7.2 have not been satisfied. Company and GP Bank may incorribuleus at their election, proceed with the transactions contemplated in the Agreement. If any of the conditions specified in Section 7.3 have not been satisfied, Bank may nevertheless, at its election, proceed with the transactions contemplated in this Agreement. It any party elects to proceed pursuant to the provisions bereaf, the conditions that are unsatisfied immediately prior to the Effective Time shall be deemed to be satisfied, as evidenced by a configure delivered by the electing party.

8.3 Force Majoure

Company, GT Bank and Bank agree that, notwithstanding anything to the contrary in this Agreement, in the event this Agreement is terminated as a result of a failure of a condition, which failure is due to a natural divergence or other act of God, or an act of war or of terror, and provided under party has materially failed to observe the obligations of such party under this Agreement, neither party shall be obligated to the other party to this Agreement for any expenses or otherwise be liable hereunder.

5.5 Monetary Consequences of Termination

8.5.1 Bank brickly agrees that if this Agreement is terminated (i) by Company and/or GT Bank pursuant to Section 8.1.2, because of the fullure of Bank chareholders to approve this Agreement and the transactions contomplated hereby following the withdrawal, or modification in any monner adverse to Company and/or G1 Bank, of the Board of Directors of Bank's recommendation of this Agreement and the transactions contomplated hereby. (ii) its Company and/or G7 Bank pursuant to Section 8.1.3 because of the failure of any of the

conditions set forth in Section 7.2, (ii) by Company and/or GT Bank pursuant to Section 8.1.4, or (iv) by Company and/or GT Bank pursuant to Section 8.1.6 because Bank is involved in an Acquisition Event, Bank shell promptly, and in any event within seven Business Days after such termination, pay Company and/or GT Bank 3400,000 as liquidated damages for the direct expresses matriced by Company and GT Bank related to the transactions contemplated hereby, which the parties agree is a reasonable estimate of such expenses.

8.5.2 Company and GT Bank hereby agree that if this Agreement is terminated (i) by Bank purcuant to Section 8.1.3, because of the future of Company shareholders to approve this Agreement and the transactions contemplated hereby following the withdrawal, or modification in any manner adverse to Bank, of the Board of Directors of Company's recommendation of this Agreement and the transactions contemplated hereby, (ii) by Bank pursuant to Section 8.1.3 because of the failure of any of the conditions set forth in Section 7.3, (iii) by Bank pursuant to Section 8.1.5, or (iii) by Bank pursuant to Section 8.1.6 because Company is involved in an Acquisition Event, Company and GT Bank shall promptly, and in any event within seven (7) Business Days after such termination, pay Bank the combined som of \$400,000 as liquidated domanes for the direct expenses incorred by Bank related to the transactions contemplated hereby which the parties agree is a reasonable estimate of such expenses. The payment obligations of Company and GT Back set forth in the preparities semicore to pay the combined sum shall be defined joint and several.

8.53 Freque as otherwise provided herein and in Section 5.5.1, all Expenses jocated by Company and GT Bank or Bank in connection with or related to the authorization preparation and execution of this Agricement, the collicitation of shareholder approvals and all other matters related to the closing of the transactions contemptated hereby, including, without lumitation of the generatity of the foregoing, all fees and expenses of agenci, representatives counce), and accountants employed by other of the games or its affiliates, shall be horne solely and entirely by the party which has incurred the same.

8.3.4 "Expenses" as used in this Agreement shall include all reasonable metofpocket expenses (including all fees and expenses of attractys, accountants, investment barkers, experts and consultants to the party and us Affiliates) meaned by the party or on its behalf in connection with conducted to the outportaction, preparation and execution of this Agreement, the solicitation of shareholder approvals and all other matters related to the closing of the transactions contemplated hereby.

8.6 Effect of Lemonation; Survival

Except as provided in Section 8.5, no termination under Section 8.1 for any many or in any memor shall release, or be construct as so releasing, any party briefform its obligations parsition to Sections 5.5, 8.5, 30.5 or 10.6 hereof or from any lizbility or damage to any other pury hereto arising out of, in connection with, or otherwise relating to, directly or inducedly, said party's material broach, Default or failure in performance of any of its covenants, agreements, duties or inhightions arising heretunder, or any breaches of any representation or warranty contained herein arising prior to the date of termination of this Agreement.

ARTICLE 9. EMPLOYEE BENEFITS.

9.1 Employee Broofity

To the extra permissible under ERISA and similarly applicable laws and regulations, all employees of OT Bank at the Effective Time who continue as employees of the Bank, shall be entitled to participate in all Bank employee benefits on substantially the same basis as other similarly similarly employees of the Bank. Each of these employees will be credited for eligibidity, puricipation and vesting purposes with such employee's respective years of past service with G1 Bank (or other prior service so credited by Bank) as theorgic flag had been employees of the Bank, except with respect to the Bank's stock option plan, any additionat equity compensation plan. Company may adopt and any 401(k) plan.

ARTICLE 10. GENERAL PROVISIONS

Nonsurveyed of Representations and Warrantiss

None of the representations, warranties, covenants and agreements in this Agreement or in any instrument delivered pursuant to this Agreement shall solvive the Effective Time, except for those covenants and agreements contained herein and therein which by their terms apply in whole in in part who the Effective Time or to a termination of this Agreement.

10.2 Notices

All notices and other communications herounder shall be in writing and shall be deemed given (i) on the date such notice is delivered if delivered personally, (a) foor Susienas Days after such notice is mailed by registered or certified anall (return receipt requested), (ii) one Business Day after such notice is sent if sent by continued overlight contribution from the person to whom such construite is addressed), as the case may be, to the parties at the following addresses (or any such other address for a party as shall be specified by like untitient).

If to Bank at	Bay Commercial Bank 128) Civic Drive, Suite 100 Walnot Creek, CA 91545 Fax No (925) 476-1818 c-meil: GGuarinigScS-calcorn Attention: George & Guarins, President
with a copy to	Kaesal, Young & Logan 300 Oceangate, 14 ¹⁶ Floot Long Beach, CA 20802 Fax No. (262) 436-7416 c-mail: Sandor Mayoga@kyl anto Attention: Sandor X Mayoga

If to Company and/or GT Bank at:

Global Trust Bank 700 E. El Camine Real, Strite U10 Mountam View, CA 94040 Fax No., (650) 810-9452 o-mail: Vicent Thekral@oryglobaltrastback.com Attent on: Vined Thekral, Chairman

with a copy to: Gary Steven Findley & Associates 1470 North Hundley Stree: Anaberm, CA 93836 Fax No.: (714) 630-7910 e-utvill. gs@findley-reports com Attention: Cary Steven Findley, Esq.

10.3 <u>Counterparts</u>

This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties, it being understood that all parties need not sign the same counterpart.

10.4 Entry Agreement/No Third Party Rights/Assignment

This Agreement (including the documents and instruments referred to herein) (a) questiones the entire agreement and supersedes all price agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof; (b) except as expressly set forth herein, a not intended to confer upon any person other than the parties hereto any rights or remedies hereunder; (c) shall not be assigned by a party, by operation of law on otherwise, without the consent of the other parties, and (d) subject to the foregoing, shall be binding upon and shall more to the benefit of the parties hereto and their permitted successors and assigns.

10.5 Nooffisclosure of Agreement: Fress Releases

Company, GT Back and Back agree except as tappined by law, so long as this Agreement is in effect, not to issue any public natice, disclosure or press release with respect to the vanisactions contemplated by this Agreement without seeking the content of the other party, which consent shall not be unreasonably withheld provided, however, that a party may, without the prior consent of the other party four after such consultation, to the extent procheable in the circumstances), save such press release or make such public statements as may upon the advice of outside counsel be required by saw or applicable regulations.

10.6 Confidentiativy

All Confidential information disclosed here:ofore or hereafter by any pany to this Agreement to any other party to this Agreement shall be kept confidential by such other party and shall not be about by such other party otherwise that as herein contemplated, except to the extent that (a) it is necessary or appropriate to disclose to the Federal Reserve. FDIC, CDFI or

any other Governmental Exitity having jurisdiction over any of the parties or as may be otherwise required by Rule (way disclosure of Confidential Information to a Governmental Entity shall be accompanied by a request that such Governmental Entity preserve the confidentiality of such Confidential Information), or (b) such duty as to confidentiality is maived by the other party. Such astligation as to confidentiality and nonuse shall service the termination of this Agreement precision to Article 8. In the event of such termination and on request of another party, each party shall use all reasonable efforts to (1) return to the other parties all documents (and reproductions thereof) received from such other parties that contain Confidential Information (and, in the case of reproductions, all such reproductions made by the receiving party), and (2) destroy the originals and all copies of any analyses, computations, studies or other documents prepared for the internal use of such party that included Confidential Information.

(f) 7 Governing Law: Venue

This Agreement shall be governed and construed in accordance with the laws of the State of California, without regard to any applicable conflicts of law. Any action to enforce this Agreement shall be brought only in a state or federal court located in the County of Course Costa, California.

198 Hordings/Table of Contents

The table of contents and buildings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.

.

10.9 Enforcement of Agreement

The parties hereto agree that interpatchle damage will occur in the event that any of the material provisions of this Astronment of the Merger Agreement is not performed in accordance, with its specific terms or is otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunctions of injunctions to prevent breaches of material provisions of this. Agreement and to enforce specifically the material terms and provisions hereof in any federation state court in the County of Contra Costa, California, this being in addition to any remetly to which they are entitled at law or in equity as ender this Agreement.

10.10 Severainity.

Any term or provision of this Agreement which is invalid or upenforceable in any junsdiction shall, as to that jurisdiction, he ineffective to the extent of such invalidity or upenforceability without rendering invalid or upenforceable the remaining forms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other junsdiction. If any provision of this Agreement is so how as to be strendereable, the provision shall be interpretent to be unly so how as is enforceable.

10.11 Attorneys' Fees

If any legal action or any arbitration upon mutual agreement is brought for the enforcement of this Agreement or because of an alleged dispute, breach or default to connection with this Agreement, the prevating party shall be entitled to recover reasonable attournys' fees and other costs and expenses incurred in that action on processions, in indiction to any other rebef to which it may be entitled.

10.12 Wolver; Artiendment,

Prior to the Effective Time of the Merger, any provision of this Agreement may be (i) waived in whole or in part by the party benefited by the provision or by both parties or (a) amended or modified at any time, by an agreement in writing beliween the parties hereto extended in the same manner as this Agreement, except that after a costy's share relider meeting at which the transactions contemplated in this Agreement were approved, this Agreement may not be annualful if 't would reduce the aggregate value of the consideration to be involved by the Company's shareholders in the Merger without any subsequent approval by such shareholders of be in violation of applicable law.

IN WITNESS WHEREGF, the Bank, Company and GT Bank have caused this Agreement to be signed by their respective officers thereente duly such on zee, all 48 of the date first above written.

BAY COMMERCIAL BANK

Ву:

GLOBAL BANCORP.

avi Baugen Annon

GLODAL TRUST BANK



1012 Waiver; Amendment.

Prior to the Effective Time of the Merger, any provision of this Agreement easy be (i) waived in whole or in part by the party benefited by the provision or by both parties or (ii) amended or modified at any time, by an agreement in writing between the parties hereto executed in the same mannet as this Agreement, except that after a party's shareholder meeting at which the transactions contemplated in this Agreement were approved, this Agreement may not be amended if it would reduce the aggregate value of the consideration to be received by the Company's shareholders in the Merger without any subsequent approval by such shareholders or be in violation of applicable law.

IN WITNESS WHEREOF, the Bank, Company and OT Bank have caused this Agreement to be signed by their respective officers thereanth duly authorized, all as of the care first above written.

BAY COMMERCIAL BANK

Ry _____

Be:

GLOBAL BANCORP

Ву

By:

GLOBAL TRUST BANK

<u>чу:____</u>, ____

By:

AMENDMENT NO. 1

AGREEMENT AND PLAN OF REORGANIZATION AND MERGER, dated as of May 12, 2011

This Amendment No. 1 (the "Amendment") amends that certain Agreement and Plan of Reorganization and Merger, dated as of May 12, 2011, by and among Bay Commercial Bank, Global Bancorp and Global Trust Bank (the "Agreement"). Capitalized terms used, but not defined, in this Amendment, shall have the meanings given such terms in the Agreement.

The parties intend to amend the Agreement by agreeing to the following amendments of the Agreement:

1. <u>Amendment to Section 2.</u>]. The second and third sentences of the second paragraph of Section 2.1 are amended to read in full as follows:

"The Merger shall be effective when the Merger Agreement (together with any other documents required by law to effectuate the Merger) shall have been filed with the Secretary of State of the State of California and a copy of the Merger Agreement, certified by the California Secretary of State, has been filed with the CDFL. When used in this Agreement, the term "Effective Time" shall mean when a copy of the Merger Agreement, certified by the California Secretary of State, has been filed with the CDFI as provided in Section 4887 of the California Financial Code. When used in this Agreement, the term "Downstream Merger Effective Time" shall mean when a copy of the Merger Agreement for the Downstream Merger, has been filed with the California Secretary of State."

 Amendment to Section 7.1.6. Section 7.1.6 of the Agreement is deleted in its entirely and replaced by the insertion of "[RESERVED.]"

The parties amend the Agreement by the execution of this Amendment below.

The signature page follows this page.

Signature page to Amendment No. 1 to

Agreement and Plan of Reorganization and Merger, dated May 12, 2011

BAY COMMERCIAL BANK

By /s/George J. Guatoni President & CEO

.

By /s/ Keary L. Colwel Secretary

GLOBAL TRUST BANK

- By 1si Vinod Thukral Charmon of the Board
- By /s/ Bhupen Amin Secretary

GLOBAL BANCORP.

- By 7s' Vinod Thusral Chairman of the Board
- By /s/ Bhugen Amin Secretary

APPENDIX B

Downstream Merger Agreement between Global Bancorp

and Global Trust Bank

.

.

AGREEMENT OF MERGER

This AGREEMENT OF MERGER, dated as of May 12, 2011 (this "Agreement"), is made and entered into by and between Global Bancorp, a California corporation (the "GB"), and Global Trust Bank, a California banking corporation ("Bank").

A. Bay Commercial Bank, a California state-chartered banking corporation, GB and Bank have entered into an Agreement and Plan of Reorganization and Merger, dated as of May , 2011 (the "Merger Agreement"), providing, among other things, for the merger of GB into Bank, with Bank surviving the merger (the "Merger"). Capitalized terms used but not otherwise defined herein have the meanings ascribed to such terms in the Merger Agreement.

In consideration of the premises and mutual agreements contained in this Agreement and the Merger Agreement, the parties to this Agreement hereby agree that GB shall be merged with and into Bank in accordance with the provisions of the Jaws of the State of California and upon the terms and subject to the conditions set forth as follows:

<u>The Merger</u>.

(a) The Merger shall be pursuant to the provisions of, and with the effect provided in the California General Corporation Law (the "CGCL"), and the Merger shall become effective on the date and time (the "Effective Time") a copy of this Agreement is filed with the Secretary of State of the State of California.

(b) At the Effective Time, GB shall be merged with and into Bank, and the Bank shall be the surviving corporation (the "Surviving Corporation"). All the property, rights, privileges, powers and franchises of GB shall vest in the Surviving Corporation, all debts, liabilities, obligations, restrictions, disabilities and duties of GB shall become the debts, liabilities, obligations, restrictions, disabilities and duties of the Surviving Corporation. Bank shall retain its bank charter in the Merger.

<u>Corporate Governance</u>.

(a) From and after the Effective Time and until thereafter amended in accordance with applicable law. (i) the articles of incorporation of Bank as in effect immediately prior to the Effective Time shall be the articles of incorporation of the Surviving Corporation; and (ii) the bylaws of the Bank as in effect immediately prior to the Effective Time shall be the bylaws of the Surviving Corporation.

(b) At the Effective Time, the directors and corporate officers of the Surviving Corporation shall be those persons who are the directors and corporate officers of Bank immediately prior to the Effective Time, and they shall continue to hold office from and after the Effective Time until they shall have resigned or shall have been legally removed or until respective successors shall have been elected and qualified.

3. Effect of Merger on Outstanding Shares.

As of the Effective Time, by virtue of the Merger and as provided in the Merger Agreement, and without any action on the part of the holder of any shares of common stock of Bank (the "Bank Common Stock") or the common stock of GB (the "GB Common Stock"):

(a) <u>Bank Common Stock</u>. Each share of Bank Common Stock issued and outstanding immediately prior to the Effective Time shall be cancelled and retired without payment of any consideration therefor.

(b) <u>GR_Common_Stock</u>. Each share of GB Common Stock issued and outstanding immediately prior to the Effective Time (other than any Dissenters' Shares) shall be converted into one share of Bank Common Stock.

General Provisions.

(a) <u>Termination and Agreement</u>. The obligations of the parties to effect the Merger shall be subject to all the terms and conditions contained in the Merger Agreement. Notwithstanding shareholder approval of this Agreement, this Agreement shall terminate forthwith in the event that the Merger Agreement shall be terminated as therein provided prior to the Effective Time

(b) <u>Amendment</u>. This Agreement may be amended by GB and Bank at any time prior to the Effective Time. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.

(c) <u>Successors and Assigns</u>. This Agreement shall be binding upon and enforceable by the parties hereto and their respective successors, assigns and transferees, but this Agreement may not be assigned by any party hereto without the written consent of the other.

(d) <u>Governing Law</u>. This Agreement has been executed in the state of California, and the laws of the state of California shall govern the validity and interpretation hereof and the performance by the parties hereto.

(e) <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

[The remainder of this page is intentionally blank.]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

GLOBAL BANCORP

By ______ Name: Vinod Thukral Title:_Chairman of the Board

By: ______ Name ______ Trile: <u>Executive</u> Vice President

GLOBAL TRUST BANK

By:_____ Name: Vinod Thokcal Title: <u>Chairman of the Board</u>

By:	
Name:	
Title, Executive Vice, President	

.

Certificate of Approval of Agreement of Merger

Vinoc Thukral and Bhupen Amin and certify that:

- They are the Chairman of the Board and the Secretary, respectively, of Global Trust Bank, a California banking corporation.
- The principal terms of the Agreement of Merger in the form attached were duly approved by the board of directors and by the sole shareholder of the corporation by a vote that equaled or exceeded the vote required.
- 3 The shareholder approval was by the sole shareholder of Global Trust Bank owning 100 shares, constituting 100 percent of the outstanding shares of the corporation
- 4 There is only one class of shares and the number of shares outstanding entitled to vote on the merger is _____.

We further declate under penalty of perjury under the laws of the State of California that the matters set forth in this certificate are true and correct of our own knowledge.

Date: _____

Vined Thukral, Chairman of the Board

Bhuper, Amin, Secretary

Certificate of Approval of Agreement of Merger

Vinod Thukral and Bhupen Amin certify that.

- They are the Chairman of the Board and the Secretary, respectively, of Global Bancorp, a California corporation.
- The principal terms of the Agreement of Merger in the form attached were duly approved by the board of directors and by the sheacholders of the corporation by a vote that equaled or exceeded the vote required.
- 3. The shareholder approxal was by the holders of _____ shares of the outstanding shares of the corporation constituting more than 50% of the outstanding shares of the corporation.
- There is only one class of shares and the number of shares outstanding entitled to vote on the marger is 2.115.001.

We further declare under penalty of perjury under the laws of the State of California that the matters set forth in this certificate are true and correct of our own knowledge.

Date: _____

Viood Thukral, Chairman of the Board

Bhupen Amin, Secretary

APPENDIX C

Morger Agreement between Bay Commercial Bank

and Global Trust Bank

.

MERGER AGREEMENT

THIS MERGER AGREEMENT, dated as of ______, 2011 (the "Merger Agreement") is made and entered into by and between GLOBAL TRUST BANK, a California corporation ("GTB" or the "Disappearing Corporation"), and BAY COMMERCIAL BANK, a California corporation ("BCB" or the "Surviving Corporation").

Recitals

A. GTB, BCB and Global Bancorp, a California corporation ("GB") have entered into an Agreement and Plan of Reorganization dated as of May 12, 2011 (the "Agreement and Plan of Reorganization") which provides for, among other things, the merger (the "Merger") of GTB with and into BCB pursuant to this Merger Agreement; and

B. The Boards of Directors of BCB and GTB have approved, and deem it advisable and in the best interests of BCB. GTB and their respective shareholders that BCB and GTB consummate the Merger.

NOW, THEREFORE, in consideration of the mutual agreements contained in this Merger-Agreement, and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties to this Merger Agreement hereby agree that GTB shall be merged with and into BCB in accordance with applicable federal and state laws and upon the terms and subject to the conditions set forth as follows:

1. <u>The Merger</u>. On the Effective Date (as hereinafter defined), the Disappearing Corporation shall be merged with and into the Surviving Corporation. The Merger shall become effective on the date (the "Effective Date") when a copy of the Agreement of Merger, certified by the Secretary of State, has been filed with the Commissioner of Financial Institutions as provided in Section 4887 of the California Financial Code (the "Effective Time").

 <u>Articles of Incorporation and By-laws</u>. The Articles of Incorporation and Bylaws, respectively, of the Surviving Corporation in effect immediately prior to the Effective Time shall be and remain the Articles of Incorporation and By-laws, respectively, of the Surviving Corporation until amended as provided by law.

3. <u>Officers and Directors</u>. From and after the Effective Time and until they shall have resigned or shall have been legally removed or until their respective successors shall have been elected or appointed and qualified. (a) the officers of the Surviving Corporation shall be those persons who are the officers of the Surviving Corporation immediately prior to the Effective Time, and (b) the directors of the Surviving Corporation immediately prior to the Effective Time shall continue to serve as directors of the Surviving Corporation.

Effect of the Merger.

4.1 All assets, rights, privileges, immunities, powers, franchises and interests of the Disappearing Corporation and the Surviving Corporation in and to every type of property (real, personal and mixed) and choses in action, as they exist as of the Effective Time, including.

ı.

appointments, designations and nominations and all other rights and interests as trustee, executor, administrator, registrar of stocks and bonds, guardian of estates, assignee, receiver and in every other fiduciary capacity, shall pass and be transferred to and vest in the Surviving Corporation by virtue of the Merger at the Effective Time without any deed, conveyance or other transfer. The corporate existence of the Surviving Corporation as a California corporation shall continue unaffected and unimpaired upon consummation of the Merger and the separate corporate existence of the Disappearing Corporation shall cease at and as of the Effective Time.

4.2 From and after the Effective Time, the Surviving Corporation shall be responsible and liable for all the liabilities and obligations of each of the Surviving Corporation and the Disappearing Corporation or the Disappearing Corporation or the Disappearing Corporation or the Disappearing Corporation at the Effective Time thereafter may be prosecuted as if the Merger had not taken place or, in connection with any such action or proceeding to which the Disappearing Corporation is a party, the Surviving Corporation may be substituted in place of the Disappearing Corporation. Neither the rights of creditors nor any liens upon the property of either the Surviving Corporation or the Disappearing Corporation of the Disappearing Corporation of the Merger.

Effect of Merger on Outstanding Shares.

5.1 Surviving Corporation. Each share of the common stock of the Surviving. Corporation issued and outstanding immediately prior to the Effective Time shall continue to be issued and outstanding from and after the Effective Time.

Disappearing Corporation. At the Effective Time, by virtue of the Merger and 5.2 without any action on the part of the holder of any shares of the capital stock of the Disappearing Corporation (the "GTB Common Stock"), each share of GTB Common Stock issued and outstanding immediately prior to the Effective Time shall no longer be outstanding and shall automatically be canceled and retired and shall cease to exist, and each certificate previously representing any such shares of GTB Common Stock shall thereafter represent the right to receive the authorized but anissued shares of common stock of Surviving Corporation ("BCB Common Stock"), or cash, or a combination of BCB Common Stock and cash, as follows: at the Effective Time, the holder of each share of GTB Common Stock shall be eligible to receive, at the election of such holder: (a) 0.75678 shares of BCB Common Stock; or (b) cash in the amount of \$7.25 pershare; or (c) a combination of BCB Common Stock and cash, each at the rate and in the amount pershare set forth in the preceding clauses (a) and (b), above, (the "Per Share Consideration"); provided however, that the aggregate amount of any cash payable by the Surviving Corporation to holders of GTB Common Stock who elect to receive all or a portion of their Per Share Consideration in each may not exceed an amount which is equal (or (x) 20% of the total consideration paid by the Surviving Corporation in the Merger, less (y) the amount of eash paid to holders of GTB Common Stock which are "dissenting shares" within the meaning of Section 1300(b) of the California Corporations Code. Certificates previously representing shares of GTB Common Stock shall be exchanged for shares of BCB Common Stock, cash or a combination of BCB Common Stock and easy in the amount of the Per Share Consideration upon the surrender of such certificates.

5.3 Dissenting Shares. Notwithstanding the foregoing, the holders of shares of GTB Common Stock which constitute "dissenting shares" within the meaning of Section 1300(b) of the California Corporations Code, shall have, in consideration for the cancellation of dissenting shares held by such holders, the rights given to them under applicable California law, including the right to receive the fair market value of those shares, in the manner and subject to the procedures and conditions provided by law.

5.4 <u>Restriction on Transfer</u>. From and after the Effective Date, no transfer of GTB Common Stock outstanding prior to the Effective Time shall be made on the record books of the Disappearing Corporation.

6. <u>Further Assurances</u>. From time to time as and when requested by the Surviving Corporation and to the extent permitted by law, the officers and directors of the Disappearing Corporation and the Surviving Corporation last in office shall execute and deliver such assignments, deeds and other instruments and shall take or cause to be taken such further or other action as shall be necessary in order to vest or perfect in or to confirm of record or otherwise to the Surviving Corporation title to, and possession of, all of the assets, rights, franchises and interests of the Disappearing Corporation and the Surviving Corporation in and to every type of property (real, personal and mixed) and choses in action, and otherwise to carry out the purposes of this Merger Agreement, and the proper officers and directors of the Surviving Corporation are fully authorized to take any and all such action in the name of the Disappearing Corporation or the Surviving Corporation or otherwise.

7. <u>Termination; Amendment</u>. This Merger Agreement may be terminated prior to the Effective Time by the mutual consent of the Boards of Directors of the Disappearing Corporation and the Surviving Corporation. This Merger Agreement may be amended prior to the Effective Time with respect to its principal terms by the mutual consent of the Boards of Directors of the Disappearing Corporation and the Surviving Corporation and the approval of the shareholders of the Disappearing Corporation and the Surviving Corporation. This Merger Agreement shall terminate automatically upon the termination of the Agreement and Plan of Merger prior to the Effective Time of the Merger therein provided for.

8. <u>Counterparts</u>. This Merger Agreement may be signed in any number of counterparts, each of which shall be deemed an original, and all of which shall be deemed but one and the same instrument.

9. <u>Choice of Law.</u> The validity, interpretation, and performance of this Merger Agreement shall be governed by and construed under the laws of the State of California.

IN WITNESS WHEREOF, the parties have duly executed this Merger Agreement as of the date first written above.

BAY COMMERCIAL BANK

GLOBAL TRUST BANK

Bх Name: George J Guarini Title: President

By_____ Name: Vinod Thukral Title President

By_____ Name: Kwary Coiwett Fitle: Secretary

Bv Name: Bhapen Arain Title Secretary

Certificate of Approval of Merger Agreement

George J. Guarioi and Keary Colwell certify that:

- They are the President and Secretary, respectively, of Bay Commercial Bank ("BCB"), a California banking corporation.
- This certificate is attached to the Morger Agreement, dated as of ______. 2011 by and between BCR and Global Trust Bank, a California banking corporation ("GTB"), which provides for the morger of GTB with and into BCR ("Merger").
- The Merger Agreement in the form attached was duly approved by the Board of Directors of BCB
- BCB has two classes of capital stock authorized: common stock and preferred stock. BCB has one class of capital stock, issued and outstanding consisting of 3,549,794 shares of common stock, which were entitled to vote on the Merger. No preferred shares were issued or outstanding.
- 5. The principal terms of the Agreement of Merger in the form attached were approved by the shareholders of BCB by the vote of a number of shares of its common stock which equaled or exceeded the vote required.
- The percentage vote required is more than 50%.

We declare under penalty of perjury of the laws of the California that the matters set forth in this Conflicate are true and correct of our own knowledge.

Date: _____, 2011

George J. Guarini, President

Keary L. Colwell, Secretary

Certificate of Approval of Morger Agreement

Viocid Thukral and Hhupen Amin and certify that:

- They are the Chairman of the Board and the Secretary, respectively, of Global Trust Bank ("GTB"), a California banking corporation
- This certificate is attached to the Morger Agreement, dated May 12, 2014 by and between G1B and Bay Commercial Bank, a California backing corporation ("BCB"), which provides for the merger of GTB with and into BCB ("Merger").
- The Merger Agreement in the form attached was duly approved by the Board of Directors of GTB.
- G1B has one class of shares authorized consisting of common shares and has 2,115,001 common shares outstanding which were entitled to vote on the Merger.
- The principal terms of the Agreement of Merger in the form attached were approved by GTB by the vote of a number of shares of its capital stock which equaled or exceeded the vote required.
- The percentage vote required is more (hun 50%)

We declare under penalty of perjury of the laws of the California that the matters set forth in this Certificate are true and correct of our uwn knowledge.

Date: _____ , 2011

Visiod Thukial, Chairman of the Board

Bhupen Amin, Secretary

APPENDIX D

Selected sections of Chapter 13 of the California Corporation Code

•

CALIFORNIA CORPORATIONS CODE TITLE 1. CORPORATIONS DIVISION 1. GENERAL CORPORATION DAW

CHAPTER 13. Dissemens' Rights.

§ 1300 Recognitization or Short String Merger, dissenting shares, corporate proclease at the fair market value; definitions

(a) If the approval of the outstanding shares (Section 152) of a corporation is required for a reorganization under subdivisions (a) and (b) or subdivision (e)or (f) of Section 120), each shareholder of the corporation confiled to vote on the transaction and each slowbholder of a subsidiary corporation to a substitution merger may, by complying with this depter, resplic the corporation in which the storeholder holds shares to purchase for tash at their fair market value the shareholder which are dissenting shares as defined an subdivision (b). The fair market value shall be decontined as of the day before the first announcement of the terms of the proposed reorganization or short-form merger, excluding any appreciation or depreciation in consequence of the proposed action, but adjusted for any slock split, reverse stock split, or share dividend which becomes effective (beteafter

(b) As used in this chapter, 'dissenting shares' means shares which come withough of the following descriptions:

(1) Which were put immediately prior to the reorganization of short form merger either (A) listed on any national securities exchange detailed by the Commissioner of Corporations under sold vision (o) of Section 25100 or (B) listed on the fist of OTU margin stocks issued by the Apare of Governors of the Federal Reserve System, and the total of stocks issued by the Apare of Governors of the Federal Reserve System, and the total of sections 300, 1307, 1307, and 1374, provided, however, that this provision loss not apply to any sbares with respect to which there exists any restriction on transfer imposed by the corporation of by any law or regulation; and provided, forther, that this provision does not apply to any class of shares described in subparagraph (A) in (B) of demands for payment are filed with inspect to 5 percent or more of the outstanding shares of that class.

(2) Which were outstanding on the date for the determination of shoreholders emitted to vote on the reorganization and (A) were not voted in favor of the reorganization or, (B) if described in subparagraph (A) or (B) of paragraph (I) (without regard to the provisos in that paragraph), were voted against the conganization, or which were hold of record on the effective date of a short-form merger; provided, however, that subparagraph (A) rather than subparagraph (B) of this paragraph applies in any case where the appreval required by Section 1201 is sought by written consent rather than at a meeting

(3) Which the dissenting shareholder has demanded that the corporation purchase at their fair market value, in accordance with Section 1300.

(4) Which the dissenting shareholder has submitted for endorsement, in ancordance with Section 1302. (c) As used in this chapter. "dissealing shareholder" means the record holder of disseating shares and includes a basisferer of record.

§ 1301 Notice to holders of disconting shares in reorganizations; domand for purchase; to be contents

(a) If, in the case of a reorganization, any shareholders of a corporation have a right order Section 1000, subject to couple new with paragraphs (3) and (4) of subgiveness (b) thereof, to require the corporation to purchase their shares for each, such corporation shall mail to each such shareholder a notice of the approval of the reorganization by its outstanding shares (Section 152) within 10 days after the date of such approval, accompanied by a copy of Sections (300, 1302, 1303, 1304 and this section, a statement of the price determined by the corporation to represent the tair market value of the dissenting shares, and a brief determined in the price determined to be followed.

of the shurdholder desires to express the shareholder's orgit in the rate instations. The statement of price constitutes an offer by the corporation to purchase as the price stated any disserting shares as defined in subdivision (b) of Section 1300, onlines tony love their states as disserting shares under Section 1309.

(b) Any shareholder who has a right to require the coporation to purchase the shareholder's shares for each under Section 1300, subject to compliance with paragraphs (3) and (4) of subdivision (b) thereof, and who desires the corporation to purchase such shares shall make written demand upon the corporation for the purchase of such shares and payment to the shareholder is right of their fair marker value. The demand is for effective for only purpose orders it is received by the corporation or any transfer agent thereof (1) in the case of shares described in clause (i) or (ii) of paragraph (1) of subdivision (b) of Section 1300 (without regard to the provises in that purchase), not have that the date of the shareholders' meeting to vote upon the conganization, or (2) in any other case within 30 days after the date on which the notice of the approval by the constrained purchase pursuant to subdivision (a) of the notice pursuant to subdivision (b) of Section 1500 with notice of the shareholder's meeting to the approval by the constrained ag shares pursuant to subdivision (a) or the notice pursuant to subdivision (b) of Section 1500 with the approval by the constrained ag shares pursuant to subdivision (b) of the notice pursuant to subdivision (b) of Section 1500 with any other case within 30 days after the date of which the notice pursuant to subdivision (b) of Section 1500 with any other case within 50 days after the date of the notice pursuant to subdivision (b) of Section 1500 with any other case within 50 days after the date of the notice pursuant to subdivision (i) of Section 1500 with a visco visce).

(c) The demand shall state the nonther and class of the shares held of record by the shareholder which the shareholder demands that the corporation proclease and shall contain a statement of what such shareholder claims to be the fair market value of those shares as of the day before the announcement of the proposed reorganization or short-form merger. The statement of fair market value constitutes an offer by the shareholder to sell the shares at such price.

§ 1002. Submission of share certificates for eadorsement, oncert ficated securities.

Within 30 days after the date on which notice of the approval by the outstanding shares or the notice present to successive (i) of Section 1110 was mailed to the shareholder, the shareholder shall submit to the corporation at its principal office or at the office of any transferagent thereof. (a) if the shares are certificated scentrics, the shareholder's certificates representing any shares which the shareholder demands that the corporation purchase, to be stamped or endorsed with a statement that the shares are dissenting shares or to be exchanged for certificates of appropriate demonstration so stamped or cardoraed or (b) if the shares are uncertificated scentrics, written action of the number of states which the shareholder demands that the corporation purchase. Upon subsequent transfers of the dissenting shares on the books of the corporation, the new certificates, initial transaction statement, and other written statements issued therefor shall near a like statement, together with the name of the original discorting holder of the shares.

§ 1362 Payment of agreed poice with outpress, agreement fixing fair market value; filing time of payment

(a) If the response on and the shareholder agree that the sparse is diversing shares, and agree upon the price of the shares, the dissenting shareholder is entitled to the agreed price with interest thereon at the legal rate on judgments from the date of the agreement. Any agreements focus the fair market value of any dissenting shares as between the corporation and the holders thereof shall be filed with the secretary of the corporation.

(b) Subject to the provisions of Section 1306, payment of the fair master value of distances plant. So make within 30 days after the account of the fair back beau pyrood or within 30 days after any statutory or contracted conditions to the reorganization are substract, whichever is later, and in the case of crubificiant securities, subject to succuries of the reorbacks therefor, unless provided otherwise by agreement.

§ 1304 Action to determine whether shares are dissenting shares or fair market value; timination; joinder; consolidation, determination of issues; appointment of appraisers

(a) If the corporation denies that the shares are dissualing shares, or the corporation and the shareholder field to gree open the fact market value of the states, then the shereholder domanding parthase of such states as dissenting vieres or any interacted corporation, within six months after the date on which nonce of the approval by the outstanding shares (bectors 152) monifor parsition to subdivision (*) of Section (11.1) must build to the shores (bectors 152) monifor parsition to subdivision (*) of Section (11.2) must build to the shores (bectors 152) monifor, may file a complaint to the superior court of the proper courtly proving the court to determine with the shores (to describ given build to the superior double of the table to describ give a states to both or use in court of the states (to describ give such a complaint section pending on such a complaint).

(b) Two or more discating sisteholders may join as plaint("s or be joined as definedants in any such action and two or more such actions may be consolidated.

(c) On the trial of the action, the court shall determine the issues. If the status of the shaces as dissenting shaces is in issue, the court shall first determine that issue. If the fair market value of the dissenting shares is in issue, the court shall determine, or shall appoint one or more impartial appraisers to determine, the fair market value of the shares.

§ 1305. Report of appraisers; confirmation; determination by court: judpment; payment; appeal; costs

(a) If the court appoints an appraiser or appraisers, they shall proceed forthwith to determine the four market value per share. Within the time fixed by the court, the appraisers, or a cuajority of them, shall make and file a report in the office of the cleck of the court. Theorem, on the motion of any party, the report shall be submitted to the court and considered on such evidence as the court considers relevant. If the court finds the report reasonable, the court may confine to:

(b) If a mixture of the appraisers appointed fail to make and file a report within 10 days from the date of their appointment or within such further time as may be allowed by the court or the report is not confirmed by the court, the court shall determine the fair market value of the describing sames.

(a) Subject to the provisions of Section 1306, (adgment shall be rendered exacts) the corporation for payment of an amount equal to the fair market value of each dissenting share multiplied by the number of a resenting share wheth only dissenting share to be set there are a puty, or who has intrivered, is entitled to require the corporation to purchase, with interest there are the legal rate from the date or, which judgment was entered.

(d) Any such judgment shall be payable forthwith with respect to uncertificated securities and, with respect to certificated securities, only upon the indersonant and delivery to the corporation of the certificates for the shares described in the judgment. Any party may append from the judgment.

(c) The costs of the attent, including masor able comparisation to the oppreisers to be fixed by the court, shall be assessed or apport aned as the court considers equitable, but, if the appraisal exceeds the price offered by the corporation, the corporation shall pay the costs (including in the discretion of the court attouncys) fees, here of report witnesses and interest at the legal rate on judgments from the date of compliance with Sections 1300, 1301 and 1802 if the value awarded hy the court for the shares is more than 125 percent of the price offered by the corporation under subdivision (a) of Section 3301).

§ 1366 Prevention of isotic fall payment; status as creditors: interest

To the extent that the provisions of Chapter is prevent the payment of any bolder, at dissenting shores of their bin market value, they that the come creditions at the corporation for the amount thereo hopedro with interest of the legislicite operation of it is directly payment for a subordinate to all other forditors in any high-training papers ing, such deby to be payable when permissible order the provisions of Chapter 5.

§ 1307. Dividends on disserting shares.

Cash dividends declared and paid by the corporation upon the dissenting shares after the date of approval of the corganization by the outstanding shares (Section 352) are prior to payment for the shares by the corporation shall be credited against the total amount to be paid by the corporation therefor.

§ 1308 Rights of disseming shareholders pending valuation; withdrawal and demond for payment

Except as expressly limited in this chapter, holders of dissenting shares continue to have all the rights and privileges mederated their shares, until the fair market value of their shares is agreed upon to determined. A dissenting shareholder may not withdraw a demand for payment unless the corporation consents thereto. § 1309 Termination of dissenting shareholder status.

Dissenting shares use their status as dissenting shares are the holders thereof cease to be dissenting shareholders and cause to be entitled to require the corporation to purchase their shares upon the happening of any of the following:

(a) The corporation abandons the reorganization. Upon abandonment of the thorganization, the corporation shall pay on the band to any disk at ng shandholder who it as initiated proceedings in good faith under this chapter all necessary expenses included in such proceedings and reasonable altomays' fees.

(b) The shares are transferred prior to their submission for endorsement in accordance with Section 1000 or me surrandered for conversion into shares of another class in accordance with the articles.

(a) The discentral shareholder and the corporation do not agree (pen the states of the shares as disconting shares or good the purchase proce of the shares, and nother lifes a completation intervenes in a pencing action as provided in Section 1304, with n six months offer the disc environment of the approval by the outstandard shares or not corporation to subdivision (?) of Section 1110 was mailed to the shareholder.

(d) The disserting shareholder, with the conserve of the corporation, withdraws the shareholder's demand for purchase of the disserting shares.

§ 1010 Suspension of a given to comparation or valuation proceedings: bright-or of shatchold as approval.

If 'ititation is indicated to test the sufficiency or equilatity of the roots of the shareholders in attinuing a recordenzation, any proceedings under bactions 1.06 and 13%, shall be separated until first determination of which by share

§ [311. Exempt shares]

This chapter, except Section 1312, does not apply to classes of shares whose terms and provisions specifically set forth the amount to be paid in respect to such shares in the event of pheorganization of merger.

§ 13(1). Right of dissenting shareholder to altee c, set aside or rescind morger or reorganization; restraining order or injunction; conditions.

(a) No shareholder of a corporation who has a right under this chapter to domand payment of cash for the shares held by the shareholder shallhave any right at law or in equity to anack the validity of the reorganization or short-form merger, or to have the reorganization or short form merger set as de or rescinded, except in an action to test whether the number of shares required to authorize or approve the reorganization have been legally voted in favor thereof, but any bolder of shares of a class whose terms and provisions specifically set forth the amount to be paid in respect to them in the event of a reorganization or short-form merger is existing to payment in accordance with those terms and provisions or, if the principal terms of the reorganization are approved parsuant to subthy, s(or (b) of Section 1202, is entitled to payment (nreordation with the terms and provisions of the approved storganization.

(a) If end of the parties to a reorganization of short form morger is directly or arditectly controlled by, or used to our more on drof with, enother party to the reorganization or short-form marger, subdivision (a) shell not apply to any shareholder of such party who has not demanded payment of oash for such shareholder's starts parsault to this or given by it the shareholder instruction to death the velicity of the temperature of the subgiventies or or to have the reorganization or short-form integer set aside or rescinded, the shareholder shall not thereafter have any right to demand payment of each for the shareholder's shares pursuant to this chapter. The court is any action starking the validity of the congenization or short-form merger of to have the reorganization or short-form integer set aside or rescinded, the shareholder shall not thereafter have any right to demand payment of each for the shareholder's shares pursuant to this chapter. The court is any action starking the validity of the congenization or short-form merger of to have the reorganization or short-form nerger set aside or rescinded shall not revise in or enjoin the consummation of the transaction except upon 10 days' prior notice to the congenization and begin patient by the court that clearly no other relief will adopt a progradient or the rescale of the transaction except upon 10 days' prior notice to the congenization and determination by the court that clearly no other relief will adopt a prograd by a progradie domain the constraint by the court that clearly no other relief will despise of member.

(c) If one of the nerices to a reorganization or short-form merger is directly or indirectly controlled by, or order common control with another party to the congenization or short-form merger, in any action to attack the validity of the reorganization or short-form merger or to have the reorganization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short-form merger which controls another party to the congenization or short to merger shall have the burden of proving that the transaction is just and reasonable as to the shutbodies of the next office party, and (2) a tensor who control of short reasonable as to the shutbodies of the tensor office party, and the transaction is just and reasonable as to the shutbodies of any party second effect.

APPENDIX E

Opimon of The Findley Group to Global Bancorp.

.

.

.



The Findley Group

May 12, 2011

Members of the Board of Directors Global Bancorp 700 E. El Camino Real, Suite 110 Mountain View, California 94040

Members of the Board

You have requested our opinion as investment bankers as to the fairness, from a tinang aliporation view, to the shareholders of Global Backgroup, Mountain View, Californ a ("Company") of the terms of the proposed reorganization and merger of Company and it's wholly owned subsidiary Global Trust Bank, Mountain, View, California, ("GT Bank") and Bay Commercial Bank, Walnut Creek, Californ 5 ("Bank"). It is proposed that Company and G [Bank will be merged. with and into Bank with Bank as the surviving entity (the "Morger"). Under the proposed transaction Company shareholders withreceive either shares of Sank. Common Stock in accordance with the Exchange Ratio, cash equal to the Per-Share Cash Consideration, or a combination thereof (the "Morgor Consideration") in exchange for all of the outstanding shores of Company. Common Stock as of the effective time of the Merger, as defined in the Agreement and Plan of Reorganization and Merger entered into as of May 12, 2011, (the 'Agreement') In the Agreement the Exchange Ratio is defined as 0.75678. In the Agreement the Per Share Cash Consideration is dofined as. \$7.25.

As part of its investment bonking business, the Findley Group is continually engaged in the valuation bank, bank hokling company and thrift securities in connection with mergers and acquisitions dationwide

In arriving st curopinion, we have reviewed and analyzed, amonic other things, the following: (i) the Agreement; (ii) cartain publicity available financial and other data with repood to Company and Bank, including audited heardial statements. for recent years ending December 31, 2010 and the most recent snaulted. Snand al statements for the guarter ending March 31, 2011; (#) certain other publicly available financial and other information concerning Company and Bank and the trading markets for securities of both Company and Aark; (w) publicity available information concerning other backs and back holding íor. theic securities. companies, t'ne tradiviru markets. and

3412-565

at amy sind ones.

Мадета Апалинанс & Явекрапланант Есохопис Помнома

Silvera

owano en Barbirg

1**90**-91619-014

Gh Rowney Sheet Auny No

Reis Faltoqs. Zisilyo C

10.5 AND 43

Sechi bar az 216 Levez a October

tiner Naviga ka Program

тырың Маркен Остортылі Епретиктикет

Сиру Го Февсиричина Вълга 9

Konegen en Shualas

Augustations at Meetings Accordings

Reinsennen in Caris Reinsen und der Karis

Revegicanty

542977784 (6364-9904) 542 50

Fragila de 18 a deservaria. Notes

Exservation of Nacionaly Approved ACL 1996 CONTRACT ONES

ng set sa respiration Generation Didden fai

17710 Aprile Munchey Sheet (* Anghern California 30806-1322) * 714/53077136 * Factorin e 7 4/6307510

the nature and terms of certain other merger transactions we believe relevant to our inquiry; and (v) other financial information concerning the businesses and operations of Company and Bank furnished to us by Company for purposes of our analysis. We have held discussions with senior management of Company concerning their past and current operations, financial condition and prospects.

We have reviewed with the senior management of Company earnings projections for Company, provided by Company as a stand-alone entity, assuming the Merger does not occur. We also reviewed with the senior management of Company the earnings projections for Bank that was provided by Bank, including projected does cavings prepared by the parties related to the Merger.

In conducting our review and in arriving at our opinion, we have relied upon and assumed the accuracy and completeness of the financial and other information provided to us or publicity available, and we have not assumed any responsibility for independent verification. of the same. We have relied upon the management of Company and Bank as to the reasonableness of the financial and operating forecasts, projections and projected. operating cost savings and earnings enhancement opport unities (and the assumptions and bases (herefere) provided to us, and we have accumed that such (precasts, projections) and projected operating cost savings and carnings enhancement opportunities reflect the best currently available ast mates and judgments of Company and Bank management. We have also assumed, without assuming any responsibility for the independent vertication of the same, that the appreciate allowance for loan losses for Company and Bank are apequate to pover such losses. We have not made or obtained any evaluations or appraisals of the property of either Company or Bank, nor have we examined any individual loan credit files. Hor ourboses of this opinion, we have assumed that the Merger will have the tax, accounting and legal effects described in the Agropment and assumed. the acquiracy of the disclosures set forth in the Agreement. Our opinion as expressed herein is limited to the fairness, from a financial point of view, to the holders of the shares. of Company Common Stock of the terms of the proposed mercer and reorganization of Company with Bank, with Company shareholders receiving shares of Bank Common Stock. and cash as sol forth in the Agreement and does not address Company's underlying husiness decision to proceed with the Merger.

We have considered such financial and other factors as we have deemed appropriate under the dircumstances, including among others the following: (i) the historica and current financial position and results of operations of Company and Bank, including interest income, interest expense, net interest income, net interest margin, provision for load losses, non-interest income, non-interest expense, earnings, dividends, internal capital generation, book value, intangible assets, return on assets return on shareholders' equity, capital batton, the amount and type of non-performing assets, ican losses and the reserve for den losses, all as set forth in the financial statements for Company and Bank; (ii) the assets and liabiblies of Company and Bank, including the load and investment portfolos, deposits, other fiabilities, historical and current liability sources and costs and fiquidity; and (iii) the nature and ferms of cortain other merger transactions neo virg banks are bank holding companies. We have also taken into account our assessment of generatidenomic, merkint and financial conditions and our experience in other transactions, powed as our experience in securities valuation and our experience of the banking industry Board of Directors - 3

generally. Our opinion is necessarily based upon conditions as they exist and can be evaluated on the date hereo'.

Based upon and subject to the foregoing, we are of the opinion as invostment bankers that, as of the date hereof, the terms of the morger of Company with and into Bank, with Company shareholders role ving for each share of Company Common Stock the amount of cash and number of shares of Bank Common Stock, as set forth in the Agreement, are fair, from a financial point of view, to the holders of the shares of Company Common Stock.

This oppoint may not be used or referred to by Company or public or disclosed to any person in any manner without our prior wither consent, with the exception of submission to the regulatory agencies as part of the applications and included in the proxy materials provided to shareholders of Company in relation to approval of the Merger. Units opinion is not intended to be a recommendation to any shareholder of Company as to how such shareholder should vote with respect to fine Merger.

Respectfully submitted.

y Staven ∿india Director

APPENDIX F

Opinion of Vining Sparks USG LP to Bay Commercial Bank

.



June 7, 2011

Board of Directors Bay Commercial Bank 1280 Civic Drive, Suite 100 Walnut Creek, California 94596

Members of the Board:

You have requested our opinion as to the fairness, from a financial point of view, to the lublers of the outstanding shares of common stock of Bay Commercial Bank, Walnut Creek, California ("BCB") of the consideration to be paid by BCB in the merger (the "Merger") of Global Bancorp, Mountain View, California ("Global Bancorp") and its woolly-owned subsidiary Global Trust Bank ("GT Bank") with and into BCB, pursuant to the Agreement and Plan of Reorganization and Merger by and between BCB, Global Bancorp and GT Bank (the "Agreement").

Pursuant to the terms of the Agreement, Global Bancorp will merge with and into GT Bank and subsequently, GT Bank will merge with and into BCB. Each share of Global Bancorp common stock outstanding shall become one share of GT Bank. Subsequently, each share of GT Bank shall be converted into the right to receive for each share of GT Bank common stock (i) one share of BCB common stock times the Exchange Ratio, and/or (ii) each in accordance with the Per Share Cash Consideration, at the election of the holder of such share of GT Bank common stock up to the limit set forth in the Agreement. As defined in the Agreement, the Exchange Ratio means a factor of 0.75678 and the Per Share Cash Consideration means 57.75 per share. All capitalized items used in this paragraph shall have the meanings ascribed to them in the Agreement. The terms of the Merger are more fully set forth in the Agreement.

Based on 2,115,001 shares of Global Bancorp common stock outstanding and Per Share Cash Consideration of \$7.25 per share, Global Bancorp will receive Merger consideration of \$15.3 million consisting of 20% cash and \$2% BCB common stock

For purposes of this opinion and in connection with our review of the proposed transaction, we have, among other things

Reviewed the terms of the Agreement;

Board of Directors Bay Commercial Bank June 7, 2011 Page 2

- Reviewed certain publicly available financial statements, both audited (where available) and un-audited, and related financial information of BCB and Global Bancorp, including those included in their respective annual reports for the past three years and their respective quarterly reports for the past two years;
- Reviewed certain internal financial information and financial forecasts relating to the business, earnings, cash flows, assets and prospects of each company furnished to us by BCB and Global Bancorp management;
- 4. Held discussions with members of executive and senior management of BCB and Global Bancorp concerning the past and current results of operations of BCB and Global Bancorp, their respective current financial condition and managements' opinion of their respective future prospects;
- Reviewed reported market prices and historical trading activity of BCB and Global Bancorp common stock;
- Compared the proposed financial terms of the Merger with the financial terms of certain other transactions that we deemed to be relevant;
- 7. Reviewed the potential pro forma impact of the Merger; and
- Performed such other financial studies, analyses and investigations, as we considered appropriate under the circumstances.

In giving our opinion, we have assumed and relied, without independent verification, upon the accuracy and completeness of all of the financial and other information that has been provided to us by BCB and Global Bancorp, and their respective representatives, and of the publicly available information that was reviewed by us. We are not experts in the evaluation of allowances for loan losses and have not independently verified such allowances, and have relied on and assumed that the aggregate allowance for loan losses set forth in the balance sheets of BCB and Global Bancorp at March 31, 2011 was adequate to cover such losses and complied fully with applicable law, regulatory policy and sound banking practice as of the date of such financial statements. We were not retained to and we did not conduct a physical inspection of any of the properties or facilities of BCB or Global Bancorp, did not make any independent evaluation or appraisal of the assets, liabilities or prospects of BCB or Global Bancorp, were not furnished with any such evaluation or appraisal, and did not review any individual credit files. Our option is necessarily based on economic, market, and other conditions as in effect on, and the information made available to us as of, the date hereof. Accordingly, it is important to understand that although subsequent developments may affect its opinion, we do not have any

Board of Directors Bay Commercial Bank June 7, 2011 Page 3

obligation to further update, revise, or reaffirm its opinion. We express no opinion on matters of a legal, regulatory, tax or accounting nature or the ability of the Merger, as set forth in the Agreement, to be consummated. No opinion is expressed as to whether any alternative transaction might be more favorable to holders of BCB's common stock than the Merger.

Vining Sparks IBG, L.P. ("Vining Sparks"), as part of its investment banking business, is regularly engaged in the valuation of banks and bank holding companies, thrifts and thrift holding companies, and various other financial services companies, in connection with mergers and acquisitions and valuations for other purposes. In rendering this fairness opinion, we have acted on behalf of the Board of Directors of BCB and will receive a fee for our services, which is payable upon delivery of this opinion.

Vining Sparks' opinion as expressed herein is limited to the fairness, from a financial point of view, of the Merger Consideration to be paid by BCB to the holders of Global Bancorp common stock in the Merger and does not address BCB's underlying business decision to proceed with the Merger. We have been retained on behalf of the Board of Directors of BCB, and our opinion does not constitute a recommendation to any director of BCH as to how such director should vote with respect to the Agreement. In rendering this opinion, we express no opinions with respect to the amount or nature of any compensation to any officers, directors, or employees of BCB or Global Bancorp, or any class of such persons relative to the consideration to be received by the holders of the common stock of Global Bancorp in the transaction or with respect to the fairness of any such compensation.

During the two years preceding the date of the opinion we have not had a material relationship with BCB or Global Bancorp where compensation was received or that we comemplate will be received after closing of the Merger.

Except as hereinafter provided, this opinion may not be disclosed, communicated, reproduced, disseminated, quoted or referred to at any time, to any third party or in any manner or for any purpose whatsoever without our prior written consent, which consent will not be unreasonably withheld, based upon review by us of the content of any such public reference, which shall be satisfactory to us in our reasonable judgment. This letter is addressed and directed to the Board of Directors of BCB in your consideration of the Merger and is not intended to he and does not constitute a recommendation to any shareholder as to how such shareholder should vote with respect to the Merger. The opinion herein expressed is intended solely for the benefit of the Board of Directors in connection with the matters addressed herein and may not be relied upon by any other person or entity, or for any other purpose without our written consent. This opinion was approved by the fairness opinion committee of Vining Sparks.

Board of Directors Bay Commercial Bank June 7, 2011 Page 4

Subject to the foregoing and based on our experience as investment bankers, our activities as described above, and other factors we have dremed relevant, we are of the opinion as of the date hereof that the Merger Consideration to be paid by BCB to the holders of Global Bancorp common stock is fair to the holders of BCB common stock, from a financial point of view.

Surreichy,

Vining Sparks & B.D. L.Y.

VINING SPARKS IBG, L.F.

APPENDIX G

Bay Commercial Bank Financial Statements for the Years Ended December 31, 2010 and 2009 and Independent Auditors Report



BAY COMMERCIAL BANK

FINANCIAL STATEMENTS

FOR THE YEARS ENDED DECEMBER 31, 2010 AND 2009

AND

INDEPENDENT AUDITOR'S REPORT

MOSS ADAMS or

INDEPENDENT AUDITOR'S REPORT

To The Branch of Directors and Shareholders Bay Commercial Bank

We have audited the accompanying statement of financial condition of Ray Commercial Bank (the Bank) as of December 31, 2010 and 2009, and the related statements of operations, changes in shareholders' rejuity, and cash flows for the years then ended. These financial statements are the responsibility of the Bank's management. Our responsibility is to capress an opinion on these financial statements based on our audits.

We conducted our audits in accordance with appliting standards generally accepted in the United States of America. These standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are tree of material missianement. An audit includes consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the bank's internal control over financial reporting. An audit also includes examining or a test basis, evidence supporting the amounts and disclosures to the financial statements. An audit also includes assessing the pocounting principles used and significant estimates made by management, as well as evaluating the proceil financial statement presentation. We believe that run audits provide a reasonable basis for run upholon.

In our opinion, the financial statements referred to above present fairly. In all material respects, the financial position of Bay Commercial Bank at December 31, 2010 and 2009, and the results of its operations and its cash flows for the years then ended, in conformity with accounting principles generally accepted in the United States of América.

Mess adams LLP

Stockton, California February 17, 2011

STATEMENT OF FINANCIAL CONDITION

December 31, 2010 and 2009

ASSETS		2610			2009
Cash and due from banks		\$ 1,560,536		8	2,122,895
Fodoral funds sold		33,485,702			7,445,000
Cash and cash equivalents	-	35,046,238			9,967,895
Interest bearing deposits in banks		141,431			98,194
Envestment securities available-for-aste		180,454			300,751
Pederal Home Luan Bank stock, at par		552,200			1\$\$,\$0D
J.come	\$ 116,194,799		\$ 131,819,766		
Oeferred fees, act	(112,013)		(196,857)		
Allowance for Joan (ossies	(2,365,000)		(1,650,000)		
Loans, net		133,517,486)29,972,915
Premises and equipment, net		415,298			482,338
Clines real estate owned		1,589,727			-
Deposit intangible		114,483			195,292
Interest receivable and other assets		2,379,965			2,035,234
Total Assocts	-	\$ 173,977,296	· ·	\$	143,408,319
LIABILITIBS AND SHAREHOLDERS' EQUITY					
Non-interest licening deposits		\$ 17,152,670		5	19,426,201
Interest bearing deposits	_	122,304,433			107,123, <u>987</u>
Total deposits	•	119,457,103			125,550,190
Other borrowing		415.081			-
Interest payable and other liabilities	_	608,661			823,271
Total liabilities	-	140,4±0,845			127,373,463
Commitments and contingencies (Note 11)					
Shareholdera' equity:					
Perferred stock - no par value; 10,000,000 show	es authorizad;				
no shares issued and outstanding		-			•
Common stock - no par value: 100,000,000 sha					
in 2010 and 10,000,000 shares with					
7,549,794 shares its und and outstars					
1,549,794 shares insued and nurstan	პაფ 2009	32,327,940			15,497,940
Addiuonai paid un capital		263,174			249,339
Rejgined earnistgs		902,813			282,286
Accumulated other comprehensive gain		2,524			5,291
Total shareholders' aquity		33,496,451	·		16,034,836
Total Linkilitic) and Shareholders' Equity	-	\$ 173,977,296		\$	143,408,319

.

;

;

i

:

:

The accompanying notes are en integral part of the Enancial statements.

STATEMENT OF OPERATIONS

For the years ended December 31, 2010 and 2009.

			2010		2999
Interest incom	ĸ.				
Լուցու		5	8,057,449	\$	7.157.340
Federal A	nds sold		47,835		36,815
En veçam et	n recurities and linerest bracing deposits in backs		9,851		28,649
PHL'B sto	ck dividend		1,555		· -
	Tolal interest income		8,126,690		7,274,804
Interact expan	Sec				
Deposits			1,775,169		2,328,916
Other bon	rowungs		6,563		179,993
	Total interest expense		1,781,732		2,508,909
	Net interest income		6,344,95%		4,715,895
Provision for !			1,436 <u>,926</u>		430,000
	Met interest income after provision for loan losses		4,914,032		4,265,895
Mon-interest L	oçeme:				
Other inco	ame and fees		361,822	_	299,363
	Fotal non-unterest income		365 122		299,363
Non-inferent e	хроме:				
Sulesies a	id related benefits		2,551,220		7,409,044
Occupane	y and equipment		622,741		498,216
Deta prec	ราวรับหู		290,400		306,273
Other			B13 <u>,942</u>		
	Topy] num interest expresse		4,278,309		3,954,916
	forenzas, ballose inclusio tax es		1,000,845		630,342
Nei expense fo	л ілсопа вяхая		360,31\$		270,993
-	Net income	\$	620,527	\$	359,349
Uncome par shi	11e:				
Basic:	Nei Incame	\$	0,26	\$	0.23
	Weighted average shares outstanding		3,344,315		t \$49 794
Diluted:	Net income	5	0.26	\$	0.23
	WeigMed average shares outstanding		2,344,315		1,562,057

i

:

The accompanying notes are an integral part of the financial statements.

STATEMENT OF CHANGES IN SHAREHOLDERS' EQUITY

For the years ended December 31, 2010 and 2009

Telepon December 21, 2000	Commen Stock Amount	Addibonai Puid in Capitel	Compre- headye	Retained Earnings	Accum- ulated Other Compro- bensive (Loss)Gain	Total Shuchelde=*
Pakance, December 31, 2008	\$ 11,497,940	\$ 207,055		\$ {77.063]	\$ 1,05	\$ 11,639,00k
Comprehensive income:						
Neitnune			\$ 359,349	359,349		355,349
Other compactor give income:						
Unicedized goin on available for sale of investment separates, not a	Cen ::		4,116		4,116	4.116
Yousi comprehensive iccurse			\$ 363,465	-	N(110	1,116
•			-	=		
SYCC based compensation repeate		42,280				42,283
Belance, December 34, 2009	15,497,940	349,339	-	282,166	5,291	16,014,856
lasue 2,000,000 states of commen stock	18,000,000	-			-	18,000,000
Cost of issuing common stock	(1,F10,C00)					(1,179,000)
Not another			\$ 620,527	620,577		620,427
Qaber comprehensive locevers:						
Unreatized loss on available for						
sale of investment seconties, not g	ป้าจะ		(2,767)	<u>•</u>	(2,767)	(2,367)
Total comprehens-ve 1440144			\$ 617,760			
Stock based compensation expense		15,815				(3,835
Balance, December 10, 2010	\$ 32,327,940	\$ 265,)74	-	\$ 901,013	3 2,524	\$ 33.496.451

!

;

į

The accompanying notes are on integral part of the financial statements.

STATEMENT OF CASE FLOWS

For the years ended December 31, 2010 and 2009

Cash llows from operating activities:	2013	2002
Nol negative state operating address to the second	\$ 620,127	\$ 339,349
Adjurtments to reconcide net i scome to net cash provided by operating activities:) 46649E)	a 1999
Provision for lot n latter	1,430,926	430,000
Deprectation and amonitization	162,731	127,557
Core deposis intangible amortization	80,810	47,338
Stock beest compensation	13,835	42,28)
Defended from origination fees, ast	115,462	106,426
Increase in accuraci interest acceivable and other assets	(347,498)	(1,022,621)
(Decrease) increase in accured interest propole and other lightliftes	(214,612)	322,193
Net cash provided by operating activities	1,822,189	614,325
Cluck flows from investing activities:		
(Purchase) metarity of interest bearing deposits in haning	(43,037)	1,105,325
Porchase of Federal Home Lown Back stock	(196,700)	
Principal repayment of mongage-backed securities, not of change in market value	120,277	373,484
Net i serence in loans	(6,263,6VO)	(17,284,209)
Preclare of familture, fixtures and equipment	(135,691)	(355,771)
Net each secenced from acquisition		36,017,827
Not eash (asod in) provided by investing activities	(6,520,751)	19,854,255
Cash Rows from Sinancing activities		
Not increase from issuence of common stock (not of cost to raise capital)	16,830,000	-
Net increase in domand, interest bearing and savings deposits	11,564,784	527,164
Depress in time deposits	(2,637,871)	(14,510,766)
Repayment of FHLB barrewing	<u> </u>	(5,000,000)
Net each provided by (used in) financing activities	29,736,910	(18,983,602)
Smartabe its cash and cash equivatents	25,078,343	1,285,079
Count and cash equivalents at the heginning of the year	9,567,895	8,682,816
Cash and cash equivalents it and of the year	\$ 35,646,238	6 9,957,875
Supplemental disclosure of easil flow information:		
Cush paid during the year for:		
Interess expanse	\$ 7,810,075	3 2,479,753
Income tax	0,110,000	137,377
Acquisition:		
Assets purchased	•	16,766,534
Lipbilities assumed	<u> </u>	52,779,361
Nel each second		(36,012,827)
Non-cosh investing activities:		
Net change in presided gain op inversment		
scanities available for sale	(2,767)	4,116
Transfers of loans to allor rest estate owned	(2,174,##t)	-
Assumption of Jian on other real estate owned	4 I S,Č&I	-

:

The accompanying notes are an integral part of the financial statements.

NOTES TO FINANCIAL STATEMENTS

ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES.

The seconding and reporting policies of Bay Commercial Bank (the "Bank") are in accordance with accounting principles generally accepted in the United States of America and prevailing practices within the banking industry. A summary of the significant accounting policies applied in proparation of the accompanying financial statements follows.

Organization.

The state chartered Bank was incorporated under the laws of the State of California on March 24, 2004 and opened for business on July 20, 2004. The Bank offers traditional commercial banking products and services to businesses and fadividuals through two branches located in Contra Costa and Alameda Counties.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions. These estimates and assumptions affect the reported amounts of assots and liabilities at the date of the financial statements and the reported amounts of revenues and exponses during the reporting period. Significant accounting estimates reflected in the Bank's 2010 and 2009 financial statements needed the allowance for loan losses, the valuation ellowance for deferred tax assets, the fair value of stock options, the valuation of financial and the determinetion, recognition and measurement of impained loans. Actual results could differ from these estimates.

Acquisition

The Bank acquired assets and assumed liabilities of a branch office from Community Banks of Northern California on May 15, 2009. The acquired assets and assumed liabilities were measured at estimated fair values, as required by the acquisition method of accounting for baciness combinations Financial Accounting Standards Board Accounting Standards Codification (FASB ASC) 805, Buskness Combinations. Management made significant estimates and exercised significant judgment in accounting for the acquisition. The Bank recorded an identificate intangible asset representing the value of the one deposit customer base. The deposit intangible asset represents the value ascribed to the Img-term deposit relationships acquired and is being amornized on a straight-line basis over an estimated average useful life of three years.

The Bank is required to express acquisition related costs separately from the nequisition. It also requires that any restructuring costs he expensed separately from the business combination. The Bank applies these revixed provisions when accounting for the branch acquisition.

Cesh and Cash Equivalents

Cash equivalents are defined as short-term, highly liquid investments both readily convertible into known amounts of cash and so near maturity that there is insignificant risk of change in value because of changes in interest rates. Generally, only invostments with maturities of three months or less at the time of purchase quabity as each equivalents. Cash and each equivalents include cash and due from banks, and federal funds sold. The Bapk maintains the minimum required

NOTES TO FINANCIAL STATEMENTS (continued)

amount of funds on deposit with other federally insured financial institutions under correspondent, banking agreements. At times throughout the year, balances can exceed PDFC insurance limits.

As of December 31, 2010 and 2009, the Bank bas each deposits at other financial institutions in courses of FDIC insured limits. However, as the Bank places these deposits with major financial institutions and monitors the financial condition of these institutions, management holicies the risk of loss to be minimal.

Generally, banks are required to maintain non-interest bearing cash reserves equal to a percentage of contain deposits. For the years ended Deposite 31, 2010 and 2009, \$381,000 and \$374,000 reserve balance were required, respectively.

Interest housing depreits in banks

The Bank invests in interest bearing deposits in banks with terms of up to three years. At December 31, 2010 and 2009, the Bank held interest bearing deposits totaling \$141,431 and \$98,394 with a yield of 0.05% and 0.50%, and a weighted average term to materity of less than three months and five months, respectively. Deposits totaling \$37,500 are security for performance letters of credit totaling \$37,500 issued by correspondent banks on the Bank's behalf and are restricted. The letters of credit have an original term of row year and expire in 2011.

i

Investment Securities

All of the investment securities held by the Bank are classified as available for-sole securities. Investment securities include debt securities. Investment securities are carried at estimated foir value with unrealized gains and losses reported as a separate component of shareholders' equity, accomplated other comprehensive gain or loss, until realized. The estimated fair value is based on quoted market prices or third party dealer quotes. See additional discussion order Fair Value Magaurement.

Any discounts or premiams are accreted or enumined to interest income over the expected term of the investment considering actual prepayments, if applicable. The programs and discounts are adjusted periodically to cellect actual prepayment experience. The gains or loss on all investment securities sold is based on specific identification.

Federal Home Loan Bank Stock

As of December 31, 2010 and 2009, Federal Home Loan Bank of San Francisco (FHL8) stock totaling \$352,200 and \$355,500, respectively, is recorded at cost and is redeemable at par value.

<u>Loain</u>

Leasus are stated at the principal amount outstanding, not of the allowance for loan losves, not defensed fees, and uncarried discounts, if any. The Bank holds loans receivable primarily for investment purposes. The Bank purchases and sells interests in certain loans referred to an participations. The participations sold are sold without recourse.

In 2009, the Bank acquired leans in a business combination that are recorded at estimated fair while on their purchase date. The purchaser cannot correspond the related allowance for lean losses. Purchased leans are accounted for under FASH ASC 310-30, Loans and Debt Securities

NOTES TO FINANCIAL STATEMENTS (continued)

with Deteriorated Credit Quality. As of the purchase date, none of the loans were on non-accreat or exhibited credit quality deterioration since origination. A significant portion of the Back's loan portfolio is comprised of adjustable rate loans. Interest on loans is calculated and accred daily using the simple interast method based on the daily amount of principal outstanding. The poerual of interest is discontinued and any accrued and unpaid interest is charged against current income when the payment of principal or interest is 90 days past due, unless the loan is well secured and in the process of collection.

When the ability to fully collect non-accrual toan praterpal is in doubt, each payments received are applied first to principal until such time as full collection of the remaining recorded balance is expected. Generally, loans with temporarily impaired values and loans to berrowers expendeding financial difficulties are placed on non accrual even though the berrowers continue to repay the loans as scheduled. Interest received on such loans is restored to accrual basis when principal axt interest payments are being paid currently and full payment of principal and interest is probable.

:

Loan Fees and Costal

Loan origination fees, commitment fees, direct loan origination costs and purchase premiums and discounts on loans are deferred and recognized at an adjustment of yield, to be amortized to interest income over the contractual term of the loan. Other loan fees and charges, which represent income from delinquost payment charges, and miscellaneous loan or letter of modif services, are recognized as non-interest income when collected.

Salaries, employee benefits and other expenses totaling 3185,700 and \$258,200 are defetred as loan origination casts for the years ended December 31, 2010 and 2009, respectively.

Allowners for Loss Losses

The allowance for lean losses is evaluated on a regular basis by management. Periodically, the Basik charges current carnings with provisions for estimated probable losses of loans receivable.

The flank records an adjustment to the allowance for loan loss if the total estimated allowance for loan losses exceeds the anicult of estimated losses. The provision or adjustment takes into consideration the adequacy of the total allowance for loan losses giving due consideration to specifically identified problem loans, the transcial condition of the borrowere, fair value of the underlying collateral, recourse provisions, prevailing economic conditions, and other factors. Additional consideration is given to the Bank's historical loan loss experience relative to the Bank's loan portfolio concentrations related to industry, collateral and geography. This evaluation is inherently subjective and requires estimates that are susceptible to significant change as additional allowances based on their judgments of the information regarding problem loans and credit risk available to them at the time of their examinations. All December 31, 2010 and 2009, menagement believes the allowance for loan losses adequately reflects the exactly lisk in the loan portfolio.

Generally, the allowance for toan loss concasts of various components including a component for specifically identified worknesses as a result of individual loans being impaired, a component for general non-specific workness related to historical experience, economic conditions and other factors that indicate probable loss in the loan portfolio, and an unallocated component that relates to the inherent imprecision in the use of estimates. Loans determined to be impaired ace

NOTES TO FINANCIAL STATEMENTS (continued)

individually evaluated by management for specific risk of loss.

Losses are recognized as charges to the allowance when the iosn or portion of the iosn is considered incellectible or at the time of foreclosure. Recoveries on ioans receivable proviously charged off are credited to the allowance for ioan losses.

Other Real Estate Owned

Other real estate owned ('OREO") acquired through, or in lies of, forecleaver are held-for-sale and are initially recorded at the lower of cost or fair value lass saling expenses. Any write-downs to fair value at the time of transfer are charged to the allowance for lower losses, subsequent to forecleaver.

:

:

The Bank obtains an approach or market valuation analysis on all OREO. If the periodic valuation indicates a dealine to the fair value below recorded carrying value, an additional writes down or valuation allowance for OREO losses is established as a charge to carriings. Fair value is based on current market conditions, appraisals, and estimated rules values of similar properties. Operating expenses of such properties, net of related income, are included in other expenses. The Bank may make loans to facilitate the sale of OREO. Gains and losses on financed values are recorded in other expenses income. Gains and losses on financed values are recorded in accordance with the appropriate accounting orefled, taking into consideration the bayers initial and continue investment in the property, potential subordination and transfer of ownership.

Premises and Equipment

Bank premises and equipment are stated at historical cost tess accounciled depectation or smortization. Depreciation is determined using the straight-line method over the estimated useful lives of the related assate. The useful lives of furniture, fixtures and equipment are estimated to be three to five years. Leasabold improvements are amortized over the life of the assat or the term of the related lease, whichever is shorter. When essets are cold in otherwise disposed of, the cost stic related accumulated depreciation or emortization are removed from the accounts, and ony resulting gave or loss is recognized in current income. The cost of maintenance and repairs is charged to expense as incurred.

Impairment of Assets

All assets are reviewed for impairment whenever events or changes indicate that the carrying value of the asset may not be recoverable. As of December 31, 2010 and 2009, the Book determined that no events or changes occurred during 2010 and 2009 that would indicate that the carrying value of any long-lived assets may not be recoverable.

Declines in the fair value of available-for-sale securities below their cost that are deemed to be other than tempmary are reflected in earnings as realized losses. In estimating other-thantemporary impairment losses, management considers the length of time and the extent to which the fair value has been less than cost, the financial condition and near-term prospects of the instant, and the intent and ability of the Bank to retain its investment in the issuer for a period of time sufficient to allow for any anticipated recovery in fair value.

A loan may be considered impaired when, based on current information and events, it is probable that the Bank will be unable to obliget the scheduled payments of principal or interest when due

NOTES TO FINANCIAL STATEMENTS (continued)

scanding to the contractual terms of the lose agreement. Factors considered by management in detramining implement include payment status, collateral value, and the probability of collecting scheduled principal and interest payments when due. Loses that experience insignificant payment delays and payment shortfalls generally are not classified as impaired. Management determines the significance of payment delays and payment shortfalls on a case by case basis, taking into consideration all of the commistances surrounding the loan and the borrower, including the length of the delay, the reasons for the delay, the borrower's prior payment record, and the amount of the shortfall in relation to the principal and interest owed. Impairment is measured on a loan by loan basis for commission and construction loans by either the present value of expected future cash flows disconneed at the loan's effective interest rate, the loan's obtainable market price, or the fair value of the collateral if the loan is collateral dependent. See additional discussion under <u>Pair</u> <u>Value Measurement</u>.

Transfers of financial assets

Transfers of financial assets are accounted for as sales when control over the accets has been surrendered. Control over transferred assets is decined to be surrendered when: (1) the assets have been isolated from the Bank, (2) the transferrer obtains the right (five of conditions that contain u from taking advantage of that right) to pledge or exchange the transferred assets, and (3) the Bank does not maintain effective control over the transferred assets through an agreement to repurchage them before their maturity.

Income Taxes

The Bank uses the asset and liability method to account for income taxes. Under such method, deferred tax assets and liabilities are recognized for the future tax consequences of differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax basis (temporary differences). Deferred tax assets and liabilities are measured using exacted tax rates expected to apply to laughtle income in the years in which those temporary differences are expected to be recovered on actiled. The effect on deferred tax assets and liabilities and liabilities and liabilities are measured using exacted tax rates expected to be recovered on actiled. The effect on deferred tax assets and liabilities of a change in tax rate is recognized in the period of exactment.

i

A valuation allowance is established to the extent that it is more than likely than not that the benefits associated with the determination, secondition, and measurement of deferred tax assets and the requirement for a related valuation allowance is based on estimated future taxable income.

The Bank recognizes interest accorded and penalties related to phreedgnized tax benefits in tax expense. During the years ended December 31, 2010 and 2000, the Bank recognized \$1,000 and \$4,500, in interest and penalties, respectively.

The Bank files income tax returns in the U.S. federal jurisdiction and with the State of California. The Bank is subject to U.S. federal and state income tax examinations by (ax authorities for years beginning 2007 and 2006, respectively. The Bank had no unrecognized tax benefits at December 31, 2010 or 2009

Non-igterned humanite

Fees for other client services are recorded as income when the services are performed.

NOTES TO FINANCIAL STATEMENTS (continued)

Stock Based Compensation

The Bank recognizes in the proome statement the grant-date fair value of stock options and other equity-based forms of compensation issued to employees over the employees' requisite service period (generally the vesting period).

The fair value of each option is estimated on the date of grant using the Black-Scholes options pricing model. The fair value method includes an estimate of superied volatility based on the historical volatility of the price of similar bank stocks and an ustimate of the expected option term, which is based on consideration of the vesting period and contractual term of the option. The Bank estimates the number of options expected to be functional based on historical (originaries). The rick-free interest rates are equal to the U.S. Treasury yield at the time of grant and commensurate with the contractual term of the grant.

Income Per Share

Basic income per share (EPS) is computed by dividing the net income by the weighted average number of common shares outstanding for the year. Diluted EPS reflects the potential dilution that could need if stock options or warrants were exercised. The treasury stock method is applied to determine the dilutive effect of stock options and warrants in computing diluted EPS. For the pariod ended December 31, 2010, all of the stock options and warrants were excluded from the diluted common shares because they were antidilutive. For the period ended December 31, 2009, 12,263 stock options were included in common shares because they were dilutive. There were no warrants outstanding as of December 31, 2009.

For the periods ended December 31, 2010 and 2009, total weighted average common shares, outstanding are as follows:

	2010	2009
Common Stack	2,344,315	1,549,794
Deluted effect of warrants		
Délated effect of work options		12,163
Total weight average diluted shares	2,344,315	1,362,017

Warrents

In 2010, the Bank issued 2,000,000 shares of rommon stock for \$9.00 pet share for a total of \$18.0 million. Warrants to purchase 159,817 shares of common stock were issued to certain shareholders who acquired 1,598,140 shares of common stock in the offering. A total of 159,817 warrants to purchase common stock at \$9.00 per share were issued and outstanding as of December 31, 2010.

Comprehensive Income (Loss) --

Comprehensive income (loss) includes disclosure of other comprehensive income or loss that historically has not been recognized in the calculation of net income or loss. Unreatized gains and losses on the Bank's available-for-sale investment securities are included in other comprehensive income or loss. Total comprehensive income or loss and the components of

NOTES TO FINANCIAL STATEMENTS (continued)

acconsulated other comprehensive income are presented in the statement of changes π , shoreholders' equity.

Recent Accounting Pronouncements

EASE ASULNO, 2011-01 Adoption of new accounting stondards - In January 2011, the FASE issued Accounting Standards Update (ASU) No. 2011-01, Receivables (Topie 310): Deferral of the Effective Bate of Disclosures about Troubled Debt Restructurings in Update No. 2010-20. The amendments in this Update temporarily defay the effective date for interim and annual reporting periods beginning on or after December 15, 2010. For compublic entities, the disclosures are effective for annual reporting periods ending on or after December 15, 2010. For compublic entities, the disclosures in this Update encourage, but do not require, comparative disclosures for caller reporting periods that ended before initial adoption. However, an entity should provide comparative disclosures for those reporting periods ending after initial adoption. As this ASU is disclosure-related only, the adoption of this ASU did not impact the Bank's financial condition or results of operations.

[ASB ASU NO. 2010-20 In July 2010, the FASB issued ASU No. 2010-20, Receivebles (Topic 3)0): Disclosures about the Credit Quality of Financing Receivables and the Allawance for Credit Losses. This Update amonds Topic 310 to Introve the disclosures that an only provides about the credut quality of its financing receivables and the related allowance for credit losses. As a result of these amendments, an active is required to disaggregate by portfolio argment or close contain existing disclosures and provide centally new disclosures about its financing receivables and related attownnon for credit losses. The amendments in this Unstate apply to all entities, both public and nonpublic. The amendments in this Update affect all cutities with financing receivables, excluding short term trade accounts receivable or receivables measured at fair value. or lower of cost or fair value. For public entitles, the disclosures required by this Update as of facend of a reporting period are effective for Interim and annual reporting periods ending on or after December 15, 2010. The disclosures about activity that occurs during a reporting period are officetive for interim and annual reporting periods beginning on a after December 15, 2010. For noppublic entities, the disclosures are effective for annual unputing periods ending on or after December 15, 2011. The amendments in this Update encourage, but do not require, comparative disclosures for earlier reporting periods that ended before mitial adoption. However, an enlity should provide comparative disclosures for those reputing periods ending after initial adoption. As this ASU is disclosure-related only, the eduption of this ASU did not impact the Bonk's figancial condition or results of operations.

i

I

FASB ASU No. 2010-18 to April 2010, the FASB issued ASU No. 2010-18, Receivables (3 opic J10): Effect of a Loan Modification When the Loan is Part of a Pool That is Accounted for as a Single Asset. This Update clarifies that meetifications of loans that are accounted for within a pool under Subtople 310-30, which provides guidance on seconding for acquired toans that have evidence of credit deterimination upon acquisition, do not result in the removal of those loans from the pool over if the modification would otherwise be considered a troubled debt restructuring. An entity will continue to be required to consider whether the pool of essets in which the loan is included is impaired if expected eash flows for the pool change. The amendments do not affect the accounted for individually under Subtople 310-30 that are not accounted for within pools. The amendments in this Update affect any entity that acquires leans subject to Subtople 310-30, that accounts for some or all of those loans within popls, and that subsequently modifies one or more of those leans.

NOTES TO FINANCIAL STATEMENTS (continued)

after acquisition. The amendments in this Update are effective for modifications of leans accounted for within pools under Subtopic 310-30 occurring in the first interim or annual period reating on or after July 15, 2010. The amendments are to be applied prospectively. Early application is permitted. Upon initial adoption of the guidance in this Update, an entity may make a continue election to terminate accounting for leans as a pool under Subtopic 310-30. This election may be applied on a pool-by-pool basis and does not produce an entity from applying pool accounting to subsequent acquisitions of leans with credit deterioration. Adoption of this ASU did met have an impact on the Dank's financial statements.

EASB ASU No. 2010-06 In January 2010, the FASB issued ASU No. 2010-06, Fair Value Measurements and Disclosures (Topic 820): Improving Disclosures about Fair Value Measurements. This Update requires: (1) disclosure of the amounts of significant transfers is and out of Level 1 and Level 2 fair value measurement categories and the reasons for the transfers; and (2) separate presentation of purchases, soles, issuances, and solitements in the reconciliation for thir value measurements of the following existing disclosures set forth in the Codification.

i

;

Subtypic 820-10. (1) For purposed of reporting fair value measurement for each class of assets and liabilities, a reporting entity occds to use judgment in determining the appropriate classes of assets and liabilities; and (2) a reporting entity should provide disclosures about the valuation techniques and inputs used to measure fair value for both recurring and non-recurring fair value measurements. This Update is effective for interno and annual reporting periods beginning lannary 1, 2010, except for the disclosures about purchases, sales, issuances, and settlements in the roll forward of activity in Level 3 fair value measurements, which are effective for fiscal years beginning January 1, 2011, and for interim periods within those fiscal years. As this ASU is disclosure-related only, the adoption of this ASU did not impact the Bank's financial condition on results of operations.

Subsequent Events

Management has reviewed events occurring through February 17, 2011, the date the financial statements were issued.

Reclassifications

Certain prior year amounts may have been reclassified to conterts with the current year presentation.

ACQUISITIONS

On May 15, 2009, the Back acquired certain loans and deposits related to a branch office located in Castro Volley, California. The nanoaction was recorded as a business combination. The Bank acquired \$16.4 million in feans and assumed \$52.7 million in deposit liabilities. The Bank assumed the loave obligation related to the branch facility. The Bank paid a premium on transaction deposit accounts totaling \$242.431. In 2009, the Bank recognized a total of \$65,060 in transaction related expenses.

NOTES TO FINANCIAL STATEMENTS (continued)

The following table summarizes the fair value of the assets acquired and including assembed at the acquisition date:

	May 15, 2009
Pair value of assets acquired:	
Cosh	\$ 35,012,827
Loans	16,442,103
Premises and equipment	29,091
Core deposit intanguble	742,431
Other assess	52,909
Total fair value of essets explored	52,779,361
Fair value of liabilities assumed:	
an aug sa	
Demand	4,398,899
Interest-bearing domand and sevings	8,558,656
Time certificales	39,255,765
'Fotal deposits	52,721,320
Accrued interest	58,041
Total fair value of Jiabilities assumed	\$2,779,961
Gabdwill	5 .

The deposit intengible asset represents the value ascribed to the long term deposit relationships sequired and is being amortized on a straight-line basis over an estimated average useful life of three years. The core deposit intangible is estimated not to have a significant residual value. The carrying amount of the loans acquired and liabilities assumed were equal to fair value. No goodwill was seconded in the transaction. No impairment loss was recognized to connection with the core deposit intengible during 2010 m 2009.

i

i

ł

ł

INVESTMENT SECURITIES

The amortized cost and estimated market value of available-for-sale investment securities by contractual maturity at December 31, 2010 and 2009 consisted of the following.

2010	Américes Cost	Grots Unreal:act Gain	Conss Uprovileed Lass	Estimated Michael Vicus
Fixed rate mortgage-backed scoolities; After one year through 1 ve years Total Investment remaining	\$ 176,484 \$ 176,484	<u>s 4,290</u> s 4,290		\$ 180,474 \$ 380,174
2009: Si sedi tammangaga-backed securities: After one year through five years Tauti toopsment persidens	\$ 291,806 \$ 291,80d	S <u>8,943</u> S <u>8,943</u>	\$	<u>\$ 000,751</u> <u>\$ 000,751</u>

Expected matmities will differ from contracteal maturities because the underlying mortgages can

NOTES TO FINANCIAL STATEMENTS (continued)

be repaid with or without prepayment penalties. Factors such as prepayments and interest rates may affect the yield and carrying value of mortgage-backed accurities. The weighted average yield on the investment scennifics is 4.21% and 4.13% for the years ended Docember 31, 2010 and 2009, respectively. As of December 35, 2010 and 2009, the investment securities had a weighted average maturity of less than two years, respectively.

Management periodically evaluates each investment security in an unrealized loss position to determine if the impairment is temporary or other-than-temporary. For the period ended December 31, 2010 and 2009, none of the investment securities were to a loss position and management has determined that no investment security is other than temporarily impaired.

Not unrealized gain on available-for-sale investment securities totaling \$2,524 is recorded as other comprehensive income within shareholders' equity at December 31, 2010 and a net unrealized gain totaling \$5,291 is recorded as other comprehensive income within shareholders' equity at December 31, 2009. The Bank recorded no gains or losses on the value of investment securities during 2010 and 2009.

i

:

ļ

LOANS

The Bank's loan portions at December 30, 2010 and 2009 is summarized below:

	20)0	2009
Commercial	\$ 39,251,671	\$ 48,157,565
Construction and land	13,280,581	12 170,575
Commercial real estate	72,938,951	63,728.(35
Regident of root estate	10,492,536	7,756,193
Consumer	291,056	7,300
Tetal inans	\$ 136.194.799	\$ 131,319,765

For the years ended December 31, 2010 and 2009, the Bank had \$3,931,000 and \$2,175,000, respectively, of impaired loans on non-actual. The average balance of non-accual loans and impaired loans was \$1,138,300 and \$1,126,200, respectively, for the periods ended December 31, 2010 and 2009. For the period ended December 31, 7010 and 2009, if interest had been accrued such income would have been approximately \$64,800 and \$96,683, respectively.

15

NOTES TO FINANCIAL STATEMENTS (continue))

As of December 31, 2010, all of the Bank's impaired and son-secrut leans have a related allowance for loss as follows

	Recorded Lavestment	Unpaid Principal Balance	Related Alivarance	A conset Accorded Layertanicat	Interest Income Recognized
With an allemance recorded	1				
Comencecial	\$ 1,821,603	\$ 1,824,803	\$ \$00,000	5 2,052,475	\$ 100,303
Construction and land					
Commencial real estate	-				
Residennal	108,843	100,043	75,000	100,859	5,026
Consumes				-	
Тоы	F 1,930,646	1,930,646	1 575,0%	\$ 2.135.332	\$ 103,320

As of December 31, 2010 and 2009, the Back had no restructured loans and no loans 90 days delinquent and still accruing interest. As of December 31, 2010 and 2009, no loans and two loans totaling \$340,300, respectively, that are more than 30 days but less than 89 days past due are still accruing interest.

L

.

As of December 31, 2010 and 2009, fixed rate loads total \$3.6 million and \$16.2 million, respectively, and variable rate loads total \$127.6 million and \$113.6 million, respectively. As of December 31, 2010, variable rate loads with interest rate caps total \$34.9 million none of which have resched their caps, and a total of \$107.5 million have interest rate flows all of which are at their flows. More than 90% of the variable interest rate loans are their to the Prime rate as reported by the Wall Street Journal and can adjust monthly based on changes in the Prime rate. At December 31, 2010 and 2009, a total of \$1.1 million and \$2.0 million, respectively, of variable rate loans are field to the treasury constant maturity rate (CMT) as published by the Federal Reserve and adjust every two or five years.

Loans are made primarily for business, personal, and real estate purposes concentrated in Contra Costa and Alameda Counties. As of December 31, 2010, the Bank's loans outstanding comprised 54,6% term mortgage-type loans secured primarily by commercial teal estate, 1.9% for the purpose of constructing commercial and residential property, 6.8% for the purpose of holding or acquiring unimproved land, 5.2% term mortgage-type toans secured by residential property, and 28.8% for general commercial uses including protessional, retait, and small business. Less that 1% of Bank's loans are consumer loans.

As of December 31, 2010, the Bank's unsecured toans outstanding totaled 7.9% of total loans. Real estate loans are secured by real property. Secured commercial and other loans are secured by deposits, or business or personal assets. The Bank's policy for requiring colluteral is based on analysis of the borrower, the borrower's traductry and the economic environment in which the loan is granted. The loans are expected to be repaid from each flows or proceeds from the sale of selected assets of the borrower.

As of December 31, 2010 and 2009, the single largest loan totaled \$7.0 million and \$4.1 million, respectively, and are second by commercial real calate. As of December 31, 2010 and 2009, undisbursed commitments and \$22.4 million and \$22.4 million, respectively. As of December 31, 2010 and 2009, the Bank sold participations in toans totaling none and \$4.0 million, respectively. No gain or loss was recorded related to participations \$6.6

NOTES TO FINANCIAL STATEMENTS (continued)

The Bank evaluates and assigns a risk grade to each loan based on cartain criteria to assess the credit quality of each loan. The assignment of a risk rating is done for each individual toan. Loans are graded from inception and on a continuing basis until the dobt is repaid. Any adverse or beneficial trends will trigger a review of the toan risk rating. Each loan is assigned a risk grade based on its cherecteristics. Loans with low to average credit risk are assigned a lower risk grade than those with higher credit risk as determined by the individual toan characteristics.

The Bank's pass nans includes litera with acceptable business or individual credit risk where the barrower's operations, each flow or financial condition provides evidence of line to average levels of risk. Losse that are assigned higher risk grades are losse that exhibit the following characteristics:

A special mantion asset has potential weaknesses that deserve clove attrutices. If left uncorrected, these potential weaknesses may result in a deterioration of the repayment prospects for the asset or in the Bank's credit position at some future date. Special Montion assets are not adversely classified and do not expose the Bank to sufficient risk to warrant adverse classification. Loans in this category would be characterized by any of the following situations.

- Uredit that is currently protected but is potentially a weak asset.
- Credit that is difficult to manage because of an inadequate loan agreement, the condition of and/or control over collateral, failure to obtain proper documentation, or any other deviation from product lending practices.

:

Adverse financial trends.

Special Mention should be a temporary rating, pending the occurrence of an event that would cause the risk rating to either approve or to be dowograded.

A substandard asset is inadequately protected by the current net worth and paying capacity of the obliger or of the collateral pledged. Assets so clossified must have a well-defined weakness or weaknesses that juppardize the liquidation of the debt. Assets are characterized by the distinct possibility that the Dark will sustain some loss it the deficiencies are not corrected. The potential loss does not have to be recognizable of an individual credit for that credit to be risk total substandard. A lean can be fully and adequately secured and still be considered substandard. Some characteristics of substandard loans are:

- Inability to service debt from ordinary and recurring cash flow.
- Chronic delinquency.
- Reliance upon alternative sources of repayment.
- Tensi lossis that are granted on fiberal terms because the borrower cannot service normal payments for that type of debt
- Repayment dependent upon the liquidation of collateral.
- Inability to perform as agreed, but adequately protected by collateral.
- Necessity to renegatiate payments to a non-standard level to ensure performance.
- The borrower is bankrupt, or for any other reason, future repayment is dependent on court solion.

Any assot classified doubtful has all the weaknesses inherent in one classified substandard with the added characteristic that the weaknesses make collection or figuidation in full, on the basis of currently existing tasts, conditions, and value, highly questionable and improbable. Doubtful assots have a high probability of loss, yet certain important and reasonably specific proxing.

NOTES TO FINANCIAL STATEMENTS (continued)

factors may work toward the strengthening of the asset. The Bank had an assets classified as doubtful as of December 31, 2010.

Assets classified loss are considered uncollectible and of minimal value. Assets classified loss are charged off against the allowance for loss losses.

The following table summarizes the Bark's losn portfolio by credit quality and preduct and/or collateral type as of December 31, 2010:

			Special			
	Pass		Mention	<u>Ş</u>	u'assendard	Total
Commercia]	\$ 33,333,916	\$	1,270,169	\$	2,627,588	\$ 39,251,673
Construction and land	13,280,583		-		-	13,280,583
Commercial real estate	72,938,951				-	72,938,951
Residential real asiate	10,383,693		-		108,543	10,492,536
Contunter	231,056		•		-	231,056
Tutel	\$1.32,198,199	5	1.270.169	\$	2,736,431	\$ 136,194,759

The following table provides an aging of the Bank's loans receivable as of December 31, 2010.

.

	30-29 Days Nati Dus	60-89 Days Pas' Duc	Gacales Then 90 Daya	Tratel Past	Cuprent	Total Louns Receivable	Recorded Inversiment > 90 Days and According
Commercial	\$ 300,000	\$ 291,917	5 1,326,836	\$ 1,821,803	1 17,429,870	\$ 39,251.673	
Consultation and land		-		-	13,180,583	13,230,580	
Commercial real estate				-	72,418,551	72,938.951	-
Residentia)	-		108.843	NG8,843	10,383,690	10,492,536	
Consyner					231,036	231.016	-
Total	2 300,000	<u>\$2P4.947</u>	\$ 1.333.699	\$ 1.930.646	1 134 264 152	\$136,194,299	· · · · · · ·

5. ALLOWANCE FOR LOAN LOSSES

Changes in the Bank's allowance for Ioan lesses for the years ended December 31, 2010 and 2009 is summarized below;

	2010	 2009
Balance at the beginning of the year	\$ 1,650,000	\$ 1,220,000
Provision	1,400,926	430,000
The begrand states.	(716,559)	-
Recoveries	633	 -
Balance al the end of the year	\$ 2,363,000	\$ 1,659,000

Impayed loons are loons for which it is probable that the Bank will not be able to collect all omounts due in accordance with the loan terms. Impaired loans totaled \$1,931,000 and \$2,175,000 at December 31, 2010 and 2009, respectively, and had related valuation allowances of

NOTES TO FINANCIAL STATEMENTS (continued)

\$\$75,000 and zero, respectively.

The following table summarizes the Bank's showance for loan losses for the year ended. December 31, 2010 by loan product and pollateral type:

Allowerse for togs lostes.	Commercial	Construction and Land	Commercial Real Extide	Besidentis	Consumer	Upellocated	Τοιοί
Beginning balance Tharge-utts	\$ \$40,000 (496,559)	\$ 170,000	\$ 795,000	\$ 135,000 (220,000)	\$.	dc0,01 2	1,650,000 (716,559)
Reconcises Provision Suding balance	633 <u>520,976</u> <u>5 595,030</u>	<u></u>	171,000 \$ <u>270,000</u>	293,030 \$ 210,000	<u>\$,000</u>	\$ 19.000	633 1,430,925 <u>\$ 2,365,000</u>
Priding hatanra	5 19 251,623	313,260,583	\$72.938.951	\$10.492,536	\$ 231,056		\$ 146,194,729
Allowence for loan load retated to loans individue evaluated for impairment	15 <u>5 546,000</u>	<u></u>	£	<u> </u>	<u>- 2</u>	<u>i:</u>	<u>\$ 575,000</u>
Deletes of loans individually evaluated for ampairment		<u> </u>	<u></u> -	<u>_\$_103.840</u>	<u></u>		\$ 1,930,546
Allowance for loan tess related to loans collective evaluted for impairment	y <u>5 495,000</u>	<u>\$ 175,000</u>	<u>\$ 970,000</u>	2 122000	<u>\$ 5,000</u>	<u> 19,005</u>	1.750.006
Balance of loans collectively evaluated for impairment	<u>\$ 37.419.870</u>	\$10.289,583	<u>_\$72,931,951_</u>	<u>_510.513.691</u>	<u>\$ 231.0%</u>		<u>\$ 104.264.550</u>

-

L

PREMISES AND EQUIPMENT.

Promises and equipment consisted of the following st Depember 31, 2010 and 2009:

		2010	_	2009
Fornitore, fixtures and equipment	\$	800,221	5	765,768
Leasehold improvements		279,007		279,081
Less accumulated depreciation and amortization		(623,930)		(\$62,012)
Lictal premises and equipment, net	5	455.298	\$	482.338

Depreciation and amertization included in occupancy and equipment expense total \$162,700 for the year ended December 31, 2010 and \$127,600 for the year ended December 31, 2009.

NOTES TO FINANCIAL STATEMENTS (continued)

The Bank leaves its branches and administration office under noncancelable operating leases. These leases expire on various dates through 2014. All leases have options to renew for five years. Forture minimum lease payments are as follows:

Year Easting December 31,

2011	5	361,438
2012		3\$9,984
2013		2\$5,537
2014		60,74D
Thereafter		-
	3	1,120,714

Rental expense included in occupancy and equipment expresse totals \$419.200 and \$337,900 for the years ended December 31, 2010 and 2009, respectively. As of December 31, 2010, a performance iener of credit totaling \$37,500 is notstanding for the benefit of one of the Bank's landlords. The letter of credit is second by a time deposit totaling \$37,500.

i

OTHER REAL ESTATE OWNED.

Other real estate owned as of December 31, 2030 and 2009 consisted of the following:

		2010		2009
Residenta.	8	1,589,722	ş	-
Allowence for loss		-	-	
Tota	\$	1,589,722	\$	

DEPOSITS

Deposits consisted of the following at December 31, 2010 and 2039:

Demand deposits S 17,152,670 \$ 29,426,203 NGW accounts and Savings 7.686,437 6,943,518	_
-	-
	;
Nithey marker 47,259,961 30,164,363	
Time – Jess John \$100,000 17,627,635 44,692,844	
Time \$100,000 without 49,730,400 25,324,052	;
Total deposits \$ 139,457,103 \$ 126,550,190	

At December 31, 2010 and 2009, the weighted average stated rate is 1.15% and 1.50%, respectively . At December 31, 2010, approximately \$22.5 million, or 16.1%, of the Bank's deposits are derived from nine (9) depositors. At December 31, 2009, approximately \$22.5 million, or 17.6%, of the Bank's deposits are derived from ten depositors.

The Bank accepts deposits related to real estate transactions qualifying under the Internal Revenue Code Section 1031, Tax Deferred Exchanges. These deposits fluctuate as the soliters of the real estate have up to six months to invest in replacement real estate to define the income tax on the property sold. Deposits related to this activity total \$30.3 million and \$2.6 million ist December 31, 2010 and 2009, respectively. Average deposit balances for this activity totaled \$2.6 million and \$3.0 million during 2010 and 2009, respectively

NOTES TO FINANCIAL STATEMENTS (contritued)

At December 31, 2010, appropriate annual maturities of time deposits are as follows:

Year Ending, December 31,

2091	\$ 13.076,239
2012 eori eDer	<u>14,281,796</u>
Total time deposits	\$ 67,05B,035

interest expense, net of early withdrawal penalty, recognized on interest-bearing deposits for the years ended December 31, 2010 and 2009 convists of the following:

....

!

I

	2010	2009
NOW eccounts and Sawings	S 20.795	\$ 12,231
Money market	482,821	486,033
Tune - Jess share \$100,000	352,62)	933,615
Tane - \$100,000 or more	7L8,931	\$99,039
Total inferest expense	\$ 1,775,169	\$ 2,328,916

9. OTHER BORROWINGS.

Office housewings for the period ending and as of December 33, 3010 and 2009 are as follows:

	2010			2009	
Oststanding balance	\$	415,081	5	-	
Interest rate		6.50%		0.00%	
Average balance	\$	103,07D	s	4,520,548	
Average interest rate		6.50%		3,95%	
Maximum balance	5	415,081	\$	5,000,000	

In 2010, the Bank assumed the senior lien on one of its OREO properties. The Bank intends to repay this lien in 2011.

The Bank has an approved we used bour owing facility with the FHLB for up to 25% of total essets. for a term not to curred five years under a blanket lion of certain types of loans. As of December 31, 2010, the total principal balance of such loans totals \$83.7 million and the total grincipal bebuice of investment securities plotged is \$176,184. There were no borrowings outstanding under this facility at December 31, 2010 and 2009.

The Bank has two Federal Funds (ines with available balances totaling \$9.0 million with two correspondent banks. There are no amounts outstanding under these facilities at December 31, 2010 and 2009.

NUTES TO FINANCIAL STATEMENTS (continued)

10. INCOME TAXES

Income taxes expense for the years ended December 33, 2010 and 2009 are as follows:

		1010		<u>7009</u>
Conent income laves	5	746,205	3	387,407
Delened income taxes, not		(366,017)		(126,414)
Total prevision for Income taxes	- 5	320,318	5	270.993

The provision for income tax differs from the amounts computed by applying the statutory Federal and State income tax rates. The significant terms comprising these differences for the years ended December 31, 2010 and 2009 consist of the following:

	2010		2009				
		A M Graph L		Rate %		Amo <u>urt</u>	Rate %
Reduced statutory fax rate	S	340,287		34,03%	\$	214,116	34,00%
State standory tax rate, not of							
Federal effective tax rate		71,604		7.15%6		45,097	7.15%
Stock based compensation		3,360		0.35%		4,990	0.79%
Other		(441,86)	-	-3.51%		6,590	1.05%
Fotal income tax expense	5	386,316	_	38.0036	5	270,993	42.99%

Deferred tax assets at December 31, 2010 and 2009, included as a component of other assets in the Statement of Financial Condition, consisted of the following:

Defoned tax assets, not of Fabilities:	2010	2009
Allowance for loan losses	2 K29,188	\$ 583,389
Defensed lower fees net of costs	128,531	RI,D13
Stock based compensation	33,324	31,191
FIILB SLOCK dividend	(12,717)	(12,717)
Other	2,340	(40,227)
Total deterred lax assols	\$ 1,010,666	5 644,649

As of December 31, 2010 and 2009, there is no valuation allowance based on management's estimate that the Bank will more likely then not be able to utilize all of the deferred tax assets. As of December 31, 2010 and 2009, the Bank has no federal or state not operating loss carryforwards.

÷

The Dank Eles income tax returns in the U.S. federal jurisdiction and in California. The Bank is no langer subject to income tax examinations by taxing authorities for years before 2007 for its federal filings and 2006 for its California tillings. The Bank's polley is to recognize penalties and interest as income tax expanse.

COMMITMENTS AND CONTINGENCIES

Landing and Lefter of Credit Commitments

In the normal course of business, the Bank enters into various commitments to extend oredit, which are not reflected in the financial statements. These commitments consist of the undishorsed ladace on personal and commercial lines of great and of undishorsed funds on

NOTES TO FINANCIAL STATEMENTS (continued)

construction and development loans. At December 31, 2010 and 2009, undispursed communents total 322,386 000 and \$22,434,000, respectively. In addition, at December 31, 2010 and 2009, the Bank issued latters of credit totaling \$483,000 and \$550,000, respectively, which represent guarantees of obligations of Bank chents. As of December 31, 2010 and 2009, the Bank pledged a time deposit totaling \$37,500 and none, respectively, to a correspondent bank as vecurity related to one of its branch offices.

The actual liquidity needs or the credit risk that the Bank will experience will be lower than the contractual amount of commitments to extend credit because a significant portion of these commitments are expected to expire without being drawn upon. The Bank's outstanding loan commitments are made using the same inderwriting standards as comparable outstanding toans. As of December 31, 2010 and 2009, the reserve associated with these commitments is zero.

Local Agency Deposite

In the normal course of business, the Bank accepts deposite from local agencies. The Bank is required to provide collateral for contain total agency deposits. As of December 31, 2010 and 2009, the FHLM issued a letter of credit on behalf of the Bank totaling \$4,250,000 and \$3,650,000, respectively, as collateral for local agency deposits.

I

12, EMPLOYEE BENEFIT PLANS

Effective January 1, 2005, the Bank adopted a qualified 401(k) profit shading plan (401(k) Plan) that covers substantially all full-time employees. The 401(k) Plan permits voluntary contributions by participants and provides for voluntary matching contributions by the Bank. For the years cuded December 31, 2010 and 2009 the Bank made contributions to the plan of \$33,500 and \$12,000, respectively.

See Note 13, Stock Option Plan for discussion of the Bank's stock option plan.

STOCK OPTION PLAN.

In 2004, the Board of Directors adopted the 2004 Strick Option Plan (Plan) subject to shareholder approval. The shareholders approved the Plan at the 2005 Annual Meeting. The Plan requires that the option price may not be less than the fair market value of the stock at the date the option is granted, and that the stock must be paul for in full at the time the option is exercised. All options expire on a date determined by the Row d of Directors, but not later than ton years from the date of grant. Upon grant, options yest ratebly over a three or four year period.

As of December 31, 2010, pursuant to the Plan, 354,979 shares of common stock are reserved for issuance to employees, organizers and Directors under measure and constantiony agreements. To date options to acquire 9,794 shares of common stock have been exercised.

NOTES TO FINANCIAL STATEMENTS (continued)

No options were granted in 2010. The following are the weighted average assumptions used to estimate the fair value of stock options granted during 2009;

		2009
Expected dividend yield		0.0%
Risk-free interest rate		1.74%
Expected volatility tector		26.3%
Excepted team		4.5 years
Weighted average exercise price	5	7.38
Weighted average fair value		1.79
Number of options granted		15,000
Pair value of opticits granted	\$	26,850

The following table provides the stock option activity for the year ending December 31, 2010:

	20	2010 Weighted		
	Number of Shares		vezege. remine Pric <u>e</u>	
Bulance at heginning of the year	346,852	\$	10.32	
Ouented	-		•	
Exercised	-		-	
Сируссі	(5,000)		9 9 8	
Dalance at end of the year	341,832	\$	10.32	

The following table provides the weighted-average fair and intrinsic values, and the weighted average remaining contractual life for stock option activity as of and for the years ending. December 31, 2010 and 2009:

2010

2009

		2010		7009
Weighted-average fear value of options granted during the year			s	1.79
Intrinsic value of options exercised		•		
Options exercisable at year each.		329,352		316,685
Weighterl-average cateroise price	\$	10.37	5	11,70
Intrinsic value		-		•
Weighted-average remaining				
contractual life		3.5 years	Ŕ	.,5 уелга
Options outstanding at year end:		141,852		346,852
Weighted average exercise prace	\$	10.32	5	10.32
Internation value.		-		-
Weighted average remaining				
contractual life		3.6 years	2	l.6 years
Weighterl-average caeseise price Intrinsic value Weighted-average remaining contractual life Options outstanding at year end: Weighted-average exercise prace Intrinsic value. Weighted-average remaining	\$	10.37 3.5 years 041,852 10.32	5	11,70 .5 years 346,852 10,32

NOTES TO FINANCIAL STATEMENTS (continued)

As of December 31, 2010, 214,200 incentive stock options and 130,652 non-qualified stock options are outstanding. As of December 31, 2010, there is \$9,100 of total unrecognized compensation cost related to non-vested stock options which is expected to be recognized over a weighted average period of two years. No stock options were exercised in 2010 and 2009. A total of \$2,133 and none, respectively, of tax benefits related to non-qualified stock options were recorded during 2010 and 2009.

34. REGULATORY MATTERS

Dividende.

The California Financial Code restricts the total dividend payment of any state banking association in my calcular year to the lesser of the Bank's retained earnings or the Bank's net income for its last three fiscal years, less distributions made to shareholders during the sense flow-year period. At December 31, 2010, 3902,310 is free from such restrictions

ł

;

Regulatory Capital

The Bank is subject to certain regulatory capital requirements administered by fuderal and state banking agencies. Failure to meet these ministrum capital requirements convinitiate certain mandatory and possibly additional discretionary, actions by regulators that, if materiaken, could have a direct material effect on the Bank's financial statements. Under capital adequacy guidelines and the regulatory framework for prompt corrective action, the Bank must meet specific capital guidelines that involve quantitative measures of the Bank's assets, liabilities and certain off-balance-sheet items as calculated under regulatory accounting practices. The Bank's capital amounts and classification are also subject to qualitative judgments by the regulators about components, risk weightings and other factors.

Quantitative measures established by regulation to ensure capital adequacy require the Bank to maintain minimum amounts and ratios of total and Tier 1 capital to risk-weighted assets and of Tier 1 capital to average assets. Each of these components is defined in the regulations. As of December 31, 2010 and 2009, management believes that the Bank meets all its capital adequacy acquirements. The Bank received notification from the FDIC categorizing the Bank as Well Capitalized under the framework of prompt corrective action regulations.

NOTES TO FINANCIAL STATEMENTS (continued)

To be entergoised as well capitalized, the Bank must maintain minimum total risk-based, Tier 1 risk-based and Tier 1 leverage ratios as set forth below.

		2010			300	s
Leverage Ratio		Sollers .	Ratio		Jaikes	Ratio
Day Commercial Bank	5	\$3,379	20.2%	8	15,835	10.4%
Minintum requirement for "Well-Capitalized"		8,247	5,0%		7,635	5 195
Minimum regulatory requirement		6,597	4.0%		6,106	4.0%
Tier) Risk-Based Capital Ratio						
Bay Commercial Bank	ş	33,379	23,7%	5	15.835	11.4%
Minimum requirement for "Well-Capitalized"		8,448	6.0%		8,151	6.0%
Minimum regulatory requirement		5,632	4.0%		5,366	4 D%
Total Righ-Based Copies! Rulig						
Bay Commercial Bank	\$	35,155	25.0%	\$	17,4\$5	126%
Minimum requirement for "Well-Capitalized"		14,090	10,0%		13,219	10.6%
Minimum regulatory requirement		11,264	8.0%		11,135	5.0%

15. RELATED PARTY TRANSACTIONS

In the ordinary course of business, the Bank may enter into transactions with related parties, including Directors, shareholdere, officers and their associates. These transactions include borrowings from the Bank with substantially the same terms, including rates and collateral, as to any loans to unrelated parties and do not involve more then usernal risk of collection.

The following is a summary of the aggregate form activity involving related party borrowers for the years ending December 31, 2010 and 2009:

:

	2010 2039
Balance, beginning	S 3,249,496 S 4,846,832
Disbursements	1,251.657 11,263
Amounts repaid	(1,508,598)
Ralance, ending	\$ 3,475,532 \$ 3,249,496
Undistanced commitments to related parties	\$ 3,094,221 \$ 2,463,402
Letters of credit issued for related partics.	<u>\$ 14,482 \$ 14,482</u>

At December 31, 2010 and 2009, the Bank's deposity included deposits from related parties which total approximately \$6.0 million and \$5.3 million, respectively.

NOTES TO FINANCIAL STATEMENTS (continued)

16 OTHER EXPENSES

For the years ended December 31, 2010 and 2009, respectively, other expenses consist of the following:

	2010	2009
Insurance including FDIC Preenties	\$ 191,911	\$ 272,384
Professional fees	231,203	133,913
Stationery and supplies	85,726	100,201
Marketing and promotions	73,387	55,740
Communication and postage	50,013	45,749
Bank service charges	39,264	38,009
Director feez	36,000	36,000
Courler Expense	19,404	24,202
(Joan Gefaul) related cuponses	24,873	
Other	42,161	35,143
Total other expenses	\$ 813,942	\$ 743,343

The Bank expenses marketing and promotions cover as flory at incutred. Advartising expense, included in marketing and promotions, total \$71,413 and \$11,831 for the years ended December 31, 2010 and 2009, respectively.

17. FAIR VALUE OF FINANCIAL INSTRUMENTS

Fair Value Measurement

Accounting guidance obtifies the definition of fair value, describes includes used to oppropriately measure fair value in accordance with generally scorptized accounting principles and expands fair value disclosure requirements. This guidance applies whenever other accounting pronouncements require or permit this value measurements.

The fair value hierarchy under ASC 820-10-05 prioritizes the inputs to valuation techniques used to measure fair value into three broad levels (Level 1, Level 2, and Level 3).

Level 2 - Inputs are mentjusted quitted prices in active markets (as defined) for (dentical assets or liabilities that the reporting entity has the ability to access at the measurement date.

Level 2 - Inputs are inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly.

Level 3 Inputs are unobservable inputs for the assot or listility, and reflect the reporting ontity's own assumptions about the assumptions that market participants would use in pricing the asset or liability (including assumptions about risk).

Investment securities available-for-sale - Securities classified as available-for-sale are recorded at fair value on a recurring basis. Fair value measurement is based upon quoted prices, if available if quoted prices are not available, fair values are measured using independent pricing models or other model-based valuation techniques such as the present value of force cash flows, adjusted for the security's credit rating, prepayment accumptions and other factors such as credit loss.

CONTRACTOR STATEMENTS (Continued)

assumptions. The recorded amount of securities measured at fair value on a recurring basis as of December 31, 2010 and 2009 are as follows:

2010	Total	<u>Level I</u>	<u>Level 2</u>	icyali
Fixed rate marigage-backed securities	8 160,474	\$ -	\$ 180,474	S
Tiotal assets measured at fair value	\$ 180,474	<u> </u>	\$ 180,474	<u>s</u> .
2009	Total	<u>Loval I</u>	<u>Level 2</u>	s .
Pixed rate mortgage-backed scewitics	S JCX5,753	S -	5 300,751	
Total assets measured at fair value	\$ 100,751	<u> </u>	\$ 300,751	3

Impaired Loans - The Bank does not eccord foans at fair value on a recorring basit. However, from time to time, a loan may be considered impaired and an allowance for loan tosses is established. Loans for which it is probable that payment of interest and principal will not be made in accordance with the contractual terms of the loan agreement are considered impaired. Once a lean is identified as individually impaired, management measures impairment in accordance with "Accounting by Creditors for Impairment of a Loan" (ASC 210-10). The fair value of impaired feans is estimated using one of several methods, including collateral value, market value of similar debt, enterprise value, liquidation value and discounted each flows. Those impaired heave not requiring an allowance represent leans for which the fair value of the expected repayments or collateral exceed the recorded investments in such loans. When the fair value of the collateral is based on an observable market price or a current appraised value which uses substantially observable data, the Bank records the impaired lean is non requiring Level 2. When an appraired value is not available or management determines the fair value of the collateral is further impaired below the appraised value, or the appraised value contains a significant assumption, and there is no observable market price, the Bank records the impaired lean as non-recurring Level 3.

<u>Other Real Estate Dwned</u> - The Bank records OREO at fair value on a non-recurring basis. The recorded amount of impaired loans measured at fair value on a non-recurring basis as of December 31, 2010 is as follows:

2010		<u>Lotal</u>	<u>L</u> g	<u>yel I</u>	44	vel 2		Level 1
OREO	Σ	1,389,792	s	•	8	•	\$	1.589,772
Speedie teserve		-				-		-
Total assets measured at few value	5	1,589,772	3		5	•	5	1.582,772

The Bank had no OREO as of December 31, 2009.

NOTES TO FINANCIAL STATEMENTS (continued)

The carrying amounts and fair values of the Bank's financial instruments at December 31, 2010, and 2009 are presented below.

		10	2009		
	Canying Antoants	Pair Value	Салтурд Атношаль	Fair Value	
Fintencial assora:					
Cash and each equivalents	\$ 35,187,669	\$ 35,187,669	\$ 10,066,289	\$ 10,065,289	
Investmenta	732,574	731,674	656,251	655,251	
Loans, net	133,517,486	133,700,000	129,972,915	179,983,000	
Accred interest receivable	445,191	443,791	491,948	497,948	
Firsencial lindflit dat					
Deposts	139,457,103	139,291,000	126,000,190	126,777,000	
Other borrowings	415,051	435,081	-	-	
Account interest payable	92,495	52,A95	120,788	120,768	
Commissions	-				

The following methods and assumptions were used to estimate the fair value of each class of Spaceial instruments (ASC 825-18-50).

<u>Cash and Cash Equivalents</u> - Cash and tesh equivalents include cash and due from banks, interest beauting deposits in banks, and Fed funds sold, and are valued at their corrying amounts because of the short-term nature of these instruments.

<u>Investments</u> - Investment securities are valued at the quoted market prices. See Note 1 for further analysis. The carrying value of the FHLB stock approximates the fair value because the stock is redeemable at par.

Loans Loans with variable interest rates are valued at the optrent carrying value, hecause these loans are tegularly adjusted to market rates. The fair value of fixed rate with recusioning maturities in excess of one year is estimated by discounting the future cash flows using correct rates at which similar loans would be made to borrowers with similar credit ratiogs for the serve remaining maturities.

<u>Accrued Interest Reveloable and Payable</u> - The accrued interest receivable and payable balance, approximates its fair value.

<u>Deputy</u>: The fair value of non-interest bearing deposits, interest licering transaction accounts and sayings accounts is the amount payable on demand at the reporting date. The fair value of time deposits is estimated by disconsiting the future cash flows using current rates offered for deposits of similar remaining matricities.

Other Reprovings - The fair value is estimated by discussions the future cash flows using current rates offered for similar homowings.

<u>Commitments</u> - The fair value of commitments represents the carrying amount of the related manuratized loan fees and is not material.

APPENDIX H

Global Bancorp and Subsidiary

Report of Independent Auditors and Consolidated Financial Statements

as of and for the two years ended December 31, $2010\,$

Report of Independent Auditors and Consolidated Financial Statements

.

Global Bandorp and Subsidiary

.

December 31, 2510 and 2009

CONTENTS

REPORT OF INDÉPENDENT AUDITORS	1
CONSOLIDATED FINANCIAL STATEMENTS	
Consolidated balance sheets	2
Consolidated statements of operations	3
Consolidated statements of changes in shareholders' equity	4
Consolidated statements of cash flows	5
Notes to consolid ated financial statements	6 - 23

PAGE

.

WAY HISSACARS SPA



REPORT OF INDEPENDENT AUDITORS

To the Board of Directors and Shareholders. Global Baucory and Subsidiary

We have addited the accompanying consolidated balance sheets of Glubal Bancorp and solvidiary (the Company) as of December 31, 2010 and 2009, and the related consolidated statements of operations, changes in shareholders' equity, and cash flows for the years ended December 31, 2010 and 2009. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our sudit.

We conducted our addits in accordance with additing standards generally accepted in the United States of America. Those standards require that we plan and perform our addits to obtain reasonable assurance about whether the consolidated financial statements are free of material infistatement. An oudit includes exam ning, on a test basis, evidence supporting the amounts and disclosures in the enosolidated financial statements. An addit also includes assessing the accounting principles used and significant estimates made by management as well as evaluating the overall consolidated financial statement presentation. We believe that our audits provide a reasonable basis for our opiaion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of G obai Bancorp and subsidiary as of December 31, 2010 and 2009, and the consolidated results of their operations and their cash flows for the years ended December 31, 2010 and 2009, in conformity with accounting principles generally accepted in the United States of America.

ma felome UP

Stockton, California March 24, 2011



GLOBAL BANCORP AND SUBSIDIARY

.

GLOBAL BANCORP AND SUBSIDIARY CONSOLIDATED BALANCE SHEETS

ASSETS

	DECEM9ER 31.			
	\$010	2009		
Cash and due from banks	\$ 3.640,814	\$ 5,046,549		
Federal fonds with	+ 6 <u>10,000</u>	3.500.000		
Cash and cash equivalents	8,250,814	8,576,549		
Interest bearing deposits in panks	19,798,000	18,478,000		
Loads, net	36.604.734	23,259,256		
Premises and equipment, net	513,657	729,946		
Interest in evable and other assets	410.1 <u>60</u>	581,122		
	5 65,560,373	5 34,624,873		

UABILITIES AND SHAREHOLDERS' EQUITY

Deposits Interest payable and other liabilities	\$ 49,239,356 <u>196,956</u>	\$ 3.3,772.306 261,5.23
Total liabilities	40,436,322	
Commutativity and Contingencies (Notes B and 10)		
Shareholder s' equity		
Common stock, \$.01 par value, 20,000,000 shares authorized; 2,115.001 shares is used		
and outstanding at December 31, 2010 and 2009	⊉1,150	21,150
Additional paid-in capital	71,704,660	21.583.960
Accumulated deficit	(5,581.759)	[4,014,066]
Total shareholders' equity	16,144,051	17,591,044
	5 65,580.373	\$ 51,624,873

GLOBAL BANCORP AND SUBSIDIARY CONSOLIDATED STATEMENTS OF DEERATIONS

	YEARS ENDED DECEMBER 31,			
	2010	2009		
INTERESTINGOME				
Interest and fees on loans	\$ 1,200,935	\$ 740,635		
(merest on federal /unde sold	3.259	6,820		
followerst on interest hearing deposits in banks	409,918	216,548		
	2,402,152	964.003		
INTEREST EXPENSE				
foreness enjærise og degarsits	575.24+	297,324		
Interest expense on borrowings	106,193	· .		
	681,437	297,324		
Net laterest lacome	1,770,715	666,679		
PROVISION FOR LOAN LOSSES	299,035	475,150		
Net Interest Income after provision for				
loan losses	1,421,660	19 L,529		
NON-INTEREST INCOME				
Cate on sole of leases	349,801	522,709		
Service charges on deposits	16,885	1,843		
Other non-interest income	20.797	· .		
	395,485	524,552		
NON-UNTERFST EXPENSES				
Salarxes and employee banefits	2,080,375	1,776,973		
Occupancy expense	430,028	469,480		
Professional fees	251,072	262.394		
Data processing fees	279,810	262,299		
Business development expense	80,080	120,939		
Incurance and assessment expense	91,956	74.280		
Supplies expense	23,655	44,177		
Other operating expenses	137,281			
	3,383,258	3,091,621		
Loss before income taxes	(1,566,093)	[2,375,740]		
Provision for income taxes	1,600	1,600_		
Net loss	\$ (1,567,693)	\$ (2,377,340)		
NET LOSS PER SHARF - BASIC	5 (0.743	<u>\$ [1.12]</u>		
WEIGHTED AVERAGE SHARES OUTSTANDING	2,115,001	2,115,001		

GLOBAL BANCORP AND SUBSIDIARY CONSOLIDATED STATEMENTS OF CHANGES IN SHAREHOLDERS' EQUITY FOR THE YEARS ENDED DECEMBER 31, 2010 AND 2009

	Compon Stack		Additional	Accomulated	
	Shares	ÁMCL NU	Faid-In Capital	Deficit	Total
Balances at Becember 31, 2008	2 115,001	\$ 21,150	\$ 21,453.304	\$ [1,635726]	\$ 19837,726
Stock-based compensation		-	130,656		130.656
Ne: loss	<u> </u>		<u> </u>	[2,377.340]	(2377,340)
3% paces at December 31, 2009	≥115, 00 1	21,150	21,533.960	[5,011066]	17591,044
Stock-based compensation		-	120,700	-	120,700
Net lass	<u></u>		·	[1,567.693]	[1367,693]
Balances at December 31, 2010	2115.001	\$ 21,150	\$ 21,704,660	\$ [5,581759]	\$ 15149D51

GLOBAL BANCORP AND SUBSIDIARY CONSOLIDATED STATEMENTS OF CASH FLOWS

	YEARS END AD DECEMBER 31		
	2010	2009	
CASH FLOWS FROM OPERATING ACTIVITIES:			
Net loss	\$ [7,567,693]	\$ (2,377,340)	
Adjustments to reconcile net loss to net cash	1,1,1,1,1,1,1	• [•]0.7,1340]	
from operating activities:			
Depreciation and amortization of premises			
and equipment	215,463	217,071	
Provision for loan losses	299,035	475,350	
Stock-based compressation.	120,700	130.656	
Increase in deferred Ican (ees	33,924	279,929	
Gain on sale of Joans	(349,601)	(522,209)	
Proceeds from sales of loans	3,866,601	10,001,948	
Louaces originated for sale	(5.516.800)	(9,213,177)	
(Decrease) increase in interest payable			
and other habilities	(64,557)	193,360	
Decrease (increase) in interest receivable and			
other assets	167,554	[351,482]	
Net cash from operating activities	(793,174)	(1,186,594)	
CASH FLOWS FROM INVESTING ACTIVITIES:			
Net increase in loans	(13,678,437)	(22,711,490)	
Net increase in interest bearing deposits in banks	(1,320,000)	(634,000)	
Purchase of premises and equipment	[3,174]	[334,339]	
Net cash (rom investing activities	(15,001,611)	(23,679,829)	
CASH FLOWS FROM FINANCING ACTIVITIES:	·		
Net increase in demand deposits and			
savings accounts	4 591,511	2,736,892	
Net increase in money market accounts	8,193,523	12,948,180	
Net increase in time deposits	2 682,016	16.762.635	
Repayment of Joan from organizers		(250,000)	
Net cash from financing activities	15 467.050	32,197,707	
NET [DECREASE] INCREASE IN CASH			
AND CASU EQUIVALENTS	(325.735)	7,351,284	
CASH AND CASH EQUIVALENTS, beginning of year	8,576,549	1,725,265	
CASH AND CASH EQUIVALENTS, control year	\$ 8.250,014	5 8.576.549	
SUPPLEMENTAL DISCLOSURES			
OF CASH FLOW INFORMATION:			
Cash baid during the period for:			
lateresi	\$ 691,831	\$ 286,050	
Income taxes	\$ 1,60C	5 3,600	
	* · · · · · ·		

NOTE 1 - ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES.

Global Bancorp and Subsidiary (the Company) is a bank holding company incorporated under the laws of the state of California on May 3, 2007, to organize and serve as the holding company for Global Trust Bank (the Bank). In conjunction with the incorporation of the Company, a total of 20,000,000 shares of common stock were buthorized with a \$.01 par value. The authorized shares of common stock were further divided into 10,000,000 shares of voting common stock and 10,000,000 shares of non-voting common stock. At December 31, 2010 and 2009, no shares of non-voting common stock, bad been issued.

On June 4, 2008, the California Commissioner of Financial Institutions approved the Bank's application for organization. The Bank also received approval of an application that it filed for insurance of bank deposit accounts with the Federal Deposit Insurance Comparation (FDIC) on December 3, 2008. The state-chartered bank was incorporated under the laws of the state of California on July 21, 2008, and opened for business on December 3, 2008. The Bank offens traditional commercial banking products and services to husinesses and individuals through one branchillocated in Santa Clana County. The Bank's target market is focused toward small to medium-steed, privately-held business customers, with an emphasis on serving the South Asian community of Indian-American individuals, professionals, and small, business ewners.

The consolidated financial statements include the accounts of Global Bancorp and its wholly-owned subsolutry, Global Trow Bank. There is non-mataritizity at the folding company level as the investment in the Bank is the only significant asset of the holding company. Intercompany accounts and transactions have been eliminated in consolidation. References to the Company are references to Global Bancorp. Including the holding company and the Bark.

The accounting and reporting policies of the Company conform with generally accepted accounting principles and general practices within the teacking industry. A summary of the significant accounting policies applied in the preparation of the accompanying consolidated financial statements follows.

Estimates – In preparing the consolidated force ial statements in conformity with generally accepted accounting principles, management is required to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the consolidated financial statements and revenues and expenses during the reported period. Actual results could differ from those estimates.

NOTE 1 -- ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Estimates (continued) The allowance for loan losses is the most significant accounting estimate reflected in the Company's consolidated financial statements. The allowance for loan losses includes charges to reduce the recorded balances of loans receivable to their estimated net realizable value, as appropriate. The allowance is based on estimates, and ultimate losses may vary from turnent estimates. These estimates for losses are based on individual assets and their related cash flow forecasts, sales values, independent appraisals, the volatility of certain real estate markets, and concern for disposing of real estate in distressed markets. Although management of the Company believes the estimates underlying the calculation of specific allowances are reas(mable, there can be no assurances that the Company could ultimately realize these values. In addition to providing valuation allowances on specific allowances for losses based on the overall portfolio composition, general market conditions, concentrations, and prior loss experience.

Other significant management judgments and accounting estimates reflected in the Company's 2010 consolidated financial statements include:

- Decisions regarding the tuning and placement of loans on non-accrual;
- Determination, recognition, and measurement of impaired loans; and
- Determination and evaluation of deferred tax assets and habilities.

Concentrations of credit risk - Assets and liabilities that subject the Company to concentrations of credit risk consist of interest-bearing deposits at other banks, customer loans, and customer deposits. Most of the Company's customers are located within Santa Clara County and the surrounding areas. The Company's primary lending products are discussed in Note 3 to the consolidated financial statements. The Company did not have any significant concentrations in its business with any one costomer or industry. The Company obtains what it believes to be sufficient collateral to secure the carrying value of loans. The extent and value of collateral varies based on the details underlying each loan agreement.

As of December 31, 2010, the Company does not have cash deposits at other financial institutions in excess of FDIC insured limits. However, the Company places these deposits with major financial institutions and monitors the financial condition of these institutions. Management believes the risk of loss to be minimal.

Cash and cash equivalents - Cash and rash equivalents include rash on hand, amounts due from banks, money market funds, and federal funds sold. Generally, federal funds are sold for one-day periods. The Company maintains the minimum required amount of funds on deposit with other federally insured financial institutions under correspondent banking agreements. Generally, banks are required to maintain non-interest bearing cash reserves equal to a percentage of certain deposits. At times throughout the year, balances can exceed FDIC insurance limits. At December 31, 2010 and 2009, the Company was required to maintain a \$100,000 minimum balance with its correspondent banks.

NOTE 1 - ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Interest bearing deposits in banks - The Company holds certificates of deposit with other financial institutions. The certificates have an original maturity date greater than 90 days and are carried at cost, which approximates for value.

Loans and allowance for loan losses - Loans are reported at the principal amount outstanding, net of deferred loan fees and costs and the allowance for loan losses. Uncarned discounts on installment loans are recognized as income over the terms of the loans. Interest on other loans is calculated by using the simple interest method on the daily balance of the principal amount outstanding.

Loan fees, net of certain direct costs of origination, are deferred and amountized over the contractual term of the loan as an adjustment to the interest yield. During the years ended December 31, 2010 and 2009, salaries, employee benefits, and other expenses totaling \$189,766 and \$175,418, respectively, were deferred as loan origination costs.

Loads on which the accrual of interest has been discontinued are designated as non-accrual loans. Accrual of interest on loans is discontinued either when reasonable doubt exists as to the full and timely collection of interest or principal or when a loan becomes contractually past due by 90 days or more with respect to interest or principal. When a loan is placed on non-accrual status, all interest proviously actrued, but not collected, is reversed against current period interest income, income on such loans is then recognized only to the extent that cash is received and where the future collection of principal is prohable. Interest accruals are resumed on such loans only when they are brought fully current with respect to interest and principal and when, in the judgment of management, the loans are estimated to be fully collectible as to both principal and interest.

The allowance for loan losses is established through a provision for loan losses charged to operations. Loans are charged against the allowance for loan losses when management believes that the collectability of the principal is unlikely. Subsequent recoveries of previously charged off amounts, if any, are credited to the allowance

The allowance for loan losses is evaluated on a regular basis by management and is based on management's periodic review of the collectability of the loans in light of historical experience, the nature and volume of the loan portfolio, adverse subations that may affect the borrower's ability to repay, estimated value of any underlying collateral, and prevailing economic conditions. This evaluation is inherently subjective as it requires estimates that are susceptible to significant revision as more information becomes available.

The allowance consists of specific, general, and unallocated components. The specific component relates to loans that are classified as impaired. Impaired loans, as defined, are measured based on the present value of expected future cash flows discounted at the loan's effective interest rate or the fair value of the collateral if the loan is collateral dependent. The general component relates to non-impaired loans and is based on historical loss experience adjusted for qualitative factors. An unallocated component is maintained to cover uncertainties that could affect management's estimate of probable losses. The unallocated component of the allowance reflects the margin of imprecision inherent in the underlying assumptions used in the methodologies for estimating specific and general losses in the portfolio.

NOTE 1 - ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICICS (CONTINUED)

Loans and allowance for loan losses (continued) - The Company considers a loan impaired when it is probable that all amounts of principal and interest due, according to the contractual terms of the loan agreement, will not be collected, which is the same criteria used for the transfer of loans to non-accrual status. Interest income is recognized on impaired loans in the same manier as non-accrual hours. Factors considered by management in determining impairment include payment status, collateral value, and the probability of collecting, scheduled principal and interest payments when doe, Loans that experience insignificant payment delays and payment shortfalls generally are not classified as impaired. Management determines the significance of payment delays and payment when it is basis, taking into consideration all of the circounstances surrounding the loan and the horizower, including the length of the delay, the reasons for the delay, the borrower's prior payment record, and the amount of the circuital moduli interest rowed.

Premises and equipment - Premises and equipment are stated at cost less accumulated deprectation and amortization. Depreciae on and amortization are provided for in amounts sufficient to relate the cost of depreciable assets to operations over their estimated service lives using the straight-line inclined. The estimated lives used in determining depreciation are:

Forniture, fixtures, and equipment	5 × 10 years
Computer software and equipment	<i>i</i> years
Leasehold corprovements	S years

Coasehold improvements are amortized over the lesser of the useful life of the asset or the term of the lesser. The straight-line method of depreciation is followed for all assets for financial reporting purposes, but accelerated methods are used for tax purposes. Deferred income taxes have been provided for the resulting temporary differences.

(ncome taxes - The Company uses the liability method to account for income taxes. Finder such method, deferred tax assets and liabilities are recognized for the future tax consequences of differences between the financial statement cataging accounts of existing assets and liabilities and their respective tax basis (temporary differences). Deferred tax assets and liabilities are reflected at currently enacted income tax rates applicable to the period in which the deferred tax assets or tiabilities are expected to be realized or settled. As changes in tax laws or rates are enacted, deferred tax assets and liabilities are called to be realized to through the provision for income taxes in the period of enactment.

A valuation allowance is established to the extent that it is more likely than not that the bedefits associated with the deferred tax assets will not be fully realized.

In accordance with accounting standards, the Company has assessed its tax positions and has concluded, that there are minumeringuized tax becefits at December 31, 2010 and 2009.

NOTE 1 - ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Income taxes (continued) - The Company recognizes interest accreed and penalties related to unrecognized tax benefits in tax expense. During the years ended December 31, 2010 and 2009, the Company recognized no interest and penalties.

The Company files income tax returns in the U.S. federal jurisdiction and with the state of California. The Company is subject to U.S. federal or state income tax examinations by tax acthorities for the years beginning with 2007.

Net loss per common share - Basic net loss per share amounts are computed by dividing net loss available to shareholders by the weighted average number of common shares outstanding during the year. Diluted earnings per share reflects the potential dilution that could occur if securities or other contracts to issue common stock, such as stock options, result in the issuance of common stock that share in the earnings of the Company. The Treasury stock method is applied to determine the dilutive effect of stock options when computing diluted earnings per share. However, dilutive earnings per share amounts are not presented when a net loss occurs because the conversion of potential common stock is antidilutive.

Stock-based compensation - The Company accounts for its stock based compensation by recognizing in the income statement the grant-date fair value of stock options and other equity-based forms of compensation issued to employees and directors over the requisite service period (generally the vesting period).

The fair value of each option grant is estimated as of the grant date using the Black-Scholes option pricing model. The assumptions used in this model include an estimate of expected volatility, which is based on the historical volatility of the price of similar bank stocks, and an estimate of the expected option term, which is based on consideration of the vesting period and contractual term of the option. In addition, the Company estimates the number of options expected to be forfeited based on bistorical forfeitore rates. The risk-free interest rates are equal to the US. Treasury yield at the time of the grant and commensurate with the expected term of the grant. There were no stock option grants during the year ended December 31, 2010. The weighted average assumptions used to estimate the value of stock options granted during the year ended December 31, 2009 included the following. dividend yield of 0.00%, expected volatility of 20.00%, risk-free interest rate of 1.55%, and expected option term of 5.03 years.

Transferred assets or third-party beneficial interests related to those transferred assets. No gain or loss has been surrendered to the set assets or third-party beneficial interests and (3) the Bank does not maintain effective control over the transferred assets or third-party beneficial interests related to those transferred assets. No gain or loss has been recognized on the sale of these participation interests for the years ended December 31, 2010 and 2009.

NOTE 1 – ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Transfers of financial assets (continued) - The Company has sold the guaranteed portion of certain Singli Business Administration (SBA) and United States Department of Agriculture (USDA) loans (with servicing retained) for cash proceeds equal to the principal amount of loans, as adjusted to yield interest. to the investor based upon the current market rates. The Company accounts for these transfers of Sinancial assets as sales when control over the transferred financial assets has been suprendered and any warranty period or recourse provision expires. The warranty period associated with the loan vales. typically extends for 90 days from the date of the sale and requires the Company to reimburse the purchaser for certain fees paid to the Company if the underlying borrower prepays or defaults on the Joan. During the warranty period, the Company does not developmize the loan from its consolidated. balance sheets. The loans are reported as a component of loans held for sale on the consolidated balance. sheets. The proceeds received in the sale are reflected as secured borrowings on the consolidated balance sheets and any gain or losy to be realized on the vale is deferred. The Company records interest income on the retained loan and interest expense on the secured borrowing. Upon the expiration of the warranty period, the sale is recognized by the Company, the Joan subject to the sale and the related. secured borrowing are derecognized from the balance shoets, and any deferred gain or loss is recognized in the consolidated statements of operations. At December 31, 2010, no loan sales were inprocess and accounted for as vectored borrowings.

When the loan sale is recorded, the carrying value of a sold loan is allocated between the guaranteed portion of the loan and the non-guaranteed portion of the loan, based on their relative fair values. The unguaranteed portion typically generates a loan discount that is amortized as an adjustment of the yield over the life of the loan. The cost of performing the servicing approximates the servicing income received. Accordingly, no servicing asset is recorded related to these loan values with servicing retained.

Advertising costs — The Company expenses marketing costs as they are incurred. Advertising expenses was \$33,098 and \$49,060 for the years ended December 31, 2010 and 2009, respectively.

Loans held for safe - The Company originates SHA and USDA loans with the intent to sell. Accordingly, cash Rows resulting from these loan sales are reflected as a component of operating activities on the consolidated statements of tash flows. Loans held for sale are reported at the lower of cost or fair value. Cost generally approximates fair value, given the short duration of these assets. Gains or losses on the sale of loans that are held for sale are recognized at the time of the sale and determined by the difference between net sale proceeds and the net houk value of the loans, net of discount related to the inguaranteed portion, less the estimated fair value of any retained mortgage servicing rights. These were no loans held for sale at Detember 31, 2010 or 2009.

Reclassifications — Certain reclassifications have been made to the 2009 convolidated financial statements to conform to the classifications used in 2010.

NOTE 1 - ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (CONTINUED)

Impact of new financial accounting standards - In January 2011, the FASD issued Accounting Standards Update (ASU) No. 2011-01, Receivables (Topic 310): Deferral of the Effective Date of Disclosures about Troubled Debt Restructurings in Update No. 2019-20. The amendments in this Update temporarily delay the effective date of the disclosures about the bled debt restructurings in ASU No. 2010-20, Receivables (Topic 310): Disclosures about the Credit Quanty of Financing Receivables and the Allowance for Credit Losses for public entities. The delay is intended to allow the Board time to complete its deliberations on what constitutes a troubled debt restructuring. The effective date of the new disclosures about troubled debt restructuring without be coordinated for credit, the guidance for determining what constitutes a troubled debt restructuring without be coordinated for credit, the guidance is anticipated to be effective for interve and annual periods ending after June 15, 2011. The amendments in this update apply to all public-entity creditors that modify financing inceivables within the scope of the disclosure requirements about troubled debt restructurings in Update 2010-20. The amendments in this update do not affect non-public entities that modify financing inceivables within the scope of the disclosure requirements about troubled debt restructurings in Update 2010-20. The amendments in this update do not affect non-public entities that modify financing inceivables within the scope of the disclosure requirements about troubled debt restructurings in Update 2010-20. The amendments in this update do not affect non-public entities As this ASU is disclosure-related only, our adoption of this ASU is 2011 will not import on financial cord tius or results of operations.

in July 2010, the FASB issued ASL No. 2010-20, Receivables (Tapic 316): Disclosures about the Credit Suality of Financing Receivables and the Allowance for Credit Losses. This Update amends Topic 310 to improve the disclosures that an entry provides about the credit quality of its financing reneivables and the related allowance for credit losses. As a result of these amendments, an entity is required to disaggregate by portfolio segment or class certain existing disclosures and provide certain new disclosures about its financing receivables and velocit allowance for credit losses. The amendments in this update apply to all entities, both public and non-public. The amendments in this update affect alentities with financing receivables, excluding short-cent, trade accounts receivable or nece vables. measured at fair value or lower of cost or fair value. For public entities, the disclosures required by this update as of the end of a reporting period are effective for interim and anomal reporting periods ending on or after December 15, 2010. The disclosures about activity that occurs during a reporting period ate effective for interim and annual reporting periods beginning on or after December 15, 2010. For nonpublic entities, the disclosures are effective for annual reporting periods ending on or after December. 15, 2011. The amendments in this update encourage, but do not require, comparative disclosures for earlier reporting periods that ended before (pitia, adoption. However, an eatity should provide comparative disclosures for those reporting periods ending after initial adoption. As this ASU is disclosure-related only, our planned adoption of this ASU during 2011 will not import our financial condition or results of operations.

NOTE 3 -- ORGANIZATION AND SUMMARY OF SIGNIFICANE ACCOUNTING POLICIES (CONTINUED)

Impact of new financial accounting standards (continued) - In April 2010, the FASB issued ASU No 2010-10. Receivables (Topic 310): Effect of a Loan Madification: When the Loan is Part of a Pool That is Accounted for as a Single Asset. This update clariflet that modifications of loans that are accounted for within a pool under Subtopic 310-30, which provides guidance on accounting for acquired loans that have evidence of credit deterior at on upon acquisition, do not result in the removal of toose loans from the pool even of the modification would otherwise be considered a troubled debt restructuring. An entity will continue to be required to consider whether the pool of assets in which the load is included is impaired if expected cash flows for the peol change. The amendments do not affect the alcounting for loans under the scope of Subtopic 310-30 that are not accounted for within pools. Loans accounted for individually under Subtagin 310-36 contoute to be subject to the troubled debt restructuring accounting. provisions within Subtopic 310-90. The amendments in this update affect any orbity that acquires loans subject to Subtopic 310-30, that accounts for some or all of those loans within pools, and that subtequently modifies one or more of those loans after acquisition. The amendments in this update are effective for modifications of loans accounted for within pools under Subtopic 310-30 occurring in the first interior or annual puriod ending on or after July 1S, 2010. The anocodments are to be applied prospectively. Early application is permitted, Upon initial adoption of the Ruidance in this update, an entity may make a one-time election to terminate accounting for loans as a pool ander Subtopic 313-30. This election may be applied on a pool-by-pool basis and does not preclude an entity from applying pouaccounting to subsequent acquisitions of Joans with credit delerioration. Adoption of this ASU did not have an impact on one financial condition or results of operations.

In jonuary 2010, the FASB issued ASU No. 2010-06, Fair Value Measurements and Disclosures (Taple 020): Improving Disclosures about Fair Value Measurements. This update requires: (1) disclosure of the amounts of significant transfers in and out of Level 1 and Level 2 fair value measurement (u) rgor ex and the reasons for the transfers; and (2) separate presentation of purchases, sales, issuances, and settlements in the reconciliation for fair value measurements using significant anohservable mouts (Level 3). In addition, this update can fies the requirements of the following existing disclosures set forth in the Codification Subtopic 820-10. (1) for purposes of reporting fair value measurement for each class of assets and liabilities; an eporting entity needs to use judgment in determining the appropriate classes of assets and liabilities; and (2) a reporting entity should provide disclosures about the valuation techniques and inputs used to measurements, which are offerning periods beginning fair value measurements. This update is effective for interim and annual reporting periods beginning January 1, 2010, except for the disclosures about purchases, sales, issuances, and settlements in the roll forward of artivity in Level 3 fair value measurements, which are effective for fixed years beginning January 1, 2011, and for interim periods within those fiscal years. As this ASU is disclosure-related only, our adoption of this ASU did not impact our fluancial condition or results of operations.

GLOBAL BANCORP AND SUBSIDIARY NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

NOTE 2 - CASH AND DUE FROM BANKS

Cash and due from panks includes balances with the Federal Reserve Bank and other correspondent banks. The Company is required to maintain specified reserves by the Federal Reserve Bank. The average reserve requirements are based on a percentage of the Company's depusit lish littles.

NOTE 3 LOANS

Major classifications of loans are as follows:

	ресьми	6R 31,
	2010	2009
Land and consurance on	5 2,478,895	\$. ž
Residential real estate	1,903,334	
Commercial real estate	14,110,650	10.732,165
Commercial	15.872.121	10 318,126
Consumer and other	3 349,822	2.983,694
	37,712,622	24,034,185
Allowance (or loan josses	(794,035)	(495,000)
Deferred loan fees and costs, net	(313,853)	(279,929)
Цомпъ, не	5 36 604,734	<u>5 23.259.256</u>

The Company's customers are primarily boated in Santa Clara and San Jose Counties. Approximately 37% of the Bank's loans are commercial real estate loans. Approximately 42% of the Bank's loans are other commercia, loans. The remaining 21% are in agricultural production, land and construction, residential real estate, and consumer loans. Generally, real estate loans are collateralized by real property while commercial and other loans are collateralized by funds on deposit, business, or personal assets. Repayment is generally expected from the proneeds of the sales of property for real estate construction loans and from cash flows of the borrower for other loans.

NOTE 3 + LOANS (CONTINUED)

Charges in the allowance for loan lowers any as follows:

	DECEMBER 31.			
		2010		2009
Balance, beginning of period	5	495.000	\$	19,850
Provision for Joan Josses		299,035		475,150
Loans charged off		-		-
Recoveries of loans previously charged off		·		-
Balance, and of period	\$	794,0 <u>35</u>	\$	495,000

There were no impaired loans, including troubled dept restructures, at or during the years including becomber 31, 2010 or 3009. There were no loans places on non-accrual status and no loans were past due greater than 90 days and still accruing interest at December 31, 2010 or 2009.

NOTE 4 - PREMISES AND EQUIPMENT

Premises and equipment consists of the following:

	DECEMBER 31,		1,
	2010		2009
Formiture, Socioces, and equiproph	\$ 177, <i>4</i>	83 5	177 634
Computer soltware and equipment	359,4	33	356356
leasehold improvements	429,3	47	429,242
	966,4	106	963,232
Less accumulated depreciation and			
ar90ml281l00	(452.7	(49)	(233.286)
	\$ 5 <u>134</u>	57 5	729 946

Depreciation and amortization expense totaled \$219,463 and \$217,071 for the years ended. December 31, 2010 and 2009, respectively.

NOTE 5 - INTEREST RECEIVABLE AND OTHER ASSETS

Interest receivable and other assets are as follows.

	DECEMBER 31.			
	2010		2009	
interest receivable - loans	5	109,195	\$	73,384
Interest recreatile - interest hearing deposits in banks		16,004		19,566
Prepaid FDIC assessment		10,383		101,357
Prepaid expenses and other assets		277,586		386,835
	\$	413,16 <u>8</u>	5	581,122

NOTE 6 - DEPOSITS.

fluxturiner deposits putstanding are as follows

	DECEMBER 31,		
	2010	2009	
Demand, non-interest bearing	\$ 4,990,322	\$ 2,750,573	
NOW	2.750,077	407.941	
Sevings	20,969	11,392	
Money market	21,717,538	13,524,015	
Time - less than \$100,000	2.012.293	1,275,273	
Linve - greater than \$100,000	17,748,159	15,803,162	
	\$ 19,239,356	\$ 33,772,306	

Time deposit issued and their remaining maturities are as follows:

Years ending December 31,

2011	\$ 9,961,396
2012	9,192,702
2013	306,353
	19,760.451

.

NOTE 6 - DEPOSITS (CONTINUED)

Interest expense, not of early withdrawal penalty recognized on interest bear ng doposits, consisted of the following:

	YEARS ENDED DECEMBER 31,			186R 31,
		2010		2009
WOW	2	3,956	\$	666
Savings		75		30
Money market		219,111		102,443
Time - less than \$100,000		24,27B		21,752
ilme – greater than \$100,000		327,824		172,433
Taral interest expense on deposits	\$	575,744	5	297,324

NOTE 7 - INCOME TAXES

The provision for income taxes for the years ended December 31, 2010 and 2009 was \$1,600.

Oeferred tax assets as Generaber 31, 2010 and 2009, consisted of the following.

	DECEMBER 31,		
	2010	2009	
Deferred tax assets:			
Net operating losses	5 1.360.000	\$ 903,000	
Drganszation and start-up costs	385,000	41S,000	
Accountion ash	-	11.000	
Allowance (or Joan Joss	297,000	161,000	
Charjual de contributions	4,000	2.000	
Stock-based compensation	194,000	175,000	
Other	2,000		
	2,250,000	1,687,000	
Deleared tax hobil (105:			
Acquinulated depoet intim		(53.000)	
Accesal to cash	(5,000)	·	
	(5,000)	(\$3,000)	
Valuation allowance	(2.245.000)	(1,634,000)	
Net deferred income tax asset	<u> </u>	\$	

NOTE 7 - INCOME TAXES (CONTINUED)

As of December 31, 2010, a valuation allowance equal to the amount of realizable deferred tax assets was recorded based on the determination that the Company is more likely that not unable to utilize the deferred tax assets. At December 32, 2010 and 2009, the Company had federal and state net operating loss carryforwards totaling approximately \$3,344,000 and \$2,223,000 and \$3,227,000 and \$465,000, respectively. These net operating loss carryforwards will begin to explicit in 2028 for federal income tax purposes and in 2018 for state income tax purposes it not previously utilized. Utilization of the net operating loss may be subject to substantial annual for itation due to ownership change limitation could result in the internal Revenue Code and similar state provisions. Such an annual limitation could result in the expiration of the upcrating loss carryforwards before utilization.

The valuation allowance was increased by approximately \$611,000 and \$963,000, respectively, during the years ended December 31, 2010 and 2009.

NOTE 8 - FINANCIAE INSTRUMENTS

The Company is a party to linuncial instruments with off-halance-sheet risk in the normal course of husiness to operative financing peerls of its customers. These financial instruments in ude compatinents to extend credit in the form of loans or through standby letters of credit. These instruments involve, to varying degrees, elements of credit and interest cate risk in excess of the amount recognized in the conso seated balance speet.

The Company's exposure to credit loss in the event of non-performance by the other party to the financial instrument for commitments so extend credit and standby letters of condit is represented by the contractual amount of those costruments. The Company uses the some credit policies in making commitments and conditional only gations as it does for our balance-sheet instruments.

	Contract
	Amount
Financial instruments whose contract	
ampunts represent credit risk:	
Undisbursed loan commitments	<u>5 4.291 390</u>

Conorditionants to extend conditione agreements to bord to a costoneer as long as there is no violation of any condition established in the contract. Commitments generally have fixed expiration dates or other termination classes and may require payment of a fee. Since many of the commitments are expected to expire without being drawn upon, the total commitment amounts do not necessarily represent future cash requirements. The Company evaluates each customer's creditworthmess on a case-by-case basis. The anomation of collateral obtained, if deemed benevary by the Company upon extension of credit is based on management's credit evaluation. Collateral held varios but may include accounts receivable, inventory, property, plant and equipment, and income-producing commercial properties.

Standby letters of credic are conditional commitments issued by the Company to guarantee the performance of a customer to a third party. The credit r sk involved in issuing letters of credit is essentially the same as that involved in extending loans to customers.

NOTE 9 - RELATED PARTY TRANSACTIONS

The Company, in the normal course of basiness, makes loans to and receives deposits from its directors, officers, principal sharebuilders, and their associates. In management's opinion, these transactions are on substantially the same terms as comparable transactions with other customers of the Company.

The Company's related party loan activity is summarized below:

	DECEMBER 31,		
	20 LC	2009	
Aggregate amount outstanding, beginning of the year	\$ 2,389,902	s .	
New name or advances during the year	795.000	2,550,000	
Repayments during the year	(592,963)	(160,098)	
Aggregate account constanting, earl of year	5 7,491,9 <u>39</u>	\$ 2,389,402	
l oan commitments	\$ 255,000	\$ <u>420,000</u>	

Related party deposits amounted to \$1,142,125 and \$725,162 as of December 31, 2010 and 2009, respectively

NOTE 19 - COMMITMENTS AND CONTINGENCIES

The Company is addigated for reactal payments under certain operating lease agreements, some of which contain renewal options and escolation clauses that provide for increased zentals. The Company's lease agreements expire in 2013. Total rental expenses for the years ended Detember 31, 2010 and 2009, was \$170,018 and \$205,208, respectively.

As of December 31, 2010, the luture minimum route, payments under non-cance lable operating leases are as follows:

Years ending December 31.

2011 2012 2013	٤	214,011 218,291 186,909
	5	619,211

The Company has an unsecured federal funds line totaling \$2 million with its correspondent bank, Pacific Coast Bankers' Bank. There were no amounts outstanding under these facilities at Dimember 31, 2010 and 2009.

NOTE 11 - REGULATORY MATTERS.

The Company and the Back are subject to various regulatory capital requirements administered by federal and state banking agencies. Follure to meet minimum capital requirements can initiate certain mandatory and possibly additional discretionary actions by regulators that, if undertaken, could have a direct material effect on the Company's consolidated financial statements. Under capital adequacy guidelines and the regulatory framework for prompt corrective action, the Company and the Bank must over specific regulator specific regulators that involve quantitative measures of the Company's assets, halolities, and certain off-balance-sheet tems as calculated under regulatory occolaring practices. The Company's capital amounts and classification are also subject to qualitative judgments by the regulators about components risk weightings, and other factors.

Quantitative measures established by regulation to ensure capital adequaty require the Bank to maintain combining amounts and ratios (set forth in the table on the following (age) of total and Tier I capital (as defined in the regulations) to risk-weighted assets (as defined), and of Tier I capital [as defined) to average assets (as defined). The capital guidelines do not apply to the Company on a coreso idated basis because its consolidated total assets are less than \$500 million and it is not engaged in significant off balance sheet activities. *Management* believes, as of December 31, 2010, that the Bank met all capital adequacy requirements to which it is subject.

As of Geocomber 31, 2010, the most recent notification from the FDIC categorized the Company and the Bank as well capitalized under the regulatory framework for prompt contective action. To be categorized as well capitalized, the Bank must maintain minimum total risk based. The first based, and Tier is leverage ratios as set for their the table on the following page. There are no conditions or events since that notification that management behaves have changed the Bank's category.

NOTE 11- REGULATORY MATTERS (CONTINUED)

The Bank's actual capital amounts and ratios are presented in the following table. The Company's actual capital amounts and ratios are substantially the same as the Bank's.

		Actual		For capital adequasy porposes			To be well capitalized under prompt corrective action provisions				
		ភាទារពង្	Ratio	<u>A</u>	mount		Ratio		mount		Hatio
(dottars in thousends) As of December 31, 2010. Total Capical (co Rice)											
Weighted Assets) Turnf Capital (in 8isk-	5	16,547	39%	s	3.374	2	8.0%	\$	4,218	2	10.0%
Weighted Assols) Tier (Capital (to	\$	16,017	38%	\$	1,687	7	4.0%	\$	2,595	2	6.0%
Average Assets)	5	16,017	2596	٩	2,538	2	4.0%	٢	3,173	£	5.0%
As of December 31, 2009. Total Capital (to Risk-											
Weighted Assets] Tier I Capital (to Risk-	\$	17,779	54%	\$	2,623	٢	H 046	5	3,279	2	10.0%
Weighted Assets) Tion I Cepital (to	\$	17,457	53%	\$	1,911	2	4.0%	2	1,967	2	6.0%
Average Assets)	5	17,457	42%	s	1,679	Z	40%	\$	2,009	۶	5.6%

The California Financial Code restricts the total dividend payment of any state banking association in any calendar year to the textur of the Bank's retained earnings on the Bank's net income for its last three fiscal years, less distributions made to shareholders during the same three-year period. At December 31, 2010, no amounts were available for dividend distributions.

_ . ..

NOTE 12 - STOCK OPTION PLAN

During 2008, the Company's Board of Directors approved a fixed stock option plan (the Plan) under which increasive and non-qualified stock options may be granted to key employees, directors, and organizers to purchase up to 495,000 shares of the authorized and non-scored communications of the formpany at a prior equal to the fair market value on the date of grant. The Plan provides that the incentive options are exercisable in equal increments over a three-year period from the date of the grant or over any other schedule approved by the Board of Directors. In addition, the Plan provides that all non-qualified stock options issued to the Company's organizers are fully vested upon issuance. Non-qualified stock options prants made to the Company's Board of Directors and employees are exercisable in equal increments organizers are fully vested upon issuance. Non-qualified stock options grants made to the Company's Board of Directors and employees are exercisable in equal increments over a three-year period from the date of upon issuance. Non-qualified stock options grants made to the Company's Board of Directors and employees are exercisable in equal increments over a three-year period from the date of grant All incentive and non-qualified stock options expire no later than teo years from the date of grant. The Plan was ratified by the shareholders as part of the approval of the private planement memorandum.

A summary of the status of the Company's fixed stock option during the year ended December 31, 2030, and 2009, menuding charges during the year ended December 31, 2031, are presented bylow.

		2010			
		Shares		Weighted- Average Exercise Price	
Outstanding bog mong of year Granted		357.500	5 5 5	10.00	
Exercised Forfaited		(30.000)	\$:0.00	
Outstanding end of year		377,500	ŝ	10.00	
		2010		2000	
Weighted average fair value of options granted Suring the period	\$	-	\$	Z. L L	
hitricsic value alloptions exercised	Ŧ		•		
Options controschie at year end:		258,767		202,167	
Weighted average erese ve price	\$	10.00	\$	10.00	
Infinitsé, value Weighted-average remaining contractual life		7.94 years		- B 97 years	
Options cutstanding at year end:		327,500		857,500	
Weighted-average exercise price Intrinsic value	\$	10.00	2	10.00	
Weighted average rentaming contractual life		7.96 years		897 years	
Componisation expresse to be recorded related to wire-vested stock options	\$	146.175	\$	296,275	
Weighted average period of time remaining compensation expenses will be recognized		L 07 years		2 05 years	

NOTE 13 - FAIR VALUE MEASUREMENTS

The Company adopted the fair value measurement accounting standard related to its financial and nonfinancial assets and liabilities. The accounting standard clavifies the definition of fair value, describes methods generally used to appropriately measure fair value in accordance with generally eccepted accounting principles, and expands fair value disclosure requirements. Fair value is defined in the accounting standard as the price that would be received to sell an asset or paid to transfor a liability in an orderly transaction between masket participants at the measurement date. The statement applies whenever other accounting pronouncements require or permit fair value measurements.

The fair value hierarchy under the fair value measurement accounting standard prioritizes the inputs to valuation techniques used to measure fair value into three broad levels (Level 1, Level 2, and Level 3). Level 1 inputs are unadjusted quoted prices in active markets (as defined) for identical assets or liabilities that the Company has the ability to arress at the measurement date, Level 2 inputs are inputs other than quoted prices included within Level 1 that are observable for the asset or liability, enhec directly or indirectly. Level 3 inputs are unobservable inputs for the asset or liability, and reflect the Company's own assumptions about the assumptions that market participants would use or pricing the asset or liability (including assumptions about nok) in principal market.

None of the Company's assets or liabilities are measured and recorded at fair value up a recurring or non-recurring basis at December 31, 2010 and 2009.

NOTE 14 - SUBSEQUENT EVENTS

The Bank has evaluated the effects of subsequent events that have occurred after the period ending December 31, 2010 and through March 24, 2011, which is the date the consolidated financial statements were issued.